ANNALY CAPITAL MANAGEMENT INC Form 424B3 June 04, 2018 Table of Contents

> Filed Pursuant to Rule 424(b)(3) Registration No. 333-224968

Offer by

MOUNTAIN MERGER SUB CORPORATION,

a direct wholly owned subsidiary of

ANNALY CAPITAL MANAGEMENT, INC.

to Exchange Each Outstanding Share of Common Stock of

MTGE INVESTMENT CORP.

for

\$9.82 in Cash and

0.9519 Shares of Common Stock of Annaly Capital Management, Inc.

or

\$19.65 in Cash

or

1.9037 Shares of Common Stock of Annaly Capital Management, Inc.

(subject in each case to the election procedures and, in the case of an all-cash

election or an all-stock election, to the proration procedures described

in this document and related letter of election and transmittal)

THE OFFER AND THE WITHDRAWAL RIGHTS WILL EXPIRE AT 5:00 P.M., EASTERN TIME, ON JUNE 18, 2018, UNLESS EXTENDED.

Annaly Capital Management, Inc., a Maryland corporation (Annaly), through its direct wholly owned subsidiary Mountain Merger Sub Corporation, a Maryland corporation (the Offeror), is offering (the offer), upon the terms and subject to the conditions set forth in this prospectus/offer to exchange and in the accompanying letter of election and transmittal, to exchange for each outstanding share of common stock, par value \$0.01 per share (the MTGE common stock), of MTGE Investment Corp., a Maryland corporation (MTGE), validly tendered and not validly withdrawn in the offer:

\$9.82 in cash; and

0.9519 shares of Annaly common stock, par value \$0.01 per share (the Annaly common stock), together with cash in lieu of any fractional shares of Annaly common stock. We refer to the above as the mixed consideration.

In lieu of receiving the mixed consideration, holders of shares of MTGE common stock may elect to receive, for each share of MTGE common stock that they hold, (1) \$19.65 in cash (we refer to this election as the all-cash election and this amount as the all-cash consideration) or (2) 1.9037 shares of Annaly common stock, together with cash in lieu of any fractional shares of Annaly common stock (we refer to this election as the all-stock election and this amount as the all-stock consideration). The mixed consideration, the all-cash consideration and the all-stock consideration (as applicable) will be paid without interest and less any applicable withholding taxes.

MTGE common stockholders who validly tender and do not validly withdraw their shares of MTGE common stock in the offer but who do not make a valid election will receive the mixed consideration for their shares of MTGE common stock. MTGE common stockholders who make the all-cash election or the all-stock election will be subject to proration so that approximately 50.0% of the aggregate consideration in the offer will

be paid in shares of Annaly common stock and approximately 50.0% of the aggregate consideration in the offer will be paid in cash. See The Offer Elections and Proration for a description of the proration procedure.

The purpose of the offer is for Annaly to acquire control of, and ultimately the entire common equity interest in, MTGE. The offer is the first step in Annaly s plan to acquire all of the issued and outstanding shares of MTGE common stock. If the offer is completed, promptly following the consummation of the offer, Annaly intends to consummate a merger of MTGE with and into the Offeror, with the Offeror surviving the merger (which we refer to as the merger). The purpose of the merger is for Annaly to acquire all of the issued and outstanding shares of MTGE common stock that it did not acquire in the offer. In the merger, each issued and outstanding share of MTGE common stock that was not acquired by Annaly or the Offeror will be converted into the mixed consideration or, at the election of the holder of such shares, the all-cash consideration or all-stock consideration, subject to proration so that approximately 50.0% of the aggregate consideration in the merger will be paid in shares of Annaly common stock and approximately 50.0% of the aggregate consideration in the merger will be paid in cash. In addition, in the merger, each issued and outstanding share of MTGE s 8.125% Series A Cumulative Redeemable Preferred Stock, \$0.01 par value per share (the MTGE Series A preferred stock) will be automatically converted into the right of the holders thereof to receive one newly issued share of Annaly s 8.125% Series H Cumulative Redeemable Preferred Stock, \$0.01 par value per share, which will have rights, preferences, privileges and voting powers substantially the same as those of the MTGE Series A preferred stock. After the merger, the MTGE business will be held in a wholly owned subsidiary of Annaly, and the former MTGE stockholders will no longer have any direct ownership interest in the surviving corporation.

NOTICE OF OFFER AND MERGER UNDER SECTION 3-106.1 OF THE MARYLAND GENERAL CORPORATION LAW

In accordance with Section 3-106.1(e) of the Maryland General Corporation Law (the MGCL), notice of the offer and the merger and the transactions contemplated thereby is hereby given by Annaly and the Offeror. The Articles of Merger, pursuant to which the Merger will become effective, will be filed with the State Department of Assessments and Taxation of Maryland not earlier than 30 days after the date of this preliminary prospectus/offer to exchange.

The Offeror s obligation to accept for exchange, and to exchange, shares of MTGE common stock for cash and shares of Annaly common stock in the offer is subject to a number of conditions, including that at least a majority of the outstanding shares of MTGE common stock have been validly tendered (and not validly withdrawn) in the offer. See The Offer Conditions of the Offer for a description of all of such conditions.

Annaly common stock is listed on the New York Stock Exchange (NYSE) under the symbol NLY, and MTGE common stock is listed on the Nasdaq Stock Market (Nasdaq) under the symbol MTGE.

The merger will not entitle MTGE stockholders to dissenters rights or rights of objecting stockholders under the MGCL and MTGE s charter.

For a discussion of certain factors that MTGE common stockholders should consider in connection with the offer, please read the section of this document entitled <u>Risk Factors</u> beginning on page 23.

You are encouraged to read this entire document and the related letter of election and transmittal carefully, including the annexes and information referred to or incorporated by reference in this document.

Neither Annaly nor the Offeror has authorized any person to provide any information or to make any representation in connection with the offer other than the information contained or incorporated by reference in this document, and if any person provides any information or makes any representation of this kind, that information or representation must not be relied upon as having been authorized by Annaly or the Offeror.

Neither the U.S. Securities and Exchange Commission (the SEC) nor any state securities commission has approved or disapproved of these securities or passed upon the adequacy or accuracy of this document. Any representation to the contrary is a criminal offense.

The date of this preliminary prospectus/offer to exchange is May 31, 2018.

TABLE OF CONTENTS

	Page
QUESTIONS AND ANSWERS ABOUT THE OFFER AND THE MERGER	1
<u>SUMMARY</u>	9
SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF ANNALY	18
SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF MTGE	20
COMPARATIVE PER SHARE DATA (UNAUDITED)	22
<u>RISK FACTORS</u>	23
Risk Factors Relating to the Offer and the Merger	23
Risk Factors Relating to Annaly as the Combined Company	25
Risks Related to Annaly s Business	26
<u>Risks Related to MTGE s Business</u>	26
FORWARD-LOOKING STATEMENTS	27
THE COMPANIES	28
Annaly	28
Offeror	28
MTGE	28
THE OFFER	29
General	29
Background of the Offer and the Merger	29
Annaly s Reasons for the Offer and the Merger	38
MTGE s Reasons for the Offer and the Merger; Recommendation of the MTGE Board of Directors	40
Opinion of the MTGE Special Committee s Financial Advisor	45
MTGE Unaudited Prospective Financial Information	52
Annaly Unaudited Prospective Financial Information	55
Elections and Proration	57
Consequences of Tendering with No Election	60
Distribution of Offering Materials	60
Expiration of the Offer	60
Extension, Termination and Amendment	61
Exchange of Shares; Delivery of Cash and Annaly Shares	62
Withdrawal Rights	63
Procedure for Tendering	63
No Guaranteed Delivery	65
Grant of Proxy	65
Fees and Commissions	65
Matters Concerning Validity and Eligibility	65
Announcement of Results of the Offer	66
Ownership of Annaly Common Stock After the Offer and the Merger	66
Purpose of the Offer and the Merger; Dissenters Rights	66
Plans for MTGE	67
Effect of the Offer on the Market for the MTGE Shares; Nasdaq Listing; Registration Under the Exchange	
Act; Margin Regulations	67
Conditions of the Offer	68
Certain Legal Matters: Regulatory Approvals	71

Interests of Certain Persons in the Offer and the Merger	72
Certain Relationships With MTGE	73
Management Agreement Termination	74
Source and Amount of Funds	74
Fees and Expenses	74

i

Accounting Treatment	75
Stock Exchange Listing	75
Resale of Annaly Common Stock	75
Exchange Agent Contact Information	75
MERGER AGREEMENT	76
The Offer	76
The Merger	70
Completion and Effectiveness of the Merger	77
Merger Consideration for Common Stock	77
<u>No Rights of Objecting Stockholders</u>	81
Treatment of Shares of MTGE Preferred Stock	81
Exchange of MTGE Stock Certificates or Book-Entry Shares for the Merger Consideration	81
Fractional Shares	81
Treatment of MTGE Restricted Stock Units	82
Representations and Warranties	82
Material Adverse Effect	84
No Solicitation of Other Offers by MTGE	85
Change of Recommendation	87
Conduct of Business Before Completion of the Merger	88
Additional Dividends	92
Access	93
Existing Financing Cooperation	93
Other Agreements	93
Employee Matters	93
Directors and Officers Indemnification	94
Conditions to the Merger	95
Termination of the Merger Agreement	95
Termination Fee and Expenses	96
Effect of Termination	97
Amendments; Enforcement and Remedies; Extensions and Waivers	97
COMPARATIVE MARKET PRICE AND DIVIDEND MATTERS	98
Market Price History	98
Dividends	98
MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES	100
DESCRIPTION OF ANNALY CAPITAL STOCK	131
General	131
Description of Common Stock	131
Description of Annaly Series H Preferred Stock	132
Certain Provisions of Maryland Law and Annaly s Charter and Bylaws	144
Restrictions on Ownership and Transfer	147
COMPARISON OF STOCKHOLDERS RIGHTS	149
LEGAL MATTERS	160
EXPERTS	160
WHERE TO OBTAIN MORE INFORMATION	161

Annex A	Agreement and Plan of Merger	A-1
Annex B	Opinion of Barclays Capital	B-1

<u>Annex C</u> <u>Directors and Executive Officers of Annaly and the Offeror</u>

This document incorporates by reference important business and financial information about Annaly, MTGE and their respective subsidiaries from documents filed with the SEC that have not been included in or delivered with this document. This information is available without charge at the SEC s website at www.sec.gov, as well as from other sources. See Where To Obtain More Information.

You can obtain the documents incorporated by reference in this document by requesting them in writing or by telephone at the following address and telephone number:

Annaly Capital Management, Inc.

1211 Avenue of the Americas

New York, New York 10036

Attention: Investor Relations

1-888-8ANNALY

http://www.annaly.com/investors

In addition, if you have questions about the offer or the merger, or if you need to obtain copies of this document, the letter of election and transmittal or other documents incorporated by reference in this document, you may contact the information agent for this offer listed below. You will not be charged for any of the documents you request.

D.F. King & Co., Inc.

48 Wall Street, 22nd Floor

New York, New York 10005

Bankers & Brokers Call: (212) 269-5550

All Others Call Toll-Free: (800) 669-5550

If you would like to request documents, please do so by June 11, 2018, in order to receive them before the expiration of the offer.

Information included in this document relating to MTGE, including but not limited to the descriptions of MTGE and its business and the information under the headings Selected Historical Consolidated Financial Data of MTGE, The Offer Background of the Offer and the Merger, The Offer MTGE s Reasons for the Offer and the Merger; Recommendation of the MTGE Board of Directors, The Offer Opinion of the MTGE Special Committee s Financial Advisor, The Offer MTGE Unaudited Prospective Financial Information and The Offer Interests of Certain Persons in the Offer and the Merger appears in the Solicitation/Recommendation Statement on Schedule 14D-9 dated May 16, 2018 and filed by MTGE with the SEC (the Schedule 14D-9). The Schedule 14D-9 was mailed to holders of shares of MTGE common stock on or about May 16, 2018.

iii

QUESTIONS AND ANSWERS ABOUT THE OFFER AND THE MERGER

Below are some of the questions that you as a MTGE stockholder may have regarding the offer and the merger and answers to those questions. You are encouraged to carefully read the remainder of this document and the related letter of election and transmittal, including the annexes and information referred to or incorporated by reference in this document, because the information contained in this section and in the section entitled Summary is not complete. Additional important information is contained in the remainder of this document, and the related letter of election and transmittal. See Where To Obtain More Information. As used in this document, unless otherwise indicated or the context requires, Annaly, we or us refers to Annaly Capital Management, Inc., a Maryland corporation, and its consolidated subsidiaries; the Offeror refers to MUTGE Investment Corp., a Maryland corporation, and its consolidated subsidiaries. The acceptance time for purposes of this document and the merger agreement is the time that Annaly will accept for payment, subject to the terms and conditions of the merger agreement, all shares of MTGE common stock, par value \$0.01 per share (the MTGE common stock), that are validly tendered and not validly withdrawn in the offer.

Who is offering to buy my shares of MTGE common stock and why I am receiving this document?

Annaly Capital Management, Inc. and the Offeror are making this offer to exchange cash and shares of Annaly common stock, par value \$0.01 per share (the Annaly common stock) for shares of MTGE common stock pursuant to the Agreement and Plan of Merger, dated as of May 2, 2018, entered into by Annaly, the Offeror and MTGE, which is referred to as the merger agreement . You are receiving this document because you own shares of MTGE common stock.

Annaly is a leading diversified capital manager that invests in and finances residential and commercial assets. Annaly s principal business objective is to generate net income for distribution to its shareholders and to preserve capital through prudent selection of investments and continuous management of its portfolio. Annaly has elected to be taxed as a real estate investment trust, or REIT, for U.S. federal income tax purposes. Annaly is externally managed by Annaly Management Company LLC.

What are the classes and amounts of MTGE securities that Annaly is seeking to acquire in the offer?

Annaly is seeking to acquire all of the issued and outstanding shares of MTGE common stock in the offer.

What will I receive for my shares of MTGE common stock in the offer?

Annaly and the Offeror are offering to exchange for each outstanding share of MTGE common stock validly tendered and not validly withdrawn in the offer:

\$9.82 in cash; and

0.9519 shares of Annaly common stock together with cash in lieu of any fractional shares of Annaly common stock.

We refer to the above as the mixed consideration.

In lieu of receiving the mixed consideration, holders of shares of MTGE common stock may elect to receive, for each share of MTGE common stock that they hold, (1) \$19.65 in cash (we refer to this election as the all-cash election and this amount as the all-cash consideration) or (2) 1.9037 shares of Annaly common stock, together with cash in lieu of any fractional shares of Annaly common stock (we refer to this election as the all-stock election and this amount as the all-stock consideration). The mixed consideration, the all-cash consideration and the all-stock consideration (as applicable) will be paid without interest and less any applicable withholding taxes (such consideration, the common transaction consideration).

MTGE common stockholders who validly tender and do not validly withdraw their shares of MTGE common stock in the offer but who do not make a valid election will receive the mixed consideration for their shares of MTGE common stock. MTGE common stockholders who make the all-cash election or the all-stock election will be subject to proration so that approximately 50.0% of the aggregate consideration in the offer will be paid in shares of Annaly common stock and approximately 50.0% of the aggregate consideration in the offer will be paid in cash. See The Offer Elections and Proration for a description of the proration procedure.

MTGE common stockholders should consider the potential effects of proration and should obtain current market quotations for shares of MTGE common stock and shares of Annaly common stock before deciding whether to tender pursuant to the offer and before electing the form of consideration they wish to receive. The market value of the stock portion of the common transaction consideration will change as the market value of Annaly common stock fluctuates during the offer period and thereafter. Please see Risk Factors Risk Factors Relating to the Offer and the Merger.

What will happen to my shares of MTGE preferred stock?

If you own shares of MTGE s 8.125% Series A Cumulative Redeemable Preferred Stock, par value \$0.01 per share (the MTGE Series A preferred stock), you do not need to do anything in connection with the offer, as the offer is not applicable to the MTGE Series A preferred stock. If the offer is completed, in connection with the completion of the merger, each issued and outstanding share of MTGE Series A preferred stock will be automatically converted into the right to receive one share of Annaly s 8.125% Series H Cumulative Redeemable Preferred Stock, par value \$0.01 per share (the Annaly Series H preferred stock) on substantially the same terms.

Will I have to pay any fee or commission to exchange my shares of MTGE common stock?

If you are the record owner of your shares of MTGE common stock and you tender these shares in the offer, you will not have to pay any brokerage fees, commissions or similar expenses. If you own your shares of MTGE common stock through a broker, dealer, commercial bank, trust company or other nominee and your broker, dealer, commercial bank, trust company or other nominee tenders your shares of MTGE common stock on your behalf, your broker or such other nominee may charge a fee for doing so. You should consult your broker, dealer, commercial bank, trust company or other nominee to determine whether any charges will apply.

Why is Annaly making this offer and what is the purpose of the merger?

The purpose of the offer is for Annaly to acquire control of, and ultimately own the entire common equity interest in, MTGE. The offer is the first step in Annaly s plan to acquire all of the outstanding shares of MTGE common stock. Annaly intends to consummate the merger promptly following the completion of the offer. The purpose of the merger is for Annaly to acquire all of the issued and outstanding shares of MTGE common stock that it did not acquire in the offer. After the merger, the MTGE business will be held in a wholly owned subsidiary of Annaly, and the former MTGE stockholders will no longer have any direct ownership interest in this entity.

What does the MTGE board of directors recommend?

The MTGE board of directors, acting upon the unanimous recommendation of a special committee of the MTGE board of directors comprised of three independent directors (the MTGE special committee), has resolved to recommend that the MTGE stockholders accept the offer and tender their shares of MTGE common stock to Annaly in the offer. The MTGE board of directors, upon the unanimous recommendation of the MTGE special committee, also determined that the terms of the merger agreement and the transactions, including the offer, the merger and the

issuance of shares of Annaly common stock in connection therewith, are in the best interests of MTGE and its stockholders, and declared the offer, the merger and the other transactions contemplated by the merger agreement advisable.

See The Offer MTGE s Reasons for the Offer and the Merger; Recommendation of the MTGE Board of Directors for more information. A description of the reasons for this recommendation is also set forth in MTGE s Solicitation/Recommendation Statement on Schedule 14D-9 (the Schedule 14D-9) that is being mailed to you together with this document.

What are the most significant conditions of the offer?

The offer is conditioned upon, among other things, the following:

- 1. *Minimum Tender Condition* MTGE common stockholders having validly tendered and not validly withdrawn in accordance with the terms of the offer and prior to the expiration of the offer a number of shares of MTGE common stock that, together with any shares of MTGE common stock then owned by Annaly and the Offeror, represents at least a majority of the then-outstanding shares of MTGE common stock at any expiration of the offer (the minimum tender condition);
- 2. *Effectiveness of Form S-4* The registration statement on Form S-4, of which this document is a part, having become effective under the U.S. Securities Act of 1933, as amended (the Securities Act), and must not be the subject of any stop order or proceeding seeking a stop order, at any expiration of the offer;
- 3. No MTGE Material Adverse Effect There not having occurred and continuing prior to the expiration of the offer any change, effect, development, circumstance, condition, state of facts, event or occurrence after the date of the merger agreement that has had or would reasonably be expected to have, individually or in the aggregate, a material adverse effect on MTGE and its subsidiaries (with such term as defined in the merger agreement and described under Merger Agreement Material Adverse Effect);
- 4. *No Annaly Material Adverse Effect* There not having occurred and continuing prior to the expiration of the offer any change, effect, development, circumstance, condition, state of facts, event or occurrence after the date of the merger agreement that has had or would reasonably be expected to have, individually or in the aggregate, a material adverse effect on Annaly and its subsidiaries (with such term as defined in the merger agreement and described under Merger Agreement Material Adverse Effect);
- 5. *Accuracy of MTGE s Representations and Warranties* The representations and warranties of MTGE contained in the merger agreement being true and correct as of the expiration of the offer, subject to specified materiality standards;
- 6. *MTGE s Compliance with Covenants* MTGE must have, in all material respects, performed or complied with its obligations, agreements and covenants required to be performed or complied with by it under the merger agreement at or prior to the expiration of the offer;

Accuracy of Annaly s and the Offeror s Representations and Warranties The representations and warranties of Annaly and the Offeror contained in the merger agreement being true and correct as of the expiration of the offer, subject to specified materiality standards;

- 8. *Annaly s and the Offeror s Compliance with Covenants* each of Annaly and the Offeror must have, in all material respects, performed or complied with their agreements and covenants required to be performed or complied with by it under the merger agreement at or prior to the expiration of the offer;
- 9. *Listing of Annaly Common Stock and Annaly Preferred Stock* The shares of Annaly common stock to be issued in the offer and the merger and the shares of Annaly Series H preferred stock to be issued in the merger having been approved for listing on the NYSE, subject to official notice of issuance;
- 10. *No Legal Prohibition* No law, order or injunction (whether temporary, preliminary or permanent) restraining or enjoining or otherwise prohibiting the consummation of the offer or the merger must have been enacted, issued, promulgated or granted by a governmental entity of competent jurisdiction;

- 11. *Regulatory Approvals* The approvals of the U.S. Department of Housing and Urban Development (HUD), the HUD lenders/servicers of certain loans financed by HUD and the Federal National Mortgage Association (Fannie Mae), if required, must have been obtained at or prior to the expiration of the offer;
- 12. *Transaction Tax Opinions* The receipt of a written opinion by each of Annaly and MTGE from its legal counsel, dated as of the expiration date, to the effect that the offer and the merger, taken together, will qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (the Code); and
- 13. *REIT Tax Opinions* The receipt of a written opinion by each of Annaly and MTGE from the other party s tax counsel, dated as of the expiration date, to the effect that at all times since a specified date and, with respect to Annaly, through the expiration date or, with respect to MTGE, through and including the taxable year that ends on the effective time of the merger, such company has been organized and operated in conformity with the requirements for qualification as a REIT under the Code and, with respect to Annaly, that the proposed method of operation of Annaly will enable Annaly to continue to meet the requirements for qualification as a REIT under the Code and REIT under the Code.

The offer is subject to certain other conditions set forth below in the section entitled The Offer Conditions of the Offer. The conditions to the offer are for the sole benefit of Annaly and the Offeror and may be asserted by Annaly or the Offeror regardless of the circumstances giving rise to any such condition (other than as a result of any action or inaction by Annaly or the Offeror that is completely within the control of Annaly or the Offeror), and may be waived by Annaly or the Offeror, by express and specific action to that effect, in whole or in part at any time and from time to time. However, certain specified conditions (including all the conditions noted above other than the conditions related to a material adverse effect on MTGE, accuracy of MTGE s representations, MTGE s compliance with covenants and receipt by Annaly of the transaction and REIT tax opinions noted above) may only be waived by Annaly or the Offeror with the express written consent of MTGE. Pursuant to the merger agreement, MTGE has the right to require that Annaly and the Offeror waive the conditions set forth in the fourth, seventh, eighth, twelfth and thirteenth paragraphs above (no material adverse effect on Annaly, accuracy of Annaly s and the Offeror s representations and warranties, Annaly s and the Offeror s compliance with covenants, MTGE s receipt of the transaction tax opinion and MTGE s receipt of the REIT tax opinion). There is no financing condition to the offer.

How long will it take to complete the offer and the merger?

The offer and the merger are expected to be completed during the third quarter of 2018, subject to the satisfaction or waiver of the conditions described in the sections entitled The Offer Conditions of the Offer and Merger Agreement Conditions to the Merger.

How long do I have to decide whether to tender my shares of MTGE common stock in the offer?

The offer is scheduled to expire at 5:00 p.m., Eastern Time, on June 18, 2018, unless further extended or terminated. Any extension, delay, termination, waiver or amendment of the offer will be followed as promptly as practicable by public announcement thereof to be made no later than 9:00 a.m., Eastern Time, on the next business day after the previously scheduled expiration date. During any such extension, all of the shares of MTGE common stock previously tendered and not validly withdrawn will remain subject to the offer, subject to the rights of a tendering stockholder to withdraw such stockholder s shares. Expiration date means 5:00 p.m., Eastern Time, on June 18, 2018, unless and until the Offeror has extended the period during which the offer is open, subject to the terms and conditions of the merger agreement, in which event the term expiration date means the latest time and date at which the offer, as so extended by

the Offeror, will expire.

Subject to the provisions of the merger agreement and the applicable rules and regulations of the Securities and Exchange Commission (SEC), and unless MTGE consents otherwise or the merger agreement is otherwise

terminated, the Offeror must (1) extend the offer for one or more successive periods of up to 10 business days each (or for such longer period as may be agreed by MTGE) in order to further seek to satisfy the conditions to the offer in the event that any of the offer conditions (other than the minimum tender condition) have not been satisfied or validly waived as of any then scheduled expiration of the offer, (2) extend the offer for up to two successive periods of 10 business days each (or such longer or shorter period as may be agreed by MTGE) if each of the offer conditions (other than the minimum tender condition) has been satisfied or validly waived and the minimum tender condition has not been satisfied as of the scheduled expiration of the offer, and MTGE requests that the Offeror so extend the offer, and (3) extend the offer for the minimum period required by any rule, regulation, interpretation or position of the SEC or its staff or NYSE or Nasdaq that is applicable to the offer, the merger, the Schedule TO or the related offer documents. However, the Offeror is not required to extend the offer beyond February 2, 2019 (which is the outside date).

Any decision to extend the offer will be made public by an announcement regarding such extension as described under the section entitled The Offer Extension, Termination and Amendment.

How do I tender my shares of MTGE common stock?

To tender your shares of MTGE common stock represented by physical certificates in the offer, you must deliver the certificates representing such shares, together with a completed letter of election and transmittal and any other documents required by the letter of election and transmittal, to Computershare Trust Company, N.A. (Computershare), the depositary and exchange agent for the offer, not later than the expiration date. The letter of election and transmittal is enclosed with this document.

To tender your shares of MTGE common stock in electronic book-entry form, you must deliver an agent s message in connection with a book-entry transfer, and any other required documents, to the exchange agent at its address set forth elsewhere in this document and follow the other procedures for book-entry tender set forth herein, not later than the expiration date.

If your shares of MTGE common stock are held in street name (*i.e.*, through a broker, dealer, commercial bank, trust company or other nominee), these shares can be tendered by your nominee by book-entry transfer through The Depository Trust Company.

We are not providing for guaranteed delivery procedures and therefore you must allow sufficient time for the necessary tender procedures to be completed during normal business hours of The Depository Trust Company prior to the expiration date. Tenders received by the exchange agent after the expiration date will be disregarded and of no effect. In all cases, you will receive your consideration for your tendered shares of MTGE common stock only after timely receipt by the exchange agent of certificates for such shares (or of a confirmation of a book-entry transfer of such shares) and a properly completed and duly executed letter of election and transmittal and any other required documents.

For a complete discussion on the procedures for tendering your shares of MTGE common stock, see The Offer Procedure for Tendering.

Until what time can I withdraw tendered shares of MTGE common stock?

You may withdraw your previously tendered shares of MTGE common stock at any time until the offer has expired and, if the Offeror has not accepted your shares of MTGE common stock for payment by July 15, 2018, you may withdraw them at any time on or after that date until the Offeror accepts shares for payment. Once the Offeror accepts

your tendered shares of MTGE common stock for payment upon expiration of the offer, however, you will no longer be able to withdraw them. For a complete discussion of the procedures for withdrawing your shares of MTGE common stock, see The Offer Withdrawal Rights.

How do I withdraw previously tendered shares of MTGE common stock?

To withdraw previously tendered shares of MTGE common stock, you must deliver a written notice of withdrawal with the required information to the exchange agent at any time that you have the right to withdraw shares. If you tendered shares of MTGE common stock by giving instructions to a broker, dealer, commercial bank, trust company or other nominee, you must instruct such broker, dealer, commercial bank, trust company or other nominee to arrange for the withdrawal of your shares of MTGE common stock and such broker, dealer, commercial bank, trust company or other nominee must effectively withdraw such shares at any time that you have the right to withdraw shares. For a discussion on the procedures for withdrawing your shares of MTGE common stock, including the applicable deadlines for effecting withdrawals, see The Offer Withdrawal Rights.

When and how can I expect to receive the offer consideration in exchange for my tendered shares of MTGE common stock?

The Offeror will exchange all validly tendered and not validly withdrawn shares of MTGE common stock promptly after the expiration date, subject to the terms thereof and the satisfaction or waiver of the conditions to the offer, as set forth in The Offer Conditions of the Offer. The Offeror will deliver the consideration for your validly tendered and not validly withdrawn shares through the exchange agent, that will act as your agent for the purpose of receiving the common transaction consideration from the Offeror and transmitting such consideration to you. In all cases, you will receive the consideration for your tendered shares of MTGE common stock only after timely receipt by the exchange agent of certificates representing such shares of MTGE common stock (or a confirmation of a book-entry transfer of such shares as described in the section entitled The Offer Procedure for Tendering) and a properly completed and duly executed letter of election and transmittal and any other required documents for such shares.

Why does the cover page to this document state that this offer is preliminary and subject to change, and that the registration statement filed with the SEC is not yet effective? Does this mean that the offer has not commenced?

No. Completion of this document and effectiveness of the registration statement are not necessary to commence this offer. The offer was commenced on May 16, 2018, the date of the initial filing of the registration statement on Form S-4 of which this document is a part. The Offeror cannot, however, accept for exchange any shares of MTGE common stock tendered in the offer or exchange any shares until the registration statement is declared effective by the SEC and the other conditions to the offer have been satisfied or waived.

What happens if I do not tender my shares of MTGE common stock?

If, after consummation of the offer, Annaly and the Offeror own a majority of the outstanding shares of MTGE common stock, Annaly and the Offeror intend to promptly complete the merger. Upon consummation of the merger, each issued and outstanding share of MTGE common stock that has not been tendered and accepted for exchange in the offer will be converted in the merger into the right to receive, at the election of the holder, the all-cash consideration, the all-stock consideration or the mixed consideration, but the all-cash consideration and all-stock consideration will be subject to proration so that approximately 50.0% of the aggregate consideration in the merger will be paid in shares of Annaly common stock and approximately 50.0% of the aggregate consideration in the merger will be paid in cash. A letter of election and transmittal will be sent to you following the merger to make these elections. The election deadline to make these elections will be 5:00 p.m. Eastern Time, on the 20th calendar day following the date on which such forms of election and transmittal are mailed. If you do not make a valid election, you will be treated as if you had made an election to receive the mixed consideration.

If the offer is completed, will MTGE continue as a public company?

No. Annaly is required, on the terms and subject to the satisfaction or waiver of the conditions set forth in the merger agreement, to consummate the merger promptly following the purchase of shares of MTGE common

stock in the offer. Upon the effective time of the merger, MTGE will no longer be publicly traded. If, for some reason, the merger does not take place, and Annaly (through the Offeror) purchases all of the shares of MTGE common stock validly tendered and not validly withdrawn, there may be so few remaining stockholders and publicly held shares that MTGE common stock will no longer be eligible to be traded through Nasdaq or other securities exchanges, there may not be an active public trading market for MTGE common stock, and MTGE may no longer be required to make filings with the SEC or otherwise comply with the SEC rules relating to publicly held companies.

Will the offer be followed by a merger if all of the shares of MTGE common stock are not tendered in the offer?

Yes, unless the conditions to the merger are not satisfied or waived. If the Offeror accepts for payment all of the shares of MTGE common stock validly tendered and not validly withdrawn pursuant to the offer, and the other conditions to the merger are satisfied or waived, the merger will take place promptly after the consummation of the offer. If the merger takes place, Annaly will own 100% of the equity of MTGE, and all of the remaining holders of MTGE common stock, other than Annaly and the Offeror, will have the right to receive the mixed consideration, the all-cash consideration or the all-stock consideration, in each case without interest and less any applicable withholding taxes, with the form of such consideration to be subject to further proration as described in this document (such consideration, the common transaction consideration).

Because the merger will be governed by Section 3-106.1 of the Maryland General Corporation Law (the MGCL), no stockholder vote will be required to consummate the merger in the event that the offer is consummated. Annaly is required, on the terms and subject to the satisfaction or waiver of the conditions set forth in the merger agreement, to consummate the merger promptly following the consummation of the offer. As such, Annaly does not expect there to be a significant period of time between the consummation of the offer and the consummation of the merger.

Does Annaly have the financial resources to complete the offer and the merger?

Yes. The common transaction consideration will consist of Annaly common stock and cash. The offer and the merger are not conditioned upon any financing arrangements or contingencies.

What are the U.S. federal income tax consequences of receiving shares of Annaly common stock and/or cash in exchange for my shares of MTGE common stock in the offer and/or the merger?

The offer and the merger, taken together, are intended to qualify as a reorganization within the meaning of Section 368(a) of the Code. If the offer and the merger, taken together, qualify as a reorganization within the meaning of Section 368(a) of the Code, the U.S. federal income tax consequences to U.S. holders (as defined herein) of MTGE common stock who receive shares of Annaly common stock and/or cash in exchange for their shares pursuant to the offer and/or the merger generally will be as follows:

if a MTGE stockholder receives solely shares of Annaly common stock in exchange for its shares of MTGE common stock, such stockholder generally will not recognize any gain or loss, except with respect to cash received in lieu of a fractional share of Annaly common stock;

if a MTGE stockholder receives solely cash in exchange for its shares of MTGE common stock, such stockholder generally will recognize gain or loss equal to the difference between the amount of cash received

and the stockholder s tax basis in its MTGE common stock surrendered; and

if a MTGE stockholder receives a combination of Annaly common stock and cash in exchange for its shares of MTGE common stock, such stockholder generally will recognize gain (but not loss) in an amount equal to the lesser of (1) the sum of the cash and the fair market value of the Annaly common stock received, less the stockholder s tax basis in its MTGE common stock surrendered, and (2) the amount of cash received.

Non-U.S. holders (as defined herein) who receive a combination of Annaly common stock and cash in exchange for shares of MTGE common stock pursuant to the offer and/or the merger may be subject to U.S. withholding tax with respect to the cash consideration.

Each MTGE stockholder should read the discussion under Material U.S. Federal Income Tax Consequences and should consult its own tax advisor regarding the U.S. federal, state, local and non-U.S. tax consequences of the offer and the merger to such stockholder in light of its particular circumstances.

Will I have the right to have my shares of MTGE common stock appraised?

No. Appraisal rights, rights of objecting stockholders or dissenters rights are not available in connection with the offer or the merger. See The Offer Purpose of the Offer and the Merger; Dissenters Rights.

Who should I call if I have questions about the offer?

You may call D.F. King & Co., Inc., the information agent, at (800) 669-5550 (toll free for stockholders) or (212) 269-5550 (banks and brokers).

Where can I find more information about Annaly and MTGE?

You can find more information about Annaly and MTGE from various sources described in the section entitled Where To Obtain More Information.

SUMMARY

This section summarizes certain information presented in greater detail elsewhere in this document. However, this summary does not contain all of the information that may be important to MTGE stockholders. You are urged to carefully read the remainder of this document and the related letter of election and transmittal and the other documents to which we have referred because the information in this section and in the Questions and Answers About the Offer and the Merger section is not complete. See Where To Obtain More Information.

The Offer (Page 29)

Annaly and the Offeror are offering, upon the terms and subject to the conditions set forth in this prospectus/offer to exchange and in the accompanying letter of election and transmittal, to exchange for each outstanding share of MTGE common stock validly tendered and not validly withdrawn in the offer:

\$9.82 in cash; and

0.9519 shares of Annaly common stock, together with cash in lieu of any fractional shares of Annaly common stock.

We refer to the above as the mixed consideration.

In lieu of receiving the mixed consideration, holders of shares of MTGE common stock may elect to receive, for each share of MTGE common stock that they hold, either (1) \$19.65 in cash (we refer to this election as the all-cash election and this amount as the all-cash consideration) or (2) 1.9037 shares of Annaly common stock, together with cash in lieu of any fractional shares of Annaly common stock (we refer to this election as the all-stock election and this amount as the all-stock consideration). The mixed consideration, the all-cash consideration and the all-stock consideration (as applicable) will be paid without interest and less any applicable withholding taxes (such consideration, the common transaction consideration).

MTGE common stockholders who tender their shares of MTGE common stock in the offer but who do not make a valid election will receive the mixed consideration for their shares of MTGE common stock. MTGE common stockholders who make the all-cash election or the all-stock election will be subject to proration so that approximately 50.0% of the aggregate consideration in the offer will be paid in shares of Annaly common stock and approximately 50.0% of the aggregate consideration in the offer will be paid in cash. See The Offer Elections and Proration for a description of the proration procedure.

MTGE common stockholders will not receive any fractional shares of Annaly common stock in the offer or the merger. No fractional shares of Annaly common stock will be issuable in the offer or the merger and each MTGE stockholder who otherwise would be entitled to receive a fraction of a share of Annaly common stock pursuant to the offer or the merger will be paid an amount in cash (without interest) equal to such fractional part of a share of Annaly common stock multiplied by the volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the time Annaly accepts for payment shares of MTGE common stock validly tendered and not validly withdrawn pursuant to the offer. See The Merger Agreement Fractional Shares.

Purpose of the Offer; The Merger; The Merger Consideration (Pages 66, 77 and 77)

Table of Contents

The purpose of the offer is for Annaly to acquire control of, and ultimately the 100% of the common equity interests in, MTGE. The offer is the first step in Annaly s plan to acquire all of the outstanding shares of MTGE common stock. Annaly intends to consummate the merger promptly after the consummation of the offer. The purpose of the merger is for Annaly to acquire all of the issued and outstanding shares of MTGE common stock that it did not acquire in the offer.

In the merger, each outstanding share of MTGE common stock that was not acquired by Annaly or the Offeror will be converted into the mixed consideration or, at the election of the holder of such shares, the all-cash consideration or all-stock consideration, subject to proration so that approximately 50.0% of the aggregate consideration in the merger will be paid in shares of Annaly common stock and approximately 50.0% of the aggregate consideration in the merger will be paid in cash. Following the merger, a letter of election and transmittal will be mailed to such non-tendering stockholders to make these elections. The election deadline to make these elections will be 5:00 p.m. Eastern Time on the 20th calendar day following the date on which the forms of election and transmittal are mailed. If you do not make an election, you will be treated as if you had made an election to receive the mixed consideration.

In addition, in the merger, each issued and outstanding share of MTGE Series A preferred stock will be automatically converted into the right to receive one newly issued share of Annaly Series H preferred stock, which will have rights, preferences, privileges and voting powers substantially the same as those of the MTGE Series A preferred stock.

After the merger, the MTGE business will be held by the Offeror, and the former MTGE common stockholders will no longer have any direct ownership interest in such entity.

Annaly expects to consummate the merger promptly after the consummation of the offer in accordance with Section 3-106.1 of the MGCL, and no stockholder vote to approve the offer or the merger, adopt the merger agreement or approve any other action by the MTGE stockholders will be required in connection with the merger. See The Offer Purpose of the Offer and the Merger; Dissenters Rights.

Treatment of MTGE Restricted Stock Units (Page 82)

At the effective time of the merger, each MTGE restricted stock unit, whether vested or unvested, will automatically be cancelled, with the holder of such restricted stock unit becoming entitled to receive (a) the mixed consideration in respect of each MTGE share underlying such restricted stock unit immediately prior to the effective time, and (b) a cash payment in lieu of any fractional share of Annaly common stock that such holder would otherwise be entitled to receive, in each case, less applicable tax withholdings.

The Companies (Page 28)

Annaly

Annaly Capital Management, Inc.

1211 Avenue of the Americas

New York, New York 10036

Annaly is a leading diversified capital manager that invests in and finances residential and commercial assets. Annaly s principal business objective is to generate net income for distribution to its shareholders and to preserve capital through prudent selection of investments and continuous management of its portfolio. Annaly has elected to be taxed as a real estate investment trust, or REIT, for U.S. federal income tax purposes. Annaly is externally managed by Annaly Management Company LLC. Annaly common stock trades under the ticker symbol NLY on the NYSE.

Offeror

Mountain Merger Sub Corporation

Table of Contents

c/o Annaly Capital Management, Inc.

1211 Avenue of the Americas

New York, New York 10036

The Offeror, a Maryland corporation, is a wholly owned subsidiary of Annaly. The Offeror is newly formed and was organized for the purpose of making the offer and consummating the merger. The Offeror has not engaged in any business activities to date, and it has no material assets or liabilities of any kind, other than those incidental to its formation and those incurred in connection with the merger agreement and the transactions contemplated thereby, including the offer and the merger.

MTGE

MTGE Investment Corp.

2 Bethesda Metro Center, 12th Floor

Bethesda, Maryland 20814

MTGE Investment Corp. is a real estate investment trust that invests in and manages a leveraged portfolio of agency mortgage investments, non-agency mortgage investments and other real estate-related investments. MTGE is externally managed and advised by MTGE Management, LLC, an affiliate of AGNC Investment Corp. (Nasdaq: AGNC).

MTGE s common stock trades under the ticker symbol MTGE on Nasdaq.

Annaly s Reasons for the Offer and the Merger (Page 38)

The purpose of the offer is for Annaly to acquire control of, and ultimately the entire common equity interest in, MTGE. Annaly and the Offeror are making the offer and Annaly plans to complete the merger because it believes that the acquisition of MTGE by Annaly will expand and diversify Annaly s investment portfolio, supporting the continued growth of Annaly s businesses.

Expiration of the Offer (Page 60)

The offer is scheduled to expire at 5:00 p.m., Eastern Time, on June 18, 2018, unless further extended or terminated. Expiration date means 5:00 p.m., Eastern Time, on June 18, 2018, unless and until the Offeror has extended or terminated the period during which the offer is open, subject to the terms and conditions of the merger agreement, in which event the term expiration date means the latest time and date at which the offer, as so extended by the Offeror, will expire.

Extension, Termination and Amendment (Page 61)

Subject to the provisions of the merger agreement and the applicable rules and regulations of the SEC, and unless MTGE consents otherwise or the merger agreement is otherwise terminated, the Offeror must (1) extend the offer for one or more successive periods of up to 10 business days (or such longer period as may be agreed by MTGE) each in order to further seek to satisfy the conditions to the offer in the event that any of the offer conditions (other than the minimum tender condition) have not been satisfied or validly waived as of any then scheduled expiration of the offer, (2) extend the offer for up to two successive periods of 10 business days each (or for such longer or shorter period as may be agreed by MTGE) if each of the offer conditions (other than the minimum tender condition) has been satisfied or validly waived as of the scheduled expiration of the offer, and MTGE requests that the Offeror so extend the offer, and (3) extend the offer for the minimum period required by any rule, regulation, interpretation or position of the SEC or its staff or NYSE or Nasdaq which is applicable to the

offer or the merger or to the extent necessary to resolve any comments of the SEC or its staff applicable to the offer, the merger, the Schedule TO or the related offer documents. However, the Offeror is not required to extend the offer beyond the outside date.

1	1
	L

The Offeror will effect any extension, termination, amendment or delay by giving oral or written notice to the exchange agent and by making a public announcement as promptly as practicable thereafter as described under The Offer Extension, Termination and Amendment.

In the case of an extension, any such announcement will be issued no later than 9:00 a.m., Eastern Time, on the next business day following the previously scheduled expiration date. Subject to applicable law (including Rules 14d-4(c) and 14d-6(d) under the U.S. Securities Exchange Act of 1934, as amended (the Exchange Act), which require that any material change in the information published, sent or given to stockholders in connection with the offer be promptly disseminated to stockholders in a manner reasonably designed to inform them of such change) and without limiting the manner in which the Offeror may choose to make any public announcement, the Offeror assumes no obligation to publish, advertise or otherwise communicate any such public announcement of this type other than by issuing a press release. During any extension, shares of MTGE common stock previously tendered and not validly withdrawn will remain subject to the offer, subject to the right of each MTGE stockholder to withdraw previously tendered shares of MTGE common stock.

The merger agreement provides that the merger agreement may be terminated if the offer has not been consummated on or before 11:59 p.m., Eastern Time, on February 2, 2019, and the Offeror may not extend the offer beyond such date without the prior written consent of MTGE.

No subsequent offering period will be available following the expiration of the offer.

Conditions of the Offer (Page 68)

The offer is subject to certain conditions, including:

that MTGE stockholders have validly tendered and not validly withdrawn in accordance with the terms of the offer and prior to the expiration of the offer a number of shares of MTGE common stock that, together with any shares of MTGE common stock then owned by Annaly and the Offeror, represents at least a majority of the then-outstanding shares of MTGE common stock at any expiration of the offer;

the effectiveness of the registration statement on Form S-4 of which this document is a part;

no material adverse effect on MTGE and its subsidiaries (as defined in the merger agreement and described under The Merger Agreement Material Adverse Effect) having occurred and continuing prior to the expiration of the offer;

no material adverse effect on Annaly and its subsidiaries (as defined in the merger agreement and described under The Merger Agreement Material Adverse Effect) having occurred and continuing prior to the expiration of the offer;

the truth and accuracy of MTGE s representations and warranties made in the merger agreement, subject to specified materiality standards;

MTGE being in material compliance with its covenants under the merger agreement;

the truth and accuracy of Annaly s and the Offeror s representations and warranties made in the merger agreement, subject to specified materiality standards;

Annaly and the Offeror being in material compliance with their covenants under the merger agreement;

the listing on the NYSE of the shares of Annaly common stock to be issued in the offer and the merger and the shares of Annaly Series H preferred stock to be issued in the merger;

lack of legal prohibitions;

receipt of approvals of HUD, lenders/servicers of certain loans financed by HUD and Fannie Mae, if required;

the receipt of opinions by each of Annaly and MTGE from their respective legal counsel regarding the U.S. federal income tax treatment of the offer and the merger; and

the receipt of opinions by each of Annaly and MTGE from the other party s respective legal counsel regarding each of Annaly s and MTGE s qualification as a REIT.

Subject to applicable SEC rules and regulations, the Offeror also reserves the right, in its sole discretion, at any time or from time to time to waive any condition identified as subject to waiver in The Offer Conditions of the Offer by giving oral or written notice of such waiver to the exchange agent. However, certain specified conditions (including all conditions noted in the immediately preceding list other than third, fifth, sixth, eleventh, twelfth and thirteenth conditions listed above and other than the receipt by Annaly of the opinions noted above) may only be waived by Annaly or the Offeror with the express written consent of MTGE. Pursuant to the merger agreement, MTGE has the right to require that Annaly and the Offeror waive the fourth, seventh and eighth conditions listed above and the conditions listed of the opinions listed above.

Withdrawal Rights (Page 63)

Tendered shares of MTGE common stock may be withdrawn at any time prior to the expiration date. Additionally, if the Offeror has not agreed to accept the shares for exchange on or prior to July 15, 2018, MTGE common stockholders may thereafter withdraw their shares from tender at any time after such date until the Offeror accepts the shares for exchange pursuant to the offer, all tenders not previously withdrawn become irrevocable.

Procedure for Tendering (Page 63)

To validly tender shares of MTGE common stock pursuant to the offer, MTGE common stockholders must:

deliver a properly completed and duly executed letter of election and transmittal, along with any required signature guarantees and any other required documents, and certificates for tendered shares of MTGE common stock to the exchange agent at its address set forth elsewhere in this document, all of which must be received by the exchange agent prior to the expiration date; or

deliver an agent s message in connection with a book-entry transfer, and any other required documents, to the exchange agent at its address set forth elsewhere in this document, and shares must be tendered pursuant to the procedures for book-entry tender set forth herein (and a confirmation of receipt of that tender received), and in each case be received by the exchange agent prior to the expiration date.

MTGE common stockholders who hold shares of MTGE common stock in street name through a bank, broker or other nominee holder, and desire to tender their shares of MTGE common stock pursuant to the offer, should instruct the nominee holder to do so prior to the expiration date.

Exchange of Shares; Delivery of Cash and Annaly Shares (Page 62)

Table of Contents

Upon the terms and subject to the satisfaction or waiver of the conditions of the offer (including, if the offer is extended or amended, the terms and conditions of any extension or amendment), promptly following the expiration date, the Offeror will accept for exchange, and will exchange, all shares of MTGE common stock validly tendered and not validly withdrawn prior to the expiration date.

Elections and Proration (Page 57)

By indicating their elections in the applicable section of the letter of election and transmittal, MTGE common stockholders may elect to receive the mixed consideration, the all-cash consideration or the all-stock consideration in exchange for each share of MTGE common stock validly tendered and not validly withdrawn pursuant to the offer, subject in each case to the election procedures and, in the case of elections of the all-cash consideration or the all-stock consideration, to the proration procedures described in this document and the related letter of election and transmittal. If a MTGE common stockholder decides to change its election after tendering its shares of MTGE common stock, it must first validly withdraw the tendered shares of MTGE common stock and then re-tender (and not validly withdraw) the shares prior to the expiration date, with a new letter of election and transmittal that indicates the revised election. MTGE common stockholders who tender their shares of MTGE common stock in the offer but who do not make a valid election will receive the mixed consideration for their shares of MTGE common stock.

Certain Legal Matters; Regulatory Approvals (Page 71)

Annaly and MTGE conduct operations in a number of jurisdictions where regulatory filings or approvals may be required or advisable in connection with the completion of the offer and the merger. In particular, because certain subsidiaries of MTGE are borrowers under loans insured by HUD or serviced loans insured by the U.S. Federal Housing Administration, an office of HUD, or owned by Fannie Mae, the approval of HUD and Fannie Mae may be required to close the offer. Additionally, it is a condition to the closing of the offer that these regulatory approvals, if required, are obtained, which condition may be waived by Annaly in its sole discretion. Annaly and MTGE have initiated a process to obtain these approvals.

Annaly has been advised that the offer and the merger are exempt from the pre-notification and waiting period requirements of the Hart-Scott-Rodino Antitrust Improvements Act of 1976 (the HSR Act). Therefore, we are not attempting to comply with those requirements. The fact that a transaction is exempt from the requirements of the HSR Act does not preclude the Department of Justice or the Federal Trade Commission from seeking to prevent the transaction on the ground that it violates the United States antitrust laws. However, we do not believe that the offer or the merger will be viewed as violating the antitrust laws.

Interests of Certain Persons in the Offer and the Merger (Page 72)

In considering the recommendation of the MTGE board of directors that you accept the offer and tender your shares of MTGE common stock to the Offeror pursuant to the offer, you should be aware that the MTGE directors may have interests that are different from, or in addition to, the interests of MTGE stockholders generally, including the treatment of outstanding restricted stock units held by directors pursuant to the merger agreement, effective as of the closing of the transactions contemplated by the merger agreement.

Management Agreement Termination (Page 73)

In connection with the execution of the merger agreement, MTGE, MTGE TRS, LLC, the MTGE external manager, and Annaly entered into an amendment to the amended and restated MTGE management agreement dated as of July 1, 2016 (the MTGE management agreement), which provides that one month following completion of the merger, the MTGE management agreement will terminate (such time, the termination time). Pursuant to the termination provisions set forth in the MTGE management agreement, in the event the MTGE external manager is terminated without cause, it is entitled to a termination fee equal to three times the average annual management fee earned by the MTGE external manager during the 24-month period immediately preceding the most recently completed month prior to the termination. Accordingly, MTGE will pay the MTGE external manager \$36,659,461.54 million upon the

completion of the merger, and \$5,000,000 at the termination

time (for an aggregate termination fee of \$41,659,461.54). The amendment provides that in addition to the termination fee, MTGE will pay the MTGE external manager for accrued management fees through and until the termination time and will reimburse the MTGE external manager for expenses it incurs prior to the termination time in the ordinary course of business and consistent with past practice to the extent reimbursable pursuant to the MTGE management agreement. In addition, the indemnification obligations included in the management agreement shall survive the termination. Subject to completion of the transactions contemplated by the merger agreement, Annaly has guaranteed payment of the termination fee and amounts that become owed to the Manager under the management agreement.

The amendment to the MTGE management agreement also provides that the MTGE external manager may not intentionally and knowingly take any action, directly or indirectly, that is prohibited to be taken by MTGE representatives as it relates to such representatives non-solicitation obligations under the merger agreement, and the MTGE external manager is required to otherwise comply with the non-solicitation obligations applicable to MTGE representatives under the merger agreement. However, to the extent that MTGE is permitted to, and in fact does, engage in discussions or negotiations regarding an acquisition proposal in accordance with the merger agreement, the MTGE external manager may assist MTGE in such discussions or negotiations. The MTGE external manager further agreed that, other than those actions that MTGE is permitted to take under the merger agreement, or as required by law, it will not intentionally and knowingly take any action that would reasonably be expected to cause any of the conditions to the offer or the merger to fail to be satisfied.

Source and Amount of Funds (Page 74)

The offer and the merger are not conditioned upon any financing arrangements or contingencies.

Annaly estimates the aggregate amount of cash consideration required to purchase the outstanding shares of MTGE common stock and consummate the merger will be approximately \$450.1 million, plus related fees and expenses. Annaly anticipates that the funds needed to complete the transactions will be derived from available cash on hand. Neither Annaly nor the Offeror have any specific alternative financing plans in connection with the offer or the merger. See The Offer Source and Amount of Funds.

Dissenters Rights (Page 66)

No appraisal rights, rights of objecting stockholders or dissenters rights are available in connection with the offer or the merger. See The Offer Purpose of the Offer and the Merger; Dissenters Rights.

Comparative Market Price and Dividend Matters (Page 98)

Annaly common stock is listed on the NYSE under the symbol NLY and MTGE common stock is listed on Nasdaq under the symbol MTGE. On May 1, 2018, the trading day prior to public announcement of the merger agreement, the closing price per share of MTGE common stock on Nasdaq was \$18.15, and the closing price per share of Annaly common stock on the NYSE was \$10.34. On May 15, 2018, the most recent trading date prior to the mailing of this document, the closing price per share of MTGE common stock on Nasdaq was \$19.65, and the closing price per share of Annaly common stock on the NYSE was \$10.33.

The market value of the stock portion of the common transaction consideration will change as the market value of Annaly common stock fluctuates during the offer period and thereafter. MTGE common stockholders should obtain current market quotations for MTGE common stock and Annaly common stock before deciding whether to tender their shares of MTGE common stock in the offer and before electing the form of common transaction consideration they wish to receive. See Comparative Market Price and Dividend Matters.

Ownership of Annaly Common Stock After the Offer and the Merger (Page 66)

Annaly estimates that former MTGE common stockholders would own, in the aggregate, approximately 4% of the shares of Annaly common stock outstanding after the merger. For a detailed discussion of the assumptions on which this estimate is based, see The Offer Ownership of Annaly Common Stock After the Offer and the Merger.

Comparison of Stockholders Rights (Page 149)

The rights of Annaly common stockholders are different in some respects from the rights of MTGE common stockholders. Therefore, MTGE common stockholders who become Annaly common stockholders as a result of the offer and/or the merger will have different rights once they become Annaly common stockholders. The differences are described in more detail under Comparison of Stockholders Rights.

Material U.S. Federal Income Tax Consequences (Page 100)

The offer and the merger, taken together, are intended to qualify as a reorganization within the meaning of Section 368(a) of the Code. If the offer and the merger, taken together, qualify as a reorganization within the meaning of Section 368(a) of the Code, the U.S. federal income tax consequences U.S. holders (as defined herein) of MTGE common stock who receive shares of Annaly common stock and/or cash in exchange for their shares pursuant to the offer and/or the merger generally will be as follows:

if a MTGE stockholder receives solely shares of Annaly common stock in exchange for its shares of MTGE common stock, such stockholder generally will not recognize any gain or loss, except with respect to cash received in lieu of a fractional share of Annaly common stock;

if a MTGE stockholder receives solely cash in exchange for its shares of MTGE common stock, such stockholder generally will recognize gain or loss equal to the difference between the amount of cash received and the stockholder s tax basis in its shares surrendered; and

if a MTGE stockholder receives a combination of Annaly common stock and cash in exchange for its shares of MTGE common stock, such stockholder generally will recognize gain (but not loss) in an amount equal to the lesser of (1) the sum of the cash and the fair market value of the Annaly common stock received, minus the stockholder s tax basis in its shares surrendered, and (2) the amount of cash received.

Non-U.S. holders (as defined herein) who receive a combination of Annaly common stock and cash in exchange for shares of MTGE common stock pursuant to the offer and/or the merger may be subject to U.S. withholding tax with respect to the cash consideration.

Each MTGE stockholder should read the discussion under Material U.S. Federal Income Tax Consequences and should consult its own tax advisor regarding the U.S. federal, state, local and non-U.S. tax consequences of the offer and the merger to such stockholder in light of its particular circumstances.

Accounting Treatment (Page 74)

In accordance with accounting principles generally accepted in the United States (GAAP), Annaly will account for the acquisition of shares through the offer and the merger under the acquisition method of accounting for business combinations.

1	6

Questions about the Offer and the Merger

Questions or requests for assistance or additional copies of this document may be directed to the information agent at the telephone number and addresses set forth below. Stockholders may also contact their broker, dealer, commercial bank, trust company or other nominee for assistance concerning the offer.

The information agent for the offer is:

D.F. King & Co., Inc.

48 Wall Street, 22nd Floor

New York, New York 10005

Bankers & Brokers Call: (212) 269-5550

All Others Call Toll-Free: (800) 669-5550

SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF ANNALY

The following table sets forth summary consolidated financial data for Annaly as of and for each of the five years ended December 31, 2017, 2016, 2015, 2014 and 2013 and as of and for each of the three months ended March 31, 2018 and 2017. All references to fiscal years, unless otherwise noted, refer to the twelve-month fiscal year.

The summary consolidated financial data as of December 31, 2017 and 2016, and for the years ended December 31, 2017, 2016 and 2015, were derived from Annaly s audited consolidated financial statements included in its Annual Report on Form 10-K for the period ended December 31, 2017, previously filed with the SEC on February 16, 2018 and incorporated by reference into this document. The summary consolidated financial data as of December 31, 2015, 2014 and 2013, and for the years ended December 31, 2014 and 2013, were derived from Annaly s audited consolidated financial statements not included in or incorporated by reference into this document. The summary consolidated financial data as of and for the three months ended March 31, 2018 was derived from Annaly s unaudited consolidated financial statements included in its Quarterly Report on Form 10-Q for the period ended March 31, 2018, previously filed with the SEC on May 3, 2018 and incorporated by reference into this document. The summary consolidated financial data as of and for the three months ended March 31, 2017 was derived from Annaly s unaudited consolidated financial statements included in its Quarterly Report on Form 10-Q for the period ended March 31, 2018, previously filed with the SEC on May 3, 2018 and incorporated by reference into this document. The summary consolidated financial statements included in its Quarterly Report on Form 10-Q for the period ended March 31, 2017, previously filed with the SEC on May 5, 2017 and which is not included in or incorporated by reference into this document.

Such financial data should be read together with, and is qualified in its entirety by reference to, Annaly s historical consolidated financial statements and the accompanying notes and the Management s Discussion and Analysis of Financial Condition and Results of Operations, which are set forth in the above-referenced Annual Report on Form 10-K.

in thousands,	, For the Three Months Ended March 31,							For the Years Ended December 31,							
re data)		2018	,	2017		2017		2016		2015	2014				
ent of														20	
ons Data:															
	\$	879,487	\$	587,727	\$	2,493,126	\$	2,210,951	\$	2,170,697	\$	2,632,398	\$	2,9	
expense		367,421		198,425		1,008,354		657,752		471,596		512,659		e	
rest income		512,066		389,302		1,484,772		1,553,199		1,699,101		2,119,739		2,2	
l and ed gains						100 100				(1.001.075)		(2 = 24 2 2 2 2			
(1)		844,689		74,265		199,493		84,204		(1,021,351)		(2,791,399)		1,5	
come (loss)		34,023		31,646		115,857		44,144		(13,717)		44,044			
tal general inistrative				FQ 0.000						000 010		000.000			
		62,510		53,828		224,124		250,356		200,240		209,338		2	
(loss) before taxes and from equity investment		1 200 200		441.005		1 575 000		1 401 101		462 702		(026.05.1)		~ ~ ~	
ite		1,328,268		441,385		1,575,998		1,431,191		463,793		(836,954)		3,7	
come taxes		564		977		6,982		(1,595)		(1,954)		5,325			
ome (loss)		1,327,704		440,408		1,569,016		1,432,786		465,747		(842,279)		3,7	
ome (loss) able to rolling															
		(96)		(103)		(588)		(970)		(809)		(196)			
ome (loss) able to		1,327,800		440,511		1,569,604		1,433,756		466,556		(842,083)		3,7	
ds on		1,527,600		J10,J11		1,507,004		1,755,750		T00,550		(0+2,003)		5,1	
	\$	33,766	\$	23,473	\$	109,635	\$82,260		\$	71,968	\$	71,968	\$		
ome (loss) e (related) to															
lders		1,294,034		417,038		1,459,969		1,351,496		394,588		(914,051)		3,6	

	Edgar Filing: ANNALY CAPITAL MANAGEMENT INC - Form 424B3													
ome (loss) per vailable) to common lders:														
	\$	1.12	\$	0.41	\$	1.37	\$	1.39	\$	0.42	\$	(0.96)	\$	
	\$	1.12	\$	0.41	\$	1.37	\$	1.39	\$	0.42	\$	(0.96)	\$	
ed average of common utstanding:														
	1	,159,617,848		1,018,942,746		1,065,923,652		969,787,583		947,062,099	Ģ	947,539,294	9	47,3
	1,160,103,185			1,019,307,379		1,066,351,616		970,102,353		947,276,742	Ç	947,539,294	9	95,5
inancial														
sets	\$	100,382,233	\$	84,658,957	\$	101,760,050	\$	87,905,046	\$	75,190,893	\$	88,355,367	\$	81,9
uity	\$	13,942,935	\$	12,647,349	\$	14,871,573	\$	12,575,972	\$	11,905,922	\$	13,333,781	\$	12,4
ds declared mon share		0.30		0.30		1.20		1.20		1.20		1.20		
earnings to arges ⁽¹⁾⁽²⁾		4.11x		2.38x		2.06x		2.16x		1.36x		0.32x		
earnings to d fixed and preferred														
vidends ⁽¹⁾⁽²⁾		3.88x		2.28x		1.98x		2.08x		1.34x		0.36x		

ar Filing: ANNALY CARITAL MANAGEMENT INC Earm 40400

(1) Includes unrealized gains (losses) on investments and/or derivatives.

(2) Fixed charges include realized gains (losses) on interest rate swaps.

19

SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF MTGE

The following table sets forth summary consolidated financial data for MTGE as of and for each of the five years ended December 31, 2017, 2016, 2015, 2014 and 2013 and as of and for each of the three months ended March 31, 2018 and 2017. All references to fiscal years, unless otherwise noted, refer to the twelve-month fiscal year.

The summary consolidated financial data as of December 31, 2017 and 2016, and for the years ended December 31, 2017, 2016 and 2015 was derived from MTGE s audited consolidated financial statements included in its Annual Report on Form 10-K for the period ended December 31, 2017, previously filed with the SEC on February 27, 2018 and incorporated by reference into this document. The summary consolidated financial data as of December 31, 2013, and for the years ended December 31, 2014 and 2013, were derived from MTGE s audited consolidated financial statements not included in or incorporated by reference into this document. The summary consolidated financial data as of and for the three months ended March 31, 2018 was derived from MTGE s unaudited consolidated financial statements included in its Quarterly Report on Form 10-Q for the period ended March 31, 2018, previously filed with the SEC on May 10, 2018 and incorporated by reference into this document. The summary consolidated financial statements included in its Quarterly Report on Form 10-Q for the period ended March 31, 2017, previously filed with the SEC on May 10, 2018 and incorporated by reference into this document. The summary consolidated financial statements included in its Quarterly Report on Form 10-Q for the period ended March 31, 2017, previously filed with the SEC on May 9, 2017 and which is not included or incorporated by reference into this document.

Such financial data should be read together with, and is qualified in its entirety by reference to, MTGE s historical consolidated financial statements and the accompanying notes and the Management s Discussion and Analysis of Financial Condition and Results of Operations, which are set forth in the above-referenced Annual Report on Form 10-K.

	For the Qua Marc			For the Year Ended December 31,								
	2018	2017	2017	2016	2015	· · · · · · · · · · · · · · · · · · ·						
		(all n	umbers in th	ousands, ex	cept per shar	e data)						
Statement of operations												
data:												
Interest income	\$ 39,630	\$ 33,757	\$143,648	\$147,770	\$ 174,118	\$183,358	\$ 255,699					
Interest expense	(17,053)	(10,165)	(52,526)	(39,582)	(30,800)	(28,631)	(38,754)					
Net interest income	22,577	23,592	91,122	108,188	143,318	154,727	216,945					
Net servicing loss	(200)	(2,427)	(5,337)	(17,690)	(17,367)	(15,213)	(4,139)					
Net healthcare investment												
income	1,965	662	5,862	659								
Other gains (losses)	(40,742)	22,069	90,158	(23,844)	(139,062)	45,265	(270,676)					
Expenses	(4,967)	(5,095)	(21,029)	(23,767)	(25,205)	(25,362)	(25,921)					
Benefit from (provision												
for) income tax, net			550	125	(42)	(238)	(679)					
Net Income (loss)	(21,367)	38,801	161,326	43,671	(38,358)	159,179	(84,470)					
Dividend on preferred												
stock	(1,117)	(1,117)	(4,468)	(4,468)	(4,468)	(2,718)						

Edgar Filing: ANNALY CAPITAL MANAGEMENT INC - Form 424B3										
Noncontrolling interest in net income	5	(2)	9	(2)						
Net income (loss) to common shareholders	\$ (22,479)	\$ 37,682	\$ 156,867	\$ 39,201	\$ (42,826)	\$ 156,461	\$ (84,470)			
Net income (loss) per common share basic and diluted	\$ (0.49)	\$ 0.82	\$ 3.42	\$ 0.85	\$ (0.85)	\$ 3.06	\$ (1.59)			
Weighted average common shares basic	45,810	45,798	45,805	46,005	50,506	51,176	53,015			
Weighted average common shares diluted	45,822	45,806	45,811	46,008	50,519	51,192	53,015			
Dividends declared per common share	\$ 0.50	\$ 0.45	\$ 1.85	\$ 1.60	\$ 1.80	\$ 2.60	\$ 3.05			

	For the Quarter Ended March 31,					For the Year Ended December 31,								
	2018 2017					2017	L	2014		2013				
					um	bers in th	ous	ands, exce	pt p	2015 oer share o	data))	-	
Balance Sheet														
Data														
Agency securities, at fair value	\$3	3,660,403	\$2	2,997,725	\$:	3,758,181	\$2	2,803,168	\$3	,217,252	\$4,	384,139	\$5,	641,682
Non-agency														
securities, at fair														
value	\$	833,681	\$	948,495	\$	872,084	\$1	1,134,469	\$1	,557,671	\$1,	168,834	\$1,	011,217
Healthcare real														
estate assets	\$	282,816	\$,	\$	262,833	\$	93,266	\$		\$		\$	
Total assets	\$5	5,857,808	\$ 5	5,619,182	\$:	5,953,036	\$4	1,797,155	\$5	,482,402	\$7,	031,252	\$8,	397,865
Financing														
arrangements		3,945,422		3,271,342		4,050,219		3,311,043		,107,615		423,630		158,192
Total liabilities	\$4	4,897,212	\$ 4	4,669,039	\$ 4	4,947,414	\$3	3,864,110	\$4	,491,290	\$5,	855,283	\$7,	295,145
Total stockholders	.		<i>•</i>	0 40 0 0 -	1	.	
equity	\$	959,797	\$	949,837	\$	1,005,107	\$	932,730	\$	991,112	\$1,	175,969	\$1,	102,720
Net asset value per	¢	10 70	¢	10.54	¢	20.75	¢	10.17	¢	10.00	¢	01.01	¢	01.47
common share	\$	19.76	\$	19.54	\$	20.75	\$	19.17	\$	19.66	\$	21.91	\$	21.47
Other Financial														
Data Dividends declared														
per common share	\$	0.50	\$	0.45	\$	1.85	\$	1.60	\$	1.80	\$	2.60	\$	3.05
Ratio of earnings	φ	0.50	φ	0.45	φ	1.03	φ	1.00	φ	1.00	Φ	2.00	Φ	5.05
to fixed charges $^{(1)}$		(0.3)x		4.7x		4.0x		2.0x		(0.3)x		6.2x		(1.2)x
Ratio of earnings		(0.5)X		4./A		4.0X		2.0X		(0.3)X		0.2X		(1.2)X
to combined fixed														
charges and														
preferred stock														
dividends $^{(2)(3)}$		(0.2)x		4.3x		3.7x		1.9x		(0.2)x		5.8x		(1.2)x
										()				. ,

(1) Fixed charges consist of primarily interest expense, as defined under U.S. generally accepted accounting principles, on our repurchase agreements.

(2) Fixed charges consist of primarily interest expense, as defined under U.S. generally accepted accounting principles, on our repurchase agreements.

(3) There were no shares of preferred stock outstanding prior to 2014, therefore, there are no amounts for preferred stock dividends included in the above calculations.

COMPARATIVE PER SHARE DATA

(UNAUDITED)

The following table reflects historical information about basic and diluted earnings per share and cash dividends per share of common stock for the three months ended March 31, 2018 and the fiscal year ended December 31, 2017, on a historical basis. In addition, the following table reflects historical information about book value per share for the three months ended March 31, 2018, on a historical basis.

MTGE stockholders should read the information presented in the following table together with the historical financial statements of Annaly and MTGE and the related notes which are incorporated herein by reference.

	nnaly storical		TGE torical
Net income (loss) per share attributable to common			
stockholders for the three months ended March 31,			
2018:			
Basic earnings per share	\$ 1.12	(\$	0.49)
Diluted earnings per share	\$ 1.12	(\$	0.49)
Cash dividends declared per share common stock for			
the three months ended March 31, 2018	\$.30	\$	0.50
Book value per share as of March 31, 2018	\$ 10.53	\$	19.76
Net income per share attributable to common			
stockholders for the year ended December 31, 2017:			
Basic earnings per share	\$ 1.37	\$	3.42
Diluted earnings per share	\$ 1.37	\$	3.42
Cash dividends declared per share of common stock			
for the year ended December 31, 2017	\$ 1.20	\$	1.85

RISK FACTORS

MTGE stockholders should carefully read this document and the other documents referred to or incorporated by reference into this document, including in particular the following risk factors, in deciding whether to tender their shares pursuant to the offer. Additional risks and uncertainties not presently known to Annaly or MTGE or that are not currently believed to be important may have adverse effects on the offer, the merger and the combined company.

Risk Factors Relating to the Offer and the Merger

The stock portion of the common transaction consideration is fixed and will not be adjusted. Because the market price of Annaly common stock may fluctuate, MTGE common stockholders cannot be certain of the market value of the common transaction consideration they will receive in exchange for their shares of MTGE common stock in connection with the transactions or the value of Annaly common stock thereafter.

In connection with the offer and the merger, MTGE common stockholders will receive, at their election, the mixed consideration, the all-cash consideration or the all-stock consideration. The mixed consideration and all-stock consideration provide for a fixed number of shares of Annaly common stock for each share of MTGE common stock. In addition, MTGE common stockholders who make the all-cash election or the all-stock election will be subject to proration so that approximately 50.0% of the aggregate consideration in the offer and merger will be paid in shares of Annaly common stock and approximately 50.0% of the aggregate consideration in the offer will be paid in cash. As a result, a portion of the consideration that MTGE common stockholders who make the all-cash election will receive in the offer and the merger may be a fixed number of shares of Annaly common stock. Because the number of shares of Annaly common stock being offered as part of the portion of the common transaction consideration will not vary based on the market value of Annaly common stock, the market value of Annaly common stock will vary based on the price of such stock at the time you receive the common transaction consideration. The market price of Annaly common stock may decline after the date of this document, after you tender your shares and/or after the offer and the merger are completed.

A decline in the market price of Annaly common stock could result from a variety of factors, some of which are beyond Annaly s control, including, among other things, an increase in interest rates, the possibility that Annaly may not achieve the expected benefits of the acquisition of MTGE as rapidly or to the extent anticipated, MTGE s portfolio and businesses may not perform as anticipated following the transactions, the effect of Annaly s acquisition of MTGE on Annaly s financial results may not meet the expectations of Annaly, financial analysts or investors, or the addition and integration of MTGE s business may be unsuccessful, take longer or be more disruptive than anticipated, as well as numerous factors affecting Annaly and its businesses that are unrelated to MTGE, including the price of the securities and loans in Annaly s existing portfolio.

Because the offer will not be completed until certain conditions have been satisfied or waived, a significant period of time may pass between the commencement of the offer, the time you tender your shares and the time that the Offeror accepts your shares for payment. See The Offer Conditions of the Offer. Therefore, at the time you tender your shares of MTGE common stock pursuant to the offer, you will not know the exact market value of the stock portion of the common transaction consideration that will be issued if the Offeror accepts such shares for payment.

See Comparative Market Price and Dividend Matters of this document. You are urged to obtain current market quotations for MTGE common stock and Annaly common stock.

MTGE common stockholders may not receive all consideration in the form elected.

Table of Contents

MTGE common stockholders electing to receive either the all-cash consideration or the all-stock consideration in the offer will be subject to proration so that approximately 50.0% of the aggregate consideration

in the offer will be paid in shares of Annaly common stock, and approximately 50.0% of the aggregate consideration in the offer will be paid in cash. Similarly, MTGE common stockholders electing to receive either the all-cash consideration or the all-stock consideration in the merger will be subject to proration so that approximately 50.0% of the aggregate consideration in the merger will be paid in shares of Annaly common stock, and approximately 50.0% of the aggregate consideration in the merger will be paid in cash. Accordingly, some of the consideration a MTGE common stockholder receives in the offer or the merger may differ from the type of consideration selected and such difference may be significant. This may result in, among other things, tax consequences that differ from those that would have resulted if the MTGE common stockholder had received solely the form of consideration elected. A discussion of the proration mechanism can be found under the heading The Offer Elections and Proration and a discussion of the material U.S. federal income tax consequences of the offer and the merger can be found under Material U.S. Federal Income Tax Consequences.

The offer remains subject to conditions that neither Annaly nor MTGE can control.

The offer is subject to conditions, including that at least a majority of the outstanding shares of MTGE common stock have been validly tendered in the offer (and not validly withdrawn), receipt of required regulatory approvals, lack of legal prohibitions, no material adverse effect (with such term as defined in the merger agreement and described under the section entitled Merger Agreement Material Adverse Effect) having occurred with respect to MTGE, Annaly and their respective subsidiaries, the truth and accuracy of MTGE s and Annaly s and the Offeror s representations and warranties made in the merger agreement, subject to specified materiality standards, MTGE, Annaly and the Offeror being in material compliance with their covenants under the merger agreement, the listing of the shares of the Annaly common and Series H preferred stock to be issued in the offer and the merger being authorized for listing on the NYSE, the receipt of opinions by each of Annaly and MTGE from their respective legal counsel to the effect that the offer and the merger, taken together, will qualify as a reorganization within the meaning of Section 368(a) of the Code and that each of Annaly and MTGE (as applicable) meet and, in the case of Annaly, will continue to meet after closing of the merger, the requirements for qualification as a REIT under the Code, and the registration statement on Form S-4 of which this document is a part, becoming effective. There are no assurances that some or all of the conditions to the offer will be satisfied or that the conditions will be satisfied in the time frame expected. If the conditions to the offer are not met, then Annaly may, subject to the terms and conditions of the merger agreement, allow the offer to expire, or amend or extend the offer. See The Offer Conditions of the Offer for a discussion of the conditions to the offer.

MTGE common stockholders who receive Annaly common stock in the offer will become Annaly common stockholders. Annaly common stock may be affected by different factors and Annaly common stockholders will have different rights than MTGE common stockholders.

Upon consummation of the offer, MTGE common stockholders receiving shares of Annaly common stock will become common stockholders of Annaly. Annaly s business differs from that of MTGE, and Annaly s results of operations and the trading price of Annaly common stock may be adversely affected by factors different from those that would affect MTGE s results of operations and stock price.

In addition, holders of shares of Annaly common stock will have rights as Annaly common stockholders that differ from the rights they had as MTGE common stockholders before the offer or the merger. For a detailed comparison of the rights of Annaly common stockholders to the rights of MTGE common stockholders, see Comparison of Stockholders Rights.

The receipt of shares of Annaly common stock in the offer and/or the merger may be taxable to MTGE common stockholders.

The offer is conditioned upon the receipt of an opinion by each of Annaly and MTGE from their respective legal counsel to the effect that the offer and the merger, taken together, will qualify as a reorganization within

the meaning of Section 368(a) of the Code. However, if the offer and the merger are not treated as component parts of an integrated transaction for U.S. federal income tax purposes, if the merger is not completed or if the transaction otherwise fails to qualify as a reorganization within the meaning of Section 368(a) of the Code, the exchange of shares of MTGE common stock for shares of Annaly common stock in the offer and/or the merger will be a taxable transaction for U.S. federal income tax purposes.

MTGE common stockholders should consult their tax advisors regarding the tax consequences to them of the offer and the merger, including any U.S. federal, state, local, non-U.S. or other tax consequences, and any tax return filing or other reporting requirements, in light of their particular circumstances.

Risk Factors Relating to Annaly as the Combined Company

Annaly, as the combined company, may fail to realize all of the anticipated benefits of the merger or those benefits may take longer to realize than expected.

The full benefits of the transactions may not be realized as expected or may not be achieved within the anticipated time-frame, or at all. Failure to achieve the anticipated benefits of the transactions could adversely affect Annaly s results of operations or cash flows, cause dilution to the earnings per share or book value per share of Annaly, decrease or delay the expected accretive effect of the transactions, and negatively impact the price of Annaly common stock.

In addition, Annaly and MTGE will be required to devote significant attention and resources prior to closing to prepare for the post-closing operation of Annaly, as the combined company, and Annaly will be required post-closing to devote significant attention and resources to successfully integrate the MTGE portfolio and operating businesses into the existing Annaly structure. In particular, prior to the acquisition, Annaly will have limited experience managing MTGE s healthcare real estate assets. These businesses present additional regulatory constraints and pose operational risks different from those that Annaly has successfully managed in the past. This integration process, coupled with managing new business lines, may disrupt Annaly s businesses and, if ineffective, would limit the anticipated benefits of the merger and could adversely affect Annaly s results of operations or cash flows, cause dilution to the earnings per share or book value per share of Annaly, decrease or delay the expected accretive effect of the transactions, and negatively impact the price of Annaly common stock.

Annaly and MTGE will incur direct and indirect costs as a result of the offer and the merger.

Annaly and MTGE will incur substantial expenses in connection with and as a result of completing the offer and the merger and, following the completion of the merger, Annaly expects to incur additional expenses in connection with integrating the MTGE portfolio and operating businesses into the existing Annaly structure. Factors beyond Annaly s control could affect the total amount or timing of these expenses, many of which, by their nature, are difficult to estimate accurately.

REITs are subject to a range of complex organizational and operational requirements.

To qualify as a REIT, Annaly must distribute with respect to each taxable year at least 90% of its net income (excluding capital gains) to its stockholders. A REIT must also meet certain other requirements, including with respect to the nature of its income and assets, and the ownership of its stock. For any taxable year that Annaly fails to qualify as a REIT, it will not be allowed a deduction for dividends paid to its stockholders in computing its net taxable income and thus would become subject to U.S. federal, state and local income tax as if it were a regular taxable corporation. In such an event, Annaly could be subject to potentially significant tax liabilities. Unless entitled to relief under

certain statutory provisions, Annaly would also be disqualified from treatment as a REIT for the four taxable years following the year in which it lost its qualification. If Annaly failed to qualify as a REIT, the market price of its common stock may decline, and Annaly may need to reduce substantially the amount of distributions to its stockholders because of its increased tax liability.

Risks Related to Annaly s Business

You should read and consider the risk factors specific to Annaly s business that will also affect Annaly, as the combined company, after the merger. These risks are described in Part I, Item 1A of Annaly s Annual Report on Form 10-K for the fiscal year ended December 31, 2017, and in other documents that are incorporated by reference into this document. See Where To Obtain More Information for more detail on the information incorporated by reference in this document.

Risks Related to MTGE s Business

You should read and consider the risk factors specific to MTGE s business that will also affect Annaly, as the combined company, after the merger. These risks are described in Part I, Item 1A of MTGE s Annual Report on Form 10-K for the fiscal year ended December 31, 2017, and in other documents that are incorporated by reference into this document. See Where To Obtain More Information for more detail on the information incorporated by reference in this document.

FORWARD-LOOKING STATEMENTS

Information both included and incorporated by reference in this document may contain forward-looking statements, concerning, among other things, Annaly s and MTGE s outlook, financial projections and business strategies, all of which are subject to risks, uncertainties and assumptions. These forward-looking statements are identified by their use of terms such as intend, should, believe. could, estimate, plan, may, will, anticipate, forecast, ex potential, opportunity, project and similar terms. These statements are based on certain assumptions and analyses that we believe are appropriate under the circumstances. Should one or more of these risks or uncertainties materialize, or should the assumptions prove incorrect, actual results may differ materially from those expected, estimated or projected. Management believes that these forward-looking statements are reasonable. However, we cannot guarantee that we actually will achieve these plans, intentions or expectations, including completing the offer and the merger on the terms summarized in this document. Forward-looking statements speak only as of the date they are made, and we undertake no obligation to publicly update or revise any of them in light of new information, future events or otherwise. Factors that could have a material adverse effect on Annaly s operations and future prospects or the consummation of the offer and the merger include, but are not limited to:

failure to satisfy the conditions to consummate the offer and the merger;

the occurrence of any event, change or other circumstance that could give rise to the termination of the merger agreement;

the failure of the offer or the merger to close in a timely manner or at all for any other reason, including failure to receive required regulatory approvals;

the amount of the costs, fees, expenses and charges related to the offer and the merger (collectively, the transactions);

the ability to successfully integrate MTGE following completion of the transactions;

failure to realize the expected benefits of the transactions in a timely manner or at all;

effects of the pendency of the transactions on relationships with employees and business partners;

general economic and business conditions;

global economic growth and activity;

industry conditions;

changes in interest rates, interest rate spreads and the yield curve;

changes in prepayment rates;

the availability of mortgage-backed securities and other securities for purchase;

the availability of financing, and, if available, the terms of any financing;

the ability to maintain qualification as a REIT for U.S. federal income tax purposes;

the ability to maintain an exemption from registration under the Investment Company Act of 1940, as amended;

changes in program requirements of HUD or Fannie Mae; and

changes in laws or regulations.

These risks and uncertainties, along with the risk factors discussed or referenced under Risk Factors in this document, should be considered in evaluating any forward-looking statements contained in this document. All forward-looking statements speak only as of the date of this document. All subsequent written and oral forward-looking statements attributable to us or any person acting on our behalf are qualified by the cautionary statements in this section.

THE COMPANIES

Annaly

Annaly is a leading diversified capital manager that invests in and finances residential and commercial assets. Annaly s principal business objective is to generate net income for distribution to its shareholders and to preserve capital through prudent selection of investments and continuous management of its portfolio. Annaly has elected to be taxed as a real estate investment trust, or REIT, for U.S. federal income tax purposes. Annaly is externally managed by Annaly Management Company LLC.

Annaly s common stock trades under the ticker symbol NLY on the NYSE.

The address of Annaly s principal executive offices is 1211 Avenue of the Americas, New York, New York 10036. Annaly s telephone number is (212) 696-0100.

Annaly also maintains an Internet site at www.annaly.com. Annaly s website and the information contained therein or connected thereto shall not be deemed to be incorporated herein, and you should not rely on any such information in making an investment decision.

Offeror

Mountain Merger Sub Corporation, a Maryland corporation, is a wholly owned subsidiary of Annaly. The Offeror is newly formed, and was organized for the purpose of making the offer and consummating the merger. The Offeror has not engaged in any business activities to date and it has no material assets or liabilities of any kind, other than those incidental to its formation and those incurred in connection with the merger agreement and the transactions contemplated thereby, including the offer and the merger.

The Offeror s address is c/o Annaly Capital Management, Inc., 1211 Avenue of the Americas, New York, New York 10036.

MTGE

MTGE Investment Corp. is a REIT that invests in and manages a leveraged portfolio of agency mortgage investments, non-agency mortgage investments and other real estate-related investments. MTGE is externally managed and advised by MTGE Management, LLC, an affiliate of AGNC Investment Corp. (Nasdaq: AGNC).

MTGE s common stock trades under the ticker symbol MTGE on Nasdaq.

The address of MTGE s principal executive offices is 2 Bethesda Metro Center, 12th Floor, Bethesda, Maryland 20814. MTGE s telephone number is (301) 968-9220.

MTGE also maintains an Internet site at www.mtge.com. MTGE s website and the information contained therein or connected thereto shall not be deemed to be incorporated herein, and you should not rely on any such information in making an investment decision.

THE OFFER

General

Annaly and the Offeror are offering, upon the terms and subject to the conditions set forth in this prospectus/offer to exchange and in the accompanying letter of election and transmittal, to exchange for each outstanding share of MTGE common stock validly tendered and not validly withdrawn in the offer:

\$9.82 in cash; and

0.9519 shares of Annaly common stock, together with cash in lieu of any fractional shares of Annaly common stock.

We refer to the above as the mixed consideration.

In lieu of receiving the mixed consideration, each holder of shares of MTGE common stock may elect to receive, for each share of MTGE common stock that it holds, (1) \$19.65 in cash (we refer to this election as the all-cash election and this amount as the all-cash consideration) or (2) 1.9037 shares of Annaly common stock, together with cash in lieu of any fractional shares of Annaly common stock (we refer to this election as the all-stock election and this amount as the all-stock consideration). The mixed consideration, the all-cash consideration and the all-stock consideration (as applicable) will be paid without interest and less any applicable withholding taxes (such consideration, the common transaction consideration).

Each MTGE common stockholder who validly tenders and does not validly withdraw its shares of MTGE common stock in the offer that does not make a valid election will receive the mixed consideration for their shares of MTGE common stock. MTGE common stockholders who make the all-cash election or the all-stock election will be subject to proration so that approximately 50.0% of the aggregate consideration in the offer will be paid in shares of Annaly common stock and approximately 50.0% of the aggregate consideration in the offer will be paid in cash. See The Offer Elections and Proration for a description of the proration procedure.

The purpose of the offer is for Annaly to acquire control of, and ultimately the entire common equity interest in, MTGE. The offer is the first step in Annaly s plan to acquire all of the issued and outstanding shares of MTGE common stock. If the offer is completed, Annaly intends to consummate promptly following the consummation of the offer a merger of MTGE with and into the Offeror, with the Offeror surviving the merger (which we refer to as the merger). The purpose of the merger is for Annaly to acquire all of the issued and outstanding shares of MTGE common stock that it did not acquire in the offer. In the merger, each outstanding share of MTGE common stock that was not acquired by Annaly or the Offeror will be converted into the mixed consideration or, at the election of the holder of such shares, the all-cash consideration or all-stock consideration, subject to proration so that approximately 50.0% of the aggregate consideration in the merger will be paid in shares of Annaly common stock and approximately 50.0% of the aggregate consideration in the merger will be paid in cash.

In addition, in the merger, each outstanding share of MTGE Series A preferred stock will be automatically converted into the right to receive one newly issued share of the Annaly Series H preferred stock, which will have rights, preferences, privileges and voting powers substantially the same as those of the MTGE Series A preferred stock.

After the merger, the MTGE business will be held by the Offeror, and the former MTGE stockholders will no longer have any direct ownership interest in the surviving corporation.

Background of the Offer and the Merger

MTGE is a hybrid mortgage REIT that invests in agency mortgage-backed securities, non-agency mortgage investments and other real estate-related assets, including skilled nursing and senior living facilities operated by third parties.

MTGE is externally managed, meaning that all of its day-to-day operations and affairs are managed by a third-party manager, the MTGE external manager, subject to the oversight and direction of the MTGE board of directors. MTGE has no senior management employees. Rather, under the management agreement between MTGE and the MTGE external manager provides MTGE with executive officers and other services and personnel. The MTGE external manager is wholly owned by AGNC Investment Corp. (AGNC). AGNC owns approximately 5.7% of outstanding MTGE common stock. All but one of the members of the MTGE board of directors during the relevant period were not affiliated with the MTGE external manager or AGNC. Gary Kain, who was MTGE s Chief Executive Officer and a member of the AGNC board of directors. As described in more detail below, all relevant actions and decisions of the MTGE board of directors concerning the transaction were undertaken by and at the direction of the MTGE special committee formed by the MTGE board of directors to review a potential transaction, which was comprised entirely of independent members of the MTGE board of directors not affiliated with the MTGE external of directors to review a potential transaction, which was comprised entirely of independent members of the MTGE board of directors not affiliated with the MTGE external manager or AGNC.

The MTGE board of directors regularly reviews MTGE s operations, financial performance, capital structure, strategic initiatives and product portfolio and pipeline. As part of this ongoing review, the MTGE board of directors has regularly evaluated MTGE s long-term strategy, as well as other strategic alternatives that might be available to enhance stockholder value.

On or about January 11, 2018, David Finkelstein, Chief Investment Officer of Annaly, contacted Steve Abrahams (at the time a member of the MTGE board of directors and an independent director) on an informal basis regarding Annaly s potential interest in a strategic transaction with MTGE. The independent members of the MTGE board of directors discussed the communication from Annaly in an executive session of independent directors at an in person meeting of the MTGE board of directors on January 30, 2018. The independent directors conferred by phone again on February 5, 2018, with a representative of Cooley LLP (Cooley), then outside legal counsel to the independent directors, also participating. The directors determined that the general expression of interest without specific terms was not yet actionable and no response was warranted at that time. They informed Mr. Kain of the contact from Annaly in his capacity as MTGE s CEO and a member of the MTGE board of directors. After Mr. Abrahams resigned from the MTGE board of directors on February 12, 2018, Mr. Finkelstein contacted Julia Coronado, a member of the MTGE board of directors and an independent director, on an informal basis on or about February 16, 2018, regarding Annaly s continuing potential interest in a strategic transaction, but acknowledged that Annaly was free to submit a more specific proposal.

Subsequently, on February 22, 2018, Annaly delivered an unsolicited non-binding written proposal to Randy Dobbs, Chairman of the MTGE board of directors and an independent director, to acquire 100% of MTGE s capital stock based on a valuation of 95% of the book value of MTGE common stock (the Annaly February 22 Proposal). The Annaly February 22 Proposal was based on publicly-available information and subject to completion of customary due diligence and other conditions. Annaly indicated that it was prepared to pay the purchase price in a consideration mix of cash and Annaly common stock to be agreed by the parties. With respect to MTGE s outstanding shares of preferred stock, Annaly proposed to issue a new series of Annaly preferred stock on substantially similar terms. As part of its proposal, Annaly also requested a 30-day exclusive negotiation period.

On February 26, 2018, the independent directors of the MTGE board of directors held a telephonic meeting to further discuss the process for determining whether to continue as an independent company or to pursue a strategic transaction, including the Annaly February 22 Proposal or an alternative transaction. Mr. Kain did not attend or participate in this meeting. Representatives of Cooley also attended this meeting and reviewed with the MTGE independent directors their legal duties in connection with their consideration of acquisition proposals. The

independent directors were cognizant of the fact that AGNC, as the parent entity of the MTGE external manager, could be a logical acquirer of the company and may have interest in making an acquisition proposal if MTGE were to consider a strategic transaction. The independent directors were also mindful that Annaly or

another acquirer would likely plan to terminate MTGE s management relationship with the MTGE external manager without cause in connection with an acquisition and therefore MTGE would be required to pay a termination fee of approximately \$42 million to the MTGE external manager pursuant to the termination provisions of the management agreement. The amount of the termination fee is prescribed in the management agreement and calculated based on three times the average annual management fee for the prior two-year period, and it was thought that the amount of the fee could affect the prices that Annaly or another acquirer might pay relative to AGNC if it were to make a proposal. Given these considerations and the potential conflicts of interest of AGNC and Mr. Kain, the independent directors and Cooley also discussed the formation of a special committee of the MTGE board of directors consisting of the independent directors to evaluate the Annaly February 22 Proposal and other strategic alternatives. As a next step, the independent directors of the MTGE board of directors instructed Mr. Dobbs to seek additional clarification from Annaly about certain terms of the Annaly February 22 Proposal (including the proposed cash-stock consideration mix, treatment of the potential termination fee payable to the MTGE external manager and integration and operational issues). Mr. Dobbs contacted Kevin Keyes, Chairman, Chief Executive Officer and President of Annaly, by email indicating that the board needed additional information before it could act on the proposal and consider a change in strategic direction.

On March 1, 2018, Annaly delivered a presentation to the independent directors of the MTGE board of directors affirming its proposal to acquire MTGE based on a valuation of 95% of a mutually agreed upon book value of MTGE common stock and proposing an acquisition consideration mix of 50% cash and 50% Annaly common stock. Annaly also clarified that it was prepared to pay the full termination fee to the MTGE external manager. After reviewing this presentation, the independent directors of the MTGE board of directors further discussed the process for considering the proposal and alternatives, including the engagement of a financial advisor. Subsequently, Mr. Dobbs contacted Mr. Keyes by email to inform him that the independent directors were evaluating Annaly s proposal.

On March 15, 2018, the MTGE board of directors, including Mr. Kain, unanimously approved the formation of a special committee of independent directors comprised of Mr. Dobbs, Ms. Coronado and Rob Couch (the MTGE special committee) in connection with the consideration of the Annaly February 22 Proposal and strategic alternatives generally. In particular, the MTGE board of directors authorized the MTGE special committee to consider whether to pursue a strategic transaction with Annaly or other interested parties or remain independent, to negotiate the terms and conditions of a possible transaction, to reject or recommend to the MTGE board of directors for approval of a possible transaction, and to engage legal and financial advisors. The MTGE board of directors retained the authority to recommend any strategic transaction to MTGE s stockholders. The MTGE board of directors appointed Mr. Dobbs as Chairman of the MTGE special committee.

On March 22, 2018, the MTGE special committee held a telephonic meeting and conducted separate phone interviews with three investment banks (including Barclays) to act as financial advisor to the MTGE special committee in connection with the consideration of a potential strategic transaction. The MTGE special committee discussed the potential advisor candidates and their industry knowledge and experience and that the MTGE special committee in determining which alternative would best enhance shareholder value. Representatives of Cooley also attended the meeting and discussed with the MTGE special committee its legal duties in connection with the consideration of a potential strategic transaction.

On March 23, 2018, Annaly submitted an updated non-binding written proposal to Mr. Dobbs providing additional detail on the terms on which Annaly would be prepared to acquire MTGE based on its continuing analysis of MTGE (the Annaly March 23 Proposal). Annaly proposed to acquire MTGE in an all-cash transaction for 95% of MTGE s book value per share, which equated to \$19.71 per share based on MTGE s book value per share of \$20.75 as of December 31, 2017. This valuation represented a 10.7% premium to MTGE s share price of \$17.80 as of March 22,

2018. Annaly indicated that the final price would be based on the book value as of a more recent valuation date and indicated a willingness to include Annaly common shares in the mix

of consideration if the MTGE special committee preferred a stock component. Annaly reiterated that it was prepared to proceed and complete due diligence expeditiously. Annaly again requested a 30-day exclusive negotiation period. Subsequent to receipt of the Annaly March 23 Proposal, Mr. Dobbs contacted Mr. Keyes by email to inform him that the independent directors were evaluating Annaly s updated proposal.

During this period the MTGE special committee concluded discussions with the financial advisor candidates, and on March 25, 2018, the MTGE special committee informed Barclays of its decision to engage Barclays as its financial advisor in connection with the consideration of a potential strategic transaction. The MTGE special committee entered into the formal engagement agreement with Barclays on March 29, 2018.

On April 2, 2018, the MTGE special committee held a telephonic meeting, with representatives of Barclays and Cooley also attending, to discuss the Annaly March 23 Proposal. The MTGE special committee, together with input from Cooley, discussed the importance of adequately considering strategic alternatives in addition to the Annaly March 23 Proposal in connection with fulfilling its fiduciary duties. The MTGE special committee authorized Barclays to inform Annaly that it had been retained by the MTGE special committee. The MTGE special committee also instructed Barclays to consider additional prospective acquirers, including AGNC, that may be interested in a strategic transaction.

On April 2, 2018, representatives of Barclays held a telephonic conference call with Mr. Keyes and other representatives of Annaly to clarify certain aspects of the Annaly March 23 Proposal.

On April 6, 2018, the MTGE special committee held a telephonic meeting to further discuss the Annaly March 23 Proposal, with representatives of Barclays and Cooley also attending. Barclays made a presentation to the MTGE special committee to assist its consideration of the Annaly March 23 Proposal. Barclays reviewed MTGE s current positioning and outlook relative to its peer group and provided an initial framework for analyzing MTGE s valuation. Barclays also reviewed reasons for the MTGE board of directors to continue pursuing MTGE s diversification strategy as a stand-alone company and reasons to pursue potential strategic alternatives. Barclays also reviewed the current M&A market for the REIT industry and the potential interest of other parties, including AGNC, in a possible strategic transaction with MTGE. Barclays and Cooley also discussed with the MTGE special committee potential sale process considerations and strategic transaction. The MTGE special committee decided to continue its consideration of the various alternatives and instructed Barclays to continue its analysis of the Annaly March 23 Proposal and consideration of other prospective acquirers.

On April 9, 2018, the MTGE special committee held a telephonic meeting, with representatives of Barclays and Cooley also attending, to continue its consideration of the Annaly March 23 Proposal and alternatives. Barclays made a presentation to the MTGE special committee comparing Annaly s proposal to a hypothetical acquisition by AGNC. The MTGE special committee and Barclays viewed AGNC as a logical acquirer with the ability and potential willingness to pay a higher price for MTGE. Barclays also reviewed various process alternatives, and the MTGE special committee instructed Barclays to solicit additional parties, including AGNC, that were believed to be the most able and willing to pay the highest price for MTGE.

On April 10, 2018, Annaly delivered an updated non-binding written proposal to Mr. Dobbs that increased Annaly s proposed price to approximately 98.0% of the book value of MTGE common stock (which equated to \$20.33 per common share net of expenses based on MTGE s book value per share as of December 31, 2017), subject to due diligence and other conditions (the Annaly April 10 Proposal). The Annaly April 10 Proposal assumed that Annaly would bear the full cost of the \$42 million management termination fee and up to \$20 million of additional transaction expenses. Annaly reiterated that it was prepared to pay 100% of the consideration in cash or include a stock

component in the consideration mix if the MTGE board of directors preferred. Annaly also indicated that its proposal would expire on April 13, 2018 unless the parties entered into an exclusivity agreement prior to that time.

The MTGE special committee held a telephonic meeting on April 11, 2018 to discuss the Annaly April 10 Proposal, with representatives of Barclays and Cooley also attending. Barclays presented its updated valuation analysis of MTGE and Annaly s latest proposal, including the desirability and relative value of a stock consideration component and the potential for a higher after-tax return for stockholders. The MTGE special committee also discussed Annaly s request for exclusivity and the desirability of obtaining an alternative acquisition proposal from AGNC and other potentially interested parties. The MTGE special committee instructed Barclays to contact AGNC again to encourage it to make an alternative proposal, and to reach out to other potential acquirers.

On April 11, 2018, Barclays reached out to two large REITs (Company A and Company B, respectively) whom Barclays considered would be the most interested and able to engage in a potential strategic transaction. Company A initially declined to pursue an acquisition of MTGE based on its relationship with AGNC. Company B expressed an interest in pursuing an acquisition of MTGE, and Barclays encouraged Company B to work to be in a position to respond with a proposal regarding valuation of MTGE by April 13, 2018. Barclays also contacted AGNC again, who indicated that it would respond with a proposed valuation of MTGE common stock by April 13, 2018.

On April 12, 2018, in a follow up call with Barclays, Company A expressed interest in a potential acquisition but did not indicate a proposed valuation and never submitted an acquisition proposal.

On April 12, 2018, AGNC delivered a non-binding written proposal to Barclays to acquire 100% of the shares of MTGE common stock at 95% of the book value of MTGE common stock, which represented a price of \$19.72 per share net of the termination fee and other expenses based on the book value of MTGE common stock as of December 31, 2017 (the AGNC April 12 Proposal). AGNC proposed a consideration mix of cash (20%) and AGNC common stock (80%) for MTGE common stock, with MTGE preferred stock being exchanged for AGNC preferred stock containing substantially similar terms.

On April 13, 2018, Company B verbally indicated that it was considering an offer representing approximately 100% of book value of MTGE common stock based on the latest book value estimate.

On April 13, 2018, the MTGE special committee held a telephonic meeting, with representatives of Barclays and Cooley also attending, to discuss the process and the status of the various proposals. Barclays reviewed and compared the Annaly April 10 Proposal and the AGNC April 12 Proposal. From a financial point of view, the MTGE special committee considered the Annaly April 10 Proposal more favorable to MTGE stockholders based on the higher price and higher mix of cash consideration. Barclays also updated the MTGE special committee on the status of potential proposals from Company A and Company B. Company A had not followed up with an indicative valuation. Barclays noted that the verbal proposal from Company B was based on deal consideration comprised 100% of stock of the acquirer and would likely require a vote of the stockholders of both Company B and MTGE as a condition to closing. The MTGE special committee and Barclays considered the proposal as less favorable to MTGE stockholders. Barclays advised the MTGE special committee that it believed it was unlikely that MTGE would receive an improved offer from Company B in advance of Annaly s deadline to enter into exclusivity.

Based on the status of the various proposals, the MTGE special committee considered the risk probability that Annaly may withdraw its proposal, which the MTGE special committee considered to be the most favorable from a financial point of view, if MTGE did not enter into an exclusivity agreement with Annaly and considered the terms and conditions on which the MTGE special committee may be willing to enter into an exclusive negotiating period with Annaly. The MTGE special committee instructed Barclays to encourage AGNC to increase its price and submit its best offer and to explore an exclusivity arrangement with Annaly based on mutual agreement on certain key terms, including: (1) the diligence and regulatory approval timetables, (2) shortening the exclusivity period to 10-14 days, (3) retaining flexibility for MTGE to elect an all cash transaction or a cash-stock mix based on further analysis and

reverse due diligence, and (4) early termination of exclusivity if Annaly adversely changed its price or other material terms of its proposal.

Later that day, the MTGE special committee held another telephonic meeting, with representatives of Barclays and Cooley also attending. Barclays reported that AGNC indicated a willingness to increase its price to 96.0%-96.5% of book value contingent upon the satisfaction of certain conditions but that was likely its best offer and any final agreement at this price would be contingent on a number of factors, including AGNC s stock price performance. Barclays also reported that Company A and Company B had not submitted any further proposals. Following discussion, the MTGE special committee instructed Barclays and Cooley to attempt to negotiate an acceptable exclusivity agreement with Annaly.

Following that meeting and over the weekend, Barclays and Cooley discussed and negotiated the terms of a confidentiality and exclusivity agreement with Annaly s financial advisors, Wells Fargo Securities (Wells Fargo) and Sandler O Neill & Partners, L.P. (Sandler O Neill), and counsel, Wachtell, Lipton, Rosen & Katz (Wachtell Lipton). On April 16, 2018, MTGE and Annaly signed a confidentiality agreement that included mutual confidentiality restrictions, mutual standstill covenants and a 14-day exclusive negotiating period consistent with the terms authorized by the MTGE special committee. The agreement also included information sharing protocols that would facilitate the MTGE special committee s ability to rely on the MTGE external manager during its strategic review process (and, as necessary, during negotiations with third parties) without sharing confidential information with the MTGE external manager regarding the terms of Annaly s proposal and without sharing with the MTGE external manager reverse due diligence information concerning Annaly. The parties also discussed the potential regulatory approvals required for a transaction between the parties and a process for conducting due diligence and reverse due diligence. Certain employees of the MTGE external manager were identified to assist the MTGE special committee and its advisors in facilitating information sharing and Annaly s due diligence review of MTGE, negotiation of certain operationally-specific sections of the merger agreement, the preparation of disclosure schedules to the merger agreement, and the provision of financial projections based on MTGE s business plan to provide a baseline against which the MTGE special committee, with the assistance of Barclays, could evaluate MTGE s prospects, stand-alone value and potential strategic transactions, including Annaly s proposal.

Barclays opened a data room to Annaly and its advisors on April 16, 2018 containing information for their diligence process.

Later that day, on April 16, 2018, Cooley delivered a first draft of a merger agreement to Wachtell Lipton. Among other terms, the initial draft provided for a two-step transaction with an exchange offer followed by a merger pursuant to which all outstanding common shares would be exchanged for cash or a mix of cash and Annaly common stock to be agreed, all outstanding preferred shares would be exchanged for a new class of Annaly preferred stock on substantially the same terms, the full acceleration of all non-executive director equity awards, and customary deal protection terms and exceptions, including rights for the MTGE board of directors to respond to unsolicited acquisition proposals received after the announcement of a transaction, to change its recommendation to MTGE s stockholders following receipt of a superior proposal and to terminate the merger agreement to accept a superior proposal, subject in certain cases to the payment of a termination fee to Annaly equal to 2.5% of the transaction value of MTGE s common equity.

On April 17, 2018, the MTGE special committee held a telephonic meeting, with representatives of Barclays and Cooley also attending. Barclays and Cooley reported on the execution of the confidentiality and exclusivity agreement and the advisors reviewed the status and timeline of a potential transaction.

On April 17, 2018, Barclays and Cooley received access to a data room from Annaly for the purposes of reverse due diligence. On April 18, 2018, representatives of Barclays, Annaly, Wells Fargo and Sandler O Neill met at Annaly s offices in New York City, with representatives of Cooley and Wachtell Lipton joining telephonically, to discuss their respective due diligence investigations. Representatives of the MTGE external manager also joined telephonically, but

only for the portion of the meeting covering Annaly s due diligence investigation of MTGE.

Between April 18, 2018 and April 23, 2018, Wachtell Lipton and Cooley exchanged drafts of the merger agreement and had various phone calls to discuss the drafts and open issues. Among other issues, Annaly

proposed a higher termination fee (equal to 3.5% of the transaction value of MTGE s common equity) if the MTGE board of directors changed its recommendation or terminated the agreement to accept a superior proposal. Annaly also requested that the MTGE external manager enter into an amendment to the management agreement concurrently with MTGE s entry into a merger agreement to memorialize the parties understanding that the management agreement would be terminated without cause early in connection with the transaction closing.

Also during this time, MTGE and Annaly held a series of due diligence and reverse due diligence calls with their respective advisors. These calls covered, among other topics, financial diligence on MTGE and Annaly, accounting diligence on MTGE and Annaly, diligence on Annaly s financial projections, diligence on MTGE s healthcare real estate assets and diligence on Annaly s commercial real estate assets.

On April 20, 2018, the MTGE special committee had a telephonic meeting, with representatives of Barclays and Cooley also attending. Barclays made a presentation regarding the valuation of Annaly s common stock and how a hypothetical cash stock election could operate and be valued based on a 50%-50% mix and proration. The MTGE special committee and their advisors continued this discussion at a subsequent telephonic meeting on April 22, 2018. They discussed the value to MTGE s stockholders of an all cash transaction versus a mix of cash and stock consideration at various trading prices for Annaly s common stock and the potential desirability of a collar to protect against a drop in the trading price of Annaly s common stock between announcement and closing of a potential transaction. They also discussed Annaly s preference to announce a transaction prior to its quarterly earnings call scheduled for May 3, 2018 and the potential impact of the earnings release and a transaction announcement on Annaly s stock price. Barclays also reported that no new unsolicited proposals had been received from AGNC, Company A, Company B or any other party since MTGE entered into exclusive negotiations with Annaly. After discussion, the MTGE special committee instructed Barclays to communicate to Annaly that MTGE would continue to negotiate expeditiously but would not be unduly rushed and to explore with Annaly s advisors whether Annaly would agree to a stock exchange ratio collar.

On April 24, 2018, the MTGE special committee had a telephonic meeting, with representatives of Barclays and Cooley also attending. Cooley reviewed the material terms of the proposed merger agreement, including the expected timing of the exchange offer and regulatory approvals, the deal protection provisions and termination fee, and the closing conditions. The MTGE special committee and advisors also reviewed the economic terms of Annaly s proposal and expressed a preference for a mix of cash and stock consideration. After discussion, the MTGE special committee instructed Barclays to provide further analysis of a cash-stock election and provided guidance to Cooley on the open merger agreement issues. The MTGE special committee also noted the potential for extending the exclusive negotiation period and determined it would be appropriate to extend for seven days to provide for an additional period of negotiation if needed after Annaly s earnings announcement on May 3, 2018. Later that day, representatives of MTGE and Annaly had a call, with Cooley and Wachtell Lipton (along with regulatory counsel) present, to negotiate the representations and covenants in the merger agreement. That evening, Cooley and Wachtell Lipton continued to discuss the merger agreement and timing of regulatory approvals.

On April 25, 2018 and April 28, 2018, Cooley and Wachtell Lipton had calls with MTGE s and Annaly s outside federal and HUD regulatory counsel, respectively, to discuss required approvals and timelines. Cooley sent Wachtell Lipton a revised draft of the merger agreement later in the day on April 25, 2018.

In the evening of April 25, 2018, MTGE agreed to Annaly s request to extend exclusivity for seven additional days to May 7, 2018.

On April 26, 2018, AGNC delivered an updated unsolicited non-binding written proposal to Barclays that increased its price to 98.25% of book value, which equated to \$19.50 per share net of the termination fee and other expenses

based on the \$19.85 book value of MTGE s common stock as of April 19, 2018 (the AGNC April 26 Proposal). AGNC continued to propose a consideration mix of cash (20%) and AGNC common stock (80%). The AGNC April 26 Proposal was higher than the Annaly April 10 Proposal (98.0%) based on percentage

of book value, but neither the MTGE special committee nor its advisors were permitted to engage with AGNC with respect to the April 26 Proposal because MTGE was under exclusivity with Annaly. Barclays reached out to Annaly to inform them that MTGE had received an unsolicited offer from AGNC with a higher price. Barclays did not disclose the terms of the revised AGNC proposal but indicated that the MTGE special committee was no longer in a position to recommend the Annaly April 10 proposal.

On April 27, 2018, Wachtell Lipton sent a revised draft of the merger agreement to Cooley and a draft of Annaly s disclosure letter.

On April 28, 2018, Cooley sent a draft of MTGE s disclosure letter to Wachtell Lipton.

On April 29, 2018, the MTGE special committee held a telephonic meeting, with representatives of Barclays and Cooley also attending, to discuss the AGNC April 26 Proposal. At the beginning of the meeting, Barclays informed the MTGE special committee that it had just received an enhanced verbal proposal from Annaly to increase its price from 98.0% to 100% of the book value of MTGE s common stock based on a consideration mix of cash (50%) and Annaly common stock (50%) (the Annaly April 29 Proposal). Barclays reviewed a presentation on the AGNC April 26 Proposal with the MTGE special committee that the Annaly April 29 Proposal was \$0.35 per share higher than the AGNC April 26 Proposal after adjusting both prices to reflect a \$19.75 per fully diluted share book value valuation as of March 31, 2017. Barclays also informed the MTGE special committee that timesation prior to its earnings call scheduled for the morning of May 3, 2018. To facilitate that timetable, Annaly was also prepared to negotiate the amendment with the MTGE external manager and deliver a termination notice after the merger agreement was signed and announced.

On April 30, 2018, Cooley and Wachtell Lipton discussed the merger agreement terms. Over the course of that discussion, Cooley expressed its view that the merger agreement should be revised to clarify that submission of an acquisition proposal by AGNC after announcement of a merger agreement with Annaly should not be attributed to MTGE (because AGNC is a related-party through the management relationship) or deemed a breach of MTGE s covenant not to solicit acquisition proposals. Cooley also proposed a termination fee of 3.0% of the transaction value of MTGE s common equity. Subsequently, Wachtell Lipton sent Cooley a revised draft of the merger agreement and the disclosure letters. Wachtell Lipton s draft of the merger agreement did not address the acquisition proposal clarification and proposed a 3.33% termination fee.

That evening, the MTGE special committee held a telephonic meeting, with representatives of Barclays and Cooley also attending, to review the status of the open economic and legal terms. Barclays reported that Annaly rejected the concept of a stock exchange ratio collar but was willing to consider offering an independent member of the MTGE board of directors a seat on the Annaly board of directors if that was important to the MTGE special committee. The MTGE special committee instructed Barclays to continue to negotiate for the collar and board seat with Annaly and gave guidance to Cooley with respect to addressing the open legal terms. Cooley and Wachtell Lipton had another call following the MTGE special committee meeting to discuss the open issues in the merger agreement. Following this call, Wachtell Lipton sent Cooley a further revised draft of the merger agreement.

On May 1, 2018, Cooley and Wachtell Lipton had another call to discuss the merger agreement. Wachtell Lipton communicated that, in light of AGNC s proposal, Annaly s position was that MTGE should pay a higher termination fee (of 6%) in the event MTGE changes its recommendation or terminates the merger agreement in connection with an AGNC superior proposal. Cooley responded that a 6% termination fee and a two-tier termination fee would not be acceptable to the MTGE special committee. Later that day Cooley also sent a revised draft of MTGE s disclosure letter to Wachtell Lipton.

On May 1, 2018, the MTGE external manager provided Barclays (who subsequently provided to Annaly, Wells Fargo and Sandler O Neill) an updated estimate of the book value of MTGE s common stock of \$19.65 per

fully diluted share as of April 30, 2018. Representatives of Annaly, Wells Fargo, Sandler O Neill, Barclays and the MTGE external manager held an additional telephonic conference that evening to conduct additional due diligence on this updated estimate.

Over the course of the evening of May 1, 2018 and the morning of May 2, 2018, the advisors of the MTGE special committee and Annaly continued to negotiate resolutions of the open issues. Subject to final approval by the MTGE special committee, Barclays, Wells Fargo and Sandler O Neill agreed on the cash and exchange ratio calculations based on a transaction price of \$19.65 per share price net of expenses (representing 100% of the book value of MTGE s common shares as of April 30, 2018) and an Annaly stock price of \$10.32 per share (based on a volume weighted average trading price for Annaly s common stock for the ten trading day period ended April 30, 2018), with MTGE stockholders having the right to elect, for each share of MTGE common stock, \$19.65 in cash, 1.9037 shares of Annaly common stock, or \$9.82 in cash and 0.9519 shares of Annaly common stock, without a collar and subject to proration so that 50% of the aggregate consideration is payable in cash and 50% of the aggregate consideration is payable in shares of Annaly common stock; that an acquisition proposal submitted by AGNC would be treated like an acquisition proposal submitted by any other party; and that there would be a single termination fee of 3.9% of the transaction value of MTGE s common equity (or approximately \$35.1 million).

During the evening of May 1, 2018, representatives of Annaly and representatives of the MTGE external manager also spoke concerning the termination of the management agreement and Annaly s need for a short transition period after closing. They agreed to negotiate an amendment to the management agreement that could be executed concurrently with the execution of the merger agreement.

Also on May 1, 2018 and May 2, 2018, representatives of Annaly and their advisors held several telephonic conferences with representatives of MTGE, MTGE s advisors and the MTGE external manager to complete due diligence.

On the morning of May 2, 2018, the MTGE special committee held a telephonic meeting, with representatives of Barclays and Cooley also attending. Representatives of Barclays reviewed the sales process and its financial analysis of the proposed economic terms. At the conclusion of its presentation, Barclays rendered to the MTGE special committee its oral opinion, which was subsequently confirmed by delivery of a written opinion, dated as of May 2, 2018, that, as of the date of such opinion, and based upon and subject to the matters and the limitations set forth therein, the aggregate consideration to be paid to the holders of MTGE common stock (other than as specified in such opinion) is fair from a financial point of view to such holders. For a more detailed discussion of Barclays opinion, please see below under the caption Opinion of the MTGE Special Committee s Financial Advisor. The opinion of Barclays is attached to MTGE s Schedule 14D-9 as Annex A. Representatives of Cooley then reviewed the material terms of the proposed merger agreement and the proposed authorizing resolutions for approval by the MTGE special committee. Cooley also reported that Annaly and the MTGE external manager were in the process of negotiating an amendment to the management agreement, which needed to be completed prior to final MTGE special committee approval. Following discussion, the meeting was adjourned until later in the day.

Following the meeting of the MTGE special committee, the MTGE board of directors held a telephonic meeting, with representatives of Cooley also attending. At this meeting, consistent with the board s prior determination, Mr. Kain informed the other members of the MTGE board of directors that he would recuse himself from the board s deliberations and vote on the proposed transaction in light of his role with AGNC. After Mr. Kain left the meeting, the MTGE board of directors then suspended its meeting agreeing to reconvene later in the day once the final terms of the transaction were negotiated.

During the day of May 2, 2018, MTGE, Annaly, the MTGE external manager, Cooley and Wachtell Lipton finalized the merger agreement, disclosure schedules and the amendment to the management agreement.

At 4:00 p.m. on May 2, 2018, the MTGE special committee held a telephonic meeting to approve the final terms of the transaction, with representatives of Barclays and Cooley also attending. During this meeting, Cooley

updated the MTGE special committee on the final terms of the merger agreement and amendment to the management agreement. Following consideration of the merger agreement and the transactions contemplated by the merger agreement, including consideration of the factors described in MTGE s Reasons for the Offer and the Merger; Recommendation of the MTGE Board of Directors, the MTGE special committee unanimously recommended that the MTGE board of directors (a) determine that the merger agreement and the transactions contemplated thereby, including the offer and the merger, are advisable to, and in the best interests of, MTGE and its stockholders, (b) agree that the merger shall be effected under Section 3-106.1 of the MGCL, (c) approve the execution, delivery and performance by MTGE of the merger agreement and the consummation of the transactions contemplated thereby, including the offer and the merger, and (d) resolve to recommend that the stockholders of MTGE tender their shares in connection with the offer.

The prior meeting of the MTGE board of directors reconvened immediately following the meeting of the MTGE special committee. During this meeting, the independent directors of the MTGE board (a) determined that the merger agreement and the transactions contemplated thereby, including the offer and the merger, are advisable to, and in the best interests of, MTGE and its stockholders, (b) agreed that the merger shall be effected under Section 3-106.1 of the MGCL, (c) approved the execution, delivery and performance by MTGE of the merger agreement and the consummation of the transactions contemplated thereby, including the offer and the merger, and (d) resolved to recommend that the stockholders of MTGE tender their shares in connection with the offer.

Following the MTGE board meeting, the definitive merger agreement was executed and publicly announced on the afternoon of May 2, 2018 after the close of the market.

Annaly s Reasons for the Offer and the Merger

In reaching its decision to approve the merger agreement and the transactions contemplated thereby, including the offer and the merger, Annaly s board of directors consulted with Annaly s senior management team and external manager, as well as Annaly s outside advisors, and considered a number of factors, including the following material factors (not in any relative order of importance), which it viewed as supporting its decision to approve the merger agreement and the transactions contemplated thereby, including the offer and the merger:

the fact that MTGE s portfolio consists primarily of agency and non-agency residential backed mortgage securities and healthcare real estate assets and the expectation that these assets will be complementary to Annaly s existing portfolio;

the expectation that the combined company s enhanced capital base will support the continued growth of all businesses;

the expectation that the acquisition would create incremental efficiency and growth opportunities;

the fact that the offer and merger are expected to be accretive to Annaly s core earnings and to provide a competitive cash-on-cash return;

the expectation that the merger and offer would further enhance Annaly s leadership position in the fragmented mortgage REIT market;

the amount and form of consideration to be paid in the offer and merger, including the fact that the exchange ratio and aggregate mix of consideration to be paid are fixed;

the view that the terms and conditions of the merger agreement and the transactions contemplated thereby, including the representations, warranties, covenants, closing conditions and termination provisions, are comprehensive and favorable to completing the proposed transaction;

the fact that the merger agreement places limitations on MTGE s ability to seek a superior proposal and requires MTGE to pay Annaly a termination fee of \$35,118,500 if Annaly or MTGE terminates the merger agreement under certain circumstances, including if MTGE consummates or enters into an agreement with respect to a competing acquisition proposal within a specified time period after termination of the merger agreement under certain circumstances;

the scope and results of the due diligence investigation of MTGE conducted by Annaly management and outside advisors, and the findings of that investigation;

current financial market conditions and forecast and the current and historical market prices, and trading information with respect to, shares of Annaly and MTGE common stock; and

the recommendation of Annaly s management and external manager in favor of the offer and the merger. The Annaly board of directors also considered a variety of uncertainties and risks and other potentially negative factors concerning the transactions, including the following (not in any relative order of importance):

the risk that the acquisition of MTGE might not be completed in a timely manner or at all and the attendant adverse consequences for Annaly s and MTGE s businesses as a result of the pendency of the acquisition and operational disruption;

costs associated with the offer and the merger;

the risk that the offer and merger may not be consummated despite the parties efforts or that the closing of the transactions may be unduly delayed due to regulatory processes or other factors outside of the parties control;

the risks associated with the occurrence of events which may materially and adversely affect the operations or financial condition of MTGE and its subsidiaries, which may not entitle Annaly to terminate the merger agreement;

the risk that the potential benefits of the acquisition may not be fully or partially achieved, or may not be achieved within the expected time frame;

the challenges and difficulties relating to combining the portfolios and operations of Annaly and MTGE;

the risk of diverting Annaly s management s focus and resources from other strategic opportunities and from operational matters while working to implement the transaction with MTGE, and other potential disruption associated with combining the companies, and the potential effects of such diversion and disruption on the businesses of Annaly and MTGE;

the effects of general competitive, economic, political and market conditions and fluctuations on Annaly, MTGE or the combined company; and

various other risks associated with the acquisition and the businesses of Annaly, MTGE and the combined company, some of which are discussed or referenced under Risk Factors.

The Annaly board of directors concluded that the potential negative factors associated with the acquisition were outweighed by the potential benefits that it expected Annaly to achieve as a result of the offer and the merger. Accordingly, the Annaly board of directors approved the merger agreement and the transactions contemplated thereby, including the offer and the merger.

The foregoing discussion of the information and factors considered by the Annaly board of directors is not intended to be exhaustive, but includes the material factors considered by the Annaly board of directors. In view of the variety of factors considered in connection with its evaluation of the acquisition, the Annaly board of directors did not find it practicable to, and did not, quantify or otherwise assign relative weights to the specific factors considered in reaching its determination. In addition, individual directors may have given different weights to different factors. The Annaly board of directors did not undertake to make any specific determination as to whether any factor, or any particular aspect of any factor, supported or did not support its ultimate determination. The Annaly board of directors based its determination on the totality of the information presented.

MTGE s Reasons for the Offer and the Merger; Recommendation of the MTGE Board of Directors

The MTGE board of directors, acting upon the unanimous recommendation of the MTGE special committee, has (i) determined that the terms of the offer, the merger and the transactions contemplated by the merger agreement, are in the best interests of MTGE and its stockholders, (ii) declared the offer, the merger and the other transactions contemplated by the merger agreement advisable, (iii) approved the execution and delivery by MTGE of the merger agreement, the performance by MTGE of its covenants and agreements contained in the merger agreement and the consummation of the offer, the merger and the other transactions contemplated by the merger agreement on the terms and subject to the conditions contained therein, and (iv) resolved to recommend that MTGE stockholders accept the offer and tender their shares of MTGE common stock to Annaly in the offer.

In evaluating the merger agreement and the transactions contemplated by the merger agreement, including the offer, the merger and the issuance of shares of Annaly common stock in connection therewith, the MTGE special committee (consisting entirely of independent directors) and the MTGE board of directors consulted with their financial and legal advisors, including Barclays, as financial advisor to the MTGE special committee, and Cooley, as counsel to the MTGE special committee.

In the course of reaching a determination that the offer and the merger are in the best interests of MTGE and its stockholders, and a recommendation that the holders of MTGE common stock accept the offer and tender their shares of MTGE common stock to Annaly in the offer, the MTGE special committee and the MTGE board of directors considered numerous factors, including the following material factors and benefits of the offer and the merger, each of which the MTGE special committee and the MTGE board of directors believed supported its determination and recommendation:

Recommendation of the MTGE Special Committee. The MTGE board of directors considered the unanimous recommendation of the MTGE special committee that the MTGE board of directors determine that the terms of the merger agreement and the transactions contemplated by the merger agreement, including the offer, the merger and the issuance of shares of Annaly common stock in connection therewith, are in the best interests of MTGE and its stockholders and declare the offer, the merger and the other transactions contemplated by the merger agreement advisable, as well as the independence of the members of the MTGE special committee making such recommendation and the independence, experience and expertise of Barclays as the financial advisor to the MTGE special committee.

Industry and Business Considerations. The MTGE special committee and the MTGE board of directors considered the current and historical industry conditions and the financial condition, results of operations, business, and financing prospects of MTGE, including the following:

the challenges facing the residential mortgage REIT sector in general, including significant uncertainty regarding the outlook for interest rates as well as uncertainty regarding the outlook for the financial markets generally;

the challenges facing MTGE in particular, including that, since June 23, 2017, the price per share of MTGE common stock has traded at a discount to MTGE s tangible book value per share, which makes

raising equity capital to fund new investments dilutive to shareholders and has made it difficult for MTGE to significantly increase its size and scale through capital market transactions; and

the general views of the members of the MTGE special committee and the MTGE board of directors with respect to the business, financial condition, current business strategy and prospects of MTGE, including the potential challenges for MTGE to continue to access financing resources on acceptable terms.

Offer Price and Right to Participate in Annaly s Future Growth. The MTGE special committee and the MTGE board of directors considered:

the fact that the value of the consideration represents a premium of approximately 12% to the 60-day volume-weighted average price of MTGE common stock ending on April 30, 2018;

the fact that for each outstanding share of MTGE common stock accepted for payment in the offer or converted and exchanged in the merger the holder thereof will be entitled to receive approximately 50% of the consideration in Annaly common stock and, therefore, participate in the upside potential of the combined company;

the ability for stockholders to elect stock consideration which may allow them to defer a portion of their taxable gain;

the fact that in lieu of receiving the mixed consideration, holders of shares of MTGE common stock may elect to receive, for each share of MTGE common stock that they hold, the all-cash consideration or the all-stock consideration, in each case, subject to proration;

the belief of the MTGE special committee that as a result of extensive negotiations between the parties it had obtained Annaly s best and final offer for the MTGE common stock, which was increased multiple times from Annaly s initial proposal of 95% of book value to 100% of book value per share; and

the offer and the merger are expected to be immediately accretive to Annaly s earnings per share.

Evaluation of Strategic Alternatives. The MTGE special committee s and the MTGE board of directors belief that the value offered to MTGE s stockholders in the offer and merger was more favorable to MTGE s stockholders on a risk-adjusted basis than the potential value of remaining an independent public company and that the value per share obtained was the highest per share consideration that was reasonably attainable. This belief was supported in part by the results of the MTGE special committee s evaluation of strategic alternatives, through which MTGE and its financial advisors engaged with other parties that were believed to be the most able and willing to pay the highest price for MTGE and MTGE received alternative acquisition proposals that were not as favorable to MTGE and its stockholders as the offer and merger with Annaly (as more fully described above in Background of the Offer and the Merger).

Negotiations with Annaly. The MTGE special committee considered the course of negotiations between MTGE and Annaly, resulting in multiple increases in the price per share originally offered by Annaly, as well as several changes in the terms and conditions of the merger agreement from the terms and conditions proposed by Annaly that were favorable to MTGE. The MTGE special committee believed, based on these negotiations, that the offer price was the highest price per share that Annaly was willing to pay and that the

merger agreement contained the most favorable terms to MTGE to which Annaly was willing to agree.

No Financing Condition. The MTGE special committee considered the contractual representation of Annaly and Offeror that they would have access to sufficient funds to pay the amounts required to be paid under the merger agreement and that the offer and the merger are not subject to a financing condition.

Benefits of Combining with Annaly of Increased Scale, Portfolio Diversity and other Operating Considerations. The MTGE special committee and the MTGE board of directors considered the following benefits and operating considerations in combining with Annaly:

the increased portfolio diversification of the combined company, taking into account Annaly s diversified portfolio of agency and non-agency mortgage-backed securities, commercial real estate debt and equity and commercial credit and MTGE s portfolio which consists of agency and non-agency residential mortgage backed securities and healthcare real estate assets;

the potential access to capital and asset acquisition opportunities resulting from the scale and resources of the combined company, the fact that Annaly s average expenses have historically been a lower percentage of assets and equity than other residential mortgage REITs, and the potential for increased capital allocation alternatives for the combined company;

the potential access to additional financing resources that would not be available to MTGE as a stand-alone company; and

the continuing access to Annaly s external manager with broad expertise and resources to invest across a range of asset classes at a cost that has historically been a lower percentage of assets and equity than other residential mortgage REITs.

Opinion of Barclays and Related Analysis. The MTGE special committee and the MTGE board of directors considered the oral opinion of Barclays, subsequently confirmed in writing, dated May 2, 2018, that, as of such date, based upon and subject to the assumptions made, procedures followed, matters considered, and qualifications and limitations upon the review undertaken by Barclays in preparing its written opinion, as set forth in such written opinion, the Aggregate Consideration (as defined in the written opinion) to be offered to the holders of the shares of MTGE common stock (other than the stockholders of Cancelled Shares (as defined in the written opinion)) in the offer was fair from a financial point of view to such holders, as more fully described below in Opinion of MTGE s Financial Advisor. The MTGE special committee and the MTGE board of directors were aware that Barclays will become entitled to certain fees upon consummation of the offer and merger, as more fully described below in Opinion of the MTGE Special Committee s Financial Advisor.

Other Terms of the Merger Agreement. The MTGE special committee and the MTGE board of directors considered certain other terms of the merger agreement, which are more fully described under the caption *Merger Agreement.* Certain provisions of the merger agreement that the MTGE special committee and the MTGE board of directors considered important included:

the merger agreement provides for the prompt commencement of the offer, which may enable holders of MTGE common stock who tender their shares into the offer to receive their consideration more quickly than in a transaction structured as a one-step merger;

the ability to respond to unsolicited acquisition proposals by the MTGE board of directors, upon the recommendation of the MTGE special committee and determination that the failure to take such action would be inconsistent with the directors duties under applicable law, and to engage in negotiations or discussions with third parties regarding alternative transactions under certain circumstances (see Merger Agreement No Solicitation of Other Offers by *MTGE* for more information);

the right of the MTGE board of directors to change or withdraw its recommendation to holders of MTGE common stock, following receipt of an unsolicited superior proposal or upon the occurrence of

certain other intervening events and upon the recommendation of the MTGE special committee and determination that the failure to take such action would be inconsistent with the directors duties under applicable law (see Merger Agreement No Solicitation of Other Offers by MTGE for more information);

the right of the MTGE board of directors to terminate the merger agreement and to accept a superior proposal, if certain conditions are met, subject to the payment of the termination fee to Annaly (see

Merger Agreement Termination of the Merger Agreement Termination by MTGE for more information);

the fact that Offeror s obligations to purchase (and Annaly s obligation to cause the Offeror to purchase) the shares of MTGE common stock tendered in the offer and to close the merger are subject to limited conditions, and that the offer and the merger are reasonably likely to be consummated; and

the fact that the Offeror must generally extend the offer for one or more periods until the offer conditions have been satisfied.

The MTGE special committee in making its recommendation to the MTGE board of directors and the MTGE board of directors in making its determination also identified and considered the following potentially negative factors in its deliberations:

Fluctuations in the Price of Annaly Common Stock. The MTGE special committee and the MTGE board of directors considered the fact that holders of shares of MTGE common stock who receive shares of Annaly common stock in the offer or the merger will receive a fixed number of shares of Annaly common stock which is based on the MTGE common stock being valued at \$19.65 per share and the volume weighted average closing price of Annaly common stock for the ten trading day period ended April 30, 2018, and such number of shares will not be adjusted for any decrease in the trading price of shares of Annaly common stock between the date of the merger agreement and the completion of the offer or the merger, and the fact that MTGE is not permitted to terminate the merger agreement solely because of changes in the market price of shares of Annaly common stock;

Non-Solicitation Covenant. The MTGE special committee and the MTGE board of directors considered the fact that the merger agreement imposes restrictions on soliciting and responding to competing acquisition proposals from third parties;

Termination Fee. The MTGE special committee and the MTGE board of directors considered that the termination fee of \$35,118,500.00 payable in cash to Annaly if the merger agreement is terminated under certain circumstances, including if the merger agreement is terminated in order for the MTGE board of directors to accept a superior proposal, may discourage third parties that may otherwise have an interest in a business combination with, or an acquisition of, MTGE from pursuing such a transaction;

Interim Operating Covenants. The MTGE special committee and the MTGE board of directors considered that the merger agreement imposes restrictions on the conduct of the business of MTGE and its subsidiaries prior to the consummation of the merger (see Merger Agreement Conduct of Business Before Completion of the Merger Restrictions on MTGE s Operations);

Risks the Offer and Merger May Not Be Completed. The MTGE special committee and the MTGE board of directors considered the following factors in connection with the risk that the offer and the merger may not be completed:

the risk that the conditions to the offer may not be satisfied and that, therefore, shares of MTGE common stock may not be purchased pursuant to the offer and the merger may not be consummated; and

the risks and costs to MTGE of the adverse effect of the resulting public announcement of any termination of the merger agreement on the market price of shares of MTGE common stock, and operating results of MTGE, particularly in light of the diversion of management resources from operational matters and other strategic opportunities and the costs incurred in connection with the transaction;

Potential Conflict of Interest Relating to Termination of Management Agreement. The MTGE special committee and the MTGE board of directors considered the potential conflict of interest created as a result of the amendment to, and termination of, the management agreement in connection with the transaction and the resulting payment of a termination fee to the MTGE external manager;

Interests of Directors and Executive Officers. The MTGE special committee and the MTGE board of directors considered the interests of certain directors and executive officers in the merger, each as more fully described The Offer Interests of Certain Persons in the Offer and the Merger. ;

Regulatory Approvals. The MTGE special committee and the MTGE board of directors considered the period of time necessary to obtain the regulatory approvals that would be required to consummate the offer;

Cash Component of Consideration. The MTGE special committee and the MTGE board of directors considered that the holders of MTGE common stock who receive cash consideration, either because of an affirmative election or because of subsequent proration, will not be able to participate in the future benefits of the combined company and the fact that the receipt of cash will be a taxable transaction for U.S. federal income tax purposes; and

Other Risks of the Offer and the Merger. The MTGE special committee and the MTGE board of directors also considered the following additional risks:

the substantial costs to be incurred in connection with the transaction;

the absence of appraisal rights for holders of MTGE common stock under Maryland law and MTGE $\,$ s charter; and

the risks described in the section captioned Risk Factors in the prospectus/offer to exchange. Although the foregoing discussion sets forth the material factors considered by the MTGE special committee in making its recommendation to the MTGE board of directors and the MTGE board of directors in reaching its determination, it does not include all of the factors considered by either the MTGE special committee or the MTGE board of directors, and each director may have considered different factors or given different weights to different factors. In view of the variety of factors and the amount of information considered, neither the MTGE special committee nor the MTGE board of directors found it practicable to, and did not, make specific assessments of, quantify or otherwise assign relative weights to the specific factors realized that there can be no assurance about future results, including results expected or considered in the factors above. However, both the MTGE special committee and the MTGE board of directors concluded that the potential positive factors described above significantly outweighed the negative factors described above. The recommendations were made after consideration of all of the factors as a whole.

THE MTGE BOARD OF DIRECTORS, ACTING UPON THE UNANIMOUS RECOMMENDATION OF THE MTGE SPECIAL COMMITTEE, HAS DETERMINED THAT THE TERMS OF THE MERGER AGREEMENT AND THE TRANSACTION, INCLUDING THE OFFER, THE MERGER AND THE ISSUANCE OF SHARES OF ANNALY COMMON STOCK IN CONNECTION THEREWITH, WERE IN THE BEST INTERESTS OF MTGE AND ITS STOCKHOLDERS, DECLARED THE OFFER, THE MERGER AND THE OTHER TRANSACTIONS CONTEMPLATED BY THE MERGER AGREEMENT ADVISABLE, APPROVED THE EXECUTION AND DELIVERY BY MTGE OF THE MERGER AGREEMENT, THE PERFORMANCE BY MTGE OF ITS COVENANTS AND AGREEMENTS CONTAINED IN THE MERGER AGREEMENT AND THE CONSUMMATION OF THE OFFER, THE MERGER AND THE OTHER TRANSACTIONS CONTEMPLATED BY THE MERGER AGREEMENT ON THE TERMS AND SUBJECT TO THE CONDITIONS CONTAINED THEREIN, AND RECOMMENDS THAT MTGE STOCKHOLDERS ACCEPT THE OFFER AND TENDER THEIR SHARES OF MTGE COMMON STOCK TO ANNALY IN THE OFFER.

In considering the MTGE special committee s recommendation and the MTGE board of directors determination with respect to the offer and the merger, you should be aware that the MTGE external manager and certain executive officers of MTGE and the MTGE external manager have arrangements that cause them to have interests in the transaction that are different from, or are in addition to, the interests of MTGE stockholders generally The Offer Interests of Certain Persons in the Offer and the Merger.

Opinion of the MTGE Special Committee s Financial Advisor

The MTGE special committee engaged Barclays to act as its financial advisor with respect to pursuing strategic alternatives for MTGE, including a possible sale of MTGE, pursuant to an engagement letter dated March 29, 2018. On May 2, 2018, Barclays rendered its oral opinion (which was subsequently confirmed in writing) to the MTGE special committee that, as of such date and based upon and subject to the qualifications, limitations and assumptions stated in its opinion, the aggregate consideration to be offered to the stockholders of MTGE in the proposed offer and merger is fair, from a financial point of view, to such stockholders.

The full text of Barclays written opinion, dated as of May 2, 2018, is attached as Annex B to this document. Barclays written opinion sets forth, among other things, the assumptions made, procedures followed, factors considered and limitations upon the review undertaken by Barclays in rendering its opinion. You are encouraged to read the opinion carefully in its entirety. The following is a summary of Barclays opinion and the methodology that Barclays used to render its opinion. This summary is qualified in its entirety by reference to the full text of the opinion.

Barclays opinion, the issuance of which was approved by Barclays Fairness Opinion Committee, is addressed to the MTGE special committee, addresses only the fairness, from a financial point of view, of the aggregate consideration to be offered to the stockholders of MTGE and does not constitute a recommendation to any stockholder of MTGE as to whether or not such stockholder should tender the shares of common stock pursuant to the offer or how such stockholder should act with respect to the transaction. The terms of the proposed transaction were determined through arm s-length negotiations between the MTGE special committee and Annaly and were approved by the MTGE board of directors (following the unanimous recommendation of the MTGE special committee). Barclays did not recommend any specific form of consideration to MTGE or that any specific form of consideration constituted the only appropriate consideration for the proposed transaction. Barclays was not requested to address, and its opinion does not in any manner address, MTGE s underlying business decision to proceed with or effect the proposed transaction, the likelihood of the consummation of the proposed transaction, or the relative merits of the proposed transaction as compared to any other transaction in which MTGE may engage. In addition, Barclays expressed no opinion on, and its opinion does not in any manner address, the fairness of the amount or the nature of any compensation paid to the MTGE external manager or to any officers, directors or employees of any parties to the proposed transaction, or any class of such persons, relative to the aggregate consideration to be offered to the stockholders of MTGE in the proposed transaction. No limitations were imposed by the MTGE special committee upon Barclays with respect to the investigations made or procedures followed by it in rendering its opinion.

In arriving at its opinion, Barclays, among other things:

reviewed and analyzed a draft of the merger agreement, dated as of May 1, 2018, and the specific terms of the proposed transaction as set forth therein;

reviewed and analyzed publicly available information concerning MTGE that Barclays believed to be relevant to its analysis, including MTGE s Annual Report on Form 10-K for the fiscal year ended December 31, 2017;

reviewed and analyzed publicly available information concerning Annaly that Barclays believed to be relevant to its analysis, including Annaly s Annual Report on Form 10-K for the fiscal year ended December 31, 2017;

reviewed and analyzed financial and operating information with respect to the business, operations and prospects of MTGE furnished to Barclays by MTGE, including financial projections of MTGE prepared by the MTGE external manager (the MTGE Projections);

reviewed and analyzed financial and operating information with respect to the business, operations and prospects of Annaly furnished to Barclays by Annaly, including financial projections of Annaly prepared by Annaly Management Company LLC, the manager of Annaly (the Annaly Manager) (the Annaly Projections);

reviewed and analyzed a trading history of MTGE common stock from August 4, 2011 through May 1, 2018 and a trading history of Annaly common stock from August 4, 2011 to May 1, 2018, and a comparison of that trading history with those of other companies that Barclays deemed relevant;

reviewed and analyzed a comparison of the historical financial results and present financial condition of MTGE and Annaly with those of other companies that Barclays deemed relevant;

reviewed and analyzed published estimates of independent research analysts with respect to the future financial performance and net asset value of MTGE and Annaly and price targets of MTGE and Annaly;

reviewed and analyzed a comparison of the financial terms of the proposed transaction with the financial terms of certain other transactions that Barclays deemed relevant;

had discussions with the MTGE external manager and the Annaly Manager concerning Annaly s business, operations, assets, liabilities, financial condition and prospects;

had discussions with the MTGE external manager and the Annaly Manager concerning MTGE s business, operations, assets, liabilities, financial condition and prospects; and

has undertaken such other studies, analyses and investigations as Barclays deemed appropriate. In arriving at its opinion, Barclays assumed and relied upon the accuracy and completeness of the financial and other information used by Barclays without any independent verification of such information (and had not assumed responsibility or liability for any independent verification of such information). Barclays also relied upon the assurances of each of the MTGE external manager and the Annaly Manager that they were not aware of any facts or circumstances that would make such information inaccurate or misleading. With respect to the MTGE Projections, upon the advice and at the direction of the MTGE special committee, Barclays has assumed that such projections, based on the assumptions stated therein, have been reasonably prepared and reflect the most reasonable currently available estimates and judgments of the MTGE external manager as to MTGE s future financial performance and that MTGE would perform substantially in accordance with such projections. With respect to the Annaly Projections, upon the advice and at the direction of the MTGE special committee, Barclays has assumed that such projections, based on the assumptions stated therein, have been reasonably prepared and reflect the most reasonable currently available estimates and judgments of the Annaly Manager as to Annaly s future financial performance and that Annaly would perform substantially in accordance with such projections. In arriving at its opinion, Barclays assumed no responsibility for and expressed no view as to any such projections or estimates or the assumptions on which they were based. In arriving at its opinion, Barclays did not conduct a physical inspection of the properties and facilities of MTGE or Annaly and did not make or obtain any evaluations or appraisals of the assets or liabilities of MTGE or

Annaly. Barclays opinion was necessarily based upon market, economic and other conditions as they existed on, and could be evaluated as of, May 2, 2018. Barclays assumed no responsibility for updating or revising its opinion based on events or circumstances that may have occurred after May 2, 2018. Barclays expressed no opinion as to the prices at which shares of MTGE common stock would trade following the announcement of the proposed transaction or shares of Annaly common stock would trade following the announcement or consummation of the proposed transaction. Barclays opinion should not be viewed as providing any assurance that the market value of the shares of Annaly common stock to be held by the stockholders of MTGE after the consummation of the proposed transaction will be in excess of the market value of MTGE common stock owned by such stockholders at any time prior to the announcement or consummation of the proposed transaction.

Barclays assumed that the executed merger agreement would conform in all material respects to the last draft reviewed by Barclays. Additionally, Barclays assumed the accuracy of the representations and warranties

contained in the merger agreement and all the agreements related thereto. Barclays also assumed, upon the advice and at the direction of the MTGE special committee, that all material governmental, regulatory and third party approvals, consents and releases for the proposed transaction would be obtained within the constraints contemplated by the merger agreement and that the proposed transaction will be consummated in accordance with the terms of the merger agreement without waiver, modification or amendment of any material term, condition or agreement thereof. Barclays did not express any opinion as to any tax or other consequences that might result from the proposed transaction, nor did Barclays opinion address any legal, tax, regulatory or accounting matters, as to which Barclays understood MTGE had obtained such advice as it deemed necessary from qualified professionals.

In connection with rendering its opinion, Barclays performed certain financial, comparative and other analyses as summarized below. In arriving at its opinion, Barclays did not ascribe a specific range of values to the MTGE common stock but rather made its determination as to fairness, from a financial point of view, to MTGE s stockholders of the aggregate consideration to be offered to such stockholders in the proposed offer and merger on the basis of various financial and comparative analyses. The preparation of a fairness opinion is a complex process and involves various determinations as to the most appropriate and relevant methods of financial and comparative analyses and the application of those methods to the particular circumstances. Therefore, a fairness opinion is not readily susceptible to summary description.

In arriving at its opinion, Barclays did not attribute any particular weight to any single analysis or factor considered by it but rather made qualitative judgments as to the significance and relevance of each analysis and factor relative to all other analyses and factors performed and considered by it and in the context of the circumstances of the particular transaction. Accordingly, Barclays believes that its analyses must be considered as a whole, as considering any portion of such analyses and factors, without considering all analyses and factors as a whole, could create a misleading or incomplete view of the process underlying its opinion.

Summary of Material Financial Analyses

The following is a summary of the material financial analyses used by Barclays in preparing its opinion to the MTGE special committee. The summary of Barclays analyses and reviews provided below is not a complete description of the analyses and reviews underlying Barclays opinion. The preparation of a fairness opinion is a complex process involving various determinations as to the most appropriate and relevant methods of analysis and review and the application of those methods to particular circumstances, and, therefore, is not readily susceptible to summary description.

For the purposes of its analyses and review, Barclays made numerous assumptions with respect to industry performance, general business, economic, market and financial conditions and other matters, many of which are beyond the control of MTGE or any other parties to the proposed transaction. No company, business or transaction considered in Barclays analyses and review is identical to MTGE, Annaly, the Offeror or the proposed transaction, and an evaluation of the results of those analyses and reviews is not entirely mathematical. Rather, the analyses and reviews involve complex considerations and judgments concerning financial and operating characteristics and other factors that could affect the acquisition, public trading or other values of the companies, businesses or transactions considered in Barclays analyses and reviews. None of MTGE, the Manager, Annaly, the Offeror, Barclays or any other person assumes responsibility if future results are materially different from those discussed. Any estimates contained in these analyses and reviews and the ranges of valuations resulting from any particular analysis or review are not necessarily indicative of actual values or predictive of future results or values, which may be significantly more or less favorable than as set forth below. In addition, analyses relating to the value of companies, businesses or securities do not purport to be appraisals or reflect the prices at which the companies, businesses or securities may actually be sold. Accordingly, the estimates used in, and the results derived from, Barclays

inherently subject to substantial uncertainty.

The summary of the financial analyses and reviews summarized below include information presented in tabular format. In order to fully understand the financial analyses and reviews used by Barclays, the tables must be read together with the text of each summary, as the tables alone do not constitute a complete description of the financial analyses and reviews. Considering the data in the tables below without considering the full description of the analyses and reviews, including the methodologies and assumptions underlying the analyses and reviews, could create a misleading or incomplete view of Barclays and reviews.

Selected Comparable Company Analysis

In order to assess how the market values shares of similar publicly traded companies and to provide a range of relative implied equity values per share of MTGE and of Annaly, Barclays reviewed and compared specific financial and operating data relating to MTGE and Annaly, respectively, with selected companies that Barclays, based on its experience in the Mortgage REIT industry, deemed comparable to MTGE and Annaly, respectively. The selected comparable companies with respect to MTGE were:

Two Harbors Investment Corp.

Annaly Capital Management, Inc.

ARMOUR Residential REIT, Inc.

Invesco Mortgage Capital Inc. The selected comparable companies with respect to Annaly were:

AGNC Investment Corp.

ARMOUR Residential REIT, Inc.

Capstead Mortgage Corporation

Barclays calculated and compared various financial multiples and ratios of MTGE and Annaly and the selected comparable companies. As part of its selected comparable company analysis, Barclays calculated and analyzed each applicable company s (i) price per share as of May 1, 2018 as a multiple of tangible book value per share as of December 31, 2017 (P/TBV 12/31) and as of March 31, 2018, (P/TBV 3/31), (ii) ratio of its current stock price to its calendar year 2018 estimated earnings per share (commonly referred to as a price earnings ratio, or P/E), based on each of (x) the MTGE Projections ($Mgt^{(1)}$) and (y) market data, with such median based upon research reports available to Barclays (Street) and (iii) implied dividend yield by annualizing the dividends paid in the most recent quarter.

The results of this selected comparable company analysis are summarized below:

Table of Contents

MTGE Peers

	Low	Mean	Median	High
P / TBV ⁽¹⁾	0.88x	0.92x	0.92x	0.97x
P / 18E Earning ³⁾	8.1x	8.9x	9.1x	9.3x
Dividend Yield	10.0%	11.1%	11.0%	12.3%

(1) Tangible book value based on latest tangible book value publically available

(2) 2018 Earnings based on Median analyst estimates available to Barclays through FactSet

MTGE

	Metric	Selected Companies Range	Implied Share Price Range		
	Metric	Kange	Kallge		
P/TBV 3/31	\$ 19.61	0.85x 0.95x	\$	16.67	\$18.63
P/TBV 12/31	\$ 20.62	0.85x 0.95x	\$	17.53	\$19.59
P/2018E (Mgt.) ⁽¹⁾	\$ 2.29	8.5x 9.5x	\$	19.47	\$21.76
P/2018E (Street)	\$ 1.98	8.5x 9.5x	\$	16.83	\$18.81
Annualized Dividend Yield	\$ 2.00	12.0% 10.0%	\$	16.67	\$20.00

(1) Represents Net Spread, Dollar Roll and Healthcare Income, excluding catch up amortization. <u>Annaly Peers</u>

	Low	Mean	Median	High
P / TBV ⁽¹⁾	0.87x	0.93x	0.92x	1.01x
P / 18E Earning ³⁾	7.9x	10.6x	9.3x	14.5x
Dividend Yield	7.0%	9.5%	10.0%	11.4%

(1) Tangible book value based on latest tangible book value publically available

(2) 2018 Earnings based on Median analyst estimates available to Barclays through FactSet

Annaly

		Selected Companies	Imp	olied Share Price
	Metric	Range		Range
P/TBV 3/31	\$ 10.40	0.90x 1.00x	\$	9.36 \$10.40
P/2018E (Mgt.)	\$ 1.18	8.0x 10.5x	\$	9.47 \$12.43
Annualized Dividend Yield	\$ 1.20	11.5% 10.0%	\$	10.43 \$12.00

Barclays selected the comparable companies listed above because of similarities in one or more business or operating characteristics with MTGE or Annaly, as applicable. However, because no selected comparable company is exactly the same as MTGE or Annaly, as applicable, Barclays believed that it was inappropriate to, and therefore did not, rely solely on the quantitative results of the selected comparable company analysis. Accordingly, Barclays also made qualitative judgments concerning differences between the business, financial and operating characteristics and prospects of MTGE or Annaly, as applicable, and the selected comparable companies that could affect the public trading values of each in order to provide a context in which to consider the results of the quantitative analysis. These qualitative judgments related primarily to the differing sizes, growth prospects, profitability levels and degree of operational risk between MTGE or Annaly, as applicable, and the companies included in the selected company analysis.

Barclays noted that on the basis of the selected comparable company analysis, the transaction consideration of \$19.65 per MTGE share was (i) above the range of implied values per share of \$16.67 \$18.63 derived from the P/TBV 3/31

Table of Contents

calculation, (ii) above the range of implied values per share of \$17.53 \$19.59 derived from the P/TBV 12/31 calculation, (iii) within the range of implied values per share of \$19.47 \$21.76 derived from the P/2018E (Mgt.) calculation, (iv) above the range of implied values per share of \$16.83 \$18.81 derived from the P/2018E (Street) calculation and (v) within the range of implied values per share of \$16.67 \$20.00 derived from the annualized dividend yield calculation.

Barclays also noted that on the basis of the selected comparable company analysis, the transaction exchange ratio price of \$10.32 per Annaly share was within the range of implied exchange ratios per share of (i) \$9.36 \$10.40 derived from the P/TBV 3/31 calculation and (ii) \$9.47 \$12.43 derived from the P/2018E (Mgt.)

calculation, and below the range of implied exchange ratios per share of (iii) \$10.43 \$12.00 derived from the annualized dividend yield calculation. The exchange ratio price of \$10.32 is the price per share of Annaly shares which was used in the calculation of exchange ratios.

Selected Precedent Transactions Analysis

Barclays reviewed and compared the purchase prices and financial multiples paid in selected other transactions that Barclays, based on its experience with merger and acquisition transactions, deemed relevant. Barclays chose such transactions with a focus on U.S. target companies based on, among other things, the similarity of the applicable target companies in the transactions to MTGE with respect to the size, industry or business mix, margins, competitive dynamics and other characteristics of their businesses. Barclays reviewed the following transactions:

Date Announced	Acquiror	Target
April 26, 2018	Two Harbors Investment Corp.	CYS Investments, Inc.
April 11, 2016	Annaly Capital Management, Inc.	Hatteras Financial Corporation
April 7, 2016	Sutherland Asset Management Corp	ZAIS Financial Corp.
March 2, 2016	ARMOUR Residential REIT, Inc.	JAVELIN Mortgage Investment Corp.
February 26, 2016	Apollo Commercial Real Estate Finance, Inc.	Apollo Residential Mortgage

The reasons for and the circumstances surrounding each of the selected precedent transactions analyzed were diverse and there are inherent differences in the business, operations, financial conditions and prospects of MTGE and the companies included in the selected precedent transaction analysis. Although none of the selected transactions is directly comparable to the transaction, the target companies in the selected transactions were companies that, for purposes of analysis, may be considered similar to MTGE.

Accordingly, for the above selected transactions, based on information from SEC filings, SNL Financial and FactSet, Barclays calculated and reviewed the final announced transaction price as a multiple of the target company s last reported book value as of the time of announcement, which is referred to as Transaction P/BV. The following table sets forth the transactions analyzed based on such characteristics and the results of such analysis:

	Low	Mean	Median	High			
Transaction P/BV	0.85x	0.92x	0.89x	1.05x			
Barclays noted that on the basis of the selected precedent transaction analysis, the transaction consideration of \$19.65							
per share was within the range of implied values per share of \$16.67	\$20.60 cald	culated usi	ng March 31	, 2018 tang			

Dividend Discount Analysis of MTGE

Barclays performed a dividend discount analysis on MTGE using the MTGE Projections and certain publicly available information, which was used as a basis for discount rates and terminal value range. Barclays used two different methods to calculate annual discount rates that were then applied to the MTGE Projections and a range of terminal values estimated using a price to tangible book value ratio (P/TBV) methodology. In one method, Barclays calculated a range implied by the dividend yield of selected comparable companies, resulting in a selected discount range of 10.0% to 12.0%. Barclays then applied those discount rates to the projected quarterly dividends for MTGE from 2Q2018 through 4Q2020 (as set forth in MTGE Projections) and to an estimated terminal value (based on

book value.

projected tangible book value in 4Q2020). Barclays calculated the net present value of the dividends and the net present value of the terminal value, which was based on applying the value range of 0.85x to 0.95 to quarter-11 tangible book value, and then divided by the number of fully diluted shares to get an implied value per share. This analysis assumed 100% of Net Spread, Dollar Roll and Healthcare

Income, excluding catch up amortization, is paid out through dividends to common shareholders of MTGE. This analysis resulted in a range of implied present values per share of MTGE common stock of \$17.59 to \$19.86. In the other method, Barclays calculated a discount range implied by the capital asset pricing model, resulting in a selected range of 8.0% to 9.5%. Barclays then applied those discount rates to the projected quarterly dividends for MTGE from 2Q2018 through 4Q2020 (as set forth in MTGE Projections) and to an estimated terminal value (based on projected tangible book value in 4Q2020). This analysis resulted in a range of implied present values per share of MTGE common stock of \$18.55 to \$20.75

Barclays noted that on the basis of the dividend discount analysis, the transaction consideration of \$19.65 was within the range of implied values per share calculated using the MTGE Projections.

Dividend Discount Analysis of Annaly

Barclays performed a dividend discount analysis on Annaly using the Annaly Projections and certain publicly available information, which was used as a basis for discount rates and terminal value range. Additionally, based on diligence, a 2.0% decline in tangible book value per share was modeled over the forecast period. Barclays used two different methods to calculate annual discount rates that were then used in the subsequent dividend discount analysis and applied to the Annaly Projections and a range of terminal values estimated using a P/TBV methodology. In one method, Barclays calculated a discount range implied by the dividend vield of selected comparable companies, resulting in a selected range of 10.0% to 11.5%. Barclays then applied those discount rates to the Annaly Projections and calculated the net present value of the dividends and the net present value of the terminal value, which was based on applying the value range of 0.90x to 1.00x to quarter-6 tangible book value, and then divided by the number of fully diluted shares (1,160 million shares) to get an implied value per share. This analysis assumed 100% of core earnings (Ex. PAA) are paid out through dividends to common shareholders of Annaly. This analysis resulted in a range of implied present values per share of Annaly common stock of \$9.37 to \$10.43. In the other method, Barclays calculated a discount range implied by the capital asset pricing model, resulting in a selected range of 6.0% to 8.0%. Barclays then applied those discount rates to the Annaly Projections and an estimated terminal value (based on projected tangible book value in 4Q2020). This analysis resulted in a range of implied present values per share of Annaly common stock of \$9.80 to \$10.99

Barclays noted that on the basis of the dividend discount analysis, the transaction exchange ratio price of \$10.32 was within the range of implied values per share calculated using the Annaly Projections.

Other Factors

Barclays also reviewed and considered other factors, which were not considered part of its financial analyses in connection with rendering its advice, but were references for informational purposes, including, among other things, the Historical Trading and the Analyst Target Prices, each described below.

Historical Trading

To illustrate the trend in the historical trading prices of the MTGE and Annaly shares, respectively, Barclays reviewed, for informational purposes, historical data with regard to the trading prices of such shares for the period from August 4, 2011 to May 1, 2018 and compared such data with the relative stock price performances during the same periods of the respective selected companies listed under the caption Selected Comparable Company Analysis above.

Barclays noted that during the period from May 1, 2017 to May 1, 2018, the price of the MTGE common stock ranged from \$16.15 to \$20.00 and that during the period from May 1, 2017 to May 1, 2018, the price of the Annaly common stock ranged from \$10.00 to \$12.73.

Analyst Target Prices

Barclays reviewed, for informational purposes, as of May 1, 2018, the publicly available price targets of the MTGE and Annaly common stock published by equity research analysts associated with various Wall Street firms and available through FactSet and SNL Financial. The research analysts price target per share of MTGE common stock ranged from \$19.00 \$21.00 per share and the price target per share of Annaly common stock ranged from \$10.00 \$11.50 per share. The publicly available share price targets published by such equity research analysts do not necessarily reflect the current market trading price for MTGE or Annaly common stock and these estimates are subject to uncertainties, including future financial performance of MTGE and Annaly and future market conditions.

General

Barclays is an internationally recognized investment banking firm and, as part of its investment banking activities, is regularly engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, investments for passive and control purposes, negotiated underwritings, competitive bids, secondary distributions of listed and unlisted securities, private placements and valuations for estate, corporate and other purposes. The MTGE special committee selected Barclays because of its familiarity with MTGE and its qualifications, reputation and experience in the valuation of businesses and securities in connection with mergers and acquisitions generally, as well as substantial experience in transactions comparable to the proposed transaction.

Barclays is acting as financial advisor to the MTGE special committee in connection with the proposed transaction and will receive compensation for its services in connection with the proposed transaction, \$500,000 of which became payable from MTGE to Barclays upon the delivery of Barclays opinion, which is referred to as the Opinion Fee . The Opinion Fee was not contingent upon the conclusion of Barclays opinion or the consummation of the proposed transaction. A retainer fee of \$125,000 per month, which is referred to as the Retainer Fee also became payable to Barclays as of March 29, 2018 . The Retainer Fee may not exceed \$500,000. Barclays will receive compensation of \$7 million upon consummation of the offer against which the amounts paid pursuant to the Opinion Fee (but not the Retainer Fee) will be credited. In addition, MTGE has agreed to reimburse Barclays for a portion of its reasonable and documented out-of-pocket expenses incurred in connection with the proposed transaction and to indemnify Barclays for certain liabilities that may arise out of its engagement by MTGE and the rendering of Barclays opinion. Barclays has performed various investment banking services for MTGE and Annaly in the past, and is likely to perform such services in the future, and has received, and is likely to receive, customary fees for such services. Specifically, in the past two years, Barclays has not performed any investment banking and financial services for MTGE or Annaly for which it has earned investment banking fees.

Barclays, its subsidiaries and its affiliates engage in a wide range of businesses from investment and commercial banking, lending, asset management and other financial and non-financial services. In the ordinary course of its business, Barclays and its affiliates may actively trade and effect transactions in the equity, debt and/or other securities (and any derivatives thereof) and financial instruments (including loans and other obligations) of MTGE and Annaly for its own account and for the accounts of its customers and, accordingly, may at any time hold long or short positions and investments in such securities and financial instruments.

MTGE Unaudited Prospective Financial Information

MTGE does not publicly disclose projections as to future interest income, performance, earnings or other results due to, among other reasons, the uncertainty and subjectivity of the underlying assumptions and estimates as well as the high likelihood that actual results will vary from any such estimates. As a result, MTGE does not endorse the unaudited prospective financial information included in this document as a reliable indication of future results.

At the request of the special committee s financial advisor, and on behalf of the special committee, the MTGE external manager prepared the MTGE limited unaudited prospective financial information (identified below). This limited unaudited prospective financial information is being included in this document solely because it was made available to the special committee and its financial advisor for use in connection with the financial advisor s analyses and opinion. The MTGE limited unaudited prospective financial information was also made available to Annaly and its financial advisors. While the limited unaudited prospective financial information may also made available to Annaly and its financial advisors. While the limited unaudited prospective financial information presented below was reviewed by the special committee, such information was not reviewed or approved by the full MTGE board of directors. As such, the unaudited prospective financial information may vary from subsequent forecasts, financial plans, guidance and/or actual results due to a number of factors, including (but not limited to) changes in the MTGE investment portfolio and changes in interest rates, trading activity, market valuations, general market and economic conditions, capital commitments, other unexpected changes that cannot be predicted with any certainty, or at all, and the other factors described below. Although presented with numerical specificity, the limited unaudited prospective financial informations and estimates as to future events made by the MTGE external manager.

Additionally, MTGE s future financial results may also materially differ from those expressed in the unaudited prospective financial information due to numerous factors that are beyond MTGE s, Annaly s or anyone else s ability to control or predict, including with respect to the interest rate environment, industry performance, competitive factors, industry consolidation, general business, economic, regulatory, market and financial conditions, as well as matters specific to MTGE s business, including with respect to MTGE s investment and capital allocation strategy and future business initiatives. The assumptions underlying the unaudited prospective financial information may not prove to have been, or may no longer be, accurate.

The MTGE external manager estimated the unaudited prospective financial information in the context of the business, economic, regulatory, market and financial conditions that existed at that time, and the unaudited prospective financial information has not been updated to reflect revised prospects for MTGE s business and investment portfolio, changes in general business, economic, regulatory, market and financial conditions, or any other transaction or event that has occurred or that may occur and that was not anticipated at the time such unaudited prospective financial information was prepared, including costs and expenses of the merger or other obligations or limitations on its business, operations, affairs or activities imposed by the merger agreement.

Specifically, the unaudited prospective financial information as of December 31, 2017, is based on the MTGE unaudited balance sheet and unaudited investment portfolio as of March 31, 2018 and does not take into account any circumstances or events occurring after April 1, 2018, and specifically does not include any financial or other consequences resulting from the May 2, 2018 announcement of the offer and the merger or subsequent integration planning activities to follow. Additionally, the unaudited prospective financial information does not give effect to any other changes that may result from the offer, the merger or the other transactions contemplated by the merger agreement or compliance with the terms of the merger agreement.

Moreover, the unaudited prospective financial information was based upon numerous assumptions, and the realization of any or all of these assumptions is uncertain. As such, stockholders are cautioned not to place undue, if any, reliance on the unaudited prospective financial information included in this document, including in making a decision as to whether to tender their shares of MTGE common stock in the offer. The following are among the assumptions used for preparing the unaudited prospective financial information:

no material change in leverage;

fed funds rate projected increase of 50 basis points over the next four quarters;

no material change in agency mortgage backed securities and non-agency mortgage backed securities spreads;

no capital raises or repurchase or retirement of shares throughout the projection period;

no changes in common shares outstanding; and

no changes in tangible book value over the forecast period.

Given the high degree of uncertainty of the aforementioned assumptions, actual results are likely to differ from the unaudited prospective financial information, and such differences may be material. Without limiting the generality of the foregoing, nothing in the unaudited prospective financial information reflects any change in or intention to change MTGE dividend payment practices, and the merger agreement imposes limitations on the payment of dividends.

THE UNAUDITED PROSPECTIVE FINANCIAL INFORMATION PRESENTED IN THIS DOCUMENT HAS NOT BEEN AND WILL NOT BE UPDATED SINCE THE DATE OF ITS PREPARATION TO REFLECT CIRCUMSTANCES EXISTING AT OR AFTER THE DATE OF ITS PREPARATION OR TO REFLECT THE OCCURRENCE OF FUTURE EVENTS AFTER MARCH 31, 2018, EVEN IN THE EVENT THAT ANY OR ALL OF THE ASSUMPTIONS UNDERLYING SUCH PROSPECTIVE FINANCIAL INFORMATION ARE NO LONGER APPROPRIATE, EXCEPT AS MAY BE REQUIRED BY LAW.

The inclusion of this information should not be regarded as an indication that MTGE, the MTGE external manager, the MTGE board of directors, Barclays, the Offeror, Annaly, the Annaly board of directors or any other recipient of this information considered, or now considers, it to be necessarily predictive of actual future results. There can be no assurance that the prospective results will be realized or that actual results will not be significantly higher or lower than estimated.

Since the unaudited prospective financial information included in this document covers multiple quarterly periods, such information by its nature becomes less predictive with each successive quarterly period. Holders of shares of MTGE common stock are urged to review the information under the caption Risk Factors and SEC filings of MTGE for a description of risk factors with respect to the business of MTGE. See Forward-Looking Statements, Risk Factors and Where to Obtain More Information.

The unaudited prospective financial information was not prepared with a view toward public disclosure, nor was it prepared with a view toward compliance with published guidelines of the SEC or the guidelines established by the American Institute of Certified Public Accountants for preparation and presentation of prospective financial information.

Ernst & Young LLP has not audited, reviewed, compiled or performed any procedures with respect to the accompanying unaudited prospective financial information for the purpose of its inclusion herein, and accordingly, Ernst & Young LLP does not express an opinion or provide any form of assurance on such information or its achievability, and assumes no responsibility for, and disclaims any association with, the unaudited prospective financial information. The Ernst & Young LLP report incorporated by reference into this document relates to MTGE s historical financial information. It does not extend to the prospective financial information of MTGE and should not be read to do so.

The inclusion of the unaudited prospective financial information below should not be deemed an admission or representation by MTGE, the MTGE external manager, the Offeror, Annaly, Barclays, or any of their affiliates with respect to such information or that such information is or was viewed by any such person as material information regarding MTGE, and in fact MTGE views such information as non-material because such information is based on preliminary assessments of future performance that are inconsistent with past experience, practice and results and involves inherent risks and uncertainties.

The unaudited prospective financial information is not being included in this document to influence your decision whether to tender your shares of MTGE common stock in the offer, but solely because such information was provided to Barclays and/or Annaly.

The unaudited prospective financial information should be evaluated, if at all, only in conjunction with the historical financial statements and other information regarding MTGE contained in MTGE s public filings with the SEC. In light of the foregoing factors and the uncertainties inherent in the unaudited prospective financial information, stockholders are cautioned not to place undue, if any, reliance on the unaudited prospective financial information included in this document, including in making a decision as to whether to tender their shares of MTGE common stock in the offer.

MTGE Forecast

(in millions, except per share data)	Q2	-2018	Q	3-2018	Q 4	-2018	2019	2020
Net Spread, Dollar Roll and Healthcare Income,								
excluding catch up amortization	\$	27	\$	26	\$	26	\$ 106	\$ 106
Shares outstanding		46		46		46	46	46
Net Spread, Dollar Roll and Healthcare Income Per								
Common Share, excluding catch up amortization	\$	0.58	\$	0.56	\$	0.57	\$ 2.32	\$2.32
The following information was used for reference:								
(\$ per share)		Q2-20	18	Q3-2018	8	Q4-2018	2019	2020
Median Street Estimate ⁽¹⁾		\$ 0.5	50	\$ 0.50)	\$ 0.50	\$ 1.85	

(1) EPS estimated based on median analyst estimates available to Barclays through FactSet. Annaly Unaudited Prospective Financial Information

Annaly does not publicly disclose projections as to future interest income, performance, earnings or other results due to, among other reasons, the uncertainty and subjectivity of the underlying assumptions and estimates as well as the high likelihood that actual results will vary from any such estimates. As a result, Annaly does not endorse the unaudited prospective financial information included in this document as a reliable indication of future results.

Annaly is including the limited unaudited prospective financial information in this document solely because it was among the financial information made available to Barclays and the MTGE board of directors in connection with their evaluation of the offer and the merger. The limited unaudited prospective financial information presented below was not reviewed or approved by the Annaly board of directors, and is not the result of any formal internal review or process. As such, the unaudited prospective financial information may vary significantly from subsequent forecasts, financial plans, guidance and/or actual results due to a number of factors, including (but not limited to) changes in interest rates, trading activity, market valuations, general market and economic conditions, capital commitments, other unexpected changes that cannot be predicted with any certainty or at all, and the other factors described below. Although presented with numerical specificity, the limited unaudited prospective financial information reflects numerous assumptions and estimates as to future events made by the management of Annaly.

Additionally, Annaly s future financial results may also materially differ from those expressed in the unaudited prospective financial information due to numerous factors that are beyond Annaly s, the combined company s or anyone else s ability to control or predict, including with respect to the interest rate environment, industry performance, competitive factors, industry consolidation, general business, economic, regulatory, market and financial conditions, as well as matters specific to Annaly s business, including with respect to future business initiatives. The

assumptions underlying the unaudited prospective financial information may not prove to have been, or may no longer be, accurate.

Annaly s management estimated the unaudited prospective financial information in the context of the business, economic, regulatory, market and financial conditions that existed at that time, and the unaudited prospective financial information has not been updated to reflect revised prospects for Annaly s business,

changes in general business, economic, regulatory, market and financial conditions, or any other transaction or event that has occurred or that may occur and that was not anticipated at the time such unaudited prospective financial information was prepared.

Specifically, the unaudited prospective financial information does not take into account any circumstances or events occurring after April 19, 2018, the date it was prepared, and specifically does not include any financial consequences resulting from the May 2, 2018 announcement of the offer and the merger or subsequent integration planning activities to follow. Additionally, the unaudited prospective financial information does not give effect to any other changes that may result from the offer, the merger or the other transactions contemplated by the merger agreement.

Moreover, the unaudited prospective financial information was based upon several assumptions, and the realization of any or all of these assumptions is less than certain. As such, stockholders are cautioned not to place undue, if any, reliance on the unaudited prospective financial information included in this document, including in making a decision as to whether to tender their shares of MTGE common stock in the offer. The following are among the assumptions used for preparing the unaudited prospective financial information:

static interest rates through the prospective periods with no change in overall level of interest rates or shape of the yield curve;

no material changes in overall size of portfolio assets;

no material changes in leverage;

no material changes in portfolio net interest margins or net interest spreads; and

no material changes in hedge ratio or net duration measures. Given the relatively static nature of the aforementioned assumptions, actual results are likely to differ from the

unaudited prospective financial information, and such differences may be material.

THE UNAUDITED PROSPECTIVE FINANCIAL INFORMATION PRESENTED IN THIS DOCUMENT HAS NOT BEEN AND WILL NOT BE UPDATED SINCE THE DATE OF ITS PREPARATION TO REFLECT CIRCUMSTANCES EXISTING AFTER THE DATE OF ITS PREPARATION OR TO REFLECT THE OCCURRENCE OF FUTURE EVENTS, EVEN IN THE EVENT THAT ANY OR ALL OF THE ASSUMPTIONS UNDERLYING SUCH PROSPECTIVE FINANCIAL INFORMATION ARE NO LONGER APPROPRIATE, EXCEPT AS MAY BE REQUIRED BY LAW.

The inclusion of this information should not be regarded as an indication that the Offeror, Annaly, its board of directors, Barclays, MTGE, its board of directors or any other recipient of this information considered, or now considers, it to be necessarily predictive of actual future results. There can be no assurance that the prospective results will be realized or that actual results will not be significantly higher or lower than estimated.

Since the unaudited prospective financial information included in this document covers multiple quarterly periods, such information by its nature becomes less predictive with each successive quarterly period. Holders of shares of MTGE common stock are urged to review the section of this document titled Risk Factors and SEC filings of Annaly for a description of risk factors with respect to the business of Annaly. See Forward-Looking Statements, Risk Factors and Where to Obtain More Information .

The unaudited prospective financial information was not prepared with a view toward public disclosure, nor was it prepared with a view toward compliance with published guidelines of the SEC or the guidelines established by the American Institute of Certified Public Accountants for preparation and presentation of prospective financial information.

Ernst & Young LLP has not audited, reviewed, compiled or performed any procedures with respect to the accompanying unaudited prospective financial information (or unaudited prospective financial information presented under the heading MTGE Unaudited Prospective Financial Information) for the purpose of its inclusion herein, and accordingly, Ernst & Young LLP does not express an opinion or provide any form of assurance on such information or its achievability, and assumes no responsibility for, and disclaims any association with, the unaudited prospective financial information. The Ernst & Young LLP report incorporated by reference into this document relates to Annaly s historical financial information. It does not extend to the prospective financial information of Annaly and should not be read to do so.

The inclusion of the unaudited prospective financial information below should not be deemed an admission or representation by Annaly, the Offeror, MTGE, Barclays, or any of their affiliates with respect to such information or that such information is or was viewed by any such person as material information regarding Annaly, and in fact Annaly views such information as non-material because such information is based on preliminary assessments of future performance and involves inherent risks and uncertainties.

The unaudited prospective financial information is not being included in this document to influence your decision whether to tender your shares of MTGE common stock in the offer, but because such information was provided to MTGE s board of directors and MTGE s financial advisor.

The unaudited prospective financial information should be evaluated, if at all, in conjunction with the historical financial statements and other information regarding Annaly contained in Annaly s public filings with the SEC. In light of the foregoing factors and the uncertainties inherent in the unaudited prospective financial information, stockholders are cautioned not to place undue, if any, reliance on the unaudited prospective financial information included in this document, including in making a decision as to whether to tender their shares of MTGE common stock in the offer.

The following table presents the limited unaudited prospective financial data with respect to Annaly that were made available to Barclays and MTGE in connection with their evaluation of the offer and the merger:

	For the quarter ending										
	June 30,	September 30, December 31,			March 31,	1, June 30,		September 30,			
(\$ in 000s)	2018		2018		2018	2019	2019		2019		
Interest Income	\$800,616	\$	817,063	\$	825,171	\$ 833,144	\$843,156	\$	853,468		
Core Earnings (ex. PAA) ⁽¹⁾	\$346,670	\$	341,767	\$	336,924	\$ 329,033	\$328,827	\$	330,225		

(1) Core earnings is a non-GAAP measure and is defined as net income (loss) excluding gains or losses on disposals of investments and termination or maturity of interest rate swaps, unrealized gains or losses on interest rate swaps and investments measured at fair value through earnings, net gains and losses on trading assets, impairment losses, net income (loss) attributable to noncontrolling interest, transaction expenses and certain other non-recurring gains or losses, and inclusive of TBA dollar roll income (a component of Net gains (losses) on trading assets) and realized amortization of MSRs (a component of net unrealized gains (losses) on investments measured at fair value through earnings). Core earnings (excluding PAA) excludes the premium amortization adjustment representing the cumulative impact on prior periods, but not the current period, of quarter-over-quarter changes in estimated long-term prepayment speeds related to Annaly s Agency mortgage-backed securities.

Elections and Proration

MTGE common stockholders electing the mixed consideration will not be subject to proration; however, holders electing the all-cash consideration or the all-stock consideration may receive a different form of consideration than selected. MTGE common stockholders who make the all-cash election or the all-stock election will be subject to proration so that approximately 50.0% of the aggregate consideration in the offer will be paid in cash and approximately 50.0% of the aggregate consideration in the offer will be paid in shares of

Annaly common stock. MTGE common stockholders who validly tender and do not validly withdraw their shares of MTGE common stock in the offer but who do not make a valid election will receive the mixed consideration for their shares of MTGE common stock. See Consequences of Tendering with No Election for more information.

The number of shares of MTGE common stock eligible to receive the all-cash consideration in the offer will be equal to 50.0% multiplied by the sum of the total number of shares making an all-cash election in the offer and the total number of shares making an all-stock election in the offer (such product is referred to as the maximum all-cash shares in offer).

The number of shares of MTGE common stock eligible to receive the all-stock consideration in the offer will be equal to 50.0% multiplied by the sum of the total number of shares making an all-cash election in the offer and the total number of shares making an all-stock election in the offer (such product is referred to as the maximum all-stock shares in offer).

Over-Election of Cash

If the aggregate number of shares attributable to holders of MTGE common stock making an all-cash election in the offer is greater than the maximum all-cash shares in offer, such shares will be subject to proration. To determine the amount of proration, an offer cash proration factor will apply. The offer cash proration factor will be equal to:

the maximum all-cash shares in offer divided by

the aggregate number of shares of MTGE common stock for which an all-cash election in the offer has been made.

Proration will be calculated so that for each MTGE common stockholder making an all-cash election, the number of shares of MTGE common stock entitled to the all-cash consideration will be equal to the number of shares of MTGE common stock for which such stockholder has made an all-cash election, multiplied by the offer cash proration factor (as calculated above), rounded down to the nearest share. The shares of MTGE common stock that do not receive the all-cash consideration as a result of such proration will instead receive the all-stock consideration.

If following proration a MTGE common stockholder would be entitled to receive a fractional share of Annaly common stock, such stockholder will instead receive an amount in cash (without interest) equal to the amount of such fraction multiplied by the volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time.

Over-Election of Cash Example

For purposes of this example, assume the following:

there are 100,000,000 outstanding shares of MTGE common stock;

MTGE stockholders make a mixed election with respect to 50,000,000 (or 50%) shares of MTGE common stock;

MTGE stockholders make the all-cash election with respect to 30,000,000 (or 30%) shares of MTGE common stock;

MTGE stockholders make the all-stock election with respect to 20,000,000 (or 20%) shares of MTGE common stock; and

The 10-day volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time is \$10.41 per share.

In this example, without proration, there would be an over-election of cash because the number of shares of MTGE common stock making the all-cash election is 30,000,000, which is greater than 25,000,000, which is the maximum all-cash shares in offer (calculated as follows: $0.50 \times (30,000,000 + 20,000,000)$). To adjust for the over-election, the number of shares of MTGE common stock making the all-cash election will be multiplied by the offer cash proration factor of 0.8333, with the resulting number of shares rounded down to the nearest share. The offer cash proration factor is calculated by dividing 25,000,000 (which is the maximum all-cash shares in offer) by 30,000,000 (which is the aggregate number of shares of MTGE common stock making an all-cash election in the offer).

In this example, a MTGE stockholder who makes an all-cash election for 100 shares of MTGE common stock would be entitled to receive the all-cash consideration for 83 of those shares (calculated as follows: 100 shares \times 0.8333 offer cash proration factor, rounded down to the nearest share), and the all-stock consideration for the remaining 17 of those shares. This equates to \$1,630.95 in cash (calculated as follows: 83 shares \times \$19.65 all-cash consideration per share) plus 32.3629 shares of Annaly common stock (calculated as follows: 17 shares \times 1.9037 shares of Annaly common stock, the all-stock consideration per share). Because fractional shares of Annaly common stock will be converted to cash, the number of actual shares of Annaly common stock such holder would be entitled to receive would be 32, with the additional 0.3629 shares of Annaly common stock converted into \$3.77 in cash (calculated as follows: 0.3629 \times \$10.41).

See Risk Factors Risk Factors Relating to the Offer and the Merger MTGE common stockholders may not receive all consideration in the form elected.

Over-Election of Stock

If the aggregate number of shares attributable to holders of MTGE common stock making an all-stock election in the offer is greater than the maximum all-stock shares in offer, such shares will be subject to proration. To determine the amount of proration, an offer stock proration factor will apply. The offer stock proration factor will be equal to:

the maximum all-stock shares in offer divided by

the aggregate number of shares of MTGE common stock for which an all-stock election in the offer has been made.

Proration will be calculated so that for each MTGE common stockholder making an all-stock election, the number of shares of MTGE common stock entitled to the all-stock consideration will be equal to the number of shares of MTGE common stock for which such stockholder has made an all-stock election, multiplied by the offer stock proration factor (as calculated above), rounded down to the nearest share. The shares of MTGE common stock that do not receive the all-stock consideration as a result of proration will instead receive the all-cash consideration.

If following proration a MTGE common stockholder would be entitled to receive a fractional share of Annaly common stock, such stockholder will instead receive an amount in cash (without interest) equal to the amount of such fraction multiplied by the volume weighted average closing sale price per share of Annaly common stock as reported

on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time.

Over-Election of Stock Example

For purposes of this example, assume the following:

there are 100,000,000 outstanding shares of MTGE common stock;

MTGE stockholders make the mixed election with respect to 20,000,000 (or 20%) shares of MTGE common stock;

MTGE stockholders make the all-cash election with respect to 10,000,000 (or 10%) shares of MTGE common stock;

MTGE stockholders make the all-stock election with respect to 70,000,000 (or 70%) shares of MTGE common stock; and

The 10-day volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on trading day immediately preceding the acceptance time is \$10.41 per share.

In this example, without proration, there would be an over-election of stock because the number of shares of MTGE common stock making the all-stock election is 70,000,000, which is greater than 40,000,000, which is the maximum all-stock shares in offer (calculated as follows: $0.5 \times (10,000,000 + 70,000,000)$). To adjust for the over-election, the number of shares of MTGE common stock making the all-stock election will be multiplied by the offer stock proration factor of 0.5714, with the resulting number of shares rounded down to the nearest share. The offer stock proration factor is calculated by dividing 40,000,000 (which is the maximum all-stock shares in offer) by 70,000,000 (which is the aggregate number of shares of MTGE common stock making an all-stock election in the offer).

In this example, a MTGE stockholder who makes an all-stock election for 100 shares of MTGE common stock would be entitled to receive the all-stock consideration for 57 of those shares (calculated as follows: 100 shares \times 0.5714 offer stock proration factor, rounded down to the nearest share), and the all-cash consideration for the remaining 43 of those shares. This equates to \$844.95 in cash (calculated as follows: 43 shares \times \$19.65 all-cash consideration per share) plus 108.5109 shares of Annaly common stock (calculated as follows: 57 shares \times 1.9037 shares of Annaly common stock, the all-stock consideration per share). Because fractional shares of Annaly common stock will be converted to cash, the number of actual shares of Annaly common stock such holder would be entitled to receive would be 108, with the additional 0.5109 shares converted into \$5.31 in cash (calculated as follows: 0.5109 \times \$10.41).

See Risk Factors Risk Factors Relating to the Offer and Merger MTGE common stockholders may not receive all consideration in the form elected.

Consequences of Tendering with No Election

MTGE common stockholders who validly tender and do not validly withdraw their shares of MTGE common stock in the offer that do not make an election will be deemed to have elected to receive the mixed consideration.

Distribution of Offering Materials

Table of Contents

This document, the related letter of election and transmittal and other relevant materials will be delivered to record holders of shares of MTGE common stock and to brokers, dealers, commercial banks, trust companies and similar persons whose names, or the names of whose nominees, appear on MTGE s stockholder list or, if applicable, who are listed as participants in a clearing agency s security position listing, so that they can in turn send these materials to beneficial owners of shares of MTGE common stock.

Expiration of the Offer

The offer is scheduled to expire at 5:00 p.m., Eastern Time, on June 18, 2018, unless further extended or terminated. Expiration date means 5:00 p.m., Eastern Time, on June 18, 2018, unless and until the Offeror has

extended or terminated the period during which the offer is open, subject to the terms and conditions of the merger agreement, in which event the term expiration date means the latest time and date at which the offer, as so extended by the Offeror, will expire.

Extension, Termination and Amendment

Subject to the provisions of the merger agreement and the applicable rules and regulations of the Securities and Exchange Commission (SEC), and unless MTGE consents otherwise or the merger agreement is otherwise terminated, the Offeror must (1) extend the offer for one or more successive periods of up to 10 business days each (or for such longer period as may be agreed by MTGE) in order to further seek to satisfy the conditions to the offer in the event that any of the offer conditions (other than the minimum tender condition) have not been satisfied or validly waived as of any then scheduled expiration of the offer, (2) extend the offer for up to two successive periods of 10 business days each (or such longer or shorter period as may be agreed by MTGE) if each of the offer conditions (other than the minimum tender condition) have not been satisfied or validly waived as of any then scheduled expiration of the offer, (2) extend the offer for up to two successive periods of 10 business days each (or such longer or shorter period as may be agreed by MTGE) if each of the offer conditions (other than the minimum tender condition) has been satisfied or validly waived and the minimum tender condition has not been satisfied as of the scheduled expiration of the offer, and (3) extend the offer for the minimum period required by any rule, regulation, interpretation or position of the SEC or its staff or NYSE or Nasdaq that is applicable to the offer or the merger or to the extent necessary to resolve any comments of the SEC or its staff applicable to the offer, the merger, the Schedule TO or the related offer documents. However, the Offeror is not required to extend the offer beyond the outside date.

The merger agreement prohibits the Offeror and Annaly from making certain changes to the offer or waiving certain conditions to the offer without the express written consent of MTGE. Changes to the offer that require the express written consent of MTGE include changes (i) to the terms or conditions to the offer that change the form of consideration to be paid in the offer, (ii) that decrease the amount of consideration in the offer or the number of shares sought in the offer, (iii) that extend the offer (other than extensions required by law or SEC or NYSE or Nasdag regulation, extensions (x) of up to 10 business days each (or such longer period as may be agreed by MTGE) if either any of the conditions to the offer (other than the minimum tender condition) have not been satisfied or validly waived as of the then-scheduled expiration date in order to seek the satisfaction of such conditions, (y) of 10 business days each (or such longer or shorter period as may be agreed by MTGE) if all of the conditions (other than the minimum tender condition) have been satisfied or validly waived as of the then-scheduled expiration date in order to seek the satisfaction of the minimum tender condition, or (z) for any period necessary to meet the notice requirements for a short form merger pursuant to Section 3-106.1(e)(1) of the MGCL), (iv) that impose conditions in the offer not included in the merger agreement, or (v) that amend or modify any other terms or conditions of the offer in any manner that is adverse or would reasonably be expected to be adverse to MTGE common stockholders in their capacities as such. Conditions to the offer that the Offeror and Annaly may not waive without the express written consent of MTGE include (i) the minimum tender condition, (ii) effectiveness of the registration statement on Form S-4 of which this document is a part, (iii) there not having occurred a material adverse effect (with such term as defined in the merger agreement and described under Merger Agreement Material Adverse Effect) with respect to Annaly and its subsidiaries, (iv) the accuracy of Annaly s and the Offeror s representations and warranties, (v) Annaly and the Offeror s compliance with covenants under the merger agreement, (vi) the approval for listing on the NYSE of the shares of Annaly common stock to be issued in the offer and the merger and Series H preferred stock to be issued in the merger, (vii) lack of legal prohibitions, (viii) the receipt of an opinion by MTGE from its legal counsel to the effect that the offer and the merger, taken together, will qualify as a reorganization within the meaning of Section 368(a) of the Code, and (ix) the receipt of a written opinion by MTGE from Annaly s legal counsel to the effect that since a specified date Annaly has and through the expiration date meets the requirements for REIT qualification under the Code, and that the proposed method of operation of Annaly will enable Annaly to continue to meet the requirements for qualification as a REIT under the Code.

The Offeror will effect any extension, termination, amendment or delay by giving oral or written notice to the exchange agent and by making a public announcement as promptly as practicable thereafter. In the case of an extension, any such announcement will be issued no later than 9:00 a.m., Eastern Time, on the next business day

following the previously scheduled expiration date. Subject to applicable law (including Rules 14d-4(c) and 14d-6(d) under the Exchange Act, which require that any material change in the information published, sent or given to stockholders in connection with the offer be promptly disseminated to stockholders in a manner reasonably designed to inform them of such change) and without limiting the manner in which the Offeror may choose to make any public announcement, the Offeror assumes no obligation to publish, advertise or otherwise communicate any such public announcement of this type other than by issuing a press release.

If the Offeror materially changes the terms of the offer or the information concerning the offer, or if the Offeror waives a material condition of the offer, the Offeror will extend the offer to the extent legally required under the Exchange Act. If, prior to the expiration date, the Offeror changes the percentage of shares being sought or the consideration offered, that change will apply to all MTGE common stockholders whose shares are accepted for exchange pursuant to the offer. If, at the time notice of that change is first published, sent or given to MTGE common stockholders, the offer is scheduled to expire at any time earlier than the 10th business day from and including the date that such notice is first so published, sent or given, the Offeror will extend the offer until the expiration of that 10 business day period. For purposes of the offer, a business day means any day other than a Saturday, Sunday or federal holiday and consists of the time period from 12:01 a.m. through 12:00 midnight, Eastern Time.

No subsequent offering period will be available following the expiration of the offer.

Exchange of Shares; Delivery of Cash and Annaly Shares

Annaly has retained Computershare as the depositary and exchange agent for the offer and the merger (the exchange agent) to handle the exchange of shares for the common transaction consideration in each of the offer and the merger. In addition, the exchange agent will handle the exchange of shares of MTGE Series A preferred stock for the shares of Annaly Series H preferred stock.

Upon the terms and subject to the satisfaction or waiver of the conditions of the offer (including, if the offer is extended or amended, the terms and conditions of any such extension or amendment), the Offeror will accept for exchange, and will exchange, shares validly tendered and not validly withdrawn promptly after the expiration date. In all cases, a MTGE stockholder will receive consideration for tendered shares of MTGE common stock only after timely receipt by the exchange agent of certificates for those shares, or a confirmation of a book-entry transfer of those shares into the exchange agent s account at The Depository Trust Company (DTC), a properly completed and duly executed letter of election and transmittal, or an agent s message in connection with a book-entry transfer, and any other required documents.

For purposes of the offer, the Offeror will be deemed to have accepted for exchange shares validly tendered and not validly withdrawn if and when it notifies the exchange agent of its acceptance of those shares pursuant to the offer. The exchange agent will deliver to the applicable MTGE common stockholders any cash and shares of Annaly common stock issuable in exchange for shares validly tendered and accepted pursuant to the offer promptly after receipt of such notice. The exchange agent will act as the agent for tendering MTGE common stockholders for the purpose of receiving cash and shares of Annaly common stock from the Offeror and transmitting such cash and stock to the tendering MTGE common stockholders. MTGE common stockholders will not receive any interest on any cash that the Offeror pays in the offer, even if there is a delay in making the exchange.

If the Offeror does not accept any tendered shares of MTGE common stock for exchange pursuant to the terms and conditions of the offer for any reason, or if certificates are submitted representing more shares than are tendered for, the Offeror will return certificates for such unexchanged shares without expense to the tendering stockholder or, in the case of shares tendered by book-entry transfer into the exchange agent s account at DTC pursuant to the procedures set

forth below in Procedure for Tendering, the shares to be returned will be credited to an account maintained with DTC as soon as practicable following expiration or termination of the offer.

Withdrawal Rights

MTGE common stockholders can withdraw tendered shares of MTGE common stock at any time until the expiration date and, if the Offeror has not agreed to accept the shares for exchange on or prior to July 15, 2018, MTGE common stockholders can thereafter withdraw their shares from tender at any time after such date until the Offeror accepts shares for exchange.

For the withdrawal of shares to be effective, the exchange agent must receive a written notice of withdrawal from the MTGE stockholder at one of its addresses set forth elsewhere in this document, prior to the expiration date. The notice must include the MTGE stockholder s name, address, social security number, the certificate number(s), the number of shares to be withdrawn and the name of the registered holder, if it is different from that of the person who tendered those shares, and any other information required pursuant to the offer or the procedures of DTC, if applicable.

A financial institution must guarantee all signatures on the notice of withdrawal, unless the shares to be withdrawn were tendered for the account of an eligible institution. Most banks, savings and loan associations and brokerage houses are able to provide signature guarantees. An eligible institution is a financial institution that is a participant in the Securities Transfer Agents Medallion Program.

If shares have been tendered pursuant to the procedures for book-entry transfer discussed under the section entitled

Procedure for Tendering, any notice of withdrawal must specify the name and number of the account at DTC to be credited with the withdrawn shares and must otherwise comply with DTC s procedures. If certificates have been delivered or otherwise identified to the exchange agent, the name of the registered holder and the serial numbers of the particular certificates representing the shares withdrawn must also be furnished to the exchange agent, as stated above, prior to the physical release of such certificates.

The Offeror will decide all questions as to the form and validity (including time of receipt) of any notice of withdrawal in its sole discretion, and its decision will be final and binding to the fullest extent permitted by law, subject to the rights of holders of shares of MTGE common stock to challenge such decision in a court of competent jurisdiction. None of the Offeror, Annaly, MTGE, the exchange agent, the information agent or any other person is under any duty to give notification of any defects or irregularities in any tender or notice of withdrawal or will incur any liability for failure to give any such notification. Any shares validly withdrawn will be deemed not to have been validly tendered for purposes of the offer. However, a MTGE stockholder may re-tender withdrawn shares by following the applicable procedures discussed under the section Procedure for Tendering at any time prior to the expiration date.

Procedure for Tendering

For a MTGE common stockholder to validly tender shares of MTGE common stock held of record pursuant to the offer, a MTGE common stockholder must:

if such shares are in certificated form, deliver to the exchange agent a properly completed and duly executed letter of election and transmittal, along with any required signature guarantees and any other documents required by the letter of election and transmittal, and certificates for tendered shares of MTGE common stock held in certificate form, at one of its addresses set forth elsewhere in this document before the expiration date; or

if such shares are in electronic book-entry form, deliver an agent s message in connection with a book-entry transfer, and any other required documents, to the exchange agent at its address set forth elsewhere in this document and follow the other procedures for book-entry tender set forth herein, all of which must be received by the exchange agent prior to the expiration date.

If shares of MTGE common stock are held in street name (*i.e.*, through a broker, dealer, commercial bank, trust company or other nominee), those shares may be tendered by the nominee holding such shares by book-

entry transfer through DTC. To validly tender such shares held in street name, MTGE common stockholders should instruct such nominee to do so prior to the expiration date.

The term agent s message means a message transmitted by DTC to, and received by, the exchange agent and forming a part of a book-entry confirmation, which states that DTC has received an express acknowledgment from the DTC participant tendering the shares that are the subject of such book-entry confirmation, that such participant has received and agrees to be bound by the terms of the letter of election and transmittal and that the Offeror may enforce that agreement against such participant.

The exchange agent has established an account with respect to the shares at DTC in connection with the offer, and any financial institution that is a participant in DTC may make book-entry delivery of shares by causing DTC to transfer such shares prior to the expiration date into the exchange agent s account in accordance with DTC s procedure for such transfer. However, although delivery of shares may be effected through book-entry transfer at DTC, the letter of election and transmittal with any required signature guarantees, or an agent s message, along with any other required documents, must, in any case, be received by the exchange agent at one of its addresses set forth on the back cover of this document prior to the expiration date. The Offeror cannot assure MTGE common stockholders that book-entry delivery of shares by means of delivery of MTGE share certificates. We are not providing for guaranteed delivery procedures and, therefore, you must allow sufficient time for the necessary tender procedures to be completed during normal business hours of DTC prior to the expiration date. Tenders received by the exchange agent after the expiration date will be disregarded and of no effect.

Signatures on all letters of election and transmittal must be guaranteed by an eligible institution, except in cases in which shares are tendered either by a registered holder of shares who has not completed the box entitled Special Issuance Instructions or the box entitled Special Delivery Instructions on the letter of election and transmittal or for the account of an eligible institution.

If the certificates for shares are registered in the name of a person other than the person who signs the letter of election and transmittal, or if certificates for unexchanged shares are to be issued to a person other than the registered holder(s), the certificates must be endorsed or accompanied by appropriate stock powers, in either case signed exactly as the name or names of the registered owner or owners appear on the certificates, with the signature(s) on the certificates or stock powers guaranteed by an eligible institution.

The method of delivery of MTGE share certificates and all other required documents, including delivery through DTC, is at the option and risk of the tendering MTGE stockholder, and delivery will be deemed made only when actually received by the exchange agent. If delivery is by mail, the Offeror recommends registered mail with return receipt requested and properly insured. In all cases, MTGE stockholders should allow sufficient time to ensure timely delivery.

To prevent U.S. federal backup withholding with respect to any cash payments, each MTGE stockholder that is a

United States person (as defined in the Code), other than a stockholder exempt from backup withholding who otherwise establishes its exemption from U.S. federal backup withholding, must provide the exchange agent with its correct taxpayer identification number and certify that it is not subject to backup withholding of U.S. federal income tax by completing the Internal Revenue Service (IRS) Form W-9 included in the letter of election and transmittal. Certain stockholders (including, among others, certain foreign persons) are not subject to these backup withholding requirements. In order for a MTGE stockholder that is a foreign person to qualify as an exempt recipient for purposes of U.S. federal backup withholding, the stockholder must submit an IRS Form W-8BEN or IRS Form W-8BEN-E, as applicable, or other applicable IRS Form W-8, signed under penalties of perjury, attesting to such person s exempt

status. In addition, MTGE stockholders that are foreign persons may be subject to U.S. federal withholding tax at a rate of 30% (or such lower rate as may be specified by an applicable income tax treaty) with respect to cash received pursuant to the offer and/or the merger. See the discussion under Material U.S. Federal Income Tax Consequences.

The tender of shares pursuant to any of the procedures described above will constitute a binding agreement between the Offeror and the tendering MTGE stockholder upon the terms and subject to the satisfaction or waiver of the conditions of the offer (including, if the offer is extended or amended, the terms and conditions of any such extension or amendment).

No Guaranteed Delivery

We are not providing for guaranteed delivery procedures and, therefore, MTGE common stockholders must allow sufficient time for the necessary tender procedures to be completed during normal business hours of DTC prior to the expiration date. MTGE common stockholders must tender their shares of MTGE common stock in accordance with the procedures set forth in this document. In all cases, the Offeror will exchange shares tendered and accepted for exchange pursuant to the offer only after timely receipt by the exchange agent of certificates for shares (or timely confirmation of a book-entry transfer of such shares into the exchange agent s account at DTC as described above), a properly completed and duly executed letter of election and transmittal (or an agent s message in connection with a book-entry transfer) and any other required documents.

Grant of Proxy

By executing a letter of election and transmittal as set forth above, a MTGE stockholder irrevocably appoints the Offeror s designees as such MTGE stockholder s attorneys-in-fact and proxies, each with full power of substitution, to the full extent of such stockholder s rights with respect to its shares tendered and accepted for exchange by the Offeror and with respect to any and all other shares and other securities issued or issuable in respect of those shares on or after the expiration date. That appointment is effective, and voting rights will be affected, when and only to the extent that the Offeror accepts tendered shares of MTGE common stock for exchange pursuant to the offer and deposits with the exchange agent the cash consideration or the shares of Annaly common stock consideration for such shares. All such proxies will be considered coupled with an interest in the tendered shares of MTGE common stock and therefore will not be revocable. Upon the effectiveness of such appointment, all prior proxies that the MTGE stockholder has given will be revoked, and such stockholder may not give any subsequent proxies (and, if given, they will not be deemed effective). The Offeror s designees will, with respect to the shares for which the appointment is effective, be empowered, among other things, to exercise all of such stockholder s voting and other rights as they, in their sole discretion, deem proper at any annual, special or adjourned meeting of the MTGE stockholders or otherwise.

The Offeror reserves the right to require that, in order for shares to be deemed validly tendered, immediately upon the exchange of such shares, the Offeror must be able to exercise full voting rights with respect to such shares. However, prior to acceptance for exchange by the Offeror in accordance with terms of the offer, the appointment will not be effective, and the Offeror will have no voting rights as a result of the tender of shares.

Fees and Commissions

Tendering registered MTGE stockholders who tender shares directly to the exchange agent will not be obligated to pay any charges or expenses of the exchange agent or any brokerage commissions. Tendering MTGE stockholders who hold MTGE shares through a broker or bank should consult that institution as to whether or not such institution will charge the stockholder any service fees in connection with tendering shares pursuant to the offer. Except as set forth in the instructions to the letter of election and transmittal, transfer taxes on the exchange of shares pursuant to the offer will be paid by the Offeror.

Matters Concerning Validity and Eligibility

The Offeror will determine questions as to the validity, form, eligibility (including time of receipt) and acceptance for exchange of any tender of shares, in its sole discretion, and its determination will be final and

binding to the fullest extent permitted by law, subject to the rights of holders of shares of MTGE common stock to challenge such determination in a court of competent jurisdiction. The Offeror reserves the absolute right to reject any and all tenders of shares that it determines are not in the proper form or the acceptance of or exchange for which may be unlawful. The Offeror also reserves the absolute right to waive any defect or irregularity in the tender of any shares. No tender of shares will be deemed to have been validly made until all defects and irregularities in tenders of such shares have been cured or waived by the Offeror. None of the Offeror, Annaly, MTGE, the exchange agent, the information agent nor any other person will be under any duty to give notification of any defects or irregularities in the tender of any shares or will incur any liability for failure to give any such notification. The Offeror s interpretation of the terms and conditions of the offer (including the letter of election and transmittal and instructions thereto) will be final and binding to the fullest extent permitted by law.

MTGE common stockholders who have any questions about the procedure for tendering shares in the offer should contact the information agent at the address and telephone number set forth elsewhere in this document.

Announcement of Results of the Offer

Annaly will announce the final results of the offer, including whether all of the conditions to the offer have been satisfied or waived and whether the Offeror will accept the tendered shares of MTGE common stock for exchange, as promptly as practicable following the expiration date. The announcement will be made by a press release in accordance with applicable securities laws and stock exchange requirements.

Ownership of Annaly Common Stock After the Offer and the Merger

It is estimated that former common stockholders of MTGE will own in the aggregate approximately 4% of the outstanding shares of common stock of Annaly immediately following the consummation of the offer and the merger, assuming that:

Annaly acquires through the offer and the merger 100% of the outstanding shares of MTGE common stock;

Annaly issues 43,631,378 shares of Annaly common stock as part of the consideration in the offer and the merger; and

immediately following completion of the offer and the merger, there are 1,203,328,965 shares of Annaly common stock outstanding (calculated by adding 1,159,697,587, the number of shares of Annaly common stock outstanding as of May 14, 2018, plus 43,631,378, the number of shares of Annaly common stock estimated to be issued as part of the offer and merger consideration).

Purpose of the Offer and the Merger; Dissenters Rights

Purpose of the Offer and the Merger

The purpose of the offer is for Annaly to acquire control of, and ultimately the entire common equity interest in, MTGE. The offer, as the first step in the acquisition of MTGE, is intended to facilitate the acquisition of MTGE. The purpose of the merger is for Annaly to acquire all of the issued and outstanding shares of MTGE common stock it did

Table of Contents

not acquire in the offer. If the offer is successful, Annaly intends to consummate the merger promptly after the consummation of the offer. After the merger, the MTGE business will be held in a wholly owned subsidiary of Annaly, and the former MTGE stockholders will no longer have any direct ownership interest in the surviving corporation.

No Stockholder Approval

If the offer is consummated, Annaly is not required to and will not seek the approval of MTGE s remaining public stockholders before effecting the merger. Section 3-106.1 of the MGCL provides that following

consummation of a successful tender offer for any and all of the outstanding shares of the target corporation, and subject to certain other statutory requirements, if the acquiring corporation owns at least the percentage of the shares, and of each class or series of the shares, of the target corporation that would otherwise be required to approve a merger involving the target corporation, and the other stockholders receive the same consideration for their stock in the merger as was payable in the tender offer, the acquiring corporation can effect a merger without the action of the other stockholders of the target corporation. Accordingly, if Annaly consummates the offer, it intends to effect the closing of the merger without a vote of the MTGE stockholders in accordance with Section 3-106.1 of the MGCL.

No Rights of Objecting Stockholders

Pursuant to the MGCL and MTGE s charter, no appraisal rights, rights of objecting stockholders or dissenters rights are available to MTGE stockholders in connection with the offer or the merger.

Going Private Transactions

The SEC has adopted Rule 13e-3 under the Exchange Act, which is applicable to certain going private transactions, and which may under certain circumstances be applicable to the merger or another business combination following the purchase of shares pursuant to the offer in which the Offeror seeks to acquire the remaining shares not held by it. The Offeror believes that Rule 13e-3 will not be applicable to the merger because it is anticipated that the merger will be effected within one year following the consummation of the offer and, in the merger, stockholders will receive the same consideration as that paid in the offer.

Plans for MTGE

In connection with the offer, Annaly has reviewed and will continue to review various possible business strategies that it might consider in the event that the Offeror acquires control of MTGE. Following a review of additional information regarding MTGE, these changes could include, among other things, changes in MTGE s business, operations, personnel, corporate structure and capitalization.

Delisting and Termination of Registration

If MTGE qualifies for termination of registration under the Exchange Act after the offer is consummated, Annaly intends to seek to have MTGE withdraw the MTGE common stock and preferred stock from listing on Nasdaq and to terminate the registration of MTGE common stock and preferred stock under the Exchange Act. See Effect of the Offer on the Market for MTGE Shares; Nasdaq Listing; Registration Under the Exchange Act; Margin Regulations.

Board of Directors and Management

Upon consummation of the merger, the directors and officers of the Offeror immediately prior to the merger will be the directors and officers of the surviving corporation from and after the effective time of the merger.

Effect of the Offer on the Market for the MTGE Shares; Nasdaq Listing; Registration Under the Exchange Act; Margin Regulations

Effect of the Offer on the Market for MTGE Shares

If the offer is successful, there will be no market for the MTGE common stock because the Offeror intends to consummate the merger promptly following the consummation of the offer.

Table of Contents

Nasdaq Listing

The MTGE common stock and the MTGE Series A preferred stock are currently listed on Nasdaq. Immediately following the consummation of the merger (which is expected to occur promptly following the

consummation of the offer), the MTGE common stock and the MTGE Series A preferred stock will no longer meet the requirements for continued listing on Nasdaq because the only stockholder of the surviving corporation will be Annaly. Nasdaq requires, among other things, that any listed shares have at least 450 total stockholders, provided the listed shares satisfy certain trading volume requirements, or 2,200 total stockholders otherwise. Immediately following the consummation of the merger, Annaly expects to cause the delisting of the MTGE common stock and preferred stock from Nasdaq.

Margin Regulations

The shares of MTGE common stock are currently margin securities under the Regulations of the Board of Governors of the Federal Reserve System (the Federal Reserve Board), which designation has the effect, among other effects, of allowing brokers to extend credit on the collateral of shares of MTGE common stock. Depending upon factors similar to those described above regarding the market for MTGE common stock and stock listing, it is possible that, following the offer, the shares of MTGE common stock would no longer constitute margin securities for the purposes of the margin regulations of the Federal Reserve Board and, therefore, could no longer be used as collateral for loans made by brokers.

Registration under the Exchange Act

The MTGE common stock and preferred stock are currently registered under the Exchange Act. Such registration may be terminated upon application by MTGE to the SEC if MTGE shares are neither listed on a national securities exchange nor held by 300 or more holders of record. Termination of registration of MTGE shares under the Exchange Act would substantially reduce the information required to be furnished by MTGE to its stockholders and to the SEC and would make certain provisions of the Exchange Act no longer applicable to MTGE, such as the short-swing profit recovery provisions of Section 16(b) of the Exchange Act, the requirement of furnishing a proxy statement pursuant to Section 14(a) of the Exchange Act in connection with meetings of stockholders and the related requirement of furnishing an annual report to stockholders and the requirements of Rule 13e-3 under the Exchange Act with respect to going private transactions. Furthermore, the ability of affiliates of MTGE and persons holding restricted securities of MTGE to dispose of such securities pursuant to Rule 144 promulgated under the Securities Act may be impaired. If registration of MTGE shares under the Exchange Act were terminated, MTGE shares would no longer be margin securities or be eligible for listing on Nasdaq as described above. After consummation of the offer and the merger, Annaly and the Offeror currently intend to cause MTGE to terminate the registration of MTGE shares under the Exchange Act may be impaired.

Conditions of the Offer

Notwithstanding any other provisions of the offer and in addition to the Offeror s rights to extend, amend or terminate the offer in accordance with the terms and conditions of the merger agreement, the Offeror and Annaly are not required to accept for exchange or, subject to any applicable rules and regulations of the SEC (including Rule 14e-1(c) under the Exchange Act), exchange the common transaction consideration for any tendered shares of MTGE common stock, if at any expiration of the offer any of the following conditions have not been satisfied or waived:

1. *Minimum Tender Condition* MTGE stockholders having validly tendered and not validly withdrawn in accordance with the terms of the offer and prior to the expiration of the offer a number of shares of MTGE common stock that, together with any shares of MTGE common stock then owned by Annaly and the Offeror, represents at least a majority of the then-outstanding shares of MTGE common stock at any

expiration of the offer;

2. *Effectiveness of Form S-4* The registration statement on Form S-4, of which this document is a part, having become effective under the Securities Act, and must not be the subject of any stop order or proceeding seeking a stop order, at any expiration of the offer;

- 3. *No MTGE Material Adverse Effect* There not having occurred and continuing prior to the expiration of the offer any change, effect, development, circumstance, condition, state of facts, event or occurrence after the date of the merger agreement that would reasonably be expected to have, individually or in the aggregate, a material adverse effect on MTGE and its subsidiaries (with such term as defined in the merger agreement and described under Merger Agreement Material Adverse Effect);
- 4. *No Annaly Material Adverse Effect* There not having occurred and continuing prior to the expiration of the offer any change, effect, development, circumstance, condition, state of facts, event or occurrence after the date of the merger agreement that would reasonably be expected to have, individually or in the aggregate, a material adverse effect on Annaly and its subsidiaries (with such term as defined in the merger agreement and described under Merger Agreement Material Adverse Effect);
- 5. Accuracy of MTGE s Representations and Warranties The representations and warranties of MTGE contained in the merger agreement must be true and correct as of the expiration of the offer as though made on and as of the expiration of the offer (except for representations and warranties that by their terms speak specifically as of the date of the merger agreement or another date, in which case as of such date), except where the failure of such representations and warranties to be true and correct (without giving effect to any qualification as to materiality or material adverse effect) have not had and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on MTGE (with such term as defined in the merger agreement and described under Merger Agreement Material Adverse Effect); provided that (1) MTGE s representations and warranties related to the absence of a material adverse effect on MTGE (with such term as defined in the merger agreement and described under Merger Agreement Material Adverse Effect) having occurred from December 31, 2017 through May 2, 2018 (the date of the merger agreement), its qualification as a REIT under the Code and its exemption under the Investment Company Act of 1940, as amended (the Investment Company Act) must be true and correct in all respects, (2) MTGE s representations and warranties related to its subsidiaries, restricted stock units schedule, authority, the opinion of its financial advisor, state takeover statutes and brokers must be true and correct in all material respects, and (3) MTGE s representations and warranties related to its capitalization must be true and correct in all respects, except for any *de minimis* exceptions;
- 6. *MTGE s Compliance with Covenants* MTGE must have in all material respects performed or complied with the obligations, agreements and covenants required to be performed or complied with by it under the merger agreement at or prior to the expiration of the offer;
- 7. Accuracy of Annaly s and the Offeror s Representations and Warranties The representations and warranties of Annaly and the Offeror contained in the merger agreement must be true and correct as of the expiration of the offer as though made on and as of the expiration of the offer (except for representations and warranties that by their terms speak specifically as of the date of the merger agreement or another date, in which case as of such date), except where the failure of such representations and warranties to be true and correct (without giving effect to any qualification as to materiality or material adverse effect) have not had and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on Annaly (with such term as defined in the merger agreement and described under Merger Agreement Material Adverse Effect); provided that (1) Annaly s representations and warranties related to the absence of a material adverse effect on Annaly (with such term as defined in the merger agreement and effined in the merger agreement and

described under Merger Agreement Material Adverse Effect) having occurred from December 31, 2017 through May 2, 2018 (the date of the merger agreement), its qualification as a REIT under the Code, and its exemption under the Investment Company Act must be true and correct in all respects, (2) Annaly s representations and warranties related to authority must be true and correct in all material respects, and (3) Annaly s representations related to its capitalization must be true and correct in all respects, except for any *de minimis* exceptions;

- 8. *Annaly s and Offeror s Compliance with Covenants* Annaly and the Offeror must have in all material respects performed or complied with the agreements or covenants required to be performed or complied with by them under the merger agreement on or prior to the expiration of the offer;
- 9. *Listing of Annaly Common Stock and Annaly Preferred Stock* The shares of Annaly common stock to be issued in the offer and the merger and the shares of Annaly Series H preferred stock to be issued in the merger having been approved for listing on the NYSE, subject to official notice of issuance;
- 10. *No Legal Prohibition* No law, order or injunction (whether temporary, preliminary or permanent) restraining or enjoining or otherwise prohibiting the consummation of the offer or the merger must have been enacted, issued, promulgated or granted by a governmental entity of competent jurisdiction;
- 11. *Regulatory Approvals* The required approvals of HUD, the lenders/servicers of certain loans financed by HUD and Fannie Mae, if required, must have been obtained at or prior to the expiration of the offer;
- 12. *MTGE Transaction Tax Opinion* MTGE must have received an opinion of Cooley LLP, counsel to MTGE, in form and substance reasonably satisfactory to MTGE, dated as of the expiration date, to the effect that, based on facts, representations and assumptions described or referred to in such opinion, the offer and the merger, taken together, will qualify as a reorganization within the meaning of Section 368(a) of the Code;
- 13. **REIT Tax Opinion with Respect to Annaly** MTGE shall have received a written opinion of Hunton Andrews Kurth LLP, tax counsel to Annaly, in form and substance as set forth in an exhibit to the merger agreement and with such changes as are mutually agreeable to Annaly and MTGE, dated as of the expiration date, to the effect that, based on facts, representations and assumptions described or referred to in such opinion, at all times since the taxable year ended December 31, 2014 and through the expiration date, Annaly has been organized and operated in conformity with the requirements for qualification as a REIT under the Code, and the proposed method of operation of Annaly will enable Annaly to continue to meet the requirements for qualification as a REIT under the Code;
- 14. *Annaly Transaction Tax Opinion* Annaly must have received an opinion of Wachtell, Lipton, Rosen & Katz, counsel to Annaly, in form and substance reasonably satisfactory to Annaly, dated as of the expiration date, to the effect that, based on facts, representations and assumptions described or referred to in such opinion, the offer and the merger, taken together, will qualify as a reorganization within the meaning of Section 368(a) of the Code;
- 15. **REIT Tax Opinion with Respect to MTGE** Annaly shall have received a written opinion of Hogan Lovells US LLP, tax counsel to MTGE, in form and substance as set forth in an exhibit to the merger agreement and with such changes as are mutually agreeable to Annaly and MTGE, dated as of the expiration date, to the effect that, based on facts, representations and assumptions described or referred to in such opinion, at all times commencing with its taxable year ended December 31, 2011 and through and including the taxable year that ends on the effective time of the merger, MTGE has been organized and operated in

conformity with the requirements for qualification as a REIT under the Code;

- 16. *Officer s Certificate* Annaly must have received from MTGE a certificate dated as of the expiration date and signed by MTGE s Chief Executive Officer or Chief Financial Officer certifying as to the satisfaction of the conditions related to the accuracy of MTGE s representations and warranties, MTGE s compliance with covenants and the absence of a MTGE material adverse effect; and
- 17. *No Termination of the Merger Agreement* The merger agreement will not have been terminated in accordance with its terms.

The foregoing conditions are in addition to, and not a limitation of, the rights of Annaly and the Offeror to extend, terminate or modify the offer in accordance with the terms and conditions of the merger agreement.

Except for certain conditions, the waiver of which require the consent of MTGE, the conditions to the offer are for the sole benefit of Annaly and the Offeror and may be asserted by Annaly or the Offeror regardless of the circumstances giving rise to any such condition (other than as a result of any action or inaction by Annaly or the Offeror that is completely within the control of Annaly or the Offeror) or may be waived by Annaly or the Offeror, by express and specific action to that effect, in whole or in part, at any time and from time to time, in each case. However, certain specified conditions may only be waived by Annaly or the Offeror with the express written consent of MTGE. These conditions include the minimum tender condition, the effectiveness of the registration statement on Form S-4 of which this document is a part, the absence of an Annaly material adverse effect, the accuracy of Annaly s and the Offeror s representations and warranties, Annaly s and Offeror s compliance with covenants, the shares of Annaly common and preferred stock to be issued in the offer and the merger having been approved for listing on the NYSE, lack of legal prohibitions, the receipt of an opinion by MTGE from its legal counsel to the effect that the offer and the merger, taken together, will qualify as a reorganization within the meaning of Section 368(a) of the Code, and the receipt of a written opinion by MTGE from Annaly s tax counsel, dated as of the expiration date, to the effect that at all times since December 31, 2014 through the expiration date, Annaly has been organized and operated in conformity with the requirements for qualification as a REIT under the Code and that the proposed method of operation of Annaly will enable Annaly to continue to meet the requirements for qualification as a REIT under the Code. Pursuant to the merger agreement, MTGE has the right to require that Annaly and the Offeror waive the conditions relating to the absence of an Annaly material adverse effect, the accuracy of Annaly s and the Offeror s representations and warranties, Annaly s and Offeror s compliance with covenants, the receipt of an opinion by MTGE from its legal counsel to the effect that the offer and the merger, taken together, will qualify as a reorganization within the meaning of Section 368(a) of the Code, and the receipt of a written opinion by MTGE from Annaly s tax counsel, dated as of the expiration date, to the effect that at all times since December 31, 2014 through the expiration date, Annaly has been organized and operated in conformity with the requirements for qualification as a REIT under the Code and that the proposed method of operation of Annaly will enable Annaly to continue to meet the requirements for qualification as a REIT under the Code. There is no financing condition to the offer.

Certain Legal Matters; Regulatory Approvals

Regulatory Approvals

Annaly and MTGE conduct operations in a number of jurisdictions where regulatory filings or approvals may be required or advisable in connection with the completion of the offer and the merger. In particular, because certain subsidiaries of MTGE are borrowers under loans insured by HUD or serviced loans insured by the U.S. Federal Housing Administration, an office of HUD, or owned by Fannie Mae, the approval of HUD and Fannie Mae may be required to close the offer. Additionally, it is a condition to the closing of the offer that these regulatory approvals, if required, are obtained, which condition may be waived by Annaly in its sole discretion. The merger agreement provides that the parties must use their reasonable best efforts to obtain the approvals that are necessary, proper or advisable to consummate the offer and the merger. Annaly and MTGE have initiated a process to obtain these approvals.

Although we do not expect regulatory authorities to raise any significant objections in connection with their review of the offer and merger, we cannot assure you that we will obtain all required regulatory approvals on a timely basis (or at all) or that these regulatory approvals will not contain terms, conditions or restrictions that would be detrimental to the combined company after the completion of the offer and the merger.

Annaly has been advised that the offer and the merger are exempt from the pre-notification and waiting period requirements of the HSR Act. Therefore, we are not attempting to comply with those requirements. The fact that a transaction is exempt from the requirements of the HSR Act does not preclude the Department of Justice or the

Federal Trade Commission from seeking to prevent the transaction on the ground that it violates the United States antitrust laws. However, we do not believe that the offer or the merger will be viewed as violating the antitrust laws.

Litigation

Subsequent to the public announcement of the proposed acquisition of MTGE by Annaly, three civil actions were filed challenging the adequacy of the disclosures disseminated in connection with the proposed transaction. On May 25, 2018, Jeroen Van Poeck, a purported stockholder of MTGE, commenced an action in the United States District Court for the District of Maryland against MTGE, certain current MTGE directors named therein, Annaly and the Offeror. The complaint asserts claims under Sections 14(e) and 20(a) of the Securities Exchange Act of 1934 challenging the adequacy of the public disclosures made concerning the proposed transaction. The plaintiff seeks, among other things, an injunction preventing consummation of the proposed transaction, rescission of the proposed transaction or damages in the event it is consummated, and the award of attorneys fees and expenses. On May 25, 2018, Giampaolo Dell Acqua, a purported stockholder of MTGE, commenced a putative class action in the United States District Court for the District of Maryland against MTGE, certain current and former MTGE directors named therein, Annaly and the Offeror. The complaint asserts claims under Sections 14(e) and 20(a) of the Securities Exchange Act of 1934 challenging the adequacy of the public disclosures made concerning the proposed transaction. The plaintiff seeks, among other things, an injunction preventing consummation of the proposed transaction, rescission of the proposed transaction or damages in the event it is consummated, and the award of attorneys fees and expenses. On May 30, 2018, Anthony Franchi, a purported stockholder of MTGE, commenced a putative class action in the United States District Court for the District of Maryland against MTGE, certain current MTGE directors named therein, Annaly and Offeror. The complaint asserts claims under Sections 14(e), 14(d) and 20(a) of the Securities Exchange Act of 1934 challenging the adequacy of the public disclosures made concerning the proposed transaction. The plaintiff seeks, among other things, an injunction preventing consummation of the proposed transaction, rescission of the proposed transaction or damages in the event it is consummated, and the award of attorneys fees and expenses. The defendants believe the claims asserted in the actions are without merit.

Interests of Certain Persons in the Offer and the Merger

MTGE directors may have interests in the offer, the merger, and the other transactions contemplated by the merger agreement that are different from, or in addition to, the interests of the MTGE stockholders generally. These interests may create actual or potential conflicts of interest. The MTGE board of directors was aware of these interests during its deliberations and considered them, among other matters, in approving the merger agreement and the transactions contemplated by the merger agreement, as more fully discussed below in The Offer MTGE s Reasons for the Offer and the Merger; Recommendation of the MTGE Board of Directors.

Consideration for MTGE Restricted Stock Units in the Merger

As of May 2, 2018, MTGE s directors held outstanding 38,407 shares of MTGE restricted stock units. Pursuant to, and as further described in, the merger agreement, at the effective time of the merger, each MTGE restricted stock unit held by a MTGE non-executive director that is outstanding, whether vested or unvested, immediately prior to the effective time of the merger will automatically be cancelled, with the holder of such restricted stock unit becoming entitled to receive the mixed consideration in respect each share of MTGE common stock underlying the restricted stock unit immediately prior to the effective time. MTGE non-executive directors who otherwise would be entitled to receive a fractional share of Annaly common stock in respect of their restricted stock unit will instead receive an amount in cash (without interest) equal to the amount of such fraction multiplied by the volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time, less any applicable tax withholding.

The estimated aggregate value, based on a price per share of MTGE common stock of \$19.55, which is the average closing price per share of MTGE common stock on Nasdaq over the first five business days following the first public

Table of Contents

announcement of the merger on May 2, 2018, of the unvested restricted stock units held by the MTGE non-executive directors as of May 16, 2018 is \$512,151.

MTGE Management Agreement

In connection with the execution of the merger agreement, MTGE, MTGE TRS, LLC, the MTGE external manager and Annaly entered into an amendment to the MTGE management agreement, which provides that one month following completion of the merger, the MTGE management agreement will terminate (such time, the termination time). Pursuant to the termination provisions set forth in the Management Agreement, in the event the MTGE external manager is terminated without cause, it is entitled to a termination fee equal to three times the average annual management fee earned by the MTGE external manager during the 24-month period immediately preceding the most recently completed month prior to the termination. Accordingly, MTGE will pay the MTGE external manager \$36,659,461.54 upon the completion of the merger, and \$5,000,000 one month following completion of the merger, at the termination time (for an aggregate termination fee of \$41,659,461.54). The amendment provides that in addition to the termination fee, MTGE will pay the MTGE external manager for accrued management fees through and until the termination time and will reimburse the MTGE external manager for expenses it incurs prior to the termination time in the ordinary course of business and consistent with past practice to the extent reimbursable pursuant to the MTGE management agreement. The indemnification obligations included in the management agreement shall survive the termination. Subject to completion of the transactions contemplated by the merger agreement, Annaly has guaranteed payment of the termination fee and amounts that become owed to the MTGE external manager under the management agreement. For more information, see Management Agreement Termination below.

MTGE Special Committee Fees

Each member of the MTGE special committee earns cash committee fees in connection with their service on the MTGE special committee. Each member of the MTGE special committee receives a monthly retainer in the amount of \$10,000, effective March 15, 2018 and for as long as the MTGE special committee is formed and continues to work, up to an aggregate amount of \$60,000 for each member of the special committee. As of the date of this document, Randy E. Dobbs, Julia L. Coronado and Robert M. Couch earned aggregate fees of \$90,000 for their role on the MTGE special committee.

Indemnification of Directors and Officers

The merger agreement provides that Annaly will and will cause the ultimate surviving company of the merger to indemnify and hold harmless, to the fullest extent permitted under applicable law and provided pursuant to organizational documents of MTGE or its subsidiaries, or any indemnification agreements in existence as of the time of the merger agreement that were provided to Annaly (including the MTGE management agreement), each current and former director, officer and agent of MTGE and its subsidiaries against costs and expenses in connection with claims asserted or claimed prior to, at or after the effective time of the merger, in respect of acts or omissions occurring or alleged to have occurred at or prior to the effective time of the merger, based on or arising out of the fact that such person is or was serving as an officer, director, employee, agent or fiduciary of MTGE or its subsidiaries or any other entity if such service was at the request or for the benefit of MTGE or any of its subsidiaries. In addition, for a period of six years following the effective time of the merger, Annaly and Offeror are required to maintain in effect the provisions in any organizational documents of MTGE and its subsidiaries and contracts (including the MTGE management agreement) of MTGE and its subsidiaries regarding elimination of liability, indemnification, and advancement of expenses in favor of the current and former directors, officers, and agents of MTGE and its subsidiaries that are in existence as of the time of the merger agreement and were provided to Annaly prior to the date of the merger agreement, except to the extent that any such contract provides for an earlier termination. For a more complete description of the indemnification of the officers and directors of MTGE and its subsidiaries, see The Merger Agreement Directors and Officers Indemnification.

Certain Relationships With MTGE

As of the date of this document, Annaly does not own any shares of MTGE common stock. Neither Annaly nor the Offeror have effected any transaction in securities of MTGE in the past 60 days. As of the date of this

document, David Finkelstein, Annaly s Chief Investment Officer, beneficially owns 5,525 shares of MTGE Series A preferred stock. To the best of Annaly s and the Offeror s knowledge, after reasonable inquiry, none of the other persons listed on Annex C hereto, nor any of their respective associates or majority-owned subsidiaries, beneficially owns or has the right to acquire any securities of MTGE or has effected any transaction in securities of MTGE during the past 60 days.

Management Agreement Termination

In connection with the execution of the merger agreement, MTGE, MTGE TRS, LLC, the MTGE external manager and Annaly entered into an amendment to the MTGE management agreement, which provides that one month following completion of the merger, the MTGE management agreement will terminate. MTGE will pay the MTGE external manager \$36,659,461.54 upon the completion of the merger, and \$5,000,000 one month following completion of the merger, at the termination time. The indemnification obligations included in the management agreement shall survive the termination. Subject to completion of the transactions contemplated by the merger agreement, Annaly has guaranteed payment of the termination fee and amounts that become owed to the Manager under the management agreement. The amendment provides that in addition to the termination fee, MTGE will pay the MTGE external manager for accrued management fees through and until the termination time and will reimburse the MTGE external manager for expenses it incurs prior to the termination time in the ordinary course of business and consistent with past practice to the extent reimbursable pursuant to the MTGE management agreement.

The amendment to the MTGE management agreement also provides that the MTGE external manager may not intentionally and knowingly take any action, directly or indirectly, that is prohibited to be taken by MTGE representatives as it relates to such representatives non-solicitation obligations under the merger agreement, and the MTGE external manager is otherwise required to comply with the non-solicitation obligations applicable to MTGE representatives under the merger agreement. However, to the extent that MTGE is permitted to, and in fact does, engage in discussions or negotiations regarding an acquisition proposal in accordance with the merger agreement, the MTGE external manager may assist MTGE in such discussions or negotiations. The MTGE external manager further agreed that, other than those actions that MTGE is permitted to take under the merger agreement, or as required by law, it will not intentionally and knowingly take any action that would reasonably be expected to cause any of the conditions to the offer or the merger to fail to be satisfied.

Except for those rights that, pursuant to the express terms of the amendment to the MTGE management agreement, survive the termination, including indemnification obligations and the payment of certain accrued management fees and expenses, the amendment to the MTGE management agreement provides that each party to the amendment to the MTGE management agreement and their respective affiliates fully and unconditionally release any claims or liabilities whatsoever that they may have against each other party and the affiliates of each other party arising under, or pursuant to, the MTGE management agreement.

Source and Amount of Funds

The offer and the merger are not conditioned upon any financing arrangements or contingencies.

Annaly estimates the aggregate amount of cash consideration required to purchase the outstanding shares of MTGE common stock and consummate the merger will be approximately \$450.1 million, plus related fees and expenses. Annaly anticipates that the funds needed to complete the transactions will be derived from available cash on hand. Neither Annaly nor the Offeror have any specific alternative financing arrangements or alternative financing plans in connection with the offer or the merger.

Fees and Expenses

Annaly has retained D.F. King & Co., Inc. as information agent in connection with the offer and the merger. The information agent may contact holders of shares by mail, email, telephone, facsimile and personal interview and may request brokers, dealers and other nominee stockholders to forward material relating to the offer and the

merger to beneficial owners of shares. Annaly will pay the information agent reasonable and customary compensation for these services in addition to reimbursing the information agent for its reasonable out-of-pocket expenses. Annaly agreed to indemnify the information agent against certain liabilities and expenses in connection with the offer, including certain liabilities under the U.S. federal securities laws.

In addition, Annaly has retained Computershare as exchange agent in connection with the offer and the merger. Annaly will pay the exchange agent reasonable and customary compensation for its services in connection with the offer and the merger, will reimburse the exchange agent for its reasonable out-of-pocket expenses and will indemnify the exchange agent against certain liabilities and expenses, including certain liabilities under the U.S. federal securities laws.

Annaly will reimburse brokers, dealers, commercial banks and trust companies and other nominees, upon request, for customary clerical and mailing expenses incurred by them in forwarding offering materials to their customers. Except as set forth above, neither Annaly nor the Offeror will pay any fees or commissions to any broker, dealer or other person for soliciting tenders of shares pursuant to the offer.

Accounting Treatment

In accordance with accounting principles generally accepted in the United States, Annaly will account for the acquisition of shares through the offer and the merger under the acquisition method of accounting for business combinations.

Stock Exchange Listing

Shares of Annaly common stock and preferred stock are listed on the NYSE. Annaly intends to submit a supplemental listing application to list on the NYSE the shares of Annaly common stock and Annaly Series H preferred stock that Annaly will issue in the offer and the merger. Such listing (subject to official notice of issuance) is a condition to completion of the offer.

Resale of Annaly Common Stock

All shares of Annaly common stock and/or Annaly Series H preferred stock (as applicable) received by MTGE stockholders as consideration in the offer and/or the merger will be freely tradable for purposes of the Securities Act, except for Annaly common stock and/or preferred stock received by any person who is deemed an affiliate of Annaly at the time of the closing of the merger. Annaly common stock and/or preferred stock held by an affiliate of Annaly may be resold or otherwise transferred without registration in compliance with the volume limitations, manner of sale requirements, notice requirements and other requirements under Rule 144 or as otherwise permitted under the Securities Act. This document does not cover resales of Annaly common stock and/or preferred stock received upon completion of the merger by any person, and no person is authorized to make any use of this document in connection with any resale.

Exchange Agent Contact Information

The contact information for the exchange agent for the offer and the merger is:

By First Class, Registered or Certified Mail:

Computershare Trust Company, N.A.

c/o Voluntary Corporate Actions

PO Box 43011

Providence, RI 02940-3011

By Express or Overnight Delivery:

Computershare Trust Company, N.A.

c/o Voluntary Corporate Actions

250 Royall Street, Suite V

Canton, MA 02021

MERGER AGREEMENT

The following summary describes certain material provisions of the merger agreement entered into by Annaly, the Offeror and MTGE, a copy of which is attached hereto as Annex A. This summary may not contain all of the information about the merger agreement that is important to MTGE common stockholders, and MTGE common stockholders are encouraged to read the merger agreement carefully in its entirety. The legal rights and obligations of the parties are governed by the specific language of the merger agreement and not this summary.

The Offer

Pursuant to the terms of the merger agreement, the Offeror is offering to exchange for each outstanding share of MTGE common stock validly tendered and not validly withdrawn in the offer:

\$9.82 in cash; and

0.9519 shares of Annaly common stock. We refer to the above as the mixed consideration.

In lieu of receiving the mixed consideration, holders of shares of MTGE common stock may elect to receive, for each share of MTGE common stock that they hold, (1) \$19.65 in cash (we refer to this election as the all-cash election and this amount as the all-cash consideration) or (2) 1.9037 shares of Annaly common stock (we refer to this election as the all-stock election and this amount as the all-stock consideration). The mixed consideration, the all-cash consideration (as applicable) will be paid without interest and less any applicable withholding taxes.

See The Offer Elections and Proration for a detailed description of the proration procedures applicable to the offer.

The Offeror s obligation to accept for exchange and to exchange shares of MTGE common stock validly tendered and not validly withdrawn in the offer is subject to the satisfaction or waiver by the Offeror of certain conditions, including the valid tender of at least a majority of the shares of MTGE common stock outstanding as of the expiration of the offer, including any shares of MTGE common stock then owned by Annaly and the Offeror (the minimum tender condition), as more fully described under The Offer Conditions of the Offer.

Under the merger agreement, unless MTGE consents otherwise or the merger agreement is otherwise terminated, the Offeror must extend the offer:

for any period required by law, or by any rule, regulation, interpretation or position of the SEC, the SEC s staff or NYSE or Nasdaq applicable to the offer or merger, or to the extent necessary to resolve any comments of the SEC or its staff applicable to the offer, the Schedule TO, the registration statement of which this document is a part, or any other related document;

for one or more successive periods of up to 10 business days each (or such longer period as may be agreed by MTGE) if any of the conditions to the offer (other than the minimum tender condition) have not been satisfied or validly waived as of any then-scheduled expiration of the offer in order to permit satisfaction of such condition or conditions; and

for up to two successive periods of 10 business days each (or for such longer or shorter period as may be agreed by MTGE) if each condition to the offer (other than the minimum tender condition) has been satisfied or validly waived and the minimum tender condition has not been satisfied as of any then-scheduled expiration of the offer, and MTGE requests that the Offeror so extend the offer.

No extension will impair, limit or otherwise restrict the right of the parties to terminate the merger agreement pursuant to its terms.

The merger agreement may be terminated by either Annaly or MTGE if the acceptance time has not occurred by 11:59 p.m., Eastern Time, on February 2, 2019, which is referred to as the outside date (as described below under Termination of the Merger Agreement). The Offeror will not be required to extend the offer beyond the outside date.

The acceptance time for purposes of the merger agreement is the time that the Offeror accepts for payment all shares of MTGE common stock that are validly tendered and not validly withdrawn in the offer.

For a more complete description of the offer, see The Offer.

The Merger

The merger agreement provides that, promptly following the acceptance time, the parties will effect the merger of MTGE with and into the Offeror, with the Offeror continuing as the surviving corporation in the merger. After the merger, the surviving corporation will be a wholly owned subsidiary of Annaly, and the former MTGE stockholders will not have any direct equity ownership interest in the surviving corporation.

Completion and Effectiveness of the Merger

Under the merger agreement, the closing of the merger must occur as promptly as practicable after the acceptance time, and in any case no later than the second business day after satisfaction or permitted waiver of the conditions to closing of the merger, unless Annaly or MTGE agrees otherwise in writing (see Conditions to the Merger). The merger will become effective at the time the articles of merger are accepted for record by the State Department of Assessments and Taxation of Maryland unless a later date (not to exceed 30 days after the acceptance for record of the articles of merger) is specified therein.

Merger Consideration for Common Stock

General

In the merger, MTGE common stockholders will have the opportunity to elect to receive the mixed consideration, the all-cash consideration or the all-stock consideration, subject to proration of the all-cash consideration or the all-stock consideration, in each case without interest and less any applicable withholding taxes (such consideration, the common transaction consideration).

Election Procedures

Each MTGE common stockholder as of immediately prior to the consummation of the merger will be entitled to elect to receive the mixed consideration, the all-cash consideration or the all-stock consideration. The election will be made on a form of election and transmittal that will be mailed promptly after the effective time of the merger to record holders of shares of MTGE common stock and to brokers, dealers, commercial banks, trust companies and similar persons who names, or the names of whose nominees, appear on MTGE s stockholder list or, if applicable, who are listed as participants in a clearing agency s security position listing, so that they can complete the forms of election and transmittal on behalf of beneficial owners of shares of MTGE common stock. To make such election, MTGE common stockholders must submit an effective, properly completed form of election and transmittal to the exchange agent before 5:00 p.m. Eastern Time, on the 20th calendar day following the date on which the forms of election are mailed (the merger election deadline). Holders of record of shares of MTGE common stock who hold such shares as nominees, trustees or in other representative capacities may submit multiple forms of election on behalf of their respective beneficial holders.

Any form of election and transmittal may be revoked or changed by the person who submitted such form of election by written notice received by the exchange agent prior to the merger election deadline. If a form of

election and transmittal is revoked prior to the merger election deadline, the shares of MTGE common stock represented by such form of election and transmittal will be deemed to have elected to receive the mixed consideration, unless a subsequent form of election and transmittal is properly submitted prior to the merger election deadline.

The exchange agent will have reasonable discretion to determine whether any election, revocation or change has been properly or timely made and to disregard immaterial defects in any form of election and transmittal, and any good-faith decisions of the exchange agent regarding such matters will be binding and conclusive. The exchange agent will have no obligation to notify any person of a defect in such person s form of election.

Consequences of Failing to Make an Election in the Merger

MTGE common stockholders with shares to be converted into the common transaction consideration in the merger who do not make an election will be deemed to have elected to receive the mixed consideration.

Proration

MTGE common stockholders electing the mixed consideration will not be subject to proration; however, holders electing the all-cash consideration or the all-stock consideration may receive a different form of consideration than selected. MTGE common stockholders who make the all-cash election or the all-stock election will be subject to proration so that approximately 50.0% of the aggregate consideration in the merger will be paid in cash and approximately 50.0% of the aggregate consideration in the merger will be paid in shares of Annaly common stock.

The number of shares of MTGE common stock eligible to receive the all-cash consideration in the merger will be equal to 50.0% multiplied by the sum of the total number of shares making an all-cash election in the merger and the total number of shares making an all-stock election in the merger (such product is referred to as the maximum all-cash shares in merger).

The number of shares of MTGE common stock eligible to receive the all-stock consideration in the merger will be equal to 50.0% multiplied by the sum of the total number of shares making an all-cash election in the merger and the total number of shares making an all-stock election in the merger (such product is referred to as the maximum all-stock shares in merger).

Over-Election of Cash

If the aggregate number of shares attributable to holders of MTGE common stock making an all-cash election in the merger is greater than the maximum all-cash shares in merger, such shares will be subject to proration. To determine the amount of proration, a merger cash proration factor will apply. The merger cash proration factor will be equal to:

the maximum all-cash shares in merger divided by

the aggregate number of shares of MTGE common stock for which an all-cash election in the merger has been made.

Proration will be calculated so that for each MTGE common stockholder making an all-cash election, the number of shares of MTGE common stock entitled to the all-cash consideration will be equal to the number of shares of MTGE common stock for which such stockholder has made an all-cash election, multiplied by the merger cash proration factor (as calculated above), rounded down to the nearest share. The shares of MTGE common stock that do not receive the all-cash consideration as a result of such proration will instead receive the all-stock consideration.

If following proration a MTGE common stockholder would be entitled to receive a fractional share of Annaly common stock, such stockholder will instead receive an amount in cash (without interest) equal to the amount of such fraction multiplied by the volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time.

Over-Election of Cash Example

For purposes of this example, assume the following:

there are 100,000,000 outstanding shares of MTGE common stock;

MTGE stockholders make a mixed election with respect to 50,000,000 (or 50%) shares of MTGE common stock;

MTGE stockholders make the all-cash election with respect to 30,000,000 (or 30%) shares of MTGE common stock;

MTGE stockholders make the all-stock election with respect to 20,000,000 (or 20%) shares of MTGE common stock; and

The 10-day volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time is \$10.41 per share.

In this example, without proration, there would be an over-election of cash because the number of shares of MTGE common stock making the all-cash election is 30,000,000, which is greater than 25,000,000, which is the maximum all-cash shares in merger (calculated as follows: $0.50 \times (30,000,000 + 20,000,000)$). To adjust for the over-election, the number of shares of MTGE common stock making the all-cash election will be multiplied by the merger cash proration factor of 0.8333, with the resulting number of shares rounded down to the nearest share. The merger cash proration factor is calculated by dividing 25,000,000 (which is the maximum all-cash shares in merger) by 30,000,000 (which is the aggregate number of shares of MTGE common stock making an all-cash election in the merger).

In this example, a MTGE stockholder who makes an all-cash election for 100 shares of MTGE common stock would be entitled to receive the all-cash consideration for 83 of those shares (calculated as follows: 100 shares \times 0.8333 merger cash proration factor, rounded down to the nearest share), and the all-stock consideration for the remaining 17 of those shares. This equates to \$1630.95 in cash (calculated as follows: 83 shares \times \$19.65 all-cash consideration per share) plus 32.3629 shares of Annaly common stock (calculated as follows: 17 shares \times 1.9037 shares of Annaly common stock, the all-stock consideration per share). Because fractional shares of Annaly common stock will be converted to cash, the number of actual shares of Annaly common stock such holder would be entitled to receive would be 32, with the additional 0.3629 shares of Annaly common stock converted into \$3.77 in cash (calculated as follows: 0.3629 \times \$10.41).

See Risk Factors Risk Factors Relating to the Offer and the Merger MTGE common stockholders may not receive all consideration in the form elected.

Over-Election of Stock

If the aggregate number of shares attributable to holders of MTGE common stock making an all-stock election in the merger is greater than the maximum all-stock shares in merger, such shares will be subject to proration. To determine the amount of proration, a merger stock proration factor will apply. The merger stock proration factor will be equal to:

the maximum all-stock shares in merger

divided by

the aggregate number of shares of MTGE common stock for which an all-stock election in the merger has been made.

Proration will be calculated so that for each MTGE common stockholder making an all-stock election, the number of shares of MTGE common stock entitled to the all-stock consideration will be equal to the number of shares of MTGE common stock for which such stockholder has made an all-stock election, multiplied by the merger stock proration factor (as calculated above), rounded down to the nearest share. The shares of MTGE common stock that do not receive the all-stock consideration as a result of proration will instead receive the all-cash consideration.

If, following proration, a MTGE common stockholder would be entitled to receive a fractional share of Annaly common stock, such stockholder will instead receive an amount in cash (without interest) equal to the amount of such fraction multiplied by the volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time.

Over-Election of Stock Example

For purposes of this example, assume the following:

there are 100,000,000 outstanding shares of MTGE common stock;

MTGE stockholders make the mixed election with respect to 20,000,000 (or 20%) shares of MTGE common stock;

MTGE stockholders make the all-cash election with respect to 10,000,000 (or 10%) shares of MTGE common stock;

MTGE stockholders make the all-stock election with respect to 70,000,000 (or 70%) shares of MTGE common stock; and

The 10-day volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time is \$10.41 per share.

In this example, without proration, there would be an over-election of stock because the number of shares of MTGE common stock making the all-stock election is 70,000,000, which is greater than 40,000,000, which is the maximum all-stock shares in merger (calculated as follows: $0.50 \times (10,000,000 + 70,000,000)$). To adjust for the over-election, the number of shares of MTGE common stock making the all-stock election will be multiplied by the merger stock proration factor of 0.5714, with the resulting number of shares rounded down to the nearest share. The merger stock proration factor is calculated by dividing 40,000,000 (which is the maximum all-stock shares in merger) by

70,000,000 (which is the aggregate number of shares of MTGE common stock making an all-stock election in the merger).

In this example, a MTGE stockholder who makes an all-stock election for 100 shares of MTGE common stock would be entitled to receive the all-stock consideration for 57 of those shares (calculated as follows: 100 shares × 0.5714 merger stock proration factor, rounded down to the nearest share), and the all-cash consideration for the remaining 43 of those shares. This equates to \$844.95 in cash (calculated as follows: 43 shares × \$19.65 all-cash consideration per share) plus 108.5109 shares of Annaly common stock (calculated as follows: 57 shares × 1.9037 shares of Annaly common stock, the all-stock consideration per share). Because fractional shares of Annaly common stock will be converted to cash, the number of actual shares of Annaly

common stock such holder would be entitled to receive would be 108, with the additional 0.5109 shares covered into \$5.31 in cash (calculated as follows: $0.5109 \times 10.41).

See Risk Factors MTGE common stockholders may not receive all consideration in the form elected.

No Rights of Objecting Stockholders

Pursuant to MTGE s charter, MTGE stockholders will not be entitled to rights of objecting stockholders as a result of the offer or merger.

Treatment of Shares of MTGE Preferred Stock

The merger agreement provides that each issued and outstanding share of MTGE Series A preferred stock issued and outstanding as of immediately prior to the effective time of the merger will be automatically converted into the right to receive one newly issued share of Annaly Series H preferred stock, which will have rights, preferences, privileges and voting powers substantially the same as those of the MTGE Series A preferred stock.

Exchange of MTGE Stock Certificates or Book-Entry Shares for the Merger Consideration

Annaly has retained Computershare as the depositary and exchange agent for the offer and the merger to handle the exchange of shares of MTGE common stock for the common transaction consideration, and to handle the exchange of shares of MTGE Series A preferred stock for the shares of Annaly Series H preferred stock, as applicable.

To effect the exchange of shares of MTGE common stock, promptly after the effective time of the merger, Annaly will cause the exchange agent to mail to each record holder of shares of MTGE common stock a form of election and letter of transmittal and instructions for surrendering the book-entry shares and/or the stock certificates that formerly represented shares for the common transaction consideration. See Merger Consideration for Common Stock Election Procedures. After surrender to the exchange agent of book-entry shares and/or certificates that formerly represented shares of MTGE common stock for cancellation, together with an executed form of election and transmittal, the record holder of the surrendered book-entry shares and/or certificates will be entitled to receive the applicable common transaction consideration. To effect the exchange of shares of MTGE Series A preferred stock, promptly after the effective time of the merger, the exchange agent will mail to each record holder of MTGE Series A preferred stock a form of letter of transmittal and instructions for surrendering the book-entry shares and/or stock certificates that formerly represented shares for the shares of Annaly Series H preferred stock. After surrender to the exchange agent of book-entry shares and/or certificates that formerly represented stock for cancellation, together with an executed form of letter of transmittal and instructions for surrendering the book-entry shares and/or stock certificates that formerly represented shares for the shares of Annaly Series H preferred stock. After surrender to the exchange agent of book-entry shares and/or certificates that formerly represented MTGE Series A preferred stock for cancellation, together with an executed form of letter of transmittal, the record holder of the surrendered book-entry shares and/or certificates will be entitled to receive shares of Annaly Series H preferred stock.

After the effective time of the merger, each book-entry share and stock certificate formerly representing shares of MTGE common stock and MTGE Series A preferred stock that has not been surrendered will represent only the right to receive upon such surrender the applicable common transaction consideration and shares of Annaly Series H preferred stock to which such applicable holder is entitled by virtue of the merger and any dividends or other distributions payable to such holder upon such surrender.

Fractional Shares

Annaly will not issue fractional shares of Annaly common stock in the offer or the merger. Instead, each holder of shares of MTGE common stock who otherwise would be entitled to receive a fractional share of Annaly common

Table of Contents

stock will be entitled to receive an amount of cash (without interest) equal to such fractional

part of a share of Annaly common stock multiplied by the volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time.

Treatment of MTGE Restricted Stock Units

At the effective time of the merger, each MTGE restricted stock unit whether vested or unvested, will automatically be cancelled, with the holder of such restricted stock unit becoming entitled to receive the mixed consideration in respect of each share of MTGE common stock underlying the restricted stock unit immediately prior to the effective time. Holders of MTGE restricted stock units who otherwise would be entitled to receive a fractional share of Annaly common stock will instead receive an amount in cash (without interest) equal to the amount of such fraction multiplied by the volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time, less any applicable tax withholding.

Representations and Warranties

The merger agreement contains customary representations and warranties of the parties. These include representations and warranties of MTGE with respect to:

organization and qualification;

subsidiaries;

capitalization;

corporate authority relative to the merger agreement and the transactions;

governmental consents;

no violations;

SEC filings;

financial statements;

internal controls and procedures;

absence of undisclosed liabilities;

absence of certain changes or events;

compliance with applicable laws;

permits;

employee benefit plans;

tax matters;

labor matters;

investigations;

litigation;

intellectual property;

real property;

material contracts;

mortgage-backed securities;

insurance;

information supplied;

opinion of financial advisor to the MTGE special committee;

state takeover statutes;

the Investment Company Act; and

finders and brokers.

The merger agreement also contains customary representations and warranties of Annaly and the Offeror, including among other things:

organization and qualification;

subsidiaries;

capitalization;

corporate authority relative to the merger agreement and transactions;

governmental consents;

no violations;

SEC filings;

financial statements;

internal controls and procedures;

absence of undisclosed liabilities;

absence of certain changes or events;

compliance with applicable laws;

permits;

investigations;

litigation;

information supplied;

availability of sufficient funds to consummate the offer and merger;

absence of Offeror activity prior to the date of the merger agreement;

tax matters;

Annaly s lack of ownership of shares of MTGE common stock; and

the Investment Company Act.

The representations and warranties contained in the merger agreement are generally qualified by material adverse effect, as defined in the merger agreement and described below under Material Adverse Effect. The representations and warranties contained in the merger agreement will expire at the effective time of the merger. The representations, warranties and covenants made by MTGE in the merger agreement are qualified by information contained in the confidential disclosure schedules delivered to Annaly and the Offeror in connection with the execution of the merger agreement and by filings that MTGE has made with the SEC prior to the date of the merger agreement are qualified by information contained in the confidential disclosure schedules delivered to MTGE in connection with the execution of the merger agreement and by filings that Annaly has made by Annaly and the Offeror in the merger agreement are qualified by information contained in the confidential disclosure schedules delivered to MTGE in connection with the execution of the merger agreement and by filings that Annaly has made with the SEC prior to the date of the merger agreement. Stockholders are not third-party beneficiaries of these representations and warranties under the merger agreement and should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of MTGE or any of its affiliates or of Annaly or any of its affiliates.

Material Adverse Effect

A material adverse effect with respect to Annaly or MTGE, means (1) any material adverse effect on the ability of such party to consummate the transactions contemplated by the merger agreement, including the offer and the merger, by the outside date, or (2) any change, effect, development, circumstance, condition, state of facts, event or occurrence that, individually or in the aggregate, has had or would reasonably be expected to have a material adverse effect on the financial condition, business, assets, liabilities or operations of such party and its subsidiaries, taken as a whole; provided, however, with respect to clause (2), no such change, effect, development, circumstance, condition, state of facts, event or occurrence resulting or arising from any of the following will be deemed to constitute a material adverse effect or will be taken into account when determining whether a material adverse effect exists or has occurred or is reasonably expected to exist or occur:

- (a) any changes in general U.S. or global economic conditions;
- (b) conditions (or changes therein) in any industry or industries in which such party operates;
- (c) general legal, tax, economic, political and/or regulatory conditions, or changes therein, including any changes affecting financial, credit, foreign exchange or capital market conditions;
- (d) any changes in U.S. GAAP or interpretation thereof or, with respect to MTGE, other applicable accounting regulations or principles or interpretations thereof;
- (e) any adoption, implementation, promulgation, repeal, modification, amendment, reinterpretation, change or proposal of any applicable law of or by any governmental entity;
- (f) any failure by such party to meet internal or published projections or internal budgets, plans or forecasts, in and of itself (provided that the facts or occurrences giving rise or contributing to such failure that are not otherwise excluded from the definition of a material adverse effect may be taken into account);
- (g) any change in the trading price of such party s common stock (provided that facts or occurrences giving rise or contributing to such change that are not otherwise excluded from the definition of material adverse effect may be taken into account);
- (h) any change, effect, development, circumstance, condition, state of facts, event or occurrence arising out of changes in geopolitical conditions, acts of terrorism or sabotage, war, armed hostility, weather or other force majeure events, including a material worsening of conditions threatened or existing as of the date of the merger agreement;

- (i) the execution and delivery of the merger agreement or the consummation of the offer and the merger, or the public announcement (or, with respect to MTGE, the pendency) of the merger agreement or the transactions contemplated thereby (other than for purposes of the parties representation and warranties related to governmental consents and no violations, or any other representation or warranty to the extent such representation or warranty is to address the consequences resulting from the execution and delivery of the merger agreement or the consummation of the transactions contemplated thereby);
- (j) any action expressly required to be taken by such party under the merger agreement;
- (k) any action or failure to take any action that is consented to or requested by the other party, in each case, in writing; and
- any claims, actions, investigations, suits and proceedings brought or threatened by stockholders of either Annaly or MTGE (whether on behalf of MTGE, Annaly or otherwise) asserting allegations of breach of duty relating to the merger agreement or violations of securities laws solely in connection with the offer or the merger;

provided that with respect to the exceptions in clauses (a), (b), (c), (d), (e) and (h), if the change, effect, development, circumstance, condition, state of facts, event or occurrence has had a disproportionate adverse impact on such party relative to other companies of comparable size operating in the same industry, then the

incremental impact of such event shall be taken into account for the purpose of determining whether a material adverse effect has occurred.

No Solicitation of Other Offers by MTGE

Under the terms of the merger agreement, subject to certain exceptions described below, MTGE has agreed that, from the date of the merger agreement until the earlier of the acceptance time or the date the merger agreement is validly terminated, MTGE will not, and will cause its subsidiaries, directors, officers, employees and other representatives not to, directly or indirectly:

solicit, initiate, or knowingly encourage or facilitate (including by way of providing non-public information) any inquiry, proposal or offer, or the making, submission or announcement of any inquiry, proposal or offer, in each case which constitutes or would be reasonably expected to lead to an acquisition proposal (as defined in the merger agreement and as described below);

participate in any negotiations regarding, or furnish to any person any non-public information relating to MTGE or any of its subsidiaries in connection with, an actual or potential acquisition proposal (other than to state that the terms of the merger agreement prohibit such negotiations); or

approve or recommend, or propose publicly to approve or recommend, any acquisition proposal. In addition, under the merger agreement, MTGE has agreed that it will:

immediately cease, and will cause its subsidiaries and its and their respective directors, officers, employees and other representatives to cease, any and all existing discussions or negotiations, or provision of any non-public information to any party, with respect to any acquisition proposal or potential acquisition proposal;

promptly request that each person that executed a confidentiality agreement within the eighteen months preceding the date of the merger agreement with MTGE relating to an acquisition proposal or a potential acquisition proposal promptly destroy or return to MTGE all non-public information relating to such acquisition proposal, MTGE, or MTGE s businesses or assets (provided MTGE is not required to make such request if the confidentiality agreements were executed more than one year prior to the date of the merger agreement and MTGE had previously made such a request); and

take no action to exempt any person (other than Annaly or the Offeror) from the restrictions on business combinations in any applicable state takeover statute or, with respect to a person who has made or is considering making an acquisition proposal, the stock ownership limitations contained in MTGE s charter and bylaws or otherwise cause such restrictions or limitations not to apply.

Notwithstanding the foregoing, submission of an acquisition proposal by the MTGE external manager or its affiliates and their actions related thereto will not be deemed a breach or violation of the foregoing non-solicit provisions of the

merger agreement so long as the directors and officers and other representatives of MTGE not otherwise affiliated with the MTGE external manager did not take actions that would breach the non-solicit provisions with respect to such acquisition proposal.

Under the merger agreement, MTGE is obligated to promptly notify Annaly (and in any event within the earlier of thirty-six hours and one business day) after the receipt by MTGE or any of its subsidiaries or representatives of any acquisition proposal, any proposals or inquiries that would reasonably be expected to lead to an acquisition proposal, or any inquiry or request for non-public information relating to MTGE or any subsidiary by any person who has made or would reasonably be expected to make any acquisition proposal. The notice must indicate the identity of the person making the proposal, inquiry or request and the material terms and conditions of any such proposal or offer or the nature of the information requested pursuant to any such inquiry or request, including copies of all written requests, proposals, correspondence or offers (including any proposed agreements received by MTGE or any of its subsidiaries or representatives). MTGE also must keep Annaly

informed, on a prompt and timely basis, of the status and material terms of any such acquisition proposal or potential proposal (including any amendments or proposed amendments), and as to the nature of any information requested and must provide to Annaly copies of all written materials received or sent by MTGE related to such proposal, inquiry or request. MTGE also must promptly provide Annaly with any material non-public information concerning MTGE provided to any other person in connection with any acquisition proposal that was not previously provided to Annaly.

Notwithstanding the prohibitions described above, if MTGE receives prior to the acceptance time an unsolicited, written acquisition proposal that did not result from a breach of MTGE s non-solicitation obligations, MTGE is permitted to furnish non-public information to such person and engage in discussions or negotiations with such person with respect to the acquisition proposal, as long as:

the MTGE board of directors, upon the recommendation of the MTGE special committee, determines in good faith, after consulting with MTGE s outside legal and financial advisors, that such proposal constitutes, or could reasonably be expected to result in, a superior proposal (as defined in the merger agreement and described below);

the MTGE board of directors determines in good faith, after consulting with MTGE s outside legal and financial advisors, that the failure to take such action would be inconsistent with the directors duties under applicable law; and

prior to providing any such information, the person making the acquisition proposal enters into a confidentiality agreement containing terms that are no less favorable in the aggregate to MTGE than those contained in the confidentiality agreement between Annaly and MTGE (provided that the confidentiality agreement is not required to include a standstill provision) and that does not in any way restrict MTGE or its representatives from complying with its disclosure obligations under the merger agreement.

If MTGE decides to begin providing information or to engage in discussions or negotiations concerning an acquisition proposal, it must promptly notify Annaly (and in any event within the earlier of thirty-six hours and one business day) in writing after making such decision.

An acquisition proposal for purposes of the merger agreement means any offer, proposal or indication of interest from any person or group (other than Annaly or a subsidiary of Annaly) at any time relating to any transaction or series of related transactions involving:

any acquisition or purchase of more than 20% of any class of MTGE voting or equity securities (excluding any MTGE preferred stock);

any tender offer (including a self-tender offer) or exchange offer that would result in any person or group beneficially owning more than 20% of any class of MTGE voting or equity securities if consummated (excluding any MTGE preferred stock);

any merger, consolidation, share exchange, business combination, joint venture, recapitalization or reorganization, or any similar transaction, in each case involving MTGE and any other person, if it would result in the MTGE stockholders prior to such transaction holding less than 80% of the equity interests in the surviving or resulting entity of such transaction; or

any sale, lease (other than in the ordinary course of business), exchange, transfer or other disposition to any person or group of more than 20% of the consolidated assets of MTGE and its subsidiaries (measured by their fair market value).

A superior proposal for purposes of the merger agreement means a bona fide acquisition proposal which the MTGE board of directors determines in good faith (after consultation with MTGE s outside legal and financial advisors) would result in a transaction more favorable to the MTGE stockholders from a financial point of view than the offer and the merger, taking into account all relevant factors as determined in good faith by the

MTGE board of directors (including all the terms and conditions of such proposal (including the transaction consideration, conditionality, timing, certainty of financing and likelihood of consummation of such proposals) and the merger agreement (and any changes to the terms of the merger agreement proposed by Annaly in response to any acquisition proposal)). When determining whether an offer constitutes a superior proposal, references in the definition of the term acquisition proposal to 20% will be changed to be references to 50% and references to 80% will be changed to be references to 50%.

Change of Recommendation

The merger agreement requires the MTGE board of directors to recommend that MTGE common stockholders accept the offer and tender their shares of MTGE common stock in the offer. The MTGE board of directors (including any committee) may not, and may not resolve or agree to (any of the following being a change of recommendation):

approve or recommend, or propose publicly to approve or recommend, any acquisition proposal;

withdraw, change, amend, modify or qualify, in a manner adverse to Annaly, the recommendation of the MTGE board of directors in favor of the offer and the merger, or propose publicly to do any of the foregoing;

fail to publicly recommend against an acquisition proposal within 10 business days of being requested to do so by Annaly, if such acquisition proposal is an exchange or tender offer that has been commenced (provided Annaly can only make one such request with respect to such acquisition proposal, unless the acquisition proposal is amended or supplemented, in which case Annaly can make one request with respect to such amendment or supplement);

fail to reaffirm the recommendation of the MTGE board of directors in favor of the offer and the merger within 10 business days of being requested to do so by Annaly if an acquisition proposal has been publicly disclosed (provided Annaly can only make one such request with respect to such acquisition proposal, unless the acquisition proposal is amended or supplemented, in which case Annaly can make one request with respect to such amendment or supplement); or

enter into any merger agreement, acquisition agreement, reorganization agreement, letter of intent or similar agreement or document relating to, or any other agreement or commitment providing for, any acquisition proposal (other than a confidentiality agreement entered into in accordance with the merger agreement containing terms that are no less favorable in the aggregate to MTGE than those contained in the confidentiality agreement between Annaly and MTGE).

Notwithstanding the foregoing, the MTGE board of directors may take such actions if, prior to the acceptance time:

an intervening event (as defined in the merger agreement and described below) has occurred, and the MTGE board of directors, upon the recommendation of the MTGE special committee, has determined in good faith

(after consultation with MTGE s outside financial advisors and outside legal counsel) that failure to make such change of recommendation would be inconsistent with the directors duties under applicable law; or

MTGE has received an unsolicited acquisition proposal which the MTGE board of directors, upon recommendation of the MTGE special committee, has determined in good faith (after consultation with MTGE s outside financial and legal advisors) both (a) that such unsolicited acquisition proposal is a superior proposal, and (b) that failure to make a change of recommendation and terminate the merger agreement would be inconsistent with the MTGE directors duties under applicable law, and in such case the MTGE board of directors must also cause MTGE to terminate the merger agreement in order to enter into a definitive agreement providing for such unsolicited acquisition proposal.

Prior to making a change of recommendation for any reason set forth above, MTGE must give Annaly three business days prior written notice of its intent to make a change of recommendation. The notice must specify in

reasonable detail the reasons for any change in recommendation due to an intervening event or the material terms and conditions of the acquisition proposal for any change in recommendation due to a superior proposal. In each case, MTGE must negotiate in good faith, and cause its representatives (including its executive officers) to negotiate in good faith (and to the extent Annaly desires to negotiate, and does negotiate, Annaly must cause its representatives (including its executive officers) to negotiate in good faith)), any proposal from Annaly to amend the merger agreement in a way that would eliminate the need to make a change of recommendation, and the MTGE board of directors must make the required determination regarding its duties again at the end of such three business day negotiation period after in good faith taking into account any amendments proposed by Annaly. With respect to any change of recommendation due to a superior proposal (including any revision to the amount, form or mix of consideration proposed to be received by MTGE s stockholders as a result of such superior proposal, whether or not material), MTGE must give notice to Annaly of such amendment, revision or change and the three business day period described above will be extended until at least two business days after the time Annaly receives such notice, and MTGE must not make a change of recommendation prior to the end of each such extended period.

An intervening event for purposes of the merger agreement is any event, change in circumstance or development that is material to MTGE and its subsidiaries (taken as a whole) and was not known or reasonably foreseeable by the MTGE board of directors as of or prior to the date of the merger agreement and which becomes known to the MTGE board of directors prior to the acceptance time. However, the following may not be an intervening event for purposes of the merger agreement: (a) the receipt, existence or terms of an acquisition proposal or any matter relating thereto or a consequence thereof, or (b) changes, in and of itself, in the market price or trading volume of MTGE common stock or Annaly common stock or the fact that MTGE meets or exceeds (or that Annaly fails to meet or exceed) internal or published projections, forecasts or revenue or earnings predictions for any period.

Nothing in the merger agreement prohibits the MTGE board of directors from taking and disclosing to the MTGE stockholders a position contemplated by Rules 14d-9 and 14e-2(a) or Item 1012(a) of Regulation M-A promulgated under the Exchange Act, if failure to do so would violate applicable law, or from making any stop, look and listen communication to MTGE s stockholders pursuant to Rule 14d-9(f) promulgated under the Exchange Act (except if such disclosure is a change of recommendation, the disclosure may only be made to the extent otherwise permitted under the merger agreement, as described above).

Conduct of Business Before Completion of the Merger

Restrictions on MTGE s Operations

The merger agreement provides for certain restrictions on MTGE s and its subsidiaries activities until either the completion of the merger or the termination of the merger agreement. In general, unless required by law, specifically permitted or required by the merger agreement, otherwise approved in writing by Annaly (which approval may not be unreasonably withheld, conditioned or delayed), or reasonably necessary (as determined in the reasonable judgment of MTGE upon advice of counsel) for MTGE to maintain its qualification as a REIT under the Code for any period or portion of a period ending on or prior to the effective time of the merger or to avoid incurring entity level income or excise taxes under the Code, MTGE is required to:

maintain its status as a REIT;

conduct, and cause its subsidiaries to conduct, its and their respective businesses in all material respects in the ordinary course consistent with past practice, and use commercially reasonable efforts to preserve intact its and their respective present business organizations, goodwill and ongoing businesses and its and their respective present relationships with customers, suppliers, vendors, governmental entities, program lenders, employees, and other persons with which it and they have material business relationships; and

maintain compliance with a liquidity ratio test measuring the ratio of aggregate fair value of unencumbered mortgage-backed securities issued or guaranteed by a U.S. government agency,

unencumbered factor payment receivables due from a U.S. government agency and unrestricted cash and cash equivalents to the aggregate amount of outstanding indebtedness of MTGE and its subsidiaries in respect of repurchase agreements, warehouse facilities, to be announced contracts and similar financing arrangements.

In addition, unless required by law, specifically permitted or required by the merger agreement, otherwise approved in writing by Annaly (which approval may not be unreasonably withheld, conditioned or delayed), or reasonably necessary (as determined in the reasonable judgment of MTGE upon advice of counsel) for MTGE to maintain its qualification as a REIT under the Code for any period or portion of a period ending on or prior to the effective time of the merger or to avoid incurring entity level income or excise taxes under the Code, none of MTGE nor any MTGE subsidiary may, among other things, directly or indirectly (subject to specified exceptions):

amend, waive, rescind or otherwise change its organizational documents, or with respect to any person who has made or is considering making an acquisition proposal, waive the stock ownership limitation contained in MTGE s charter (unless the merger agreement has been validly terminated);

enter into any agreement with respect to voting or registration of its capital stock or other equity interests;

authorize, declare or pay any dividends or distributions on its outstanding capital stock, other than, among other exceptions, (a) regular quarterly cash dividends on its outstanding shares of common stock with declaration, record and payment dates consistent with past practice and at a rate not to exceed a quarterly rate of \$0.50 per share, (b) dividends expressly provided in the merger agreement, as described below under

Additional Dividends, and (c) dividends on the outstanding shares of the MTGE Series A preferred stock, with declaration, record and payment dates consistent with past practice, at a rate not to exceed a quarterly rate as specified by the terms of the MTGE Series A preferred stock;

split, combine, subdivide, reduce or reclassify any shares of its capital stock or other equity interests, or redeem, purchase, or otherwise acquire any of its capital stock or other equity interests;

issue or authorize the issuance of any securities in substitution of its capital stock or other equity interests;

(a) increase the compensation or benefits payable to any of its directors, executive officers or employees, (b) grant any increase in severance pay or termination pay to any director, executive officer or employee (other than as required by certain of MTGE s benefit plans in effect as of the date of the merger agreement and made available to Annaly), (c) pay or award, or commit to pay or award, any bonuses or incentive compensation to any director, executive officer or employee, (d) enter into any employment, severance or retention agreement, (e) establish, adopt, enter into, amend or terminate any collective bargaining agreement or MTGE benefit plan or arrangement that would be a MTGE benefit plan if in existence on the date of the merger agreement, (f) take any action to amend or waive any performance or vesting criteria or accelerate vesting, exercisability or funding under any MTGE benefit plan, (g) terminate the employment of any employee other than for cause, (h) hire any new employees or (i) provide any funding for any MTGE benefit plan through any rabbi trust, or similar arrangement;

take any action that would increase the absolute notional value of the asset base of MTGE and its subsidiaries by more than 10% over the notional value of the asset base of MTGE and its subsidiaries of the date of the merger agreement, excluding any increases in the size of the asset base resulting from mark to market valuation adjustments;

acquire, authorize or announce an intention to so acquire, or enter into any agreements providing for any acquisitions of, any assets or equity interests in any person, entity or business or division thereof, or otherwise engage in any mergers, consolidations, acquisitions or business combinations, other than

⁸⁹

(a) transactions between MTGE and a wholly owned subsidiary of MTGE, (b) acquisitions of agency mortgage-backed securities (including to be announced agency mortgage-backed securities) in the ordinary course of business consistent with past practice or (c) acquisitions of non-agency mortgage-backed securities (including agency credit risk transfer securities) in the ordinary course of business consistent with past practice;

restructure, reorganize, dissolve or liquidate;

make loans, advances or capital contributions to, or investments in, any other person or entity, other than (a) loans among MTGE and its wholly owned subsidiaries, (b) advances for reimbursable employee expenses in the ordinary course of business consistent with past practices, (c) subservicing advances (including, without limitation, in respect to legal fees) in the ordinary course of business consistent with past practice and (d) transactions under repurchase agreements entered into the ordinary course of MTGE s business and consistent with past practice;

except as otherwise permitted under the merger agreement, sell, lease, license, assign, abandon, permit to lapse, transfer, exchange, swap or otherwise dispose of, or subject to a lien (other than a permitted lien), any properties, rights or assets (including shares of MTGE or its subsidiaries), other than sales of assets in the ordinary course of business so long as such dispositions do not exceed a cumulative aggregate of \$300 million in notional value in any 30-day period based on trade dates (excluding the monthly sale of agency flow new production mortgage-backed securities, to be announced agency mortgage-backed securities and transactions under repurchase agreements financing or refinancing of mortgage-backed securities and credit risk transfer securities in the ordinary course of business consistent with past practice);

enter into specified material contracts (as defined in the merger agreement); modify, amend or terminate any material contract; waive, release or assign any rights or claims thereunder; acquire any healthcare owned property (as defined in the merger agreement); or subject any healthcare owned property to any lien;

make or commit to make any capital expenditure, other than in accordance with MTGE s budget previously disclosed to Annaly and other than as may be expressly required under any operator lease or real property lease in existence as of the date of the merger agreement and made available to Annaly;

waive, release, assign, compromise or settle any claim, litigation, investigation or proceeding in excess of \$500,000 (other than (x) compromises or settlements up to the amount to the amount reserved for the applicable legal contingency as reflected in the consolidated financial statements of MTGE made available to Annaly or (y) compromises or settlement of legal contingencies for which MTGE or any MTGE subsidiary reasonably and in good faith determined based on past practices will be reimbursed in fully by investors (and where such compromise or settlement does not exceed in the aggregate \$2,000,000 excluding any amounts reimbursed));

change any financial accounting policies, practices, principles or procedures, or any method of reporting income, deductions or other material items for financial accounting purposes;

(a) make or change any material tax election or liquidate any MTGE subsidiary for U.S. federal income tax purposes, (b) adopt or change any tax accounting period or material method of tax accounting, (c) file any amended tax return if the filing of such amended tax return would result in a material increase in the taxes payable by MTGE or any of its subsidiaries, (d) settle or compromise any material liability for taxes or any tax audit or other proceeding relating to a material amount of taxes, (e) enter into any closing or similar agreement with any tax authority, (f) surrender any right to claim a material refund of taxes, or (g) except in the ordinary course of business, agree to an extension or waiver of the statute of limitations with respect to a material amount of taxes (other than, in each case, (i) to the extent necessary based on advice of counsel to preserve MTGE s qualification as a REIT or (ii) to qualify or preserve the status of any MTGE subsidiary as a disregarded entity or partnership, or as a qualified REIT subsidiary or taxable REIT subsidiary);

take any action, or fail to take any action, which action or failure to act would reasonably be expected to cause (a) MTGE to fail to qualify as a REIT or (b) any MTGE subsidiary to cease being treated as a partnership or disregarded entity for U.S. federal income tax purposes or a qualified REIT subsidiary or a taxable REIT subsidiary under the applicable provisions of Section 856 of the Code, or if such subsidiary was not treated as a taxable REIT subsidiary as of the date of the merger agreement, to be so treated;

redeem, repurchase, prepay, defease, incur, assume, endorse, guarantee or otherwise become liable for, or modify in any material respect, the terms of any indebtedness, derivatives or hedging arrangements, or issue or sell any debt securities or rights to acquire any debt securities;

fail to maintain all financial books and records in all material respects in accordance with GAAP (or any interpretation thereof);

fail to duly and timely file all material reports and other material documents required to be filed with Nasdaq, SEC or any governmental entity or program lender, subject to extensions permitted by law or applicable rules and regulations;

enter into any transactions or contracts with any affiliates or other person that would be required to be disclosed by MTGE under Item 404 of Regulation S-K of the SEC, or with the MTGE external manager or its affiliates;

enter into any transactions or contracts that would restrict the ability of Annaly and the Offeror to engage after the acceptance time or the effective time of the merger in all activities in which MTGE was engaged as of the date of the merger agreement;

take any action, or fail to take any action, which action or failure would reasonably be expected to cause MTGE or any of its subsidiaries to be required to be registered as an investment company under the Investment Company Act;

take any action or fail to take any action, which action or failure would reasonably be expected to cause MTGE or any of its subsidiaries to fail to be eligible for relief from certain provisions of the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010 related to the use of swaps as hedging instruments;

enter into any new line of business;

fail to pay the premiums on or cancel MTGE s insurance policies;

amend in any material respect the investment policy of MTGE or any of its subsidiaries in effect as of the date of the merger agreement, or fail to comply with such investment policy;

terminate, materially amend, materially restate, materially supplement or waive any material rights under certain operator leases entered into by MTGE with respect to MTGE s healthcare properties;

transfer, market, sell, lease, license, mortgage, pledge, surrender, encumber, divest, cancel, abandon, create or incur liens, or allow to lapse or expire or otherwise dispose of any real property other than in the ordinary course of business and consistent with past practice;

enter into any partnership agreement, limited liability company agreement or other similar agreement with any entity that is not a wholly owned subsidiary, or distribute or otherwise make available any offering document for purposes of, or make any commitments with respect to, obtaining equity capital; or

agree or authorize, in writing or otherwise, to take any of the foregoing actions. *Restrictions on Annaly s Operations*

The merger agreement provides for certain restrictions on Annaly s and its subsidiaries activities until either the completion of the merger or the termination of the merger agreement. In general, unless required by law, specifically permitted or required by the merger agreement, otherwise approved in writing by MTGE (which

approval may not be unreasonably withheld, conditioned or delayed), or reasonably necessary (as determined in the reasonable judgment of Annaly upon advice of counsel) for Annaly to maintain its qualification as a REIT under the Code for any period or portion of a period ending on or prior to the effective time of the merger or to avoid incurring entity level income or excise taxes under the Code, Annaly is required to maintain its status as a REIT and conduct, and cause its subsidiaries to conduct, its and their respective businesses in all material respects in the ordinary course consistent with past practice, and use commercially reasonable efforts to preserve intact its and their respective present business organizations, goodwill and ongoing businesses and its and their respective present relationships with customers, suppliers, vendors, governmental entities and other persons with which it and they have material business relations.

In addition, unless required by law, specifically permitted or required by the merger agreement, otherwise approved in writing by MTGE (which approval may not be unreasonably withheld, conditioned or delayed), or reasonably necessary (as determined in the reasonable judgment of Annaly upon advice of counsel) for Annaly to maintain its qualification as a REIT under the Code for any period or portion of a period ending on or prior to the effective time of the merger or to avoid incurring entity level income or excise taxes under the Code, none of Annaly nor any Annaly subsidiary may, among other things, directly or indirectly (subject to specified exceptions):

authorize, declare or pay any dividends or distributions on its outstanding capital stock, other than (a) regular quarterly cash dividends on its outstanding shares of common stock with declaration, record and payment dates consistent with past practice, (b) dividends expressly provided in the merger agreement, as described below in Additional Dividends, and (c) dividends on the outstanding shares of its Series C, Series D, Series F and Series G cumulative redeemable preferred stock, with declaration, record and payment dates consistent with past practice, at a rate not to exceed a quarterly rate as specified by the terms of its Series C, Series D, Series F and Series G cumulative redeemable preferred stock, respectively;

split, combine, reduce or reclassify any shares of its capital stock;

amend Annaly s charter or bylaws in a way that would prevent or materially impede or delay the consummation of the offer, the merger or the other transactions contemplated by the merger agreement or that would be material and adverse to the holders of MTGE common stock relative to the treatment of existing holders of Annaly common stock;

fail to duly and timely file all material reports and other material documents required to be filed with the NYSE, SEC or any program lender, subject to extensions permitted by law or applicable rules and regulations;

adopt a plan of complete or partial liquidation of Annaly or resolutions providing for a complete or partial liquidation, dissolution or recapitalization of Annaly;

take any action, or fail to take any action, which action or failure would reasonably be expected to cause Annaly to fail to qualify as a REIT; or

agree or authorize, in writing or otherwise, to take any of the foregoing actions. Additional Dividends

The merger agreement provides that MTGE will declare a dividend to its stockholders, with a record and payment date as of the close of business on the fourth business day prior to the day on which the acceptance time occurs, and the payment date for which will be on the business day on which the acceptance time occurs. The per share dividend amount will be the per share amount of MTGE s then-most recent quarterly dividend, prorated for the number of days between the record date of MTGE s last dividend, plus any additional amount required to satisfy the requirements for REIT distributions under the Code and to avoid the imposition of income tax and excise tax under the Code. It is expected that Annaly will declare and pay a comparable stub-period dividend, with a record date as of the close of business on the last business day prior to the day on which the acceptance time occurs.

Access

The merger agreement provides that during the period prior to the effective time of the merger, MTGE and Annaly will give each other and each other s representatives reasonable access during normal business hours and upon reasonable advance notice to all of their respective properties, offices, contracts, personnel (other than, with respect to MTGE, MTGE external manager s portfolio management personnel), books and records, and will furnish promptly to the other party all information concerning their business, properties and personnel as the other party reasonably requests, including with respect to MTGE, information about its (x) financing reporting, cash management, accounts payable and receivable and portfolio level information and (y) the duration and notional balance of MTGE s mortgaged-backed securities, to be announced agency mortgage-backed securities, credit risk transfer securities, legacy residential mortgage-backed securities and swaptions. However, neither party is required to disclose information that may not be disclosed pursuant to contractual or legal restrictions or to avoid loss of legal privileges, provided that the disclosing party will use commercially reasonable efforts to make alternative arrangements for disclosure that do not violate such restrictions or to the maximum extent possible that does not result in loss of a privilege.

Existing Financing Cooperation

The merger agreement provides that MTGE and its subsidiaries are required to, and are required to use their commercially reasonable efforts to cause their representatives to, (a) cooperate with Annaly and the Offeror in connection with the replacement, backstopping or amendment, as of the effective time of the merger, of outstanding financial guaranties, letters of credit, letters of guaranty, surety bonds and other similar instruments and obligations of MTGE and its subsidiaries, including granting any waivers in respect thereof and facilitating the migration of such financial products to the facilities of Annaly or its affiliates and the satisfaction or amendment, as of the effective time of the merger, of derivative financial instruments or arrangements (including any swaps, caps, floors, futures, forward contracts and option agreements), and (b) obtain and deliver to Annaly, no later than three business days prior to the effective time of the merger, customary payoff letters for any indebtedness of MTGE or any of its subsidiaries.

Other Agreements

Under the merger agreement, Annaly and MTGE are required to use reasonable best efforts to prepare and file or otherwise provide all documentation to effect all necessary applications, notices, petitions, filings, and other documents in order to consummate the offer or the merger; and obtain all waiting period expirations or terminations, consents, clearances, waivers, licenses, orders, registrations, approvals, permits, and authorizations necessary or advisable to be obtained from any third party and/or any governmental entity and/or program lender in order to consummate the offer or the merger.

Employee Matters

All of MTGE s executive officers are employees of the parent of the Manager, and receive no compensation from MTGE. There are no agreements or understandings, whether written or unwritten, between any executive officer of MTGE and MTGE or Annaly concerning any type of compensation, whether present, deferred or contingent, that is based upon or otherwise relates to the merger.

As of the date of this document, a subsidiary of MTGE employed four individuals. Each of the four continuing employees, who are not and have not been an officer or employee of the Manager, is party to a severance agreement providing for certain payments upon such continuing employee s termination by his or her employer, Residential Credit Solutions, Inc., an indirect subsidiary of MTGE, without cause. Pursuant to the terms of the merger agreement,

Annaly will, and will cause the surviving corporation or a subsidiary thereof to, comply with certain existing severance agreements between MTGE or a subsidiary of MTGE and certain employees of a MTGE subsidiary, provided that after the closing Annaly may amend, modify or otherwise alter each such severance agreement in accordance with its terms.

In addition, with respect to any continuing employees, Annaly has agreed (a) to ensure, or cause the surviving corporation or any subsidiary thereof to ensure, that no limitations or exclusions as to pre-existing conditions, evidence of insurability or good health, waiting periods or actively-at-work exclusions or other limitations or restrictions on coverage are applicable to any continuing employees or their dependents or beneficiaries under any welfare benefit plans sponsored by Annaly and its subsidiaries in which such continuing employees or their dependents or beneficiaries first become eligible to participate following the effective time of the merger, except to the extent that such exclusions, limitations or restrictions would apply under the analogous benefit plan in which a continuing employee was a participant or was eligible to participate immediately prior to the effective time, and (b) that any costs or expenses incurred by continuing employees (and their dependents or beneficiaries) up to (and including) the effective time of the merger shall be taken into account for purposes of satisfying applicable deductible, co-payment, coinsurance, maximum out-of-pocket provisions and like adjustments or limitations on coverage under any such Annaly plans for the plan year in which the closing date occurs, except to the extent such exclusions, limitations or restrictions would not be taken into account for such purposes under the analogous benefit plan in which any such continuing employee was a participant or was eligible to participate into account for such purposes under any such Annaly plans for the plan year in which the closing date occurs, except to the extent such exclusions, limitations or restrictions would apply or costs or expenses would not be taken into account for such purposes under the analogous benefit plan in which any such continuing employee was a participant or was eligible to participate immediately prior to the effective time of the merger.

Furthermore, with respect to each employee benefit plan, policy or practice, including severance, vacation and paid time off plans, policies or practices, sponsored or maintained by Annaly or its affiliates in which the continuing employees become eligible to participate after the effective time, Annaly has agreed to credit continuing employees for all service with MTGE and its affiliates and their predecessors prior to the closing date for purposes of eligibility to participate, vesting credit, eligibility to commence benefits, and benefit accrual, but excluding (i) benefit accrual under any defined benefit pension plan or retiree medical program and (ii) any such credit that would result in a duplication of benefits.

Directors and Officers Indemnification

Under the merger agreement, for a period of no less than six years after the effective time of the merger, Annaly and the Offeror must indemnify and hold harmless, to the fullest extent permitted under applicable law and provided pursuant to organizational documents of MTGE or its subsidiaries, or any indemnification agreements in existence as of the time of the merger agreement that were provided to Annaly (including the MTGE management agreement), each current and former director, officer, agent or fiduciary of MTGE and its subsidiaries against costs and expenses in connection with claims asserted or claimed prior to, at or after the effective time of the merger, in respect of acts or omissions occurring or alleged to have occurred at or prior to the effective time of the merger, based on or arising out of the fact that such person is or was serving as an officer, director, employee or agent of MTGE or its subsidiaries or any other entity if such service was at the request or for the benefit of MTGE or are required to maintain in effect the provisions in any organizational documents of MTGE and its subsidiaries and contracts (including the MTGE management agreement) of MTGE and its subsidiaries regarding elimination of liability, indemnification, and advancement of expenses in favor of the current and former directors, officers, and agents of MTGE and its subsidiaries that are in existence as of the time of the merger agreement and were provided to Annaly prior to the date of the merger agreement, except to the extent that any such contract provides for an earlier termination.

At or prior to the acceptance time, MTGE, in consultation with Annaly, is required to purchase a directors and officers liability insurance tail insurance policy for a period of six years after the effective time of the merger with respect to acts or omissions committed at or prior to the effective time of the merger, including the transactions contemplated by the merger agreement, provided that MTGE will, in cooperation with Annaly, use commercially reasonably efforts to obtain the most favorable pricing and most comprehensive coverage reasonably available and MTGE will not commit or spend on such tail policy more than 300% of the last aggregate annual premium paid by MTGE for its directors and

officers liability insurance prior to the date of the merger agreement. If the cost of such tail policy would be in excess of such 300% threshold, MTGE will be permitted to purchase as much coverage as reasonably practicable for such amount. Following the effective time

of the merger, Annaly will cause such tail policy to be maintained in full force and effect for its full term, and cause all obligations thereunder to be honored by the surviving corporation.

Conditions to the Merger

The respective obligations of MTGE, Annaly and the Offeror to complete the merger under the merger agreement are subject to the satisfaction or waiver of the following conditions:

the Offeror having accepted for payment all shares of MTGE common stock validly tendered in the offer and not validly withdrawn;

no governmental entity of competent jurisdiction having issued or granted any order or injunction (whether temporary, preliminary or permanent) that is in effect as of immediately prior to the effective time of the merger which has the effect of restraining, enjoining or otherwise prohibiting the consummation of the merger; and

no governmental entity of competent jurisdiction having enacted, issued or promulgated any law that is in effect as of immediately prior to the effective time of the merger that has the effect of restraining, enjoining or otherwise prohibiting the consummation of the merger.

Termination of the Merger Agreement

Termination by Annaly or MTGE

The merger agreement may be terminated at any time before the acceptance time:

by mutual written consent of Annaly and MTGE;

by either Annaly or MTGE, if:

any governmental entity of competent jurisdiction has issued a final, non-appealable order, injunction, decree or ruling permanently restraining, enjoining or otherwise prohibiting the consummation of the offer and/or the merger, provided that such termination right is not available to any party whose action or failure to fulfill any obligation under the merger agreement has proximately caused the issuance of such order, injunction, decree or ruling;

the acceptance time has not occurred on or before 11:59 p.m., Eastern Time, on February 2, 2019 (the outside date), provided that such termination right is not available to any party whose action or failure to fulfill any obligation under the merger agreement has proximately caused any of the conditions to the closing of the offer to fail to be satisfied and such action or failure to act constitutes a material

breach of the merger agreement; or

the offer shall have terminated or expired in accordance with its terms (subject to the rights and obligations of Annaly to extend the offer) without all the conditions to the offer having been satisfied, provided that such termination right is not available to any party whose action or failure to fulfill any obligation under the merger agreement has proximately caused any of the conditions to the closing of the offer to fail to be satisfied and such action or failure to act constitutes a material breach of the merger agreement.

Termination by MTGE

The merger agreement may be terminated at any time before the acceptance time by MTGE:

in order for MTGE to enter into a definitive agreement providing for a superior proposal, as long as (a) MTGE has complied with its obligations to provide notice to, and negotiate with, Annaly regarding amendments to the merger agreement, as described under Change of Recommendation prior to such termination and (b) immediately prior (and as a condition) to such termination, MTGE pays to Annaly the termination fee of \$35,118,500; or

if (a) Annaly and/or the Offeror has breached, failed to perform or violated their respective covenants or agreements under the merger agreement or any of the representations and warranties of Annaly and the Offeror have become inaccurate, in each case in a manner that would give rise to the failure of any of the conditions to the consummation of the offer related to Annaly s and/or the Offeror s compliance with their covenants and agreements or the accuracy of Annaly s and/or the Offeror s representations and warranties to be satisfied, (b) such breach, failure to perform, violation or inaccuracy is not capable of being cured by the outside date, or, if capable of being cured by the outside date, is not cured by Annaly and/or the Offeror within 30 calendar days following receipt of written notice from MTGE, and (c) MTGE is not then in material breach of the merger agreement.

Termination by Annaly and the Offeror

The merger agreement may be terminated at any time before the acceptance time by Annaly if:

MTGE or the MTGE board of directors (or any committee thereof) has made a change of recommendation or breached its non-solicitation/no change of recommendation obligations under the merger agreement in any material respect; or

(a) MTGE has breached, failed to perform or violated its covenants or agreements under the merger agreement or any of the representations and warranties of MTGE have become inaccurate, in each case in a manner that would give rise to the failure of any of the conditions to the consummation of the offer related to MTGE s compliance with its covenants and agreements or the accuracy of MTGE s representations and warranties to be satisfied, (b) such breach, failure to perform, violation or inaccuracy is incapable of being cured by the outside date or, if capable of being cured by the outside date, is not cured by MTGE within 30 calendar days following receipt of written notice from Annaly, and (c) Annaly and the Offeror are not then in material breach of the merger agreement.

Termination Fee and Expenses

Except as set forth below, all fees and expenses incurred in connection with the merger agreement, the offer and the merger will be paid by the party incurring such fee or expense.

Termination Fee

The merger agreement provides that MTGE will pay Annaly a termination fee of \$35,118,500.00 (the termination fee) if:

Annaly terminates the merger agreement because of a change of recommendation by MTGE or the MTGE board of directors (or any committee thereof);

Annaly terminates the merger agreement because MTGE or the MTGE board of directors (or any committee thereof) has breached its non-solicitation/no change of recommendation obligations under the merger agreement in any material respect;

MTGE terminates the merger agreement in order to enter into a definitive agreement providing for a superior proposal; or

(a) either Annaly or MTGE terminates the merger agreement as a result of having reached the outside date, the offer having terminated or expired in accordance with its terms without all conditions to the offer having been satisfied or Annaly terminates the merger agreement as a result of a breach, failure to perform or violation by MTGE of its covenants or agreements under the merger agreement that gives rise to the failure of any of the conditions to the consummation of the offer related to MTGE s compliance with its covenants and agreements to be satisfied (and such breach, failure, or violation is incapable of being cured by the outside date or, if capable of being cured by such time, is not cured within 30 days after receiving written notice from Annaly), (b) an acquisition proposal has been

publicly disclosed after the date of the merger agreement and prior to the date of such termination and shall not have been publicly withdrawn prior to the termination of the merger agreement, and (c) within 12 months of such termination any acquisition proposal is consummated or a definitive agreement with respect to any acquisition proposal is entered into and such acquisition proposal is thereafter consummated (with references to 20% and 80% in the definition of acquisition proposal being replaced with references to 50% for this purpose).

In no event will MTGE be obligated to pay the termination fee on more than one occasion.

Effect of Termination

In the event of the valid termination of the merger agreement prior to the acceptance time for the offer in accordance with the terms of the merger agreement, the merger agreement will become null and void, and there will be no liability or further obligation on the part of Annaly, the Offeror or MTGE, except that the confidentiality agreement and certain miscellaneous provisions of the merger agreement shall survive and provided that no party will be relieved of liability for fraud or any willful breach of the merger agreement prior to such termination or any requirement to pay the termination fee.

Amendments; Enforcement and Remedies; Extensions and Waivers

Amendments

The merger agreement may be amended, modified, or supplemented by written agreement of the parties at any time.

Enforcement and Remedies

Under the merger agreement, the parties have agreed that, prior to the valid termination of the merger agreement, each party will be entitled to:

an injunction or injunctions to prevent or remedy any breaches or threatened breaches of the merger agreement by any other party;

a decree or order of specific performance specifically enforcing the terms and provisions of the merger agreement; and

any further equitable relief. *Extensions and Waivers*

Under the merger agreement, at any time prior to the effective time of the merger, each party may:

extend the time for the performance of any of the obligations or other acts of the other parties;

waive any inaccuracies in the representations and warranties of the other party, as applicable; and

waive compliance with any of the agreements or conditions contained in the merger agreement for the benefit of the waiving party.

COMPARATIVE MARKET PRICE AND DIVIDEND MATTERS

Market Price History

Annaly common stock is listed on the NYSE under the symbol NLY, and MTGE common stock is listed on Nasdaq under the symbol MTGE. The following table sets forth, for the periods indicated, as reported by the NYSE or Nasdaq, as applicable, the per share high and low sales prices of each company s common stock.

	Annaly Common Stock			MTGE Common Stock				
	High	Low	Div	vidend	High	Low	Div	vidend
2015								
First Calendar Quarter	\$11.09	\$10.29	\$	0.30	\$19.39	\$17.76	\$	0.50
Second Calendar Quarter	\$10.55	\$ 9.19	\$	0.30	\$18.29	\$15.90	\$	0.50
Third Calendar Quarter	\$10.59	\$ 9.17	\$	0.30	\$16.89	\$14.37	\$	0.40
Fourth Calendar Quarter	\$10.35	\$ 8.98	\$	0.30	\$15.22	\$13.63	\$	0.40
2016								
First Calendar Quarter	\$10.48	\$ 8.25	\$	0.30	\$14.79	\$12.01	\$	0.40
Second Calendar Quarter	\$11.13	\$10.16	\$	0.30	\$16.11	\$14.31	\$	0.40
Third Calendar Quarter	\$11.29	\$10.33	\$	0.30	\$17.77	\$15.45	\$	0.40
Fourth Calendar Quarter	\$10.50	\$ 9.83	\$	0.30	\$17.40	\$15.40	\$	0.40
2017								
First Calendar Quarter	\$11.37	\$ 9.95	\$	0.30	\$17.05	\$15.50	\$	0.45
Second Calendar Quarter	\$12.73	\$11.09	\$	0.30	\$19.65	\$16.35	\$	0.45
Third Calendar Quarter	\$12.58	\$11.70	\$	0.30	\$20.00	\$18.45	\$	0.45
Fourth Calendar Quarter	\$12.43	\$10.97	\$	0.30	\$ 19.90	\$17.65	\$	0.50
2018								
First Calendar Quarter	\$11.92	\$10.00	\$	0.30	\$18.65	\$16.15	\$	0.50
Second Calendar Quarter (through May 30, 2018)	\$10.62	\$10.17			\$20.15	\$17.60		

On May 1, 2018, the last trading day prior to public announcement of the merger agreement, the closing price per share of MTGE common stock on Nasdaq was \$18.15, and the closing price per share of Annaly common stock on the NYSE was \$10.34. On May 15, 2018, the most recent trading date prior to the mailing of this document, the closing price per share of MTGE common stock on Nasdaq was \$19.65, and the closing price per share of Annaly common stock on the NYSE was \$10.33. The market value of the stock portion of the common transaction consideration will change as the market value of Annaly common stock fluctuates during the offer period and thereafter. MTGE common stockholders should obtain current market quotations for shares of MTGE common stock and shares of Annaly common stock before deciding whether to tender their shares of MTGE common stock in the offer and before electing the form of common transaction consideration they wish to receive.

Dividends

In accordance with our requirement for maintaining REIT status, Annaly will distribute to stockholders aggregate dividends equaling at least 90% of its REIT taxable income for each taxable year and will endeavor to distribute at least 100% of its REIT taxable income so as not to be subject to tax. Distributions of economic profits from our enterprise could be classified as return of capital due to differences between book and tax accounting rules. Annaly may make additional returns of capital when the potential risk-adjusted returns from new investments fail to exceed our cost of capital. Subject to the limitations of applicable securities and state corporation laws, Annaly can return

capital by making purchases of its own stock or through payment of dividends.

Annaly has not established a minimum dividend payment level and its ability to pay dividends may be adversely affected. In addition, unrealized changes in the estimated fair value of available-for-sale investments

may have a direct effect on dividends. All distributions will be made at the discretion of Annaly s board of directors and will depend on our earnings, our financial condition, maintenance of our REIT status and such other factors as Annaly s board of directors may deem relevant from time to time.

The merger agreement provides that MTGE will declare a dividend to its stockholders, with a record date as of the close of business on the fourth business day prior to the day on which the acceptance time occurs, and the payment date for which will be on the business day on which the acceptance time occurs. The per share dividend amount will be the per share amount of MTGE s then-most recent quarterly dividend, prorated for the number of days between the record date of MTGE s last dividend and the fourth business day prior to the acceptance time, plus any additional amount required to satisfy the requirements for REIT distributions under the Code and to avoid the imposition of income tax and excise tax under the Code. It is expected that Annaly will declare and pay a comparable stub-period dividend, with a record date as of the close of business on the last business day prior to the day on which the acceptance time occurs.

MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES

The following is a discussion of (i) the material U.S. federal income tax consequences of the offer and the merger, taken together, to holders of MTGE common stock and (ii) the material U.S. federal income tax considerations relating to Annaly s qualification and taxation as a REIT and to the ownership and disposition of Annaly common stock received pursuant to the offer and/or the merger. This summary is based on provisions of the Code, final, temporary or proposed Treasury Regulations promulgated thereunder, judicial opinions, published positions of the IRS and all other applicable authorities, all as in effect as of the date of this document and all of which are subject to change, possibly with retroactive effect. Any such change could affect the accuracy of the statements and conclusions set forth in this document.

For purposes of this discussion, the term U.S. holder means a beneficial owner of MTGE common stock or Annaly common stock, as applicable, that is, for U.S. federal income tax purposes:

an individual who is a citizen or resident of the United States;

a corporation, or other entity taxable as a corporation for U.S. federal income tax purposes, created or organized under the laws of the United States, any state thereof, or the District of Columbia;

an estate, the income of which is subject to U.S. federal income tax regardless of its source; or

a trust if (1) a court within the United States is able to exercise primary supervision over the administration of the trust and one or more U.S. persons have the authority to control all substantial decisions of the trust or (2) it has a valid election in effect under applicable Treasury Regulations to be treated as a U.S. person. For purposes of this discussion, the term non-U.S. holder means a beneficial owner of MTGE common stock or Annaly common stock, as applicable, that is neither a U.S. holder nor a partnership for U.S. federal income tax

If an entity or arrangement that is treated as a partnership for U.S. federal income tax purposes holds MTGE common stock or Annaly common stock, as applicable, the tax treatment of a partner in such entity generally will depend on the status of the partner, the activities of the partnership and certain determinations made at the partner level. If you are a partner in a partnership holding MTGE common stock or Annaly common stock, as applicable, please consult your tax advisor.

This discussion only addresses holders of MTGE common stock (or, following the offer and the merger, Annaly common stock) who hold their shares as capital assets under the Code (generally, property held for investment). Further, this summary does not address all aspects of U.S. federal income taxation that may be relevant to a holder in light of the holder s particular circumstances or that may be applicable to holders subject to special treatment under U.S. federal income tax law (including, for example, financial institutions, dealers in securities, traders in securities that elect mark-to-market treatment, insurance companies, mutual funds, tax-exempt organizations (except as described in Taxation of Tax-Exempt Holders of Annaly Common Stock below), partnerships or other flow-through entities and their partners or members, U.S. expatriates, controlled foreign corporations and passive foreign investment companies, holders liable for the alternative minimum tax, U.S. holders whose functional currency is not

purposes.

the U.S. dollar, holders who hold their MTGE common stock (or, following the offer and the merger, Annaly common stock) as part of a hedge, straddle, constructive sale or conversion transaction, holders who acquired their MTGE common stock (or, in connection with the offer and/or the merger, Annaly common stock) through the exercise of employee stock options or other compensation arrangements and, except to the extent discussed below, non-U.S. holders). This discussion does not address any tax consequences arising under the unearned income Medicare contribution tax pursuant to the Health Care and Education Reconciliation Act of 2010 (except as described in

Taxation of U.S. Holders of Annaly Common Stock below), the tax consequences of transactions occurring prior to, concurrently with or after the offer and/or merger (whether or not such transactions are in connection with the offer and/or merger) are

not discussed, including without limitation the tax consequences of the payment of dividends to holders of MTGE common stock prior to the offer and/or merger, nor does it address any tax considerations under state, local or non-U.S. laws or U.S. federal laws other than those pertaining to the U.S. federal income tax.

MTGE stockholders are urged to consult their tax advisors regarding the particular tax consequences (including the application and effect of any state, local or non-U.S. income and other tax laws) to them of the offer and the merger, and of acquiring, holding and disposing of Annaly common stock received pursuant to the offer and/or the merger, in light of their particular circumstances.

Material U.S. Federal Income Tax Consequences of the Offer and the Merger

The following is a summary of the material U.S. federal income tax consequences of the offer and the merger to holders of MTGE common stock. Except to the extent specifically discussed below, this summary does not address the tax consequences of any transaction other than the offer and the merger. In addition, no information is provided herein with respect to the tax consequences of the offer and the merger under applicable state, local or non-U.S. laws or U.S. federal laws other than those pertaining to the U.S. federal income tax.

Treatment of the Offer and the Merger as a Reorganization

It is a condition to the consummation of the offer that each of Annaly and MTGE receive an opinion from their respective legal counsel to the effect that the offer and the merger, taken together, will qualify as a reorganization within the meaning of Section 368(a) of the Code. Such opinions will be based on factual representations contained in letters provided by Annaly and MTGE, and on certain customary factual assumptions, all of which must continue to be true and accurate as of the consummation of the offer. However, no ruling has been or will be sought from the IRS as to the U.S. federal income tax consequences of the offer and the merger. Consequently, there can be no assurance that the offer and the merger, taken together, will qualify as a reorganization within the meaning of Section 368(a) of the Code. There also can be no assurance that the IRS will not disagree with, or challenge, any of the conclusions described below.

Consequences of the Offer and the Merger to U.S. Holders

If the offer and the merger, taken together, qualify as a reorganization within the meaning of Section 368(a) of the Code, the U.S. federal income tax consequences to U.S. holders who receive shares of Annaly common stock and/or cash in exchange for shares pursuant to the offer and/or the merger generally will be as follows:

U.S. Holders Who Receive Solely Annaly Common Stock

A U.S. holder who receives solely shares of Annaly common stock pursuant to the offer and/or the merger will not recognize any gain or loss for U.S. federal income tax purposes, except with respect to cash received in lieu of a fractional share of Annaly common stock (see the discussion under Cash in Lieu of a Fractional Share). The aggregate tax basis of the shares of Annaly common stock received (including any fractional shares deemed received and exchanged for cash) will be equal to the U.S. holder s aggregate tax basis in the MTGE common stock surrendered. The holding period of the Annaly common stock received (including any fractional shares deemed received and exchanged for cash) will include the U.S. holder s holding period of the MTGE common stock surrendered (see the discussion under Cash in Lieu of a Fractional Share).

If a U.S. holder acquired different blocks of MTGE common stock at different times and different prices, such U.S. holder should consult its tax advisor as to the determination of the tax bases and holding periods of the shares of

Table of Contents

Annaly common stock received in the offer and/or the merger.

U.S. holders electing to receive the all-stock consideration in the offer may be subject to proration (see the discussion under The Offer Elections and Proration), which may result in the receipt of a portion of the

1	0	1
---	---	---

merger consideration in cash, in addition to Annaly common stock. See the discussion under U.S. Holders Who Receive a Combination of Shares of Annaly Common Stock and Cash for a general description of the material U.S. federal income tax consequences to U.S. holders of the receipt of Annaly common stock and cash.

U.S. Holders Who Receive Solely Cash

A U.S. holder who receives solely cash in exchange for MTGE common stock pursuant to the offer and/or the merger will recognize gain or loss in an amount equal to the difference, if any, between (1) the amount of cash received and (2) such U.S. holder s tax basis in the MTGE common stock surrendered. Any such gain or loss will be long-term capital gain or loss if, as of the date of the exchange, the U.S. holder s holding period in the MTGE common stock surrendered exceeds one year. The deductibility of capital losses is subject to limitations.

U.S. holders electing to receive the all-cash consideration in the offer may be subject to proration (see the discussion under The Offer Elections and Proration), which may result in the receipt of a portion of the merger consideration in Annaly common stock, in addition to cash. See the discussion under U.S. Holders Who Receive a Combination of Shares of Annaly Common Stock and Cash for a general description of the material U.S. federal income tax consequences to U.S. holders of the receipt of Annaly common stock and cash.

U.S. Holders Who Receive a Combination of Shares of Annaly Common Stock and Cash

A U.S. holder who receives a combination of Annaly common stock and cash (other than cash in lieu of a fractional share of Annaly common stock) pursuant to the offer and/or the merger generally will recognize gain (but not loss) in an amount equal to the lesser of (1) the sum of the amount of cash (other than cash received in lieu of a fractional share of Annaly common stock) and the fair market value of the Annaly common stock received, less the adjusted tax basis of the MTGE common stock surrendered in exchange therefor, and (2) the amount of cash received by the U.S. holder (other than cash received in lieu of a fractional share of Annaly common stock). Any gain that the U.S. holder recognizes generally will be long-term capital gain if, as of the date of the exchange, the U.S. holder sholding period in its MTGE common stock surrendered exceeds one year.

If a U.S. holder of MTGE common stock acquired different blocks of shares at different times or at different prices, any gain or loss will be determined separately with respect to each block of MTGE common stock, and such U.S. holder s tax basis and holding period in its Annaly common stock received in the offer and/or the merger may be determined with reference to each identifiable block of MTGE common stock. U.S. holders should consult their tax advisors regarding the manner in which cash and shares of Annaly common stock received in the offer and/or the merger is allocated among different blocks of MTGE common stock and with respect to identifying the bases or holding periods of particular shares of Annaly common stock received in the offer and/or the merger.

In certain cases, if a U.S. holder actually or constructively owns Annaly common stock other than Annaly common stock received in the transaction, the gain that is recognized by the U.S. holder could be treated as having the effect of the distribution of a dividend under the tests described in Section 302 of the Code, in which case such gain would be treated as dividend income for U.S. federal income tax purposes. In such cases, corporate U.S. holders should consult their tax advisors regarding the potential applicability of the extraordinary dividend provisions of the Code.

The aggregate tax basis of the Annaly common stock received (including any fractional shares deemed received and exchanged for cash) by a U.S. holder that exchanges its MTGE common stock for a combination of Annaly common stock and cash will be equal to the U.S. holder s aggregate adjusted tax basis of the shares surrendered, reduced by the amount of cash received by the U.S. holder (excluding any cash received instead of fractional shares of Annaly common stock) and increased by the amount of gain, if any, recognized by the U.S. holder (excluding any gain

recognized with respect to cash received in lieu of fractional shares of Annaly common stock) on the exchange.

The holding period of the Annaly common stock received (including any fractional shares deemed received and exchanged for cash; see the discussion under Cash in Lieu of a Fractional Share) will include the holding period of the MTGE common stock surrendered. U.S. holders receiving a combination of Annaly common stock and cash should consult their tax advisors regarding the manner in which cash and Annaly common stock should be allocated among the U.S. holder s shares and the manner in which the above rules would apply in the U.S. holder s particular circumstances.

Cash in Lieu of a Fractional Share

A U.S. holder that receives cash in lieu of a fractional share of Annaly common stock generally will be treated as having received such fractional share in the offer and/or the merger and then as having received cash in exchange for such fractional share. Gain or loss generally will be recognized based on the difference between the amount of cash received in lieu of the fractional share and the tax basis allocated to such fractional share of Annaly common stock. Such gain or loss generally will be long-term capital gain or loss if, as of the date of the exchange, the holding period for such shares is greater than one year. The deductibility of capital losses is subject to limitations.

Consequences of the Offer and the Merger to Non-U.S. Holders

If the offer and the merger, taken together, qualify as a reorganization within the meaning of Section 368(a) of the Code, a non-U.S. holder s gain or loss from the offer and/or the merger will be determined in the same manner as that of a U.S. holder. A non-U.S. holder will not be subject to U.S. federal income taxation on any gain recognized from the receipt of the exchange consideration, unless (1) the gain is effectively connected with a U.S. trade or business of the non-U.S. holder, (2) the non-U.S. holder is an individual who has been present in the United States for 183 days or more during the taxable year of disposition and certain other conditions are satisfied, or (3) the non-U.S. holder s MTGE common stock constitutes a U.S. real property interest, (a USRPI), within the meaning of the Foreign Investment in Real Property Tax Act of 1980 (FIRPTA).

A non-U.S. holder whose gain is effectively connected with the conduct of trade or business in the United States (and, if required by an applicable income tax treaty, the non-U.S. holder maintains a permanent establishment in the United States to which such gain is attributable) will be subject to U.S. federal income tax on such gain on a net basis in the same manner as a U.S. holder. In addition, a non-U.S. holder that is a corporation may be subject to a branch profits tax equal to 30% (or lesser rate under an applicable income tax treaty) on the after-tax amount of such effectively connected gain.

A non-U.S. holder who is an individual present in the United States for 183 days or more in the taxable year of the merger and who meets certain other requirements will be subject to a flat 30% tax on the gain derived from the merger, which may be offset by U.S.-source capital losses of such non-U.S. holder, if any.

If the non-U.S. holder s MTGE common stock constitutes a USRPI under FIRPTA, such non-U.S. holder will be subject to U.S. federal income tax on the gain recognized in the exchange on a net basis in the same manner as a U.S. holder. A non-U.S. holder s MTGE common stock generally will not constitute a USRPI, and gain recognized by a non-U.S. holder generally will not be taxed under FIRPTA, if (1) MTGE is treated as a domestically controlled REIT on the effective date of the exchange, (2) the non-U.S. holder owned (after application of certain constructive ownership rules) not more than 10% of the MTGE common stock at any time during the five years preceding the effective date of the exchange, or (3) MTGE is not and has not been at any time during the shorter of (i) the five years preceding the effective date of the exchange and (ii) the non-U.S. holder s holding period for its MTGE common stock, a United States real property holding corporation (a USRPHC). MTGE believes that it is and, at the effective time of the exchange, will be a domestically controlled REIT, although there can be no assurances in that regard.

A non-U.S. holder that receives cash in lieu of a fractional share of Annaly common stock generally will be treated as having received such fractional share in the offer and/or the merger and then as having received cash in exchange for such fractional share. Gain or loss generally will be recognized based on the difference between the amount of cash received in lieu of the fractional share and the tax basis allocated to such fractional share of Annaly common stock and will be subject to U.S. federal income taxation in a manner described below in Material U.S. Federal Income Tax Considerations Applicable to Annaly s Treatment as a REIT and to Holders of Annaly Common Stock Taxation of Non-U.S. Holders Dispositions of Annaly Common Stock.

Information Reporting and Backup Withholding

Information reporting and backup withholding (currently, at a rate of 24% for payments made before January 1, 2026) may apply to payments made in connection with the offer and/or the merger. Backup withholding will not apply, however, to a holder who (a) in the case of a holder who is a U.S. person as defined in the instructions to the IRS Form W-9, furnishes a correct taxpayer identification number and certifies that it is not subject to backup withholding on IRS Form W-9, (b) in the case of a holder who is not such a U.S. person, furnishes the appropriate IRS Form W-8, or (c) is otherwise exempt from backup withholding and complies with other applicable rules and certification requirements. Backup withholding is not an additional tax. Any amounts withheld under the backup withholding rules may be allowed as a refund or a credit against such holder s U.S. federal income tax liability provided the required information is furnished to the IRS on a timely basis. Holders should consult their tax advisors regarding their qualification for an exemption from backup withholding and the procedures for obtaining such an exemption.

Material U.S. Federal Income Tax Considerations Applicable to Annaly s Treatment as a REIT and to Holders of Annaly Common Stock

General

The following is a general summary of the material U.S. federal income tax considerations relating to Annaly s treatment as a REIT and to holders of Annaly common stock who receive such stock pursuant to the offer and/or the merger. For purposes of this discussion, references to we or us refer to Annaly, and references to our common stock refer to Annaly common stock.

We elected to be taxed as a REIT under the U.S. federal income tax laws beginning with our taxable year ended December 31, 1997. We believe that we have operated in a manner qualifying us as a REIT since our election and intend to continue to so operate. Moreover, our continued qualification and taxation as a REIT depends on our ability to meet, on a continuing basis, through actual operating results, certain qualification tests in the U.S. federal income tax laws. Those qualification tests involve the percentage of our income that we earn from specified sources, the percentages of our assets that fall within specified categories, the diversity of our share ownership and the percentage of our earnings that we distribute. Accordingly, no assurance can be given that the actual results of our operations for any particular taxable year will satisfy such requirements. As a result, we may have to use one or more of the REIT savings provisions described below, which would require us to pay an excise or penalty tax (which could be material) in order to maintain our REIT qualification. For a discussion of the tax consequences of our failure to qualify as a REIT, see Failure to Qualify, below.

The sections of the Code and the corresponding regulations that govern the U.S. federal income tax treatment of a REIT and its stockholders are highly technical and complex. The following discussion is qualified in its entirety by the applicable Code provisions, rules and regulations promulgated thereunder, and administrative interpretations thereof.

Provided that we qualify for taxation as a REIT, generally we will be entitled to a deduction for dividends that we pay and therefore will not be subject to U.S. federal income tax on that portion of our net taxable income that we distribute currently to our stockholders, although taxable income generated by domestic taxable REIT subsidiaries, if any, will be subject to regular corporate income.

If we qualify as a REIT, we will nonetheless be subject to U.S. federal tax as follows:

We will pay U.S. federal income tax on our taxable income, including net capital gain, that we do not distribute to stockholders during, or within a specified time after, the calendar year in which the income is earned.

For taxable years beginning prior to December 31, 2017, under some circumstances, we may be subject to the alternative minimum tax due to our undistributed items of tax preference and alternative minimum tax adjustments.

If we have net income from prohibited transactions, which are, in general, sales or other dispositions of property held primarily for sale to customers in the ordinary course of business, other than foreclosure property, such income will be subject to a 100% tax.

If we elect to treat property that we acquire in connection with a foreclosure of a mortgage loan or from certain leasehold terminations as foreclosure property, which is, in general, property acquired by us as a result of having bid in a foreclosure or through other legal means subsequent to a default on a lease of such property or on an indebtedness secured by such property, we may thereby avoid (a) the 100% tax on gain from a resale of that property (if the sale would otherwise constitute a prohibited transaction) and (b) the inclusion of any income from such property not qualifying for purposes of the REIT gross income tests discussed below, but the income from the sale or operation of the property may be subject to U.S. corporate income tax at the highest applicable rate (currently 21%).

If due to reasonable cause and not willful neglect we fail to satisfy either the 75% gross income test or the 95% gross income test discussed below, but nonetheless maintain our qualification as a REIT because other requirements are met, we will be subject to a 100% tax on the greater of the amount by which we fail the 75% gross income test or the 95% gross income test, multiplied in either case by a fraction intended to reflect our profitability.

If we fail to satisfy the asset tests (other than a de minimis failure of the 5% asset test of the 10% vote or value test, as described below under Asset Tests) as long as the failure was due to reasonable cause and not to willful neglect, we dispose of the assets or otherwise comply with such asset tests within six months after the last day of the quarter in which we identify such failure and we file a schedule with the IRS describing the assets that caused such failure, we will pay a tax equal to the greater of \$50,000 or the amount determined by multiplying the net income generated by the non-qualifying assets by the highest U.S. federal income tax rate applicable to corporations during the periods when such assets would have caused us to fail to satisfy such asset tests.

If we fail to satisfy one or more requirements for REIT qualification, other than the gross income tests and the asset tests, and the failure was due to reasonable cause and not to willful neglect, we will be required to

pay a penalty of \$50,000 for each such failure.

We may be required to pay monetary penalties to the IRS in certain circumstances, including if we fail to meet recordkeeping requirements intended to monitor our compliance with rules relating to the composition of a REIT s stockholders, as described below in Requirements for Qualification.

If we fail to distribute during a calendar year at least the sum of:

85% of our REIT ordinary income for such calendar year;

95% of our REIT capital gain net income for such calendar year; and

any undistributed taxable income from earlier periods,

we will pay a 4% nondeductible excise tax on the excess of the required distribution over the amount we actually distributed, plus any retained amounts on which income tax has been paid at the corporate level.

We may elect to retain and pay income tax on our net long-term capital gain. In that case, a U.S. holder would include its proportionate share of our undistributed long-term capital gain (to the extent we

make a timely designation of such gain to the holder) in its income, and would receive a credit or a refund for its proportionate share of the tax we paid.

We will be subject to a 100% excise tax on amounts received by us from a taxable REIT subsidiary (or on certain expenses deducted by a taxable REIT subsidiary or income earned by a taxable REIT subsidiary) if certain arrangements between us and a taxable REIT subsidiary of ours, as further described below, do not reflect arm s-length terms.

With respect to an interest in a taxable mortgage pool or a residual interest in a real estate mortgage investment conduit, or REMIC, the ownership of which is attributed to us or to a REIT in which we own an interest, although the law on the matter is unclear as to the ownership of an interest in a taxable mortgage pool, we may be taxable at the highest corporate rate on the amount of any excess inclusion income for the taxable year allocable to the percentage of our stock that is held in record name by disqualified organizations. For a discussion of excess inclusion income, see Taxable Mortgage Pools. A disqualified organization includes:

The U.S.;

any state or political subdivision of the U.S.;

any foreign government;

any international organization;

any agency or instrumentality of any of the foregoing;

any other tax-exempt organization, other than a farmer s cooperative described in Section 521 of the Code, that is exempt both from income taxation and from taxation under the unrelated business taxable income provisions of the Code; and

any rural electrical or telephone cooperative.

We do not anticipate owning REMIC residual interests; we may, however, own 100% of the equity interests in one or more collateralized debt obligation, or CDO, offerings or one or more trusts formed in connection with our securitization transactions, but we intend to structure each CDO offering and each securitization transaction so that the issuing entity would not be classified as a TMP. See Taxable Mortgage Pools.

If we acquire any assets from a non-REIT C corporation in a carry-over basis transaction, we could be liable for specified tax liabilities inherited from that non-REIT C corporation with respect to that corporation s built-in gain in its assets. Built-in gain is the amount by which an asset s fair market value exceeds its adjusted tax basis at the time we acquire the asset. Applicable U.S. Treasury regulations, however, allow us to avoid the recognition of gain and the imposition of corporate level tax with respect to a built-in gain asset acquired in a carry-over basis transaction from a non-REIT C corporation unless and until we dispose of that built-in gain asset during the five-year period following its acquisition, at which time we would recognize, and would be subject to tax at the highest regular corporate rate on, the built-in gain.

In addition, notwithstanding our qualification as a REIT, we may also have to pay certain state and local income taxes because not all states and localities treat REITs in the same manner as they are treated for U.S. federal income tax purposes. Moreover, as further described below, any domestic taxable REIT subsidiary in which we own an interest will be subject to U.S. federal, state and local corporate income tax on its taxable income. We could also be subject to tax in situations and on transactions not presently contemplated.

Requirements for Qualification

The Code defines a REIT as a corporation, trust or association:

1. that is managed by one or more trustees or directors;

- 2. the beneficial ownership of which is evidenced by transferable shares, or by transferable certificates of beneficial interest;
- 3. that would be taxable as a domestic corporation, but for Sections 856 through 859 of the Code;
- 4. that is neither a financial institution nor an insurance company subject to certain provisions of the Code;
- 5. the beneficial ownership of which is held by 100 or more persons;
- 6. of which not more than 50% in value of the outstanding shares are owned, directly or indirectly, by five or fewer individuals (as defined in the Code to include certain entities) after applying certain attribution rules;
- 7. that makes an election to be taxed as a REIT for the current taxable year or has made such an election for a previous taxable year, and satisfies all relevant filing and other administrative requirements that must be met to elect and maintain REIT-qualification, which has not been terminated or revoked; and
- 8. that meets other tests, described below, regarding the nature of its income and assets and the amount of its distributions.

Conditions (1) through (4), inclusive, must be met during the entire taxable year. Condition (5) must be met during at least 335 days of a taxable year of 12 months, or during a proportionate part of a taxable year of less than 12 months other than the first taxable year for which an election to become a REIT is made. Condition (6) must be met during the last half of each taxable year but neither conditions (5) nor (6) apply to the first taxable year for which an election to become a REIT is made. For purposes of determining the stock ownership requirement described in (6) above, an

individual generally includes a supplemental unemployment compensation benefits plan, a private foundation, or a portion of a trust permanently set aside or used exclusively for charitable purposes. An individual, however, generally does not include a trust that is a qualified employee pension or profit sharing trust under the U.S. federal income tax laws, and beneficiaries of such a trust will be treated as holding our stock in proportion to their actuarial interests in the trust for purposes of the requirement described in (6) above. We believe that we have maintained and will maintain sufficient diversity of ownership to allow us to continue to satisfy conditions (5) and (6) above. In addition, our charter contains restrictions regarding the ownership and transfer of our stock that are intended to assist us in continuing to satisfy the share ownership requirements described in (5) and (6) above. These restrictions, however, may not ensure that we will be able to satisfy these share ownership requirements. If we fail to satisfy these share ownership requirements, we will fail to qualify as a REIT.

If we comply with regulatory rules pursuant to which we are required to send annual letters to holders of our stock requesting information regarding the actual ownership of our stock (as discussed below), and we do not know, or, exercising reasonable diligence, would not have known, whether we failed to meet requirement (6) above, we will be treated as having met the requirement.

To monitor compliance with the share ownership requirements described in (5) and (6) above, we generally are required to maintain records regarding the actual ownership of our stock. To do so, we must demand written

statements each year from the record holders of significant percentages of our stock pursuant to which the record holders must disclose the actual owners of the shares (i.e., the persons required to include our dividends in their gross income). We must maintain a list of those persons failing or refusing to comply with this demand as part of our records. We could be subject to monetary penalties if we fail to comply with these record-keeping requirements. If you fail or refuse to comply with the demands, you will be required by U.S. Treasury regulations to submit a statement with your tax return disclosing your actual ownership of our stock and other information. In addition, we must satisfy all relevant filing and other administrative requirements established by the IRS to elect and maintain REIT status, use a calendar year for U.S. federal income tax purposes, and comply with the record keeping requirements of the Code and regulations promulgated thereunder.

Ownership of Disregarded Entities and Partnership Interests. An unincorporated domestic entity, such as a partnership, limited liability company, or trust that has a single owner generally is not treated as an entity separate from its parent for U.S. federal income tax purposes, including for purposes of the gross income and asset tests applicable to REITs. An unincorporated domestic entity with two or more owners generally is treated as a partnership for U.S. federal income tax purposes. In the case of a REIT that is a partner in an entity that is treated as a partnership for U.S. federal income tax purposes, U.S. Treasury regulations provide that the REIT is deemed to own its proportionate share of the partnership s assets and to earn its proportionate share of the partnership s gross income tests applicable to REITs, as described below. However, solely for purposes of the 10% value test, described below (see

Asset Tests), the determination of a REIT s interest in partnership assets will be based on the REIT s proportionate interest in any securities issued by the partnership, excluding for these purposes, certain excluded securities as described in the Code. In addition, the assets and gross income of the partnership generally are deemed to retain the same character in the hands of the REIT. Thus, our proportionate share of the assets and items of income of partnerships in which we own an equity interest is treated as assets and items of income of our company for purposes of applying the REIT requirements described below. Consequently, to the extent that we directly or indirectly hold a preferred or other equity interest in a partnership or limited liability company, the partnership s or limited liability company s assets and operations may affect our ability to qualify as a REIT, even though we may have no control or only limited influence over the partnership.

Under the Bipartisan Budget Act of 2015, Congress revised the rules applicable to U.S. federal income tax audits of partnerships and the collection of any tax resulting from any such audits or other tax proceedings, generally for taxable years beginning after December 31, 2017. Under the new rules, the partnership itself may be liable for a hypothetical increase in partner-level taxes (including interest and penalties) resulting from an adjustment of partnership tax items on audit, regardless of changes in the composition of the partners (or their relative ownership) between the year under audit and the year of the adjustment. The new rules also include an elective alternative method under which the additional taxes resulting from the adjustment are assessed from the affected partners, subject to a higher rate of interest than otherwise would apply. Many questions remain as to how the new rules will apply, especially with respect to partners that are REITs, and it is not clear at this time what effect this new legislation will have on us. However, these changes could increase the U.S. federal income tax, interest, and/or penalties otherwise borne by us in the event of a U.S. federal income tax audit of a subsidiary partnership.

Qualified REIT Subsidiaries. If a REIT owns a corporate subsidiary that is a qualified REIT subsidiary, the separate existence of that subsidiary is disregarded for U.S. federal income tax purposes. Generally, a qualified REIT subsidiary is a corporation, other than a taxable REIT subsidiary, all of the stock of which is owned directly or indirectly by the REIT. All assets, liabilities and items of income, deduction and credit of the qualified REIT subsidiary will be treated as assets, liabilities and items of income, deduction and credit of the REIT itself. A qualified REIT subsidiary of ours is not subject to U.S. federal corporate income taxation, although it may be subject to state and local taxation in some states.

In the event that a qualified REIT subsidiary ceases to be wholly owned by us (for example, if any equity interest in the subsidiary is acquired by a person other than us or another disregarded subsidiary of us), the subsidiary s separate existence would no longer be disregarded for U.S. federal income tax purposes. Instead, it would have multiple owners and would be treated as either a partnership or a taxable corporation. Such an event could, depending on the circumstances, adversely affect our ability to satisfy the various asset and gross income tests applicable to REITs, including the requirement that REITs generally may not own, directly or indirectly, more than 10% of the value or voting power of the outstanding securities of another corporation. See Asset Tests and Income Tests.

Taxable REIT Subsidiaries. A taxable REIT subsidiary is an entity that is taxable as a corporation in which we directly or indirectly own stock and that elects with us to be treated as a taxable REIT subsidiary. In addition, if a taxable REIT subsidiary owns, directly or indirectly, securities representing 35% or more of the vote or value

of a subsidiary corporation, that subsidiary will also be treated as a taxable REIT subsidiary. However, an entity will not qualify as a taxable REIT subsidiary if it directly or indirectly operates or manages a lodging or health care facility or, generally, provides to another person, under a franchise, license or otherwise, rights to any brand name under which any lodging facility or health care facility is operated. We generally may not own more than 10%, as measured by voting power or value, of the securities of a corporation that is not a qualified REIT subsidiary unless we and such corporation elect to treat such corporation as a taxable REIT subsidiary. Overall, no more than 20% of the value of a REIT subsidiary is a stated by every between 2009 and 2017) may consist of stock or securities of one or more taxable REIT subsidiaries.

Income earned by a taxable REIT subsidiary is not attributable to the REIT. As a result, income that might not be qualifying income for purposes of the income tests applicable to REITs could be earned by a taxable REIT subsidiary without affecting our status as a REIT.

Several provisions of the Code regarding the arrangements between a REIT and its taxable REIT subsidiaries ensure that a taxable REIT subsidiary will be subject to an appropriate level of U.S. federal income taxation. For example, a taxable REIT subsidiary is limited in its ability to deduct interest payments made to affiliated REITs. In addition, we would be obligated to pay a 100% penalty tax on some payments that we receive from, or on certain expenses deducted by, a taxable REIT subsidiary if the IRS were to assert successfully that the economic arrangements between us and a taxable REIT subsidiary are not comparable to similar arrangements among unrelated parties. Any income earned by a taxable REIT subsidiary that is attributable to services provided to us, or on our behalf to any of our tenants, that is less than the amounts that would have been charged based upon arm s-length negotiations, will also be subject to a 100% penalty tax. We intend to scrutinize all of our transactions with any of our subsidiaries that are treated as a taxable REIT subsidiary in an effort to ensure that we do not become subject to this penalty tax; however, we cannot assure you that we will be successful in avoiding this penalty tax. Also, for taxable years beginning after December 31, 2017, overall limitations on the deductibility of net interest expense could apply to any taxable REIT subsidiary.

Taxable Mortgage Pools. An entity, or a portion of an entity, that does not elect to be treated as a REMIC may be classified as a taxable mortgage pool, or TMP, under the Code if:

substantially all of its assets consist of debt obligations or interests in debt obligations;

more than 50% of those debt obligations are real estate mortgages or interests in real estate mortgages as of specified testing dates;

the entity has issued debt obligations (liabilities) that have two or more maturities; and

the payments required to be made by the entity on its debt obligations bear a relationship to the payments to be received by the entity on the debt obligations that it holds as assets.

Under the U.S. Treasury regulations, if less than 80% of the assets of an entity (or a portion of an entity) consists of debt obligations, these debt obligations are considered not to comprise substantially all of its assets, and therefore the entity would not be treated as a taxable mortgage pool.

We do not intend to structure or enter into securitization or financing transactions that will cause us to be viewed as owning interests in one or more TMPs. Generally, if an entity or a portion of an entity is classified as a TMP, then the entity or portion thereof is treated as a taxable corporation and it cannot file a consolidated U.S. federal income tax return with any other corporation. If, however, a REIT owns 100% of the equity interests in a TMP, then the TMP is a qualified REIT subsidiary and, as such, ignored as an entity separate from the REIT.

If, notwithstanding our intent to avoid having the issuing entity in any of our securitization or financing transactions classified as a TMP, one or more of such transactions were so classified, then as long as we owned 100% of the equity interests in the TMP, all or a portion of the income that we recognize with respect to our investment in the TMP will be treated as excess inclusion income. Section 860E(c) of the Code defines the term

excess inclusion with respect to a residual interest in a REMIC. The IRS, however, has yet to issue guidance on the computation of excess inclusion income on equity interests in a TMP held by a REIT. Generally, however, excess inclusion income with respect to our investment in any TMP and any taxable year will equal the excess of (i) the amount of income we accrue on our investment in the TMP over (ii) the amount of income we would have accrued if our investment were a debt instrument having an issue price equal to the fair market value of our investment on the day we acquired it and a yield to maturity equal to 120% of the long-term applicable federal rate in effect on the date we acquired our interest. The term applicable federal rate refers to rates that are based on weighted average yields for Treasury securities and are published monthly by the IRS for use in various tax calculations. If we undertake securitization transactions that are TMPs, the amount of excess inclusion income we recognize in any taxable year could represent a significant portion of our total taxable income for that year.

Although we intend to structure our securitization and financing transactions so that we will not recognize any excess inclusion income, we cannot assure you that we will always be successful in this regard. If, notwithstanding our intent, we recognized excess inclusion income, then under guidance issued by the IRS we would be required to allocate the excess inclusion income proportionately among the dividends we pay to our stockholders and we must notify our stockholders of the portion of our dividends that represents excess inclusion income. The portion of any dividend you receive that is treated as excess inclusion income is subject to special rules. First, your taxable income can never be less than the sum of your excess inclusion income for the year; excess inclusion income cannot be offset with net operating losses or other allowable deductions. Second, if you are a tax-exempt organization and your excess inclusion income is subject to the unrelated business income tax, then the excess inclusion portion of any dividend you receive will be treated as unrelated business taxable income. Third, dividends paid to Non-U.S. holders who hold stock for investment and not in connection with a trade or business conducted in the U.S. will be subject to U.S. federal withholding tax without regard to any reduction in rate otherwise allowed by any applicable income tax treaty.

If we recognize excess inclusion income, and one or more disqualified organizations are record holders of shares of common stock, we will be taxable at the highest U.S. federal corporate income tax rate on the portion of any excess inclusion income equal to the percentage of our stock that is held by disqualified organizations. In such circumstances, we may reduce the amount of our distributions to a disqualified organization whose stock ownership gave rise to the tax. To the extent that our common stock owned by disqualified organizations is held by a broker/dealer or other nominee, the broker/dealer or other nominee would be liable for a tax at the highest corporate tax rate on the portion of our excess inclusion income allocable to our common stock held by the broker/dealer or other nominee on behalf of the disqualified organizations.

If we own less than 100% of the equity interests in a TMP, the foregoing rules would not apply. Rather, the entity would be treated as a corporation for U.S. federal income tax purposes and would potentially be subject to U.S. federal corporate income tax. This could adversely affect our compliance with the REIT gross income and asset tests described above. We currently do not have, and currently do not intend to enter into any securitization or financing transaction that is a TMP in which we own some, but less than all, of the equity interests, and we intend to monitor the structure of any TMPs in which we have an interest to ensure that they will not adversely affect our status as a REIT. We cannot assure you that we will be successful in this regard.

Tax-exempt investors, non-U.S. investors and taxpayers with net operating losses should carefully consider the tax consequences described above and are urged to consult their tax advisors in connection with their decision to invest in our common stock.

Income Tests

To qualify as a REIT, we must satisfy two gross income requirements, each of which is applied on an annual basis. First, at least 75% of our gross income for each taxable year generally must be derived directly or indirectly from:

rents from real property;

interest on debt secured by a mortgage on real property or on interests in real property;

dividends or other distributions on, and gain from the sale of, shares in other REITs;

gain from the sale or other disposition of real property (including interests in real property and interests in mortgages on real property) that is not treated as inventory property or property held for sale to customers in the ordinary course of business;

abatements and refunds of taxes on real property;

income and gain derived from foreclosure property (as described above);

amounts (other than amounts the determination of which depends in whole or in part on the income or profits of any person) received or accrued as consideration for entering into agreements (i) to make loans secured by mortgages on real property or on interests in real property or (ii) to purchase or lease real property (including interests in real property);

income derived from a REMIC in proportion to the real estate assets held by the REMIC, unless at least 95% of the REMIC s assets are real estate assets, in which case all of the income derived from the REMIC; and

interest or dividend income from investments in stock or debt instruments attributable to the temporary investment of new capital during the one-year period following our receipt of new capital that we raise through equity offerings or public offerings of debt obligations with at least a five-year term.
Second, at least 95% of our gross income for each taxable year must be derived from sources that qualify for purposes of the 75% gross income test, and from (i) dividends, (ii) interest and (iii) gain from the sale or disposition of stock or securities (provided that such stock or securities are not inventory property, *i.e.*, property held primarily for sale to customers in the ordinary course of business), or any combination of these.

Gross income from our sale of property that we hold primarily for sale to customers in the ordinary course of business is excluded from both the numerator and the denominator in both gross income tests. In addition, income and gain from hedging transactions that we enter into to hedge indebtedness incurred or to be incurred to acquire or carry real estate assets and that are clearly and timely identified as such will be excluded from both the numerator and the denominator for purposes of both gross income tests. In addition, certain foreign currency gains will be excluded from gross income for purposes of one or both of the gross income tests. Finally, gross income attributable to cancellation of indebtedness income will be excluded from both the numerator and the denominator for purposes of both gross both gross the specific application of the gross income tests to us. We will monitor the amount of our non-qualifying income and we will seek to manage our portfolio to comply at all times with the gross income tests, but we cannot assure you that we will be successful in this effort. The following paragraphs discuss of the gross income tests to us.

Interest. The term interest, as defined for purposes of both gross income tests, generally excludes any amount that is based in whole or in part on the income or profits of any person, however, it generally includes the following: (i) an amount that is received or accrued based on a fixed percentage or percentages of receipts or sales, and (ii) an amount that is based on the income or profits of a debtor, as long as the debtor derives substantially all of its income from the real property securing the debt by leasing substantially all of its interest in the property, and only to the extent that the amounts received by the debtor would be qualifying rents from real property if received directly by a REIT. We do not expect that any of our loans will be based in whole or in part on the income or profits of any person.

Interest on debt secured by mortgages on real property or on interests in real property, including, for this purpose, original issue discount, prepayment penalties, loan assumption fees and late payment charges that are not compensation for services, generally is qualifying income for purposes of the 75% gross income test. If a

loan is secured by real property and other property and the highest principal amount of a loan outstanding during a taxable year exceeds the fair market value of the real property securing the loan as of the date (i) we agreed to originate or acquire the loan or (ii) as discussed below, in the event of a significant modification, the date we modified the loan, a portion of the interest income from such loan will not be qualifying income for purposes of the 75% gross income test but will be qualifying income for purposes of the 95% gross income test. However, in the case of a loan that is secured by both real property and personal property, if the fair market value of such personal property does not exceed 15% of the total fair market value of all property securing the loan, then the personal property securing the loan will be treated as real property for purposes of determining the interest on such loan is qualifying income for purposes of the 75% gross income test. If apportionment is required, the portion of the interest income that will not be qualifying income for purposes of the 75% gross income test will be equal to the portion of the principal amount of the loan that is not secured by real property that is, the amount by which the loan exceeds the value of the real estate that is security for the loan.

We expect that the mortgage-backed securities, or MBS, in which we invest generally will be treated either as interests in a grantor trust or as regular interests in a REMIC for U.S. federal income tax purposes and that all interest income from such MBS will be qualifying income for the 95% gross income test. In the case of MBS treated as interests in grantor trusts, we would be treated as owning an undivided beneficial ownership interest in the mortgage loans held by the grantor trust. The interest on such mortgage loans would be qualifying income for purposes of the 75% gross income test to the extent that the obligation is secured by real property, as discussed above. In the case of MBS treated as qualifying income for purposes of the 75% and 95% gross income tests. If less than 95% of the assets of the REMIC are real estate assets, however, then only a proportionate part of our interest in the REMIC and income derived from the interest will qualify for purposes of the 75% gross income test. In addition, some REMIC securitizations include imbedded interest swap or cap contracts or other derivative instruments that potentially could produce non-qualifying income for the holder of the related REMIC securities.

We purchase and sell Agency MBS through to-be-announced forward contracts, or TBAs, and recognize income or gains from the disposition of those TBAs, through dollar roll transactions or otherwise. While there is no direct authority with respect to the qualification of income or gains from dispositions of TBAs as gains from the sale of real property (including interests in real property and interests in mortgages on real property) or other qualifying income for purposes of the 75% gross income test, we treat income and gains from our TBAs as qualifying income for purposes of the 75% REIT gross income test, any gain recognized by us in connection with the settlement of our TBAs should be treated as gain from the sale or disposition of an interest in mortgages on real property. If the IRS were to successfully challenge this opinion of counsel, we could be subject to a penalty tax or we could fail to remain qualified as a REIT if a sufficient portion of our income consists of income or gains from the disposition of TBAs.

We own interests in mezzanine loans, which are loans secured by equity interests in an entity that directly or indirectly owns real property, rather than by a direct mortgage of the real property. In Revenue Procedure 2003-65, the IRS established a safe harbor under which loans secured by a first priority security interest in the ownership interests in a partnership or limited liability company owning real property will be treated as real estate assets for purposes of the REIT asset tests described below, and interest derived from those loans will be treated as qualifying income for both the 75% and 95% gross income tests, provided several requirements are satisfied. Although the Revenue Procedure provides a safe harbor on which taxpayers may rely, it does not prescribe rules of substantive tax law. Moreover, our mezzanine loans may not meet all of the requirements for reliance on the safe harbor. To the extent any mezzanine loans that we acquire do not qualify for the safe harbor described above, the interest income from the loans will be qualifying income for purposes of the 95% gross income test, but there is a risk that such interest income will not be qualifying income for purposes of the 75% gross income test. We believe we have invested, and intend to continue to

invest, in mezzanine loans in a manner that will enable us to continue to satisfy the REIT gross income and asset tests.

We may modify the terms of our mortgage or mezzanine loans. Under the Code, if the terms of a loan are modified in a manner constituting a significant modification, such modification triggers a deemed exchange of the original loan for the modified loan. IRS Revenue Procedure 2014-51 provides a safe harbor pursuant to which we will not be required to redetermine the fair market value of the real property securing a loan for purposes of the gross income and asset tests in connection with a loan modification that is (i) occasioned by a borrower default or (ii) made at a time when we reasonably believe that the modification to the loan will substantially reduce a significant risk of default on the original loan. To the extent we significantly modify loans in a manner that does not qualify for that safe harbor, we will be required to redetermine the value of the real property securing the loan at the time it was significantly modified, which could result in a portion of the interest income on the loan being treated as non-qualifying income for purposes of the 75% gross income test. In determining the value of the real property securing such a loan, we generally will not obtain third-party appraisals but rather will rely on internal valuations.

The interest, original issue discount, and market discount income that we will receive from our mortgage-related assets generally will be qualifying income for purposes of both gross income tests. Some of our investments, such as the investments we acquire through our middle market lending activities, will not be secured by mortgages on real property or interests in real property. Our interest income from those investments will be qualifying income for purposes of the 95% gross income test, but not the 75% gross income test.

We also invest in excess mortgage servicing rights (MSR), which represent the portion of the servicing fee paid to mortgage servicers in excess of the reasonable compensation that would be charged for mortgage servicing in an arm s length transaction. In private letter rulings issued to taxpayers, the IRS has ruled substantially to the effect that interest received in respect of an excess MSR will be considered interest on obligations. Private letter rulings cannot be relied upon by persons other than the taxpayer to which they were issued. Nonetheless, we treat income from any excess MSR that have terms consistent with those described in such private letter rulings as qualifying income for purposes of the 75% gross income test. In the event that such income were determined not to qualify for the 75% gross income test, we could be subject to a penalty tax or we could fail to qualify as a REIT if such income, together with our other income that does not qualify for the 75% gross income test, were to exceed 25% of our gross income for any taxable year.

We have entered, and intend to enter, into financing arrangements that are structured as sale and repurchase agreements pursuant to which we would nominally sell certain of our assets to a counterparty and simultaneously enter into an agreement to repurchase these assets at a later date in exchange for a purchase price. Economically, these agreements are financings that are secured by the assets sold pursuant thereto. We believe that we would be treated for REIT asset and income test purposes as the owner of the assets that are the subject of any such sale and repurchase agreement notwithstanding that such agreements may transfer record ownership of the assets to the counterparty during the term of the agreement. It is possible, however, that the IRS could assert that we did not own the assets during the term of the sale and repurchase agreement, in which case we could fail to qualify as a REIT.

Hedging Transactions. We may enter into hedging transactions with respect to one or more of our assets or liabilities. Hedging transactions could take a variety of forms, including interest rate swap agreements, interest rate cap agreements, options, futures contracts, forward rate agreements or similar financial instruments. Except to the extent provided by U.S. Treasury regulations, any income from a hedging transaction we enter into (i) in the normal course of our business primarily to manage risk of interest rate or price changes or currency fluctuations with respect to borrowings made or to be made, or ordinary obligations incurred or to be incurred, to acquire or carry real estate assets, which is clearly identified as specified in U.S. Treasury regulations before the close of the day on which it was acquired, originated or entered into, including gain from the sale or disposition of such a transaction, (ii) primarily to manage risk of currency fluctuations with respect to any item of income or gain that would be qualifying income under the 75% or 95% gross income tests that is clearly identified as such before the close of the day on which it was

acquired, originated or entered into and satisfies other identification requirements, or (iii) in connection with the effective termination of certain hedging transactions described

above, will not constitute gross income for purposes of the 75% or 95% gross income tests. To the extent that we enter into other types of hedging transactions, the income from those transactions is likely to be treated as non-qualifying income for purposes of both of the 75% and 95% gross income tests. We intend to structure any hedging transactions in a manner that does not jeopardize our qualification as a REIT.

We may conduct some or all of our hedging activities through a taxable REIT subsidiary or other corporate entity, the income of which may be subject to U.S. federal income tax, rather than by participating in the arrangements directly or through pass-through subsidiaries.

Fee Income. Fee income generally will be qualifying income for purposes of both the 75% and 95% gross income tests if it is received in consideration for entering into an agreement to make a loan secured by real property and the fees are not determined by income and profits. Other fees generally are not qualifying income for purposes of either gross income test. Any fees earned by a taxable REIT subsidiary will not be included for purposes of the gross income tests.

Rents from Real Property. To the extent that we own or acquire real property or an interest therein, rents we receive will qualify as rents from real property in satisfying the gross income requirements for a REIT described above only if several conditions are met. These conditions relate to the identity of the tenant, the computation of the rent payable, and the nature of the property leased. First, the amount of rent must not be based in whole or in part on the income or profits of any person. However, an amount received or accrued generally will not be excluded from rents from real property solely by reason of being based on a fixed percentage or percentages of receipts or sales. Second, we, or an actual or constructive owner of 10% or more of the value of our stock, must not actually or constructively own 10% or more of the assets or net profits of the tenant, or, if the tenant is a corporation, 10% or more of the voting power or value of all classes of stock of the tenant. Rents received from such tenant that is a taxable REIT subsidiary, however, will not be excluded from the definition of rents from real property as a result of this condition if either (i) at least 90% of the space at the property to which the rents relate is leased to third parties, and the rents paid by the taxable REIT subsidiary are comparable to rents paid by our other tenants for comparable space, or (ii) the property is a qualified lodging or qualified health facility and such property is operated on behalf of the taxable REIT subsidiary by a person who is an independent contractor and certain other requirements are met. Third, if rent attributable to personal property leased in connection with a lease of real property is greater than 15% of the total rent received under the lease, then the portion of rent attributable to the personal property will not qualify as rents from real property. Finally, for rents to qualify as rents from real property for purposes of the gross income tests, we are only allowed to provide services that are both usually or customarily rendered in connection with the rental of real property and not otherwise considered rendered to the occupant. We may, however, render services to our tenants through an

independent contractor who is adequately compensated and from whom we do not derive revenue. We may also own a taxable REIT subsidiary which provides non-customary services to tenants without tainting our rental income from the related properties.

Even if a REIT furnishes or renders services that are non-customary with respect to a property, if the greater of (i) the amounts received or accrued, directly or indirectly, or deemed received by the REIT with respect to such services, or (ii) 150% of our direct cost in furnishing or rendering the services during a taxable year is not more than 1% of all amounts received or accrued, directly or indirectly by the REIT with respect to the property during the same taxable year, then only the amounts with respect to such non-customary services are not treated as rent for purposes of the REIT gross income tests.

In connection with the offer and the merger, we will acquire certain senior housing properties. As described above, rental revenue will generally not qualify as rents from real property for purposes of the REIT income tests if we directly or constructively hold a 10% or greater interest, as measured by vote or value, in the lessee s equity. Rents we

receive from a tenant that also is our taxable REIT subsidiary, however, will generally not be excluded from the definition of rents from real property as a result of our ownership interest in the taxable REIT subsidiary if the property to which the rents relate is a qualified lodging facility or a qualified healthcare

property and such property is operated on behalf of the taxable REIT subsidiary by a person who is an eligible independent contractor. An operator will qualify as an eligible independent contractor if it meets certain ownership tests with respect to us, and if, at the time the operator enters into the property management agreement, the operator is actively engaged in the trade or business of operating qualified healthcare properties for any person who is not a related person to us or the taxable REIT subsidiary. A qualified healthcare property means any real property (or any personal property incidental thereto) that is, or that is necessary or incidental to the use of, a healthcare facility. Certain of the senior housing properties that we own are leased to a taxable REIT subsidiary in a structure intended to satisfy the foregoing conditions. If, however, any of the above conditions were not satisfied, then the rents from such a lease would not be considered income from a qualifying source for purposes of the REIT rules, which could cause us to incur penalty taxes or to fail to qualify as a REIT.

Prohibited Transactions Tax. A REIT will incur a 100% tax on the net income derived from any sale or other disposition of property, other than foreclosure property, that the REIT holds primarily for sale to customers in the ordinary course of a trade or business. Whether a REIT holds an asset primarily for sale to customers in the ordinary course of a trade or business depends, however, on the facts and circumstances in effect from time to time, including those related to a particular asset. Nevertheless, we generally intend to conduct our operations so that no asset that we own will be treated as, or as having been, held for sale to customers, and that a sale of any such asset will not be treated as having been in the ordinary course of our business. We cannot assure you that we will comply with certain safe harbor provisions or that we will avoid owning property that may be characterized as property that we hold primarily for sale to customers in the ordinary course of a trade or business. The 100% tax will not apply to gains from the sale of property that is held through a taxable REIT subsidiary or other taxable corporation, although such income will be subject to tax in the hands of the corporation at regular corporate income tax rates.

Foreclosure Property. Foreclosure property is any real property, including interests in real property, and any personal property incident to such real property:

that is acquired by a REIT as the result of the REIT having bid in such property at foreclosure, or having otherwise reduced such property to ownership or possession by agreement or process of law, after there was a default or default was imminent on a lease of such property or on indebtedness that such property secured;

for which the related loan was acquired by the REIT at a time when the default was not imminent or anticipated; and

for which the REIT makes a proper election to treat the property as foreclosure property. However, a REIT will not be considered to have foreclosed on a property where the REIT takes control of the property as a mortgagee-in-possession and cannot receive any profit or sustain any loss except as a creditor of the mortgagor.

Property generally ceases to be foreclosure property at the end of the third taxable year following the taxable year in which the REIT acquired the property, or longer if an extension is granted by the Secretary of the Treasury. This grace period terminates and foreclosure property ceases to be foreclosure property on the first day:

on which a lease is entered into for the property that, by its terms, will give rise to income that does not qualify for purposes of the 75% gross income test, or any amount is received or accrued, directly or indirectly, pursuant to a lease entered into on or after such day that will give rise to income that does not qualify for purposes of the 75% gross income test;

on which any construction takes place on the property, other than completion of a building or any other improvement, if more than 10% of the construction was completed before default became imminent; or

which is more than 90 days after the day on which the REIT acquired the property and the property is used in a trade or business that is conducted by the REIT, other than through an independent contractor from whom the REIT itself does not derive or receive any income or a taxable REIT subsidiary.

We will be subject to tax at the maximum corporate rate on any income from foreclosure property, other than income that otherwise would be qualifying income for purposes of the 75% gross income test, less expenses directly connected with the production of that income. However, income from foreclosure property, including gain from the sale of foreclosure property held for sale in the ordinary course of a trade or business, will qualify for purposes of the 75% and 95% gross income tests.

We may have the option to foreclose on mortgage loans when a borrower is in default. The foregoing rules could affect a decision by us to foreclose on a particular mortgage loan and could affect whether we choose to foreclose with regard to a particular mortgage loan.

Foreign Currency Gain. Certain foreign currency gains will be excluded from gross income for purposes of one or both of the gross income tests. Real estate foreign exchange gain will be excluded from gross income for purposes of the 75% and 95% gross income tests. Real estate foreign exchange gain generally includes foreign currency gain attributable to any item of income or gain that is qualifying income for purposes of the 75% gross income test, foreign currency gain attributable to the acquisition or ownership of (or becoming or being the obligor under) obligations secured by mortgages on real property or an interest in real property and certain foreign currency gain attributable to certain qualified business units of a REIT that satisfies the 75% gross income test and 75% asset test on a stand-alone basis. Passive foreign exchange gain generally includes real estate foreign exchange gain as described above, and also includes foreign currency gain attributable to any item of income or gain that is qualifying income or gain that is qualifying income for purposes of the 95% gross income test and foreign currency gain attributable to the acquisition or ownership of income or gain that is qualifying income for purposes of the 95% gross income test. Passive foreign currency gain attributable to any item of income or gain that is qualifying income for purposes of the 95% gross income test and foreign currency gain attributable to the acquisition or ownership of (or becoming or being the obligor under) obligations. These exclusions for real estate foreign exchange gain and passive foreign exchange gain do not apply to any foreign currency gain derived from dealing, or engaging in substantial and regular trading, in securities. Such gain is treated as non-qualifying income for purposes of both the 75% and 95% gross income tests.

Dividends. Our dividend income from stock in any corporation (other than any REIT) and from any taxable REIT subsidiary will be qualifying income for purposes of the 95% gross income test, but not the 75% gross income test. If we own stock in other REITs, the dividends that we receive from those REITs and our gain on the sale of the stock in those REITs will be qualifying income for purposes of both gross income tests. However, if a REIT in which we own stock fails to qualify as a REIT in any year, our income from such REIT would be qualifying income for purposes of the 95% gross income test, but not the 75% gross income test.

Phantom Income. Due to the nature of the assets in which we will invest, we may be required to recognize taxable income from certain assets in advance of our receipt of cash flow from or proceeds from disposition of such assets, and may be required to report taxable income that exceeds the economic income ultimately realized on such assets.

We may acquire debt instruments or MBS in the secondary market for less than their face amount. The amount of such discount generally will be treated as market discount for U.S. federal income tax purposes. Accrued market discount is reported as income when, and to the extent that, any payment of principal of the debt instrument is made, unless we elect to include accrued market discount in income as it accrues. Principal payments on certain loans are made monthly, and consequently accrued market discount may have to be included in income each month as if the debt instrument were assured of ultimately being collected in full. If we collect less on the debt instrument than our purchase price plus the market discount we had previously reported as income, we may not be able to benefit from any offsetting loss deductions.

We may agree to modify the terms of distressed and other loans we hold. These modifications may be considered significant modifications for U.S. federal income tax purposes that give rise to a deemed

debt-for-debt exchange upon which we may recognize taxable income or gain without a corresponding receipt of cash.

Some of the loans and debt securities that we acquire may have been issued with original issue discount. In general, we will be required to accrue original issue discount based on the constant yield to maturity of the debt securities, and to treat it as taxable income in accordance with applicable U.S. federal income tax rules even though such yield may exceed cash payments, if any, received on such debt instrument. Under the recent tax legislation informally known as the Tax Cuts and Jobs Act, or TCJA, we generally will be required to take certain amounts in income no later than the time such amounts are reflected on certain financial statements. The application of this rule may require the accrual of income with respect to our debt instruments or MBS, such as original issue discount or market discount, earlier than would be the case under the general tax rules, although the precise application of this rule is unclear at this time. This rule generally will be effective for tax years beginning after December 31, 2017 or, for debt instruments or MBS issued with original issue discount, for tax years beginning after December 31, 2018. To the extent that this rule requires the accrual of income earlier than under the general tax rules, it could increase our phantom income, which may make it more likely that we could be required to borrow funds or take other action to satisfy the REIT distribution requirements for the taxable year in which this phantom income is recognized. We currently do not expect that this rule will have a material impact on the timing of accrual of our income or on the amount of our distribution requirement.

In addition, in the event that any debt instruments or debt securities acquired by us are delinquent as to mandatory principal and interest payments, or in the event payments with respect to a particular debt instrument are not made when due, we may nonetheless be required to continue to recognize the unpaid interest as taxable income. Similarly, we may be required to accrue interest income with respect to subordinated MBS at the stated rate regardless of whether corresponding cash payments are received.

Finally, we may be required under the terms of indebtedness that we incur to use cash received from interest payments to make principal payments on that indebtedness, with the effect of recognizing income but not having a corresponding amount of cash available for distribution to our stockholders.

As a result of each of these potential timing differences between income recognition or expense deduction and cash receipts or disbursements, there is a significant risk that we may have substantial taxable income in excess of cash available for distribution. In that event, we may need to borrow funds or take other action to satisfy the REIT distribution requirements for the taxable year in which this phantom income is recognized. See Annual Distribution Requirements.

Failure to Satisfy Gross Income Tests. If we fail to satisfy one or both of the 75% and 95% gross income tests for any taxable year, we may nevertheless qualify as a REIT for that year if we are entitled to relief under certain provisions of the Code. These relief provisions generally would be available if our failure to meet such tests is due to reasonable cause and not due to willful neglect, and we attach a schedule of the sources of our income to our U.S. federal income tax return. It is not possible, however, to state whether in all circumstances we would be entitled to the benefit of these relief provisions. For example, if we fail to satisfy the gross income tests because non-qualifying income that we intentionally recognize exceeds the limits on non-qualifying income, the IRS could conclude that the failure to satisfy the tests was not due to reasonable cause. If these relief provisions were inapplicable to a particular set of circumstances, we would fail to qualify as a REIT. Even if these relief provisions applied, a penalty tax would be imposed based on the amount of non-qualifying income. See General and Failure to Qualify.

Asset Tests

At the close of each quarter of our taxable year, we must satisfy the following tests relating to the nature of our assets.

At least 75% of the value of our total assets must be represented by the following:

interests in real property, including leaseholds and options to acquire real property and leaseholds, and, for taxable years beginning after December 31, 2015, personal property to the extent such personal property is leased in connection with real property and rents attributable to such personal property are treated as rents from real property ;

interests in mortgages on real property;

stock in other REITs and debt instruments issued by publicly offered REITs;

cash and cash items;

government securities;

investments in stock or debt instruments attributable to the temporary investment of new capital during the one-year period following our receipt of new capital that we raise through equity offerings or public offerings of debt obligations with at least a five-year term; and

regular or residual interests in a REMIC. However, if less than 95% of the assets of a REMIC consist of assets that are qualifying real estate-related assets under the U.S. federal income tax laws, determined as if we held such assets directly, we will be treated as holding directly our proportionate share of the assets of such REMIC.

Not more than 25% of our total assets may be represented by securities, other than those in the 75% asset class.

Except for securities in taxable REIT subsidiaries and the securities in the 75% asset class, the value of any one issuer s securities owned by us may not exceed 5% of the value of our total assets (or, the 5% asset test).

Except for securities in taxable REIT subsidiaries and the securities in the 75% asset class, we may not own more than 10% of any one issuer s outstanding voting securities (or, the 10% vote test).

Except for securities of taxable REIT subsidiaries and the securities in the 75% asset class, we may not own more than 10% of the total value of the outstanding securities of any one issuer, other than securities that qualify for the straight debt exception discussed below (or, 10% value test).

Not more than 20% (25% for our taxable years between 2009 and 2017) of the value of our total assets may be represented by the securities of one or more taxable REIT subsidiaries.

For any taxable year beginning after December 31, 2015, not more than 25% of the value of our total assets may be represented by debt instruments of publicly offered REITs that are not secured by real property or interests in real property.

Securities, for the purposes of the asset tests, may include debt we hold from other issuers. However, debt we hold in an issuer that does not qualify for purposes of the 75% asset test will not be taken into account for purposes of the 10% value test if the debt securities meet the straight debt safe harbor. Debt will meet the straight debt safe harbor if the debt is a written unconditional promise to pay on demand or on a specified date a sum certain in money, the debt is not convertible, directly or indirectly, into stock, and the interest rate and the interest payment dates of the debt are not contingent on the profits, the borrower s discretion or similar factors. In the case of an issuer that is a corporation or a partnership, securities that otherwise would be considered straight debt will not be so considered if we, and any of our

controlled taxable REIT subsidiaries as defined in the Code, hold any securities of the corporate or partnership issuer that (a) are not straight debt or other excluded securities (prior to the application of this rule), and (b) have an aggregate value greater than 1% of the issuer s outstanding securities (including, for the purposes of a partnership issuer, our interest as a partner in the partnership).

In addition, the following instruments will not be taken into account for purposes of the 10% value test: (i) a REIT s interest as a partner in a partnership; (ii) any debt instrument issued by a partnership (other than straight debt or any other excluded security) will not be considered a security issued by the partnership if at least 75% of the partnership s gross income is derived from sources that would qualify for the 75% REIT gross income test; (iii) any debt instrument issued by a partnership (other than straight debt or any other excluded security) will not be to the extent of the REIT s interest as a partner in the partnership; (iv) any loan to an individual or an estate; (v) any Section 467 rental agreement, other than an agreement with a related party tenant; (vi) any obligation to pay rents from real property ; (vii) certain securities issued by governmental entities that are not dependent in whole or in part on the profits of (or payments made by) a non-governmental entity; and (viii) any security (including debt securities) issued by another REIT. For purposes of the 10% value test, our proportionate share of the assets of a partnership is our proportionate interest in any securities issued by the partnership, without regard to the securities described clause (i) and (ii) in the preceding sentence.

For purposes of the 75% asset test, mortgage loans generally will qualify as real estate assets to the extent that they are secured by real property. Where a mortgage covers both real property and other property, an apportionment may be required in the same manner as described under Income Tests Interest. IRS Revenue Procedure 2014-51 provides a safe harbor under which the IRS has stated that it will not challenge a REIT s treatment of a loan as being, in part, a qualifying real estate asset in an amount equal to the lesser of (i) the fair market value of the loan on the relevant quarterly REIT asset testing date or (ii) the greater of (a) the fair market value of the real property securing the loan on the relevant quarterly REIT asset testing date or (b) the fair market value of the real property securing the loan on the date the REIT committed to originate or acquire the loan. It is unclear how the safe harbor in Revenue Procedure 2014-51 is affected by the subsequent legislative changes regarding the treatment of loans secured by both real property and personal property where the fair market value of the personal property does not exceed 15% of the sum of the fair market values of the real property and personal property and personal property securing the loan.

As described above, Revenue Procedure 2003-65 provides a safe harbor pursuant to which certain mezzanine loans secured by a first priority security interest in ownership interests in a partnership or limited liability company will be treated as qualifying assets for purposes of the 75% asset test (and therefore, are not subject to the 5% asset test and the 10% vote test or 10% value test). See Income Tests. We believe our mezzanine loans should be treated as qualifying assets for the 75% asset test or should be excluded from the definition of securities for purposes of the 10% vote or 10% value test. To the extent we originate or acquire mezzanine loans, we intend to do so only to the extent such loans will not cause us to fail the asset tests described above.

We expect that our investments in MBS will generally be treated as interests in a grantor trust or as regular interests in a REMIC for U.S. federal income tax purposes. In the case of MBS treated as interests in grantor trusts, we would be treated as owning an undivided beneficial ownership interest in the mortgage loans held by the grantor trust. In the case of MBS treated as an interest in a REMIC, such interests will generally qualify as real estate assets, and income derived from REMIC regular interests will generally be treated as qualifying income for purposes of the REIT income tests described above. If less than 95% of the assets of a REMIC are real estate assets, however, then only a proportionate part of our interest in the REMIC and income derived from the interest qualifies for purposes of the REIT asset and income tests. We purchase and sell Agency MBS through TBAs. While there is no direct authority with respect to the qualification of TBAs as real estate assets or U.S. Government securities for purposes of the 75% asset test, we treat our TBAs as qualifying assets for purposes of the REIT asset tests, based on an opinion of prior tax counsel substantially to the effect that for purposes of the REIT asset tests, our ownership of a TBA should be treated as ownership of real estate assets. If the IRS were to successfully challenge the opinion of our assets consists of TBAs.

We believe that most of the assets that we hold and those we expect to hold will be qualifying assets for purposes of the 75% asset test. However, our investment in other asset-backed securities, bank loans and other

instruments that are not secured by mortgages on real property will not be qualifying assets for purposes of the 75% asset test.

We will monitor the status of our assets for purposes of the various asset tests and will seek to manage our portfolio to comply at all times with such tests. There can be no assurances, however, that we will be successful in this effort. In this regard, to determine our compliance with these requirements, we will need to estimate the value of the real estate securing our mortgage loans at various times. In addition, we will be required to value our investment in our other assets to ensure compliance with the asset tests. Although we will seek to be prudent in making these estimates, no independent appraisals will be obtained to support our conclusions as to the value of our assets and securities, or in many cases, the real estate collateral for the senior loans and mezzanine loans that we hold. Moreover, values of some assets may not be susceptible to a precise determination. As a result, there can be no assurances that the IRS may not disagree with these determinations and assert that a different value is applicable, in which case we may not satisfy the 75% and the other asset tests.

We will not lose our REIT status for a de minimis failure to meet the 5% or 10% asset requirements if the failure is due to ownership of assets the total value of which does not exceed the lesser of 1% of the total value of our assets or \$10 million. If we fail to satisfy any of the asset requirements for a particular tax quarter, we may still qualify as a REIT if we (1) identify the failure on a separate schedule, (2) the failure is due to reasonable cause and not willful neglect, (3) the assets causing the failure are disposed of within six months of the last day of the quarter in which the failure occurred and (4) we pay a tax computed as the greater of either \$50,000 or the net income generated by the assets causing the failure multiplied by the highest corporate income tax rate.

After initially meeting the asset tests after the close of any quarter, we will not lose our status as a REIT if we fail to satisfy the asset tests at the end of a later quarter solely by reason of changes in the relative values of our assets. However, an acquisition of property by a REIT requires the REIT to revalue all of its assets. If the failure to satisfy the asset tests results from an increase in the value of our assets after the acquisition of securities or other property during a quarter, the failure can be cured by eliminating the discrepancy within 30 days after the close of that quarter. We intend to maintain adequate records of the value of our assets to ensure compliance with the asset tests and to take any available action within 30 days after the close of any quarter as may be required to cure any noncompliance with the asset tests. We cannot ensure that these steps always will be successful. If we fail to cure the noncompliance with the asset tests within this 30-day period, we could fail to qualify as a REIT.

Annual Distribution Requirements

To qualify as a REIT, we generally must distribute dividends (other than capital gain dividends) to our stockholders in an amount at least equal to:

the sum of (i) 90% of our REIT taxable income, computed without regard to the dividends-paid deduction and our net capital gain and (ii) 90% of our net income after tax, if any, from foreclosure property; minus

the excess of the sum of specified items of non-cash income (including original issue discount on our senior loans and mezzanine loans) over 5% of our REIT taxable income, computed without regard to the dividends-paid deduction and our net capital gain.

Distributions generally must be made during the taxable year to which they relate. Distributions may be made in the following year in two circumstances. First, if we declare a dividend in October, November or December of any year

with a record date in one of these months and pay the dividend on or before January 31 of the following year, we will be treated as having paid the dividend on December 31 of the year in which the dividend was declared. Second, distributions may be made in the following year if the dividends are declared before we timely file our tax return for the year and if made before the first regular dividend payment made after such declaration. These distributions are taxable to our stockholders in the year in which paid, even though the

distributions relate to our prior taxable year for purposes of the 90% distribution requirement. To the extent that we do not distribute all of our net capital gain or we distribute at least 90%, but less than 100% of our REIT taxable income, as adjusted, we will be subject to tax on the undistributed amount at regular corporate tax rates.

Generally, in order for distributions to be counted as satisfying the annual distribution requirements for REITs, and to provide us with a REIT-level tax deduction, the distributions must not be preferential dividends. A dividend is not a preferential dividend if the distribution is (i) pro-rata among all outstanding shares of stock within a particular class, and (ii) in accordance with the preferences among different classes of stock as set forth in our organizational documents. However, for taxable years after 2014, to the extent we are a publicly offered REIT, the preferential dividend rule will not apply to us.

If we fail to distribute during a calendar year (or, in the case of distributions with declaration and record dates falling in the last three months of the calendar year, by the end of January following such calendar year) at least the sum of (i) 85% of our ordinary income for such year, (ii) 95% of our capital gain net income for such year and (iii) any undistributed taxable income from prior years, we will be subject to a 4% excise tax on the excess of such required distribution over the sum of (x) the amounts actually distributed (taking into account excess distributions from prior years) and (y) the amounts of income retained on which we have paid corporate income tax.

We may elect to retain rather than distribute all or a portion of our net capital gains and pay the tax on the gains. In that case, we may elect to have our stockholders include their proportionate share of the undistributed net capital gains in income as long-term capital gains and receive a credit for their share of the tax paid by us. For purposes of the 4% excise tax described above, any retained amounts for which we elect this treatment would be treated as having been distributed.

We intend to make timely distributions sufficient to satisfy the distribution requirements. However, due to the nature of the assets in which we will invest, we may be required to recognize taxable income from those assets in advance of our receipt of cash flow on or proceeds from disposition of such assets. For instance, we may be required to accrue interest and discount income on mortgage loans, MBS, and other types of debt securities or interests in debt securities before we receive any payments of interest or principal on such assets. See Income Tests Phantom Income. Moreover, in certain instances we may be required to accrue taxable income that we may not actually recognize as economic income. For example, if we own a residual equity position in a mortgage loan securitization, we may recognize taxable income that we will never actually receive due to losses sustained on the underlying mortgage loans. Although those losses would be deductible for tax purposes, they would likely occur in a year subsequent to the year in which we recognized the taxable income.

Although several types of non-cash income are excluded in determining the annual distribution requirement, we will incur corporate income tax and the 4% nondeductible excise tax with respect to those non-cash income items if we do not distribute those items on a current basis. As a result of the foregoing, we may not have sufficient cash to distribute all of our taxable income and thereby avoid corporate income tax and the excise tax imposed on certain undistributed income. In such a situation, we may need to borrow funds or issue additional common stock or preferred stock.

We may satisfy the 90% distribution test with taxable distributions of our stock or debt securities. The IRS has issued a revenue procedure authorizing publicly offered REITs to treat certain distributions that are paid partly in cash and partly in stock as dividends that would satisfy the REIT annual distribution requirement and qualify for the dividends-paid deduction for U.S. federal income tax purposes. We have no current intention to make a taxable dividend payable in cash and stock.

Under some circumstances, we may be able to rectify a failure to meet the distribution requirement for a year by paying deficiency dividends to stockholders in a later year, which may be included in our deduction for dividends paid for the earlier year. Thus, we may be able to avoid being taxed on amounts distributed as

deficiency dividends. However, we will be required to pay interest based upon the amount of any deduction taken for deficiency dividends.

Failure to Qualify

If we fail to satisfy one or more requirements of REIT qualification, other than the income tests or asset requirements, then we may still retain REIT qualification if the failure is due to reasonable cause and not willful neglect, and we pay a penalty of \$50,000 for each failure.

If we fail to qualify for taxation as a REIT in any taxable year and the relief provisions do not apply, we will be subject to tax, including any applicable alternative minimum tax for taxable years beginning prior to December 31, 2017, on our taxable income at regular corporate rates. This would significantly reduce both our cash available for distribution to our stockholders and our earnings. If we fail to qualify as a REIT, we will not be required to make any distributions to stockholders and any distributions that are made will not be deductible by us. Moreover, all distributions to stockholders would be taxable as dividends to the extent of our current and accumulated earnings and profits, whether or not attributable to capital gains of ours. Unless entitled to relief under specific statutory provisions, we (and any successor entity) also would be disqualified from taxation as a REIT for the four taxable years following the year during which our qualification was lost. We intend to take advantage of any and all relief provisions that are available to us to cure any violation of the requirements applicable to REITs.

Taxation of U.S. Holders of Annaly Common Stock

The rules governing U.S. federal income taxation of U.S. holders are complex. This section is only a summary of such rules. We urge U.S. holders to consult their tax advisors to determine the impact of U.S. federal, state and local income tax laws on ownership of our common stock, including any reporting requirements.

Distributions

Ordinary Dividends. As long as we qualify as a REIT, distributions made to taxable U.S. holders of our common stock out of current or accumulated earnings and profits that are not designated as capital gain dividends generally will be taken into account by them as ordinary income taxable at ordinary income tax rates and will not qualify for the reduced capital gains rates that currently generally apply to distributions by non-REIT C corporations to certain non-corporate U.S. holders. However, for tax years beginning prior to January 1, 2026, non-corporate U.S. holders generally may deduct up to 20% of certain pass-through income, including ordinary REIT dividends that are not capital gain dividends or qualified dividend income, subject to certain limitations. Corporate stockholders will not be eligible for the dividends received deduction with respect to dividends distributed by us.

Distributions in excess of both current and accumulated earnings and profits will not be taxable to a U.S. holder to the extent that the distributions do not exceed the adjusted basis of the holder s stock. Rather, such distributions will reduce the adjusted basis of the stock. In determining the extent to which a distribution constitutes a dividend for tax purposes, our earnings and profits will be allocated first to distributions with respect to our preferred stock and then to our common stock. To the extent that distributions exceed the adjusted basis of a U.S. holder s stock, the distributions will be taxable as capital gains. A U.S. holder s initial basis in a share of our common stock is, in general, equal to the amount paid per share.

Distributions generally will be taxable, if at all, in the year of the distribution. However, if we declare a dividend in October, November or December of any year with a record date in one of these months and pay the dividend on or before January 31 of the following year, we will be treated as having paid the dividend, and the stockholder will be

treated as having received the dividend, on December 31 of the year in which the dividend was declared.

Capital Gain Dividends. We may elect to designate distributions of our net capital gain as capital gain dividends. Capital gain dividends are taxed to U.S. holders of our stock as gain from the sale or exchange of a capital asset held for more than one year. This tax treatment applies regardless of the period during which the U.S. holders have held their stock. If we designate any portion of a dividend as a capital gain dividend, the amount that will be taxable to the stockholder as capital gain will be indicated to U.S. holders on IRS Form 1099-DIV. Corporate U.S. holders, however, may be required to treat up to 20% of capital gain dividends as ordinary income. Capital gain dividends are not eligible for the dividends-received deduction for corporations.

Instead of paying capital gain dividends, we may elect to require U.S. holders to include our undistributed net capital gains in their income. If we make such an election, U.S. holders (i) will include in their income as long-term capital gains their proportionate share of such undistributed capital gains and (ii) will be deemed to have paid their proportionate share of the tax paid by us on such undistributed capital gains and thereby receive a credit or refund for such amount. A U.S. holder of our common stock will increase the basis in its shares of our common stock by the difference between the amount of capital gain included in its income and the amount of tax it is deemed to have paid. Our earnings and profits will be adjusted appropriately.

We will classify portions of our designated capital gain dividend into the following categories:

a 20% gain distribution, which would be taxable to non-corporate U.S. holders of our stock at a rate of up to 20%;

an unrecaptured Section 1250 gain distribution, which would be taxable to non-corporate U.S. holders of our stock at a maximum rate of 25%; or

a 28% gain distribution, which would be taxable to non-corporate U.S. holders of our stock at a rate of up to 28%.

We must determine the maximum amounts that we may designate as 20%, 25% and 28% capital gain dividends by performing the computation required by the Code as if we were an individual whose ordinary income were subject to a marginal tax rate of at least 28%. The IRS currently requires that distributions made to different classes of stock be composed proportionately of dividends of a particular type.

The maximum amount of dividends that we may designate as capital gain and as qualified dividend income (discussed below) with respect to any taxable year (effective for distributions in tax years beginning after December 31, 2015) may not exceed the dividends actually paid by us with respect to such year, including dividends paid by us in the succeeding tax year that relate back to the prior tax year for purposes of determining our dividends-paid deduction.

Qualified Dividend Income. Distributions that we make generally will not be eligible for the 20% reduced rate of tax currently applicable for qualified dividend income. However, provided that certain holding periods and other requirements are met, a non-corporate U.S. holder will be eligible for the 20% reduced rate with respect to (i) distributions attributable to dividends we receive from certain C corporations, such as our domestic taxable REIT subsidiaries, and (ii) distributions attributable to income upon which we have paid corporate income tax.

If we designate any portion of a dividend as a capital gain dividend or as qualified dividend income, the amount that will be taxable to the U.S. holder as capital gain or as qualified dividend income will be indicated to U.S. holders on

IRS Form 1099-DIV.

Passive Activity Loss and Investment Interest Limitation. Distributions and gain from the disposition of our common stock will not be treated as passive activity income, and therefore U.S. holders will not be able to apply any passive activity losses against such income. Dividends paid by us, to the extent they do not constitute a return of capital, will generally be treated as investment income for purposes of the investment income limitation on the deduction of the investment interest.

Other Tax Considerations. U.S. holders of our common stock may not include in their individual income tax returns any of our net operating losses or capital losses. Our operating or capital losses would be carried over by us for potential offset against future income, subject to applicable limitations.

Sales of Our Common Stock. Upon any taxable sale or other disposition of our common stock, a U.S. holder of our common stock will recognize gain or loss for U.S. federal income tax purposes on the disposition of our common stock in an amount equal to the difference between:

the amount of cash and the fair market value of any property received on such disposition; and

the U.S. holder s adjusted basis in such REIT stock for tax purposes. Gain or loss will be capital gain or loss. The applicable tax rate will depend on the holder s holding period in the asset (generally, if an asset has been held for more than one year it will produce long-term capital gain) and the holder s tax bracket.

Medicare Tax. Certain U.S. holders, including individuals and estates and trusts, are subject to an additional 3.8% Medicare tax on all or a portion of their net investment income, which includes net gain from a sale or exchange of our common stock and income from dividends paid on our common stock. The temporary 20% deduction allowed by Section 199A of the Code with respect to ordinary REIT dividends received by non-corporate taxpayers is allowed only for purposes of Chapter 1 of the Code and thus is apparently not allowed as a deduction allocable to such dividends for purposes of determining the amount of net investment income subject to the 3.8% Medicare tax, which is imposed under Chapter 2A of the Code. U.S. holders are urged to consult their tax advisors regarding the Medicare tax.

Taxation of Non-U.S. Holders of Annaly Common Stock

The rules governing U.S. federal income taxation of non-U.S. holders are complex. This section is only a summary of such rules. We urge non-U.S. holders to consult their tax advisors to determine the impact of U.S. federal, state and local income tax laws on ownership of our common stock, including any reporting requirements.

For most non-U.S. investors, investment in a REIT that invests principally in mortgage loans and MBS is not the most tax-efficient way to acquire and manage, through our subsidiaries, such assets. That is because receiving distributions of income derived from such assets in the form of REIT dividends subjects most foreign investors to withholding taxes that direct investment in those asset classes, and the direct receipt of interest and principal payments, with respect to them, would not. The principal exceptions are foreign sovereigns and their agencies and instrumentalities, which may be exempt from withholding taxes on REIT dividends under the Code, and certain foreign pension funds or similar entities able to claim an exemption from withholding taxes on REIT dividends under the terms of a bilateral tax treaty between their country of residence and the U.S.

Distributions

Distributions by us to a non-U.S. holder of our common stock that are neither attributable to gain from sales or exchanges by us of U.S. real property interests nor designated by us as capital gains dividends will be treated as dividends of ordinary income to the extent that they are made out of our current or accumulated earnings and profits. These distributions ordinarily will be subject to U.S. federal income tax on a gross basis at a rate of 30%, or a lower

rate as permitted under an applicable income tax treaty, unless the dividends are treated as effectively connected with the conduct by the non-U.S. holder of a U.S. trade or business. Under some treaties, however, lower rates generally applicable to dividends do not apply to dividends from REITs. Further, reduced treaty rates are not available to the extent the income allocated to the non-U.S. holder is excess inclusion income. Although we do not expect to recognize any excess inclusion income, if we did recognize excess inclusion income that exceeds our undistributed REIT taxable income in a particular year, it would be allocated to our

stockholders. See Taxable Mortgage Pools. Dividends that are effectively connected with a trade or business will be subject to tax on a net basis, that is, after allowance for deductions, at graduated rates, in the same manner as U.S. holders are taxed with respect to these dividends, and are generally not subject to withholding. Applicable certification and disclosure requirements must be satisfied to be exempt from withholding under the effectively connected income exception. Any dividends received by a corporate non-U.S. holder that is engaged in a U.S. trade or business also may be subject to an additional branch profits tax at a 30% rate, or lower applicable treaty rate. We expect to withhold U.S. income tax at the rate of 30% on any dividend distributions, not designated as (or deemed to be) capital gain dividends, made to a non-U.S. holder unless:

a lower treaty rate applies and the non-U.S. holder files an IRS Form W-8BEN-E or other applicable IRS Form W-8 with us evidencing eligibility for that reduced rate is filed with us; or

the non-U.S. holder files an IRS Form W-8ECI with us claiming that the distribution is income effectively connected with the non-U.S. holder s trade or business.

Distributions in excess of our current or accumulated earnings and profits that do not exceed the adjusted basis of the non-U.S. holder in our common stock will reduce the non-U.S. holder is adjusted basis in our common stock and will not be subject to U.S. federal income tax. Distributions in excess of current and accumulated earnings and profits that do exceed the adjusted basis of the non-U.S. holder in our common stock will be treated as gain from the sale of its stock, the tax treatment of which is described below. See Taxation of Non-U.S. Holders of Annaly Common Stock Sales of Our Common Stock. Because we generally cannot determine at the time we make a distribution whether or not the distribution will exceed our current and accumulated earnings and profits, we normally will withhold tax on the entire amount of any distribution at the same rate as we would withhold on a dividend.

We would be required to withhold at least 15% of any distribution to a non-U.S. holder in excess of our current and accumulated earnings and profits if our common stock constitutes a U.S. real property interest with respect to such non-U.S. holder, as described below under Sales of Our Common Stock. This withholding would apply even if a lower treaty rate applies or the non-U.S. holder is not liable for tax on the receipt of that distribution. However, a non-U.S. holder may seek a refund of these amounts from the IRS if the non-U.S. holder s U.S. tax liability with respect to the distribution is less than the amount withheld.

Distributions to a non-U.S. holder that are designated by us at the time of the distribution as capital gain dividends, other than those arising from the disposition of a U.S. real property interest, generally should not be subject to U.S. federal income taxation unless:

the investment in our common stock is effectively connected with the non-U.S. holder s trade or business, in which case the non-U.S. holder will be subject to the same treatment as U.S. holders with respect to any gain, except that a holder that is a foreign corporation also may be subject to the 30% branch profits tax, as discussed above; or

the non-U.S. holder is a nonresident alien individual who is present in the U.S. for 183 days or more during the taxable year and has a tax home in the U.S., in which case the nonresident alien individual will be subject to a 30% tax on the individual s capital gains.

Under FIRPTA, distributions to certain non-U.S. holders that are attributable to gain from sales or exchanges by us of U.S. real property interests, whether or not designated as a capital gain dividend, will cause such non-U.S. holders to be treated as recognizing gain that is income effectively connected with a U.S. trade or business. Such non-U.S. holders will be taxed on this gain at the same rates applicable to U.S. holders, subject to a special alternative minimum tax in the case of nonresident alien individuals. Also, this gain may be subject to a 30% (or lower applicable treaty rate) branch profits tax in the hands of a non-U.S. holder that is a corporation. Unless the non-U.S. holder is a

qualified shareholder or a qualified foreign pension fund (each as defined below), we will be required to withhold and remit to the IRS 21% of any distributions to non-U.S. holders that are designated as capital gain dividends, or, if greater, 21% of a distribution that could have been designated as a

capital gain dividend, whether or not attributable to sales of U.S. real property interests. Distributions can be designated as capital gains to the extent of our net capital gain for the taxable year of the distribution. The amount withheld, which for individual non-U.S. holders may exceed the actual tax liability, is creditable against the non-U.S. holder s U.S. federal income tax liability.

However, the 21% withholding tax will not apply to any capital gain dividend with respect to any class of our stock which is regularly traded on an established securities market located in the U.S. if the non-U.S. stockholder did not own more than 10% of such class of stock at any time during the one-year period ending on the date of such dividend. Instead, any capital gain dividend to such holder will be treated as a distribution of ordinary income subject to the rules discussed above under Distributions. Also, the branch profits tax will not apply to such a distribution.

Sales of Our Common Stock. Gain recognized by a non-U.S. holder upon the sale or exchange of our common stock generally would not be subject to U.S. federal income or withholding tax unless:

the investment in our common stock is effectively connected with the non-U.S. holder s U.S. trade or business, in which case the non-U.S. holder will be subject to the same treatment as U.S. holders with respect to any gain;

the non-U.S. holder is a nonresident alien individual who is present in the U.S. for 183 days or more during the taxable year and has a tax home in the U.S., in which case the nonresident alien individual will be subject to a 30% tax on the individual s net capital gains for the taxable year; or

the non-U.S. holder is not a qualified shareholder or a qualified foreign pension fund and our common stock constitutes a U.S. real property interest within the meaning of FIRPTA, as described below.
Our common stock will not constitute a U.S. real property interest if we either are not a U.S. real property holding corporation or we are a domestically-controlled REIT. Whether we are a U.S. real property holding corporation will depend upon whether the fair market value of U.S. real property interests owned by us equals or exceeds 50% of the fair market value of these interests, any interests in real estate outside of the U.S., and our other trade and business assets. The term U.S. real property interests generally does not include mortgage loans or MBS. Even if we are a U.S. real property holding corporation, the disposition of our common stock will not be subject to FIRPTA if we are a domestically-controlled REIT. Generally, a REIT is domestically controlled if, at all times during a specified testing period, less than 50% of the value of its shares is held directly or indirectly by non-U.S. persons.

Because our common stock will be publicly traded, no assurance can be given that we are or will be a domestically-controlled REIT. Even if we were a U.S. real property holding corporation and were not a domestically-controlled REIT, a sale of common stock by a non-U.S. holder would nevertheless not be subject to taxation under FIRPTA as a sale of a U.S. real property interest if:

our common stock were regularly traded on an established securities market within the meaning of applicable Treasury regulations; and

the non-U.S. holder did not actually, or constructively under specified attribution rules under the Code, own more than 10% of our common stock at any time during the shorter of the five-year period preceding the disposition or the holder s holding period.

If gain on the sale or exchange of our common stock were subject to taxation under FIRPTA, the non-U.S. holder would be subject to regular U.S. income tax with respect to any gain in the same manner as a taxable U.S. holder, subject to any applicable alternative minimum tax and special alternative minimum tax in the case of nonresident alien individuals. In such case, under FIRPTA, the purchaser of common stock may be required to withhold 15% of the purchase price and remit this amount to the IRS.

Qualified Shareholders. Subject to the exception discussed below, any distribution to a qualified shareholder who holds REIT stock directly or indirectly (through one or more partnerships) will not be subject to U.S. federal income taxation under FIRPTA and thus will not be subject to special withholding rules under FIRPTA. While a qualified shareholder will not be subject to FIRPTA withholding on REIT distributions, the portion of REIT distributions attributable to certain investors in a qualified shareholder (i.e., non-U.S. persons who hold interests in the qualified shareholder (other than interests solely as a creditor), and directly or indirectly hold more than 10% of the stock of such REIT (whether or not by reason of the investor s ownership in the qualified shareholder)) may be subject to FIRPTA withholding. REIT distributions received by a qualified shareholder that are exempt from FIRPTA withholding may still be subject to regular U.S. withholding tax.

In addition, a sale of our stock by a qualified shareholder who holds such stock directly or indirectly (through one or more partnerships) generally will not be subject to U.S. federal income taxation under FIRPTA. As with distributions, the portion of amounts realized attributable to certain investors in a qualified shareholder (i.e., non-U.S. persons who hold interests in the qualified shareholder (other than interests solely as a creditor), and directly or indirectly hold more than 10% of the stock of such REIT (whether or not by reason of the investor s ownership in the qualified shareholder)) may be subject to U.S. federal income taxation and FIRPTA withholding on a sale of our stock.

A qualified shareholder is a foreign person that (i) either is eligible for the benefits of a comprehensive income tax treaty which includes an exchange of information program and whose principal class of interests is listed and regularly traded on one or more recognized stock exchanges (as defined in such comprehensive income tax treaty), or is a foreign partnership that is created or organized under foreign law as a limited partnership in a jurisdiction that has an agreement for the exchange of information with respect to taxes with the U.S. and has a class of limited partnership units representing greater than 50% of the value of all the partnership units that is regularly traded on the NYSE or Nasdaq markets, (ii) is a qualified collective investment vehicle (within the meaning of Section 897(k)(3)(B) of the Code), and (iii) maintains records on the identity of each person who, at any time during the foreign person s taxable year, is the direct owner of 5% or more of the class of interests or units (as applicable) described in (i), above.

Qualified Foreign Pension Funds. Any distribution to a qualified foreign pension fund (or an entity all of the interests of which are held by a qualified foreign pension fund) who holds REIT stock directly or indirectly (through one or more partnerships) will not be subject to U.S. federal income taxation under FIRPTA and thus will not be subject to special withholding rules under FIRPTA. REIT distributions received by a qualified foreign pension fund that are exempt from FIRPTA withholding may still be subject to regular U.S. withholding tax. In addition, a sale of our stock by a qualified foreign pension fund that holds such stock directly or indirectly (through one or more partnerships) will not be subject to U.S. federal income taxation under FIRPTA.

A qualified foreign pension fund is any trust, corporation, or other organization or arrangement (i) which is created or organized under the law of a country other than the U.S., (ii) which is established by such country or an employer to provide retirement or pension benefits to participants or beneficiaries that are current or former employees (or persons designated by such employees) of one or more employers in consideration for services rendered (iii) which does not have a single participant or beneficiary with a right to more than 5% of its assets or income, (iv) which is subject to government regulation and with respect to which annual information reporting about its beneficiaries is provided or is otherwise available to the relevant tax authorities in the country in which it is established or operates, and (v) with respect to which, under the laws of the country in which it is established or operates, (a) contributions to such organization or arrangement that would otherwise be subject to tax under such laws are deductible or excluded from the gross income of such entity or arrangement or taxed at a reduced rate, or (b) taxation of any investment income of such organization or arrangement is deferred or such income is taxed at a reduced rate.

Taxation of Tax-Exempt Holders of Annaly Common Stock

Provided that a tax-exempt holder has not held its common stock as debt-financed property within the meaning of the Code, the dividend and interest income from us generally will not be unrelated business taxable income, referred to as UBTI, to a tax-exempt holder. Similarly, income from the sale of our common stock will not constitute UBTI unless the tax-exempt holder has held its common stock as debt-financed property within the meaning of the Code. Although we do not expect to recognize any excess inclusion income, to the extent that we, or a part of us, or a disregarded subsidiary of ours, is a taxable mortgage pool, a portion of the dividends paid to a tax-exempt stockholders that is allocable to excess inclusion income may be subject to tax as UBTI. See Taxable Mortgage Pools.

Notwithstanding the above, however, social clubs, voluntary employee benefit associations, supplemental unemployment benefit trusts and qualified group legal services plans that are exempt from taxation under special provisions of the U.S. federal income tax laws are subject to different UBTI rules, which generally will require them to characterize distributions that they receive from us as UBTI. Moreover, a portion of the dividends paid by a pension-held REIT are treated as UBTI as to any trust which is described in Section 401(a) of the Code, is tax-exempt under Section 501(a) of the Code, and holds more than 10%, by value, of the interests in the REIT. Tax-exempt pension funds that are described in Section 401(a) of the Code are referred to below as pension trusts.

A REIT is a pension-held REIT if it meets the following two tests:

it would not have qualified as a REIT but for Section 856(h)(3) of the Code, which provides that stock owned by pension trusts will be treated, for purposes of determining whether the REIT is closely held, as owned by the beneficiaries of the trust rather than by the trust itself; and

either (i) at least one pension trust holds more than 25% of the value of the interests in the REIT, or (ii) a group of pension trusts each individually holding more than 10% of the value of the REIT s stock, collectively owns more than 50% of the value of the REIT s stock.

The percentage of any REIT dividend from a pension-held REIT that is treated as UBTI is equal to the ratio of the UBTI earned by the REIT, treating the REIT as if it were a pension trust and therefore subject to tax on UBTI, to the total gross income of the REIT. An exception applies where the percentage is less than 5% for any year, in which case none of the dividends would be treated as UBTI. The provisions requiring pension trusts to treat a portion of REIT distributions as UBTI will not apply if the REIT is not a pension-held REIT (for example, if the REIT is able to satisfy the not closely held requirement without relying on the look through exception with respect to pension trusts). Our 9.8% ownership limit may make it less likely that a pension trust would hold more than 25% of the value of our capital stock or that a group of pension trusts each holding more than 10% of the value of our capital stock would hold more than 50% of the value of our capital stock. No assurance can be given, however, that we will not be a pension-held REIT because of ownership waivers or otherwise.

Backup Withholding Tax and Information Reporting

U.S. Holders of Our Common Stock. In general, information-reporting requirements will apply to payments of dividends and interest on and payments of the proceeds of the sale of our common stock held by U.S. holders, unless an exception applies. The payor is required to withhold tax on such payments if (i) the payee fails to furnish a taxpayer identification number, or TIN, to the payor or to establish an exemption from backup withholding, or (ii) the IRS notifies the payor that the TIN furnished by the payee is incorrect. In addition, a payor of the dividends or interest

on our common stock is required to withhold tax if (i) there has been a notified payee under-reporting with respect to interest, dividends or original issue discount described in Section 3406(c) of the Code, or (ii) there has been a failure of the payee to certify under the penalty of perjury that the payee is not subject to backup withholding under the Code. A U.S. holder that does not provide us with a correct TIN may

also be subject to penalties imposed by the IRS. In addition, we may be required to withhold a portion of capital gain distributions to any U.S. holders who fail to certify their U.S. status to us. Some U.S. holders of our common stock, including corporations, may be exempt from backup withholding. Any amounts withheld under the backup withholding rules from a payment to a stockholder will be allowed as a credit against the stockholder s U.S. federal income tax and may entitle the stockholder to a refund, provided that the required information is furnished to the IRS. The payor will be required to furnish annually to the IRS and to holders of our common stock information relating to the amount of dividends paid on our common stock, and that information reporting may also apply to payments of proceeds from the sale of our common stock. Some holders, including corporations, financial institutions and certain tax-exempt organizations, are generally not subject to information reporting.

Non-U.S. Holders of Our Common Stock. Generally, information reporting will apply to payments of interest and dividends on our common stock, and backup withholding described above for a U.S. holder will apply, unless the payee certifies that it is not a U.S. person or otherwise establishes an exemption.

The payment of the proceeds from the disposition of our common stock to or through the U.S. office of a U.S. or foreign broker will be subject to information reporting and backup withholding as described above for U.S. holders unless the non-U.S. holder satisfies the requirements necessary to be an exempt non-U.S. holder or otherwise qualifies for an exemption. The proceeds of a disposition by a non-U.S. holder of our common stock to or through a non-U.S. office of a broker generally will not be subject to information reporting or backup withholding if in accordance with instructions directly transmitted to such office from outside the U.S. by the non-U.S. holder, the broker s non-U.S. office completes the acts necessary to effect the sale outside the U.S. However, if the broker is a U.S. person, a controlled foreign corporation for U.S. tax purposes, a foreign person 50% or more of whose gross income from all sources for specified periods is from activities that are effectively connected with a U.S. trade or business, a foreign partnership if partners who hold more than 50% of the interest in the partnership are U.S. persons, or a foreign partnership that is engaged in the conduct of a trade or business in the U.S., then information reporting generally will apply as though the payment was made through a U.S. office of a U.S. or foreign broker.

Applicable U.S. Treasury regulations provide presumptions regarding the status of a holder of our common stock when payments to such holder cannot be reliably associated with appropriate documentation (i.e., the appropriate IRS Form W-8) provided to the payer. Because the application of these U.S. Treasury regulations varies depending on the stockholder s particular circumstances, you are advised to consult your tax advisor regarding the information reporting requirements applicable to you.

Legislative or Other Actions Affecting REITs

The present U.S. federal income tax treatment of REITs may be modified, possibly with retroactive effect, by legislative, judicial or administrative action at any time, which could affect the U.S. federal income tax treatment of an investment in us. The U.S. federal income tax rules dealing with REITs constantly are under review by persons involved in the legislative process, the IRS and the U.S. Treasury Department, which results in statutory changes as well as frequent revisions to regulations and interpretations. The TCJA significantly changes the U.S. federal income tax laws applicable to businesses and their owners, including REITs and their stockholders. Technical corrections or other amendments to the TCJA or administrative guidance interpreting the TCJA may be forthcoming. Future revisions in federal tax laws and interpretations thereof could affect or cause us to change our investments and commitments and affect the tax considerations of an investment in us.

State and Local Taxes

We and our stockholders may be subject to state or local taxation in various state or local jurisdictions, including those in which we or they transact business or reside. Our state and local tax treatment and that of our stockholders may not conform to the U.S. federal income tax treatment discussed above. Consequently,

prospective stockholders should consult their tax advisors regarding the effect of state and local tax laws on an investment in our common stock.

Tax Shelter Reporting

If a stockholder recognizes a loss as a result of a transaction with respect to our stock of at least (i) for a holder that is an individual, S corporation, trust or a partnership with at least one non-corporate partner, \$2 million or more in a single taxable year or \$4 million or more in a combination of taxable years, or (ii) for a holder that is either a corporation or a partnership with only corporate partners, \$10 million or more in a single taxable year or \$20 million or more in a combination of taxable years, the stockholder must file a disclosure statement with the IRS on IRS Form 8886. Direct stockholders of portfolio securities are in many cases exempt from this reporting requirement, but stockholders of a REIT currently are not excepted. The fact that a loss is reportable under these regulations does not affect the legal determination of whether the taxpayer s treatment of the loss is proper. Stockholders should consult their tax advisors to determine the applicability of these regulations in light of their individual circumstances.

Additional Withholding Requirements

Under Sections 1471 through 1474 of the Code (such sections commonly referred to as FATCA), a 30% U.S. federal withholding tax will apply to dividends that we pay and, beginning January 1, 2019, gross proceeds from the disposition of our common stock, in each case paid to certain foreign entities if such entities do not satisfy disclosure requirements related to U.S. accounts or ownership. Foreign entities must provide documentation evidencing compliance with or an exemption from FATCA, typically provided on IRS Form W-8BEN-E, to avoid this withholding tax. If a payment is both subject to withholding under FATCA and subject to withholding tax. Non-U.S. holders and U.S. holders holding through foreign accounts or intermediaries should consult their tax advisors to determine the applicability of FATCA in light of their individual circumstances.

DESCRIPTION OF ANNALY CAPITAL STOCK

The following discussion is a summary of the terms of the capital stock of Annaly and should be read in conjunction with the section entitled Comparison of Stockholders Rights. The summary set forth below does not purport to be complete and is subject to and qualified in its entirety by reference to relevant provisions of the MGCL, Annaly s charter and Annaly s bylaws. You are urged to read those documents carefully. Copies of Annaly s charter and Annaly s bylaws are incorporated by reference as exhibits to the registration statement on Form S-4, of which this document forms a part, and will be sent to stockholders of MTGE upon request. See Where to Obtain More Information.

General

Annaly s charter provides that the total number of shares of stock of all classes which it has the authority to issue is 2,000,000,000 shares of capital stock, par value \$0.01 per share. Of these shares of capital stock, 1,909,750,000 shares are classified as shares of common stock, 12,000,000 shares are classified as shares of 7.625% Series C Cumulative Redeemable Preferred Stock (the Annaly Series C preferred stock), 18,400,000 shares are classified as shares of 7.50% Series D Cumulative Redeemable Preferred Stock (the Annaly Series E Cumulative Redeemable Preferred Stock (the Annaly Series E Cumulative Redeemable Preferred Stock (the Annaly Series E preferred stock), 28,800,000 shares are classified as shares of 6.95% Series F Fixed-to-Floating Rate Cumulative Redeemable Preferred Stock (the Annaly Series F preferred stock) and 19,550,000 shares are classified as shares of 6.50% Series G Fixed-to-Floating Rate Cumulative Redeemable Preferred Stock (the Annaly Series G preferred stock).

Annaly s board of directors may classify and reclassify any unissued shares of capital stock by setting or changing in any one or more respects the preferences, conversion or other rights, voting powers, restrictions, limitations as to dividends, qualifications or terms or conditions of redemption of such shares of stock.

As of the effective time of the merger, Annaly s board of directors will reclassify 2,200,000 shares of capital stock currently classified as common stock into 2,200,000 shares of the Annaly Series H preferred stock.

As of April 30, 2018, Annaly had 1,159,697,587 shares of common stock outstanding. In addition, as of April 30, 2018, there were 7,000,000 shares of Annaly Series C preferred stock outstanding, 18,400,000 shares of Annaly Series D preferred stock outstanding, 0 shares of Annaly Series E preferred stock outstanding, 28,800,000 shares of Annaly Series F preferred stock outstanding and 17,000,000 shares of Annaly Series G preferred stock outstanding.

All shares of Annaly common stock to be issued in connection with the offer and the merger will be duly authorized, fully paid and nonassessable. The shares of Annaly Series H preferred stock to be issued in connection with the merger will also be duly authorized, fully paid and nonassessable.

Description of Common Stock

Voting

Each of holder of shares of Annaly common stock is entitled to one vote for each share held of record on each matter submitted to a vote of common stockholders.

Annaly s bylaws provide that annual stockholders meetings will be held on the date and at the time and place determined by Annaly s board of directors, and special meetings may be called by Annaly s board of directors, the chairman of the board of directors, the president, the chief executive officer, or generally the secretary of Annaly upon the written request of stockholders entitled to cast not less than a majority of the votes that all stockholders are entitled

to cast at the meeting. Annaly s charter may be amended if the amendment is advised by Annaly s board of directors and approved by the affirmative vote of the holders of a majority of the total number of shares outstanding and entitled to vote thereon.

Dividends; Liquidation; Other Rights

Common stockholders are entitled to receive dividends when authorized by Annaly s board of directors and declared by Annaly out of legally available assets. The right of common stockholders to receive dividends is subordinate to the rights of preferred stockholders or other senior stockholders. If Annaly has a liquidation, dissolution or winding up, its common stockholders will share ratably in all of its assets remaining after the payment of all of its liabilities and the payment of all liquidation and other preference amounts to preferred stockholders and other senior stockholders. Common stockholders have no preemptive or other subscription rights, and there are no conversion rights, or redemption or sinking fund provisions, relating to the shares of common stock.

Classification or Reclassification of Common Stock or Preferred Stock

Annaly s charter authorizes its board of directors to classify or reclassify any unissued shares of common or preferred stock into other classes or series of stock, to establish the number of shares in each class or series and to set the preferences, conversion and other rights, voting powers, restrictions, limitations as to dividends or other distributions, qualifications, and terms and conditions of redemption for each class or series.

Description of Annaly Series H Preferred Stock

This section describes the material terms and provisions of the shares of Annaly Series H preferred stock to be issued in connection with the merger, which terms and provisions are set forth in the form of Annaly s articles supplementary creating the Annaly Series H preferred stock and attached as Annex C to the merger agreement, which is attached as Annex A to this document, which forms a part of the registration statement on Form S-4. This summary may not contain all of the information about the Annaly Series H preferred stock that is important to you. Annaly and MTGE urge you to carefully read the full text of Annaly s articles supplementary, because they will be the legal documents that will govern the Annaly Series H preferred stock.

General

In connection with the merger, the Annaly board of directors will designate 2,200,000 shares of Annaly Series H preferred stock. At the effective time of the merger, each outstanding share of MTGE Series A preferred stock will be automatically converted into the right to receive one newly issued share of Annaly Series H preferred stock. It is a condition to the closing of the offer that the newly issued shares of Annaly Series H preferred stock be approved for listing on the NYSE, subject to official notice of issuance. Annaly expects that the Annaly Series H preferred stock will trade on the NYSE under the ticker symbol NLY-PrH.

Ranking

The Annaly Series H preferred stock will rank, with respect to dividend rights and rights upon the voluntary or involuntary liquidation, dissolution or winding up of our affairs:

senior to all classes or series of Annaly common stock, and to any other class or series of Annaly s capital stock expressly designated as ranking junior to the Annaly Series H preferred stock;

on parity with Annaly s Series C preferred stock, Series D preferred stock, Series F preferred stock and Series G preferred stock, and with all other equity securities issued by the Annaly with terms specifically providing that those equity securities rank on a parity with the Series H preferred stock; and

junior to any other class or series of our capital stock expressly designated as ranking senior to the Annaly Series H preferred stock, none of which exists on the date hereof.

The term capital stock does not include convertible or exchangeable debt securities, none of which is outstanding as of the date hereof, which, prior to conversion or exchange, will rank senior in right of payment to

the Annaly Series H preferred stock. The Annaly Series H preferred stock will also rank junior in right of payment to Annaly s other existing and future debt obligations.

Dividends

Subject to the preferential rights of the holders of any class or series of Annaly s capital stock ranking senior to the Annaly Series H preferred stock with respect to dividend rights, holders of shares of the Annaly Series H preferred stock are entitled to receive, when, as and if authorized by Annaly s board of directors and declared by Annaly out of funds legally available for the payment of dividends, cumulative cash dividends at the rate of 8.125% per annum of the \$25.00 liquidation preference per share of the Annaly Series H preferred stock (equivalent to the fixed annual amount of \$2.03125 per share of the Annaly Series H preferred stock).

Dividends on the Annaly Series H preferred stock will be payable to holders quarterly in arrears on or about the last day of March, June, September and December of each year or, if such day is not a business day, on either the immediately preceding business day or the next succeeding business day (provided no interest, additional dividends or other sums will accrue on the amount so payable for the period from and after such date to the next succeeding business day), in each case with the same force and effect as if made on such date. The term business day means each day, other than a Saturday or a Sunday, which is not a legal holiday or day on which banks in the State of Maryland are authorized or required by law, regulation or executive order to close.

The amount of any dividend payable on the Annaly Series H preferred stock for any dividend period, including any partial dividend period, will be computed on the basis of a 360-day year consisting of twelve 30-day months. A quarterly dividend period is the respective period commencing on, and including, the first day of January, April, July and October of each year. Dividends will be payable to holders of record as they appear in Annaly s stock records at the close of business on the applicable record date, which shall be the first day of the calendar month (whether or not a business day) in which the applicable dividend payment date falls, or on such record date set by the Annaly board of directors not more than 30 days preceding the applicable dividend payment date.

The initial accrual and dividend payment dates for the Annaly Series H preferred stock will depend on the timing of the effective time of the merger.

Dividends on the Annaly Series H preferred stock will accrue whether or not:

Annaly has earnings;

there are funds legally available for the payment of those dividends; or

those dividends are declared.

Except as described in the next two paragraphs, unless full cumulative dividends on the Annaly Series H preferred stock for all past dividend periods that have ended shall have been or contemporaneously are declared and paid in cash or declared and a sum sufficient for the payment thereof is set apart for payment, Annaly will not:

declare and pay or declare and set aside for payment of dividends, and Annaly will not declare and make any distribution of cash or other property on or with respect to any shares of Annaly common stock or shares of any other class or series of preferred stock ranking, as to dividends or upon liquidation, on parity with or junior to the Annaly Series H preferred stock, for any period; or

redeem, purchase or otherwise acquire for any consideration or pay or make available any monies for a sinking fund for the redemption of, any common stock or shares of any other class or series of preferred stock ranking, as to dividends or upon liquidation, on parity with or junior to the Annaly Series H preferred stock.

The foregoing sentence, however, will not prohibit:

the conversion into or exchange for other shares of capital stock of Annaly ranking junior to the Annaly Series H preferred stock as to dividends and upon liquidation;

any transfers pursuant to Annaly s charter to the extent necessary to preserve Annaly s qualification as a REIT as discussed under Restrictions on Ownership and Transfer below;

Unless full cumulative dividends on the Annaly Series H preferred stock have been or contemporaneously are declared and paid or declared and a sum sufficient for the payment thereof is set apart for payment for all past dividend periods, no dividends (other than in shares of common stock or in shares of any class or series of preferred stock that Annaly may issue ranking junior to the Annaly Series H preferred stock as to dividends and upon liquidation) shall be declared or paid or set aside for payment upon shares of common stock or preferred stock that Annaly may issue ranking junior to or on a parity with the Annaly Series H preferred stock as to dividends or upon liquidation. Nor shall any other distribution be declared or made upon shares of common stock or preferred stock that Annaly may issue ranking junior to or on a parity with the Annaly Series H preferred stock as to dividends or upon liquidation. In addition, any shares of our common stock or preferred stock that Annaly may issue ranking junior to or on a parity with the Annaly Series H preferred stock as to dividends or upon liquidation shall not be redeemed, purchased or otherwise acquired for any consideration (or any moneys be paid to or made available for a sinking fund for the redemption of any such shares) by Annaly (except by conversion into or exchange for other capital stock that Annaly may issue ranking junior to the Annaly Series H preferred stock as to dividends and upon liquidation and except for transfers made pursuant to the provisions of the Annaly charter relating to restrictions on ownership and transfer of Annaly capital stock). When Annaly does not pay dividends in full (or set apart a sum sufficient to pay them in full) upon the Annaly Series H preferred stock and the shares of any other class or series of capital stock ranking, as to dividends, on parity with the Annaly Series H preferred stock, Annaly will declare any dividends upon the Annaly Series H preferred stock and each such other class or series of capital stock ranking, as to dividends, on parity with the Annaly Series H preferred stock pro rata, so that the amount of dividends declared per share of Annaly Series H preferred stock and such other class or series of capital stock will in all cases bear to each other the same ratio that accumulated dividends per share on the Annaly Series H preferred stock and such other class or series of capital stock (which will not include any accrual in respect of unpaid dividends on such other class or series of capital stock for prior dividend periods if such other class or series of capital stock does not have a cumulative dividend) bear to each other. No interest, or sum of money in lieu of interest, will be payable in respect of any dividend payment or payments on the Annaly Series H preferred stock which may be in arrears.

Holders of shares of Annaly Series H preferred stock are not entitled to any dividend in excess of full cumulative dividends on the Annaly Series H preferred stock as described above. Any dividend payment made on the Annaly Series H preferred stock will first be credited against the earliest accumulated but unpaid dividends due with respect to the Series H preferred stock.

Annaly does not intend to authorize dividends on the Annaly Series H preferred stock, or pay or set apart for payment dividends on the Annaly Series H preferred stock, if the terms of any of Annaly s agreements, including any agreements relating to its indebtedness, prohibit such authorization, payment or setting apart for payment or provide that such authorization, payment or setting apart for payment or setting apart for payment or default under such an agreement. Likewise, no dividends will be authorized by Annaly s board of directors and declared by Annaly or paid or set apart for payment if such authorization, payment or setting apart for payment is restricted or prohibited by law. Annaly does not believe that these restrictions currently have any adverse impact on its ability to pay dividends on the

Annaly Series H preferred stock.

Liquidation Preference

Upon any voluntary or involuntary liquidation, dissolution or winding up of Annaly, before any distribution or payment shall be made to holders of shares of Annaly common stock or any other class or series of preferred

stock ranking, as to liquidation rights, junior to the Annaly Series H preferred stock, holders of shares of Annaly Series H preferred stock will be entitled to be paid out of Annaly s assets legally available for distribution to its stockholders, a liquidation preference of \$25.00 per share of Annaly Series H preferred stock, plus an amount equal to any accumulated and unpaid dividends to, but not including, the date of payment. If, upon Annaly s voluntary or involuntary liquidation, dissolution or winding up, Annaly s available assets are insufficient to pay the full amount of the liquidating distributions on all outstanding shares of Annaly Series H preferred stock and the corresponding amounts payable on all shares of each other class or series of capital stock ranking, as to liquidation rights, on parity with the Annaly Series H preferred stock in the distribution of assets, then holders of shares of Annaly Series H preferred stock and each such other class or series of capital stock ranking will share ratably in any distribution of assets in proportion to the full liquidating distributions to which they would otherwise be respectively entitled.

Holders of shares of Annaly Series H preferred stock will be entitled to written notice of any distribution in connection with any voluntary or involuntary liquidation, dissolution or winding up of Annaly not less than 30 days and not more than 60 days prior to the distribution payment date. After payment of the full amount of the liquidating distributions to which they are entitled, holders of shares of Annaly Series H preferred stock will have no right or claim to any of Annaly s remaining assets. Annaly s consolidation or merger of Annaly, or a statutory share exchange, with or into any other corporation, trust or entity or of any other entity with or into Annaly, or the sale, lease, transfer or conveyance of all or substantially all of Annaly s property or business, will not be deemed to constitute a liquidation, dissolution or winding up of Annaly.

In determining whether a distribution (other than upon liquidation, dissolution or winding up), whether by dividend, or upon redemption or other acquisition of shares or otherwise, is permitted under Maryland law, amounts that would be needed, if Annaly were to be dissolved at the time of the distribution, to satisfy the preferential rights upon dissolution of holders of shares of any class or series of capital stock whose preferential rights upon dissolution are superior or prior to those receiving the distribution will not be added to Annaly s total liabilities.

Optional Redemption

Except with respect to the special optional redemption described below and in certain limited circumstances relating to Annaly s ability to continue to qualify as a REIT as described in Restrictions on Ownership and Transfer, Annaly may not redeem shares of Annaly Series H preferred stock prior to May 22, 2019. On and after May 22, 2019, Annaly may, at its option, upon not less than 30 and not more than 60 days written notice, redeem the Annaly Series H preferred stock, in whole or in part, at any time or from time to time, for cash at a redemption price of \$25.00 per share, plus all accumulated and unpaid dividends to, but not including, the date fixed for redemption.

If fewer than all of the outstanding shares of the Annaly Series H preferred stock are to be redeemed, Annaly will select the shares of Annaly Series H preferred stock to be redeemed pro rata (as nearly as may be practicable without creating fractional shares), or by any other equitable method that the Annaly board of directors may determine will not violate the 9.8% Annaly Series H preferred stock ownership limit. Unless full cumulative dividends on all shares of Annaly Series H preferred stock have been or contemporaneously are declared and paid or declared and a sum sufficient for the payment thereof set apart for payment for all past dividend periods, no shares of Annaly Series H preferred stock will be redeemed, unless all outstanding shares of Annaly Series H preferred stock are simultaneously redeemed and Annaly will not purchase or otherwise acquire directly or indirectly any shares of Annaly Series H preferred stock (except by exchanging Annaly Series H preferred stock for capital stock of Annaly ranking junior to the Series H preferred stock as to dividends and upon liquidation); provided, however, that whether or not the requirements set forth above have been met, Annaly may purchase or acquire shares of Annaly Series H preferred stock (a) to the extent necessary for Annaly to preserve its REIT status for U.S. federal income tax purposes or (b) pursuant to a purchase or exchange offer made on the same terms to holders of all outstanding shares of Annaly

Series H preferred stock. See Restrictions on Ownership and Transfer below.

Notice of redemption will be mailed by Annaly, postage prepaid, not less than 30 days nor more than 60 days prior to the redemption date, addressed to the respective holders of record of the Annaly Series H preferred stock to be redeemed at their respective addresses as they appear on our stock transfer records of Annaly named in Transfer Agent and Registrar below. No failure to give such notice or any defect therein or in the mailing thereof will affect the validity of the proceedings for the redemption of any shares of Annaly Series H preferred stock except as to the holder to whom notice was defective or not given. Each notice will state:

the redemption date;

the redemption price;

the number of shares of Annaly Series H preferred stock to be redeemed;

the place or places where the certificates, if any, representing shares of Annaly Series H preferred stock are to be surrendered for payment of the redemption price;

that dividends on the shares of Annaly Series H preferred stock to be redeemed will cease to accumulate on such redemption date; and

whether the redemption is being effected in order for Annaly to preserve its REIT status for U.S. federal income tax purposes, is an option redemption made on or after May 22, 2019 or is a redemption in connection with a Change of Control (as defined below).

If fewer than all of the shares of Annaly Series H preferred stock held by any holder are to be redeemed, the notice mailed to such holder will also specify the number of shares of Annaly Series H preferred stock held by such holder to be redeemed.

If a redemption date falls after a dividend record date and prior to the corresponding dividend payment date, each holder of shares of the Annaly Series H preferred stock at the close of business of such dividend record date will be entitled to the dividend payable on such shares on the corresponding dividend payment date notwithstanding the redemption of such shares prior to such dividend payment date. Except as described above, Annaly will make no payment or allowance for unpaid dividends, whether or not in arrears, on shares of Annaly Series H preferred stock to be redeemed.

Any shares of Annaly Series H preferred stock that Annaly acquires in any manner will return to the status of authorized but unissued shares of preferred stock, without designation as to class or series, and may thereafter be reissued as any class or series of preferred stock.

Future debt instruments may prohibit Annaly from redeeming or otherwise repurchasing any shares of its capital stock, including the Annaly Series H preferred stock.

Special Optional Redemption

Table of Contents

Upon the occurrence of a Change of Control (as defined below), Annaly may, at its option, redeem the Annaly Series H preferred stock, in whole or in part within 120 days after the first date on which such Change of Control occurred, by paying \$25.00 per share, plus any accumulated and unpaid dividends to, but not including, the date of redemption. If, prior to the Change of Control Conversion Date (as defined in Conversion Rights below), Annaly has provided notice of redemption with respect to the Annaly Series H preferred stock (whether pursuant to Annaly s optional redemption right or its special optional redemption right), the holders of Annaly Series H preferred stock will not have the conversion right described below under Conversion Rights.

Annaly will mail to each record holder of the Annaly Series H preferred stock a notice of redemption not less than 30 days nor more than 60 days before the redemption date. Annaly will send the notice to the address for each record holder of Annaly Series H preferred stock shown on its stock transfer records. No failure to give

such notice or any defect therein or in the mailing thereof will affect the validity of the proceedings for the redemption of any shares of Annaly Series H preferred stock except as to the holder to whom notice was defective or not given. Each notice will state:

the redemption date;

the redemption price;

the number of shares of Annaly Series H preferred stock to be redeemed;

the place or places where the certificates, if any, representing shares of Annaly Series H preferred stock are to be surrendered for payment of the redemption price;

that dividends on the shares of Annaly Series H preferred stock to be redeemed will cease to accumulate on such redemption date;

whether the redemption is being effected in order for Annaly to preserve its REIT status for U.S. federal income tax purposes, is an option redemption made on or after May 22, 2019 or is a redemption in connection with a Change of Control;

that the Annaly Series H preferred stock is being redeemed pursuant to Annaly s special optional redemption right in connection with the occurrence of a Change of Control and a brief description of the transaction or transactions constituting such Change of Control; and

that the holders of the Annaly Series H preferred stock to which the notice relates will not be able to tender such Annaly Series H preferred stock for conversion in connection with the Change of Control and each share of Annaly Series H preferred stock tendered for conversion that is called, prior to the Change of Control Conversion Date, for redemption will be redeemed on the related date of redemption instead of converted on the Change of Control Conversion Date.

If fewer than all of the shares of Annaly Series H preferred stock held by any holder are to be redeemed, the notice mailed to such holder will also specify the number of shares of Annaly Series H preferred stock held by such holder to be redeemed.

If a redemption date falls after a dividend record date and prior to the corresponding dividend payment date, each holder of shares of the Annaly Series H preferred stock at the close of business of such dividend record date will be entitled to the dividend payable on such shares on the corresponding dividend payment date notwithstanding the redemption of such shares prior to such dividend payment date. Except as described above, Annaly will make no payment or allowance for unpaid dividends, whether or not in arrears, on shares of Annaly Series H preferred stock to be redeemed.

Any shares of Annaly Series H preferred stock that Annaly acquires in any manner will return to the status of authorized but unissued shares of preferred stock, without designation as to class or series, and may thereafter be reissued as any class or series of preferred stock.

A Change of Control is when, after the original issuance of the Annaly Series H preferred stock, the following have occurred and are continuing:

the acquisition by any person, including any syndicate or group deemed to be a person under Section 13(d)(3) of the Exchange Act, of beneficial ownership, directly or indirectly, through a purchase, merger or other acquisition transaction or series of purchases, mergers or other acquisition transactions of capital stock of Annaly entitling that person to exercise more than 50% of the total voting power of all stock of Annaly entitled to vote generally in the election of directors (except that such person will be deemed to have beneficial ownership of all securities that such person has the right to acquire, whether such right is currently exercisable or is exercisable only upon the occurrence of a subsequent condition); and

following the closing of any transaction referred to in the bullet point above, neither Annaly nor the acquiring or surviving entity has a class of common securities (or ADRs representing such securities) listed on the NYSE, the NYSE American LLC or Nasdaq or listed or quoted on an exchange or quotation system that is a successor to the NYSE, the NYSE American LLC or Nasdaq.

Conversion Rights

Upon the occurrence of a Change of Control, each holder of Annaly Series H preferred stock will have the right, unless, prior to the Change of Control Conversion Date, Annaly has provided or provides notice of its election to redeem some or all of the Annaly Series H preferred stock as described above under Optional Redemption or Special Optional Redemption, to convert some or all of the Annaly Series H preferred stock held by such holder (the Change of Control Conversion Right) on the Change of Control Conversion Date into a number of shares of Annaly common stock per share of Annaly Series H preferred stock to be converted (the Common Stock Conversion Consideration), which is equal to the lesser of:

the quotient obtained by dividing (i) the sum of the \$25.00 liquidation preference per share of the Annaly Series H preferred stock to be converted plus the amount of any accumulated and unpaid dividends to, but not including, the Change of Control Conversion Date (unless the Change of Control Conversion Date is after a record date for an Annaly Series H preferred stock dividend payment and prior to the corresponding Annaly Series H preferred stock dividend payment date, in which case no additional amount for such accumulated and unpaid dividends will be included in this sum) by (ii) the Common Stock Price (as defined below); and

a number to be determined as of the effective time of the merger (the Share Cap), equal to (A) 2.51004 multiplied by (B) a fraction in which (i) the numerator is equal to the sum of (x) the cash portion of the mixed consideration and (y) the product of (1) the stock portion of the mixed consideration and (2) the 10-day volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding the acceptance time, and (ii) the denominator is the 10-day volume weighted average closing sale price per share of Annaly common stock as reported on the NYSE for the 10 consecutive trading days ending on the trading day immediately preceding to the acceptance time.

The Share Cap is subject to pro rata adjustments for any stock splits (including those effected pursuant to a distribution of Annaly common stock to existing holders of Annaly common stock), subdivisions or combinations (in each case, a Stock Split) with respect to Annaly common stock as follows: the adjusted Share Cap as the result of a Stock Split will be the number of shares of Annaly common stock that is equivalent to the product obtained by multiplying (i) the Share Cap in effect immediately prior to such Stock Split by (ii) a fraction, the numerator of which is the number of shares of Annaly common stock outstanding immediately after giving effect to such Stock Split and the denominator of which is the number of shares of Annaly common stock outstanding immediately prior to such Stock Split.

For the avoidance of doubt, subject to the immediately succeeding sentence, the aggregate number of shares of Annaly common stock (or equivalent Alternative Conversion Consideration (as defined below), as applicable) issuable or deliverable, as applicable, in connection with the exercise of the Change of Control Conversion Right will not exceed the product of the Share Cap times the aggregate number of shares of the Annaly Series H preferred stock issued and outstanding at the Change of Control Conversion Date (or equivalent Alternative Conversion Consideration, as applicable) (the Exchange Cap). The Exchange Cap is subject to pro rata adjustments for any Stock Splits on the same

basis as the corresponding adjustment to the Share Cap.

In the case of a Change of Control pursuant to which Annaly common stock will be converted into cash, securities or other property or assets (including any combination thereof) (the Alternative Form Consideration), a holder of Annaly Series H preferred stock will receive upon conversion of such Annaly Series H preferred stock the kind and amount of Alternative Form Consideration which such holder would have owned or been

entitled to receive upon the Change of Control had such holder held a number of shares of Annaly common stock equal to the Common Stock Conversion Consideration immediately prior to the effective time of the Change of Control (the Alternative Conversion Consideration). The Common Stock Conversion Consideration or the Alternative Conversion Consideration, as may be applicable to a Change of Control, is referred to in this prospectus supplement as the Conversion Consideration.

If the holders of Annaly common stock have the opportunity to elect the form of consideration to be received in the Change of Control, the Conversion Consideration will be deemed to be the kind and amount of consideration actually received by holders of a majority of Annaly common stock that made or voted for such an election (if electing between two types of consideration) or holders of a plurality of the outstanding shares of Annaly common stock that voted for such an election (if electing between more than two types of consideration), as the case may be, and will be subject to any limitations to which all holders of Annaly common stock are subject, including, without limitation, pro rata reductions applicable to any portion of the consideration payable in the Change of Control.

Annaly will not issue fractional shares of common stock upon the conversion of the Annaly Series H preferred stock in connection with a Change of Control. Instead, Annaly will make a cash payment equal to the value of such fractional shares based on the Common Stock Price used in determining the Common Stock Conversion Consideration for such Change of Control. Within 15 days following the occurrence of a Change of Control, Annaly will mail to the record holders of Annaly Series H preferred stock a notice of occurrence of the Change of Control that describes the resulting Change of Control Conversion Right. Annaly will send the notice to the address shown on its stock transfer records, and the notice will state the following:

the events constituting the Change of Control;

the date of the Change of Control;

the last date on which the holders of Annaly Series H preferred stock may exercise their Change of Control Conversion Right;

the method and period for calculating the Common Stock Price;

the Change of Control Conversion Date;

that if, prior to the Change of Control Conversion Date, Annaly has provided or provides notice of its election to redeem all or any portion of the Annaly Series H preferred stock, holders of Annaly Series H preferred stock will not be able to convert the Annaly Series H preferred stock designated for redemption and such shares will be redeemed on the related redemption date, even if such shares have already been tendered for conversion pursuant to the Change of Control Conversion Right;

if applicable, the type and amount of Alternative Conversion Consideration entitled to be received per share of Annaly Series H preferred stock;

the name and address of the paying agent, transfer agent and the conversion agent;

the procedures that the holders of Annaly Series H preferred stock must follow to exercise the Change of Control Conversion Right; and

the last date on which holders of Annaly Series H preferred stock may withdraw shares surrendered for conversion and the procedures that such holders must follow to effect such a withdrawal.

Annaly will issue a press release for publication on the Dow Jones & Company, Inc., Business Wire, PR Newswire or Bloomberg Business News (or, if these organizations are not in existence at the time of issuance of the press release, such other news or press organization as is reasonably calculated to broadly disseminate the relevant information to the public), and post a notice on its website, in any event prior to the opening of business on the first business day following any date on which Annaly provides the notice described above to the holders of Annaly Series H preferred stock.

To exercise the Change of Control Conversion Right, the holders of Annaly Series H preferred stock will be required to deliver, on or before the close of business on the Change of Control Conversion Date, the certificates (if any) representing the shares of Annaly Series H preferred stock to be converted, duly endorsed for transfer (or, in the case of any shares of Annaly Series H preferred stock held in book-entry form through a depositary, to deliver, on or before the close of business on the Change of Control Conversion Date, the shares of Annaly Series H preferred stock to be converted through the facilities of such depositary), together with a written conversion notice completed, to Annaly s transfer agent. The conversion notice must state:

the relevant Change of Control Conversion Date;

the number of shares of Annaly Series H preferred stock to be converted; and

that the Annaly Series H preferred stock is to be converted pursuant to the applicable provisions of the Annaly charter.

The Change of Control Conversion Date is the date the Annaly Series H preferred stock is to be converted, which will be a business day that is no fewer than 20 days nor more than 35 days after the date on which Annaly mails the notice described above to the holders of Annaly Series H preferred stock.

The Common Stock Price will be (i) if the consideration to be received in the Change of Control by the holders of Annaly common stock is solely cash, the amount of cash consideration per share of Annaly common stock or (ii) if the consideration to be received in the Change of Control by holders of Annaly common stock is other than solely cash (x) the average of the closing sale prices per share of Annaly common stock (or, if no closing sale price is reported, the average of the closing bid and ask prices per share or, if more than one in either case, the average of the average closing ask prices per share) for the 10 consecutive trading days immediately preceding, but not including, the effective date of the Change of Control as reported on the principal U.S. securities exchange on which Annaly common stock is then traded, or (y) the average of the last quoted bid prices for Annaly common stock in the over-the-counter market as reported by OTC Link LLC or a similar organization for the 10 consecutive trading days immediately preceding, but not including, the date of the Change of the Change of Control, if Annaly common stock is not then listed for trading on a U.S. securities exchange.

Holders of Annaly Series H preferred stock may withdraw any notice of exercise of a Change of Control Conversion Right (in whole or in part) by a written notice of withdrawal delivered to Annaly s transfer agent prior to the close of business on the business day prior to the Change of Control Conversion Date. The notice of withdrawal must state:

the number of withdrawn shares of Annaly Series H preferred stock;

if certificated Annaly Series H preferred stock has been issued, the certificate numbers of the withdrawn shares of Annaly Series H preferred stock; and

the number of shares of Annaly Series H preferred stock, if any, which remain subject to the conversion notice.

Notwithstanding the foregoing, if the Annaly Series H preferred stock is held in book-entry from through The Depository Trust Company or a similar depositary, the conversion notice and/or the notice of withdrawal, as applicable, must comply with applicable procedures of The Depository Trust Company or such other depositary.

Annaly Series H preferred stock as to which the Change of Control Conversion Right has been properly exercised and for which the conversion notice has not been properly withdrawn will be converted into the applicable Conversion Consideration in accordance with the Change of Control Conversion Right on the Change of Control Conversion Date, unless, prior to the Change of Control Conversion Date, Annaly has provided or provides notice of its election to redeem such shares of Annaly Series H preferred stock, whether pursuant to its optional redemption right or its special optional redemption right, in which case only the shares of Series H preferred stock properly surrendered for conversion and not properly withdrawn that are not called for

redemption will be converted as aforesaid. If Annaly elects to redeem shares of Annaly Series H preferred stock that would otherwise be converted into the applicable Conversion Consideration on a Change of Control Conversion Date, such Annaly Series H preferred stock will not be so converted and the holders of such shares will be entitled to receive on the applicable redemption date \$25.00 per share, plus any accumulated and unpaid dividends to, but not including, the date of redemption, in accordance with our optional redemption right or special optional redemption right. See Optional Redemption and Special Optional Redemption above.

Annaly will deliver the applicable Conversion Consideration upon conversion no later than the third business day following the Change of Control Conversion Date.

In connection with the exercise of any Change of Control Conversion Right, Annaly will comply with all federal and state securities laws and stock exchange rules in connection with any conversion of the Annaly Series H preferred stock into shares of its common stock or other property. Notwithstanding any other provision of the Annaly Series H preferred stock, no holder of Annaly Series H preferred stock will be entitled to convert such Annaly Series H preferred stock into shares of its common stock to the extent that receipt of such common stock would cause such holder (or any other person) to exceed the stock ownership limits contained in Annaly s charter, including the articles supplementary setting forth the terms of the Annaly Series H preferred stock, unless Annaly provides an exemption from the applicable limits for such holder. See Restrictions on Ownership and Transfer below.

The Change of Control conversion feature may make it more difficult for a party to take over Annaly or discourage a party from taking over Annaly. Except as provided above in connection with a Change of Control, the Annaly Series H preferred stock is not convertible into or exchangeable for any other securities or property.

No Maturity, Sinking Fund or Mandatory Redemption

The Annaly Series H preferred stock has no maturity date and Annaly is not required to redeem the Annaly Series H preferred stock at any time. Accordingly, the Annaly Series H preferred stock will remain outstanding indefinitely, unless Annaly decides, at its option, to exercise its redemption right or otherwise repurchase the Annaly Series H preferred stock or, under circumstances where the holders of the Annaly Series H preferred stock have a conversion right, such holders convert the Annaly Series H preferred stock into Annaly common stock. The Annaly Series H preferred stock is not subject to any sinking fund.

Limited Voting Rights

Holders of shares of the Annaly Series H preferred stock will generally have no voting rights, except as set forth below or as otherwise required by law.

If dividends on the Annaly Series H preferred stock are in arrears for six or more quarterly periods, whether or not consecutive (a preferred dividend default), the number of directors constituting the board of directors will, subject to the maximum number of directors authorized under the bylaws then in effect, be automatically increased by two (if not already increased by two by reason of the election of directors by the holders of any other class or series of preferred stock that Annaly may issue and upon which like voting rights have been conferred and are exercisable and with which the Annaly Series H preferred stock is entitled to vote as a class with respect to the election of those two directors) and the holders of shares of the Annaly Series H preferred stock (voting together as a single class with the holders of all other classes or series of preferred stock upon which like voting rights have been conferred and are exercisable) will be entitled to vote for the election of a total of two additional directors to serve on Annaly s board of directors (the preferred stock directors), until all dividends accumulated for past dividend periods and the then current dividend period (if after a dividend record date and prior to the corresponding dividend payment date for the then

current dividend period) with respect to the Annaly Series H preferred stock and any other class or series of preferred stock upon which like voting rights have been conferred and are exercisable have been fully paid or declared and a sum sufficient for payment

thereof is set apart for such payment. In that case, the right of holders of the Annaly Series H preferred stock to elect any directors will cease and, unless there are other classes or series of preferred stock of Annaly upon which like voting rights have been conferred and are exercisable, any directors elected by holders of the Annaly Series H preferred stock shall immediately resign and the number of directors constituting the board of directors shall be reduced accordingly. Each preferred stock director will serve until his successor is duly elected and qualified or until the director s right to hold the office terminates, whichever occurs earlier, subject to such director s earlier death, disqualification, resignation or removal. The election will take place at:

either a special meeting called upon the written request of holders of record of at least 25% of the outstanding shares of Annaly Series H preferred stock together with any other class or series of preferred stock upon which like voting rights have been conferred and are exercisable, if this request is received more than 90 days before the date fixed for Annaly s next annual or special meeting of stockholders or, if Annaly receives the request for a special meeting within 90 days of the date fixed for its next annual or special meeting of stockholders, at its next annual or special meeting of stockholders; and

each subsequent annual meeting of stockholders until all past dividends and dividends for the then current dividend period (if after a dividend record date and prior to the corresponding dividend payment date for the then current dividend period) on the Annaly Series H preferred stock and on any other class or series of preferred stock upon which like voting rights have been conferred and are exercisable have been paid in full or declared and a sum sufficient for the payment thereof set aside for payment.

If and when all accumulated dividends on the Annaly Series H preferred stock and all other classes or series of preferred stock upon which like voting rights have been conferred and are exercisable for the past dividend periods shall have been paid in full or declared and a sum sufficient for such payment in full is set apart for payment, holders of shares of Annaly Series H preferred stock and holders of all other classes or series of preferred stock upon which like voting rights have been conferred and are exercisable shall be divested of the voting rights set forth above (subject to re-vesting in the event of each and every preferred dividend default) and each preferred stock directors will immediately resign and the number of directors constituting the board of directors will be reduced accordingly.

Any preferred stock director elected by holders of shares of Annaly Series H preferred stock and other holders of preferred stock upon which like voting rights have been conferred and are exercisable may be removed at any time with or without cause by the vote of, and may not be removed otherwise than by the vote of, the holders of record of a majority of the outstanding shares of Annaly Series H preferred stock and other parity preferred stock entitled to vote thereon when they have the voting rights described above (voting as a single class). So long as a preferred dividend default continues, any vacancy in the office of a preferred stock director may be filled by the preferred stock director remaining in office, or by a vote of the outstanding shares of Annaly Series H preferred stock when they have the voting rights described above (voting as a single class or series of preferred stock upon which like voting rights have been conferred and are exercisable).

In addition, so long as any shares of Annaly Series H preferred stock remain outstanding, Annaly will not, without the affirmative vote or consent of the holders of at least two-thirds of the outstanding shares of the Annaly Series H preferred stock together with the holders of all other shares of any class or series of preferred stock ranking on parity with the Annaly Series H preferred stock with respect to the payment of dividends and the distribution of assets upon our liquidation, dissolution or winding up (voting as a single class):

authorize or create, or increase the number of authorized or issued amount of, any class or series of stock ranking senior to such Annaly Series H preferred stock with respect to payment of dividends, or the distribution of assets upon Annaly s liquidation, dissolution or winding up, or reclassify any of Annaly s authorized shares of capital stock into any such shares, or create, authorize or issue any obligation or security convertible into or evidencing the right to purchase any such shares; or

amend, alter or repeal the provisions of Annaly s charter, whether by merger, consolidation or otherwise, so as to materially and adversely affect any right, preference, privilege or voting power of the Annaly Series H preferred stock,

except that with respect to the occurrence of any of the events described in the second bullet point immediately above, so long as the Annaly Series H preferred stock remains outstanding with the terms of the Annaly Series H preferred stock materially unchanged (or a substantially similar security to the Series H preferred stock is issued), taking into account that Annaly may not be the surviving entity, then the occurrence of such event will not be deemed to materially and adversely affect the rights, preferences, privileges or voting power of the Annaly Series H preferred stock, and in such case such holders shall not have any voting rights with respect to the events described in the second bullet point immediately above.

In addition, holders of shares of Annaly Series H preferred stock will not have any voting rights with respect to any increase in the amount of the authorized preferred stock, including the Series H preferred stock, or the creation or issuance of any additional shares of Series H preferred stock or other classes or series of preferred stock that Annaly may issue, or any increase in the amount of authorized shares of such class or series, in each case ranking on a parity with or junior to the Series H preferred stock that Annaly may issue with respect to payment of dividends or the distribution of assets upon liquidation, dissolution or winding up, shall not be deemed to materially and adversely affect the rights, preferences, privileges or voting powers of the Series H preferred stock.

In addition, the voting provisions above will not apply if, at or prior to the time when the act with respect to which the vote would otherwise be required would occur, Annaly has redeemed or called for redemption upon proper procedures all outstanding shares of Annaly Series H preferred stock.

In any matter in which Annaly Series H preferred stock may vote (as expressly provided in the articles supplementary setting forth the terms of the Annaly Series H preferred stock), each share of Annaly Series H preferred stock will be entitled to one vote, except that when shares of any other class or series of preferred stock have the right to vote with the Series H preferred stock as a single class on any matter, the Series H preferred stock and the shares of such other class or series will have one vote for each \$25.00 of liquidation preference (excluding accumulated dividends).

Information Rights

During any period in which Annaly is not subject to Section 13 or 15(d) of the Exchange Act and any shares of Annaly Series H preferred stock are outstanding, Annaly will use its best efforts to (i) transmit by mail (or other permissible means under the Exchange Act) to all holders of Annaly Series H preferred stock, as their names and addresses appear in Annaly s record books and without cost to such holders, copies of the Annual Reports on Form 10-K and Quarterly Reports on Form 10-Q that Annaly would have been required to file with the SEC pursuant to Section 13 or 15(d) of the Exchange Act if it were subject thereto (other than any exhibits that would have been required) and (ii) promptly, upon request, supply copies of such reports to any holder or prospective holder of Annaly Series H preferred stock. Annaly will use its best efforts to mail (or otherwise provide) the information to the holders of Annaly Series H preferred stock. Annaly will use its best efforts to mail (or otherwise provide) the information to the holders of Annaly Series H preferred stock within 15 days after the respective dates by which a periodic report on Form 10-K or Form 10-Q, as the case may be, in respect of such information would have been required to be filed with the SEC if Annaly were subject to Section 13 or 15(d) of the Exchange Act, in each case, based on the dates on which Annaly would be required to file such periodic reports if we were a non-accelerated filer within the meaning of the Exchange Act.

Restrictions on Ownership and Transfer

In order for Annaly to qualify as a REIT under the Code, its shares of stock must be beneficially owned by 100 or more persons during at least 335 days of a taxable year of 12 months or during a proportionate part of a

shorter taxable year. Also, no more than 50% of the value of its outstanding shares of capital stock may be owned, directly or indirectly, by five or fewer individuals (as defined by the Code to include certain entities) during the last half of any taxable year.

To assist Annaly in qualifying as a REIT, among other purposes, its charter prohibits anyone from acquiring or holding, directly or constructively, ownership of a number or value of shares of any class of its capital stock in excess of 9.8% of the outstanding shares. The Annaly Series H preferred stock articles supplementary will provide that generally no person may own, or be deemed to own by virtue of the attribution provisions of the Code, either more than 9.8% in value or in number of shares, whichever is more restrictive, of the outstanding Annaly Series H preferred stock.

The consequences of attempting to own or transfer shares of Annaly common stock or capital stock in violation of the ownership restrictions are described below under Restrictions on Ownership and Transfer. Those consequences also apply to any person who attempts to own, or would be deemed to own by virtue of the attribution provisions of the Code, more than 9.8% in value or in number of shares, whichever is more restrictive, of the outstanding Annaly Series H preferred stock. The beneficial ownership and/or constructive ownership rules under the Code are complex and may cause shares of stock owned actually or constructively by a group of related individuals and/or entities to be owned constructively by one individual or entity. See Restrictions on Ownership and Transfer.

Transfer Agent and Registrar

The transfer agent and registrar for the Annaly Series H preferred stock will be Computershare.

Certain Provisions of Maryland Law and Annaly s Charter and Bylaws

Classification of Board of Directors, Vacancies and Removal of Directors

Annaly has a classified board of directors that is divided into three classes, with terms of three years each. The number of directors in each class and the expiration of each class term are as follows:

Class I	3 Directors	Expires 2021
Class II	4 Directors	Expires 2019
Class III	4 Directors	Expires 2020

At each annual meeting of Annaly s stockholders, successors of the class of directors whose term expires at that meeting will be elected for a three year term and until their successors are duly elected and qualify and the directors in the other two classes will continue in office. A classified board of directors may delay, defer or prevent a change in control or other transaction that might involve a premium over the then-prevailing market price for Annaly common stock or other attributes that Annaly stockholders may consider desirable. In addition, a classified board of directors could prevent stockholders who do not agree with the policies of the board of directors from replacing a majority of the board of directors for two years, except in the event of removal for cause.

Any vacancy on the board of directors for any cause other than an increase in the number of directors may be filled by a majority of the remaining directors, even if such majority is less than a quorum. Any vacancy in the number of directors created by an increase in the number of directors may be filled by a majority of the entire board of directors. Any individual so elected as a director shall serve until the next annual meeting of stockholders and until his or her successor is duly elected and qualifies. A director may be removed at any time only for cause upon the affirmative

vote of at least a majority of the votes entitled to be cast in the election of directors. These provisions preclude stockholders from removing incumbent directors, except for cause and upon a substantial affirmative vote, and filling the vacancies created by such removal with their own nominees.

Indemnification

Annaly s charter and bylaws obligate Annaly to indemnify its directors and officers, including the advancement of expenses for them to the full extent permitted by Maryland law. The MGCL permits a corporation to indemnify its present and former directors and officers, among others, against judgments, penalties, fines, settlements and reasonable expenses actually incurred by them in connection with any proceeding to which they may be made or threatened to be made a party by reason of their service in those or other capacities, unless it is established that (1) the act or omission of the director or officer was material to the matter giving rise to the proceeding and (a) was committed in bad faith, or (b) was the result of active and deliberate dishonesty; (2) the director or officer actually received an improper personal benefit in money, property or services; or (3) in the case of any criminal proceeding, the director or officer had reasonable cause to believe that the act or omission was unlawful.

Limitation of Liability

The MGCL permits the charter of a Maryland corporation to include a provision limiting the liability of its directors and officers to the corporation and its stockholders for money damages, except to the extent that (1) it is proved that the person actually received an improper benefit or profit in money, property or services; or (2) a judgment or other final adjudication adverse to the person is entered in a proceeding based on a finding that the person s action, or failure to act, was the result of active and deliberate dishonesty and was material to the cause of action adjudicated in the proceeding. Annaly s charter provides for elimination of the liability of its directors and officers to Annaly or its stockholders for money damages to the maximum extent permitted by Maryland law from time to time.

Exclusive Forum

Annaly s bylaws provide that unless Annaly consents in writing to the selection of an alternative forum, the Circuit Court for Baltimore City, Maryland, or if that court does not have jurisdiction, the United States District Court for the District of Maryland, Baltimore Division, will be the sole and exclusive forum for (1) any derivative action or proceeding brought on behalf of Annaly, (2) any action asserting a claim of breach of any duty owed by any director or officer or other employee of Annaly to Annaly or its stockholders, (3) any action asserting a claim against Annaly or any director or officer or other employee of Annaly arising pursuant to any provision of the MGCL or the Annaly charter or Annaly bylaws, or (4) any other action asserting a claim against Annaly or any director or officer or other employee by the internal affairs doctrine.

Maryland Business Combination Act

The MGCL establishes special requirements for business combinations between a Maryland corporation and an interested stockholder unless exemptions are applicable. An interested stockholder is any person who beneficially owns, directly or indirectly, 10% or more of the voting power of Annaly s then-outstanding voting stock after the date on which Annaly had 100 or more beneficial owners of its stock; or is an affiliate or associate of Annaly and was the beneficial owner, directly or indirectly, of 10% or more of the voting power of Annaly s then-outstanding stock at any time within the two-year period immediately prior to the date in question and after the date on which Annaly had 100 or more beneficial owners of its stock. Among other things, the law prohibits for a period of five years a merger and other similar transactions between Annaly and an interested stockholder unless the board of directors approved the transaction prior to the party becoming an interested stockholder. The five-year period runs from the most recent date on which the interested stockholder became an interested stockholder. The law also requires a supermajority stockholder vote for such transactions after the end of the five-year period. This means that the transaction must be approved by at least:

Edgar Filing: ANNALY CAPITAL MANAGEMENT INC - Form 424B3

80% of the votes entitled to be cast by holders of outstanding voting shares; and

two-thirds of the votes entitled to be cast by holders of outstanding voting shares other than shares held by the interested stockholder or an affiliate or associate of the interested stockholder with whom (or with whose affiliate) the business combination is to be effected.

These super-majority vote requirements do not apply if the corporation s common stockholders receive a minimum price, as defined under Maryland law, for their shares in the form of cash or other consideration in the same form as previously paid by the interested stockholder for its shares.

As permitted by the MGCL, Annaly has elected not to be governed by the Maryland business combination statute. Annaly made this election by opting out of this statute in its charter. If, however, Annaly amends its charter to opt back in to the statute, the business combination statute could have the effect of discouraging offers to acquire us and of increasing the difficulty of consummating any such offers, even if the acquisition of Annaly would be in its stockholders best interests.

Maryland Control Share Acquisition Act

Maryland law provides that holders of control shares of a Maryland corporation acquired in a control share acquisition have no voting rights except to the extent approved by a vote of the other stockholders. Two-thirds of the votes entitled to be cast on the matter must vote in favor of granting the control shares voting rights. Shares owned by the acquirer, by officers or by employees who are directors of the corporation are excluded from shares entitled to vote on the matter. Control shares are voting shares of stock that, taken together with all other shares of stock the acquirer previously acquired or in respect of which the acquirer is able to exercise or direct the exercise of voting power (except solely by virtue of a revocable proxy), would entitle the acquirer to exercise voting power in electing directors within one of the following ranges of voting power:

one-tenth or more but less than one-third of all voting power;

one-third or more but less than a majority of all voting power; or

a majority or more of all voting power.

Control shares do not include shares of stock the acquiring person is entitled to vote as a result of having previously obtained stockholder approval. A control share acquisition means the acquisition of control shares, subject to certain exceptions.

If a person who has made (or proposes to make) a control share acquisition satisfies certain conditions (including agreeing to pay expenses), he may compel the Annaly board of directors to call a special meeting of stockholders to consider the voting rights of the shares. If such a person makes no request for a meeting, Annaly has the option to present the question at any stockholders meeting.

If voting rights are not approved at a meeting of stockholders or the acquiring person does not deliver an acquiring person statement as required by the statute, then, subject to certain conditions and limitations, Annaly may redeem any or all of the control shares (except those for which voting rights have previously been approved) for fair value. Annaly will determine the fair value of the shares, without regard to the absence of voting rights, as of the date of either:

the last control share acquisition; or

Edgar Filing: ANNALY CAPITAL MANAGEMENT INC - Form 424B3

the meeting of stockholders where stockholders considered and did not approve voting rights of the control shares.

If voting rights for control shares are approved at a stockholders meeting and the acquirer becomes entitled to vote a majority of the shares of stock entitled to vote, all other stockholders may obtain rights as objecting stockholders and, thereunder, exercise appraisal rights. This means that you would be able to force Annaly to redeem your stock for fair value. Under Maryland law, the fair value may not be less than the highest price per share paid in the control share acquisition. Furthermore, certain limitations otherwise applicable to the exercise of dissenters rights would not apply in the control share acquisition. The control share acquisition statute would not apply (a) to shares acquired in a merger, consolidation or share exchange if Annaly were a

party to the transaction or (b) to acquisitions approved or exempted by Annaly s charter or bylaws. Annaly has not opted out of the Maryland Control Share Acquisition Act. The control share acquisition statute could have the effect of discouraging offers to acquire Annaly and of increasing the difficulty of consummating any such offers, even if the acquisition of Annaly would be in its stockholders best interests.

Restrictions on Ownership and Transfer

To assist Annaly in qualifying as a REIT, among other purposes, its charter prohibits anyone from acquiring or holding, directly or constructively, ownership of a number or value of shares of any class of its capital stock in excess of 9.8% of the outstanding shares. For this purpose, the term ownership generally means either direct ownership or constructive ownership in accordance with the constructive ownership provisions of Section 544 of the Code, as modified in Section 856(h) of the Code.

The constructive ownership provisions of Section 544 of the Code generally (a) attribute ownership of securities owned by a corporation, partnership, estate or trust proportionately to its stockholders, partners or beneficiaries; (b) attribute ownership of securities owned by family members to other members of the same family; and (c) set forth rules for attributing securities constructively owned by one person to another person. To determine whether a person holds or would hold capital stock in excess of the 9.8% ownership limit, a person will be treated as owning not only shares of capital stock actually owned, but also any shares of capital stock attributed to that person under the attribution rules described above. Accordingly, a person who individually owns less than 9.8% of the shares outstanding may nevertheless be in violation of the 9.8% ownership limit.

Any transfer of shares of capital stock that would cause Annaly to be disqualified as a REIT or that would (a) create a direct or constructive ownership of shares of capital stock in excess of the 9.8% ownership limit, or (b) result in the shares of capital stock being beneficially owned (within the meaning of Section 856(a) of the Code) by fewer than 100 persons (determined without reference to any rules of attribution), (c) result in us being closely held within the meaning of Section 856(h) of the Code or (d) otherwise failing to qualify as a REIT, will be null and void, and the intended transferee (the purported transferee) will acquire no rights to those shares. These restrictions on transferability and ownership will not apply if the Annaly board of directors determines that it is no longer in Annaly s best interests to continue to qualify as a REIT.

Any purported transfer of shares of capital stock that would result in a purported transferee owning (directly or constructively) shares of capital stock in excess of the 9.8% ownership limit due to the unenforceability of the transfer restrictions described above will constitute excess securities. Excess securities will be transferred by operation of law to a trust that Annaly will establish for the exclusive benefit of a charitable organization, until such time as the trustee of the trust retransfers the excess securities. The trustee will be a banking institution designated by Annaly that is not affiliated with the purported transferee or Annaly. While the excess securities are held in trust, the purported transfere will not be entitled to vote or to share in any dividends or other distributions with respect to the securities. Subject to the 9.8% ownership limit, excess securities may be transferred by the trust to any person (if such transfer would not result in excess securities). Upon such a transfer, the purported transferee shall receive a price for such excess securities or other transaction), a price per share equal to the market price (as defined in Annaly s charter) for the excess securities on the date of the purported transfer that resulted in the excess securities to the new owner, at which point the excess securities will automatically cease to be excess securities.

Edgar Filing: ANNALY CAPITAL MANAGEMENT INC - Form 424B3

Upon a purported transfer of excess securities, the purported transferee will cease to be entitled to distributions, voting rights and other benefits with respect to the shares of capital stock except the right to payment of the purchase price for the shares of capital stock on the retransfer of securities as provided above. Any dividend or distribution paid to a purported transferee on excess securities prior to Annaly s discovery that

shares of capital stock have been transferred in violation of its charter, shall be paid to the trust. If these transfer restrictions are determined to be void, invalid or unenforceable by a court of competent jurisdiction, then the purported transferee of any excess securities may be deemed, at Annaly s option, to have acted as an agent on Annaly s behalf in acquiring the excess securities and to hold the excess securities on its behalf. All certificates representing shares of capital stock will bear a legend referring to the restrictions described above.

Any person who acquires shares in violation of Annaly s charter, or any person who is a purported transferee, such that excess securities results, must immediately give written notice or, in the event of a proposed or attempted transfer that would be void as set forth above, give at least 15 days prior written notice to Annaly of such event and shall provide Annaly such other information as Annaly may request in order to determine the effect, if any, of the transfer on Annaly s qualification as a REIT. In addition, every record owner of more than 5.0% (during any period in which the number of record stockholders is 2,000 or more) or 1.0% (during any period in which the number of record stockholders is greater than 200 but less than 2,000) or 0.5% (during any period in which the number of record stockholders is 200 or less) of the number or value of Annaly s outstanding shares of capital stock must send Annaly an annual written notice by January 30 stating the name and address of the record owner and the number of shares held and describing how the shares are held. Further, each stockholder is required to disclose to Annaly in writing information with respect to the direct and constructive ownership of shares as Annaly may reasonably request in order to comply with the REIT provisions of the Code, to comply with the requirements of any taxing authority or governmental agency or to determine any such compliance.

Annaly s board of directors may increase or decrease the 9.8% ownership limit unless, after giving effect to any increased ownership limit, five or fewer persons could beneficially own, in the aggregate, more than 50% in value of the shares of our stock then outstanding. In addition, to the extent consistent with the REIT provisions of the Code, the board of directors may, upon receipt of a ruling from the Internal Revenue Service or an opinion of Annaly s tax advisor or other documents or evidence satisfactory to the board of directors and upon such other conditions as the board of directors may direct, waive the 9.8% ownership limit for a purchaser of Annaly stock.

The provisions described above may inhibit market activity and may delay, defer or prevent a change in control or other transaction and the resulting opportunity for the holders of Annaly capital stock to receive a premium for their shares that might otherwise exist in the absence of such provisions. Such provisions also may make Annaly an unsuitable investment vehicle for any person seeking to obtain ownership of more than 9.8% of the outstanding shares of any class or series of Annaly s capital stock.

COMPARISON OF STOCKHOLDERS RIGHTS

As a result of the offer and the merger, holders of MTGE common stock will become holders of Annaly common stock. Both Annaly and MTGE are Maryland corporations and are governed by the MGCL. The differences between the rights of the stockholders of Annaly and the current rights of the stockholders of MTGE arise primarily from differences in their respective constituent documents.

The following is a summary of the material differences between the current rights of holders of MTGE common stock and the current rights of holders of Annaly common stock under Maryland law and their respective constituent documents. It is not a complete statement of the provisions affecting, and the differences between, the rights of Annaly and MTGE common stockholders. This summary is qualified in its entirety by reference to Maryland law and Annaly s and MTGE s respective constituent documents. To find out where copies of these documents can be obtained, see Where to Obtain More Information.

Authorized Capital Stock	MTGE The authorized capital stock of MTGE currently consists of 350,000,000 shares of stock, consisting of 300,000,000 shares of MTGE common stock and 50,000,000 shares of preferred stock, \$0.01 par value per share. Of these shares of preferred stock, 2,300,000 shares are classified as 8.125% Series A Cumulative Redeemable Preferred Stock.	Annaly The authorized capital stock of Annaly currently consists of 2,000,000,000 shares of capital stock, par value \$0.01 per share. Of these shares of capital stock 1,909,750,000 shares are classified as shares of common stock, 12,000,000 shares are classified as shares of 7.65% Series C Cumulative Redeemable Preferred Stock, 18,400,000 shares are classified as shares of 7.50% Series D Cumulative Redeemable Preferred Stock, 11,500,000 shares are classified as shares of 7.625% Series E Cumulative Redeemable Preferred Stock, 28,800,000 shares are classified as shares of 6.95% Series F Fixed-to-Floating Rate
Number of Directors and Size of Board	The MTGE charter provides that the size of the board may be increased or decreased pursuant to the bylaws, but shall never be less than the minimum number required by the MGCL. MTGE s bylaws authorize a majority of the entire board of directors to set the number of directors, provided that, the number may not be less than the minimum required by	 and 19,550,000 shares are classified as shares of 6.50% Series G Fixed-to-Floating Rate Cumulative Redeemable Preferred Stock. The Annaly charter provides that the size of the board may be increased or decreased pursuant to the bylaws. Annaly s bylaws authorize the board of directors to set the number of directors, provided that, unless the bylaws are amended the number may not be less than the minimum required by the MGCL, which is one, or more than 15.

the MGCL, which is one, or more than 15 and further provided that the number of directors shall not be increased by

	MTGE fifty percent or more in any twelve-month period without the approval of at least sixty-six percent of the entire board of directors.	Annaly
	MTGE s board of directors currently consists of three directors.	Annaly s board of directors currently consists of 11 directors.
Term of Directors	MTGE s directors are elected annually and serve until the next annual meeting of stockholders and until their successors are duly elected and qualify.	Annaly s directors serve for three-year terms and until their successors are duly elected and qualify. The directors are divided into three classes, and the terms of one class of directors expire each year.
Election of Directors	MTGE s charter does not provide for cumulative voting in the election of directors and MTGE s bylaws provide that a nominee for director may only be elected upon receipt of a majority of all votes cast for and against such nominee, except in a contested election, in which case a nominee for director may be elected by a plurality of the votes cast.	Annaly s charter does not provide for cumulative voting in the election of directors and Annaly s bylaws provide that a nominee for director may only be elected upon receipt of a majority of the total votes cast for and against such nominee, except in a contested election, in which case a nominee for director may be elected by a plurality of the votes cast.
Removal of Directors	MTGE s charter provides that, subject to the rights of holders of one or more classes or series of preferred stock to elect or remove one or more directors, a director may be removed only for cause (as defined in MTGE s charter) and only by the affirmative vote of at least two-thirds of the votes entitled to be cast generally in the election of directors.	Because Annaly has a classified board of directors, pursuant to the MGCL, Annaly directors may only be removed for cause, and then only by the affirmative vote of the holders of a majority of the votes entitled to be cast generally in the election of directors.
Vacancies	Subject to the terms of any class or series of preferred stock, vacancies on the MTGE board may be filled only by a majority of the remaining directors, even if the remaining directors do not constitute a quorum, and any director elected to fill a vacancy will hold office for the remainder of the full term of the directorship in which the vacancy occurred and until his or her successor is duly elected and qualifies.	The Annaly bylaws provide that any vacancy on the board of directors for any cause other than an increase in the number of directors may be filled by a majority of the remaining directors, even if such majority is less than a quorum. Any vacancy in the number of directors created by an increase in the number of directors may be filled by a majority of the entire board of directors. Any individual so elected as a director will serve until the next annual meeting of Annaly

MTGE

Special Stockholders Meetings

Advance Notice Requirements of Stockholder Nominations and Proposals

The MTGE bylaws provide that special meetings of MTGE stockholders may be called by the chairman of the board of directors, the chief executive officer or the board of directors, or by the secretary upon the request of stockholders entitled to cast not less than a majority of all the votes entitled to be cast at such meeting.

The MTGE bylaws provide that nominations of individuals for election to the board and for the proposal of other business to be properly brought before an annual meeting, the stockholder must (1) be a stockholder of record at the time of giving of notice by the stockholder and at the time of the annual meeting, who is entitled to vote at the meeting and who has complied with the advance notice procedures of the bylaws, and (2) deliver notice to the secretary of MTGE not earlier than the 150th day and not later than the 120th day prior to the first anniversary of the date of the proxy statement for the preceding year s annual meeting. In the event that the date of the annual meeting is advanced or delayed more than 30 days from the first anniversary of the preceding year s annual meeting, notice must be delivered not earlier than the 150th day prior to the date of the annual meeting and not later than the later of the 120th day prior to the date of the annual meeting or the 10th day following the day on which public announcement of the date of such meeting is first made.

The MTGE bylaws provide that only the business specified in the notice of the meeting may be brought before a special

Annaly

stockholders and until his or her successor is duly elected and qualifies.

The Annaly bylaws provide that special meetings of Annaly stockholders may be called by the chairman of the board of directors, the president, the chief executive officer or the board of directors, or by the secretary upon the request of stockholders entitled to cast not less than a majority of the votes entitled to be cast at such meeting.

The Annaly bylaws provide that nominations of individuals for election to the board and for the proposal of other business to be properly brought before an annual meeting, the stockholder must (1) be a stockholder of record at the record date for the meeting, at the time of giving of notice by the stockholder and at the time of the meeting (and any postponement or adjournment thereof), who is entitled to vote at the meeting and who has complied with the advance notice procedures of the bylaws, and (2) deliver notice to the secretary of Annaly not earlier than the 150th day and not later than the 120th day prior to the first anniversary of the date of the proxy statement for the preceding year s annual meeting. In the event that the date of the annual meeting is advanced or delayed more than 30 days from the first anniversary of the preceding year s annual meeting, notice must be delivered not earlier than the 150th day prior to the date of the annual meeting and not later than the later of the 120th day prior to the date of the annual meeting or the 10th day following the day on which public announcement of the date of such meeting is first made.

meeting of stockholders. Stockholder nominations of individuals for election to the board of

The Annaly bylaws provide that only the business specified in the notice of the meeting may be brought before a special meeting of stockholders.

MTGE

directors may be made at a special meeting, provided a special meeting has been called for purposes of electing directors, by a stockholder who (1) is a stockholder of record at the time of giving notice by the stockholder and at the time of the meeting, is entitled to vote at the meeting and has complied with the advance notice procedures of the bylaws, and (2) delivers notice to the secretary of MTGE not earlier than the 120th day prior to such special meeting and not later than the later of the 90th day prior to the special meeting and the 10th day following the day on which public announcement is first made of the date of the special meeting and of the nominees proposed by the board of directors to be elected at such meeting.

Annaly

Stockholder nominations of individuals for election to the board of directors may be made at a special meeting by a stockholder who has requested that a special meeting for the purpose of electing directors in accordance with Annaly s bylaws, or provided a special meeting has been called for purposes of electing directors, by a stockholder who (1) is a stockholder of record at the record date for the meeting, at the time of giving notice by the stockholder and at the time of the meeting (and any postponement or adjournment thereof), is entitled to vote at the meeting and has complied with the advance notice procedures of the bylaws, and (2) delivers notice to the secretary of Annaly not earlier than the 120th day prior to such special meeting and not later than the later of the 90th day prior to the special meeting and the 10th day following the day on which public announcement is first made of the date of the special meeting and of the nominees proposed by the board of directors to be elected at such meeting.

Stockholder Action by Written Consent	Under the MGCL, MTGE common stockholders may only act by written consent if such written consent is unanimous.	Under the stockholde consent if unanimous
Amendment of Governing Documents	Except for amendments to the provisions of MTGE charter relating to the removal of a director, and amendments to the vote required to amend these provisions (each of which must be advised by the board and approved by the affirmative vote of stockholders entitled to cast at least two-thirds of all the votes entitled to be cast on the matter) and provisions only requiring the approval of the board, the MTGE charter generally may be amended only if the amendment is first advised by the board and thereafter approved by the affirmative vote of	The Annal that an am must be ap stockholde the holder number sh and entitle charter do board of d amend the stockholde the aggreg that Annal number of

Under the MGCL, Annaly common stockholders may only act by written consent if such written consent is unanimous.

The Annaly charter generally provides that an amendment to the Annaly charter must be approved by Annaly stockholders by the affirmative vote of the holders of a majority of the total number shares of all classes outstanding and entitled to vote thereon. The Annaly charter does not provide the Annaly board of directors with the power to amend the Annaly charter, without a stockholder vote, to increase or decrease the aggregate number of shares of stock that Annaly is authorized to issue or the number of shares of stock of any class or

Edgar Filing: ANNALY CAPITAL MANAGEMENT INC - Form 424B3

stockholders

series that Annaly is authorized to issue.

Annaly

Table of Contents

MTGE

	entitled to cast a majority of all of the votes entitled to be cast on the matter. As permitted under the MGCL, the MTGE charter provides that the MTGE board of directors, with the approval of a majority of the entire board, and without action by the stockholders, may amend the MTGE charter to increase or decrease the aggregate number of shares of stock that the corporation is authorized to issue or the number of shares of stock of any class or series that the corporation is authorized to issue.	
	The MTGE bylaws provide that the MTGE bylaws may be amended, altered or repealed and new bylaws may be adopted, either by the affirmative vote of stockholders entitled to cast at least a majority of the votes entitled to be cast at any annual meeting or special meeting called for such purpose, or by a majority of the board of directors.	The Annaly bylaws provide that the Annaly bylaws may be amended, altered, repealed or replaced and new bylaws may be adopted, either by the affirmative vote of stockholders entitled to cast at least a majority of the votes entitled to be cast at any duly organized annual or special meeting, or by a majority of the board of directors, including a majority of Annaly s independent directors in office at any regular or special meeting of the board of directors.
Approval of Extraordinary Transactions	Under the MGCL, a Maryland corporation generally cannot merge, convert, sell all or substantially all of its assets or engage in a statutory share exchange, unless declared advisable by the board of directors and approved by the affirmative vote of stockholders entitled to cast at least two-thirds of the votes entitled to be cast on the matter. However, a Maryland corporation may provide in its charter for approval of these matters by a lesser percentage, but not less than a majority of all of the votes entitled to be cast on the matter. The MTGE charter provides for the approval of these matters by a majority of all the votes entitled to be cast on these matters.	Under the MGCL, a Maryland corporation generally cannot merge, convert, sell all or substantially all of its assets or engage in a statutory share exchange, unless declared advisable by the board of directors and approved by the affirmative vote of stockholders entitled to cast at least two-thirds of the votes entitled to be cast on the matter. However, a Maryland corporation may provide in its charter for approval of these matters by a lesser percentage, but not less than a majority of all of the votes entitled to be cast on the matter. The Annaly charter provides for the approval of these matters by a majority of all the votes entitled to be cast on these matters.
Ownership and Transfer Restrictions	Except with regard to persons exempted by the MTGE board of	Except with regard to persons exempted by the Annaly board of

	MTGE directors from the ownership and transfer restrictions of the MTGE charter, no person may beneficially or constructively own more than 9.8% (in value or number of shares, whichever is more restrictive) of the outstanding MTGE common stock or more than 9.8% (in value or number of shares, whichever is more restrictive) of all classes or series of MTGE capital stock.	Annaly directors from the ownership and transfer restrictions of the Annaly charter, no person may beneficially or constructively own more than 9.8% (in number or value of shares) of any class of Annaly capital stock.
Maryland Business Combination Act	Under the MGCL, certain business combinations (including a merger, consolidation, statutory share exchange or, in certain circumstances, an asset transfer or issuance or reclassification of equity securities) between a Maryland corporation and an interested stockholder (defined generally as any person who beneficially owns, directly or indirectly, 10% or more of the voting power of the corporation s outstanding voting stock or an affiliate or associate of the corporation who, at any time during the two-year period immediately prior to the date in question, was the beneficial owner of 10% or more of the voting power of the then-outstanding stock of the corporation) or an affiliate of such an interested stockholder are prohibited for five years after the most recent date on which the interested stockholder. Thereafter, any such business combination must generally be recommended by the board of directors of the corporation and approved by the affirmative vote of at least (1) 80% of the votes entitled to be cast by holders of outstanding voting stock of the corporation and (2) two-thirds of the votes entitled to be cast by holders of voting stock of the corporation, other than shares held by the interested stockholder with whom (or with whose affiliate) the business combination is to be effected or held by an affiliate or	As permitted by the MGCL, Annaly has elected in its charter not to be governed by the Maryland Business Combination Act.

associate of the interested stockholder, unless, among other conditions, the

MTGE

corporation s common stockholders receive a minimum price (as defined in the MGCL) for their shares and the consideration is received in cash or in the same form as previously paid by the interested stockholder for its shares. A person is not an interested stockholder under the statute if the board of directors approved in advance the transaction by which the person otherwise would have become an interested stockholder. A corporation s board of directors may provide that its approval is subject to compliance with any terms and conditions determined by the board of directors.

Pursuant to the statute, the MTGE board of directors has adopted a resolution exempting from the business combination provisions: (i) any business combination between MTGE and Annaly and the Offeror, (ii) any business combination between MTGE and any other person, provided that the business combination is first approved by a majority of the directors of MTGE (including a majority of the directors who are not affiliates or associates of such person) and (iii) any person acting in concert with any of the foregoing; consequently, these provisions will not apply to the offer or the merger.

Maryland Control ShareThe MTGE bylaws contain a provisionThe IAcquisition Actexempting from the control shareconacquisition statute any and allacquiacquisitions by any person of shares ofhaveMTGE stock.those

The MGCL provides that holders of control shares of a Maryland corporation acquired in a control share acquisition have no voting rights with respect to those shares except to the extent approved by the affirmative vote of at least two-thirds of the votes entitled to be cast by stockholders entitled to vote generally in the election of directors other than the person who has made or proposes to make the control share

acquisition, an officer of the corporation or an employee of the corporation who is also a director of the corporation. Control shares are voting shares of

MTGE

Annaly

stock that, taken together with all other voting shares of stock the acquirer previously acquired or in respect of which the acquirer is able to exercise or direct the exercise of voting power (except solely by virtue of a revocable proxy), would entitle the acquirer to exercise voting power in electing directors within one of the following ranges of voting power:

one-tenth or more but less than one-third of all voting power;

one-third or more but less than a majority of all voting power; or

a majority or more of all voting power.

Control shares do not include shares of stock the acquiring person is entitled to vote as a result of having previously obtained stockholder approval. A control share acquisition means the acquisition of issued and outstanding control shares, subject to certain exceptions.

If a person who has made (or proposes to make) a control share acquisition satisfies certain conditions (including agreeing to pay expenses), such person may compel the corporation s board of directors to call a special meeting of stockholders to be held within 50 days of demand to consider the voting rights of the shares. If such a person makes no request for a meeting, the corporation has

the option to present the question at any stockholders meeting.

If voting rights are not approved at a meeting of stockholders or the acquiring person does not deliver an acquiring person statement as required by the statute, then, subject to certain conditions and limitations, the corporation may redeem any or all of the control shares (except those for which voting rights have previously

MTGE

Annaly

been approved) for fair value. The corporation will determine the fair value of the shares, without regard to the absence of voting rights, as of the date of either:

the last control share acquisition; or

the meeting of stockholders where stockholders considered and did not approve voting rights of the control shares.

If voting rights for control shares are approved at a stockholders meeting and the acquirer becomes entitled to vote a majority of the shares of stock entitled to vote, all other stockholders may obtain rights as objecting stockholders and, thereunder, exercise appraisal rights. This means that stockholder would be able to force Annaly to redeem its stock for fair value. Under the MGCL, the fair value may not be less than the highest price per share paid in the control share acquisition. Furthermore, certain limitations otherwise applicable to the exercise of dissenters rights would not apply in the context of a control share acquisition. The control share acquisition statute would not apply to shares acquired in a merger, consolidation or share exchange if the corporation were a party to the transaction.

Annaly has not opted out of the Maryland Control Share Acquisition Act.

Limitation of Liability and Indemnification of Directors and Officers The MTGE charter provides for elimination of the liability of its directors and officers to MTGE or its stockholders for money damages to the maximum extent permitted by Maryland law from time to time. The MGCL permits a Maryland corporation to limit the liability of its directors and officers to the corporation and its stockholders for The Annaly charter provides for elimination of the liability of its directors and officers to Annaly or its stockholders for money damages to the maximum extent permitted by Maryland law from time to time. The MGCL permits a Maryland corporation to limit the liability of its directors and officers to the corporation and its stockholders for

MTGE

money damages except to the extent that (1) it is proved that the person actually received an improper benefit or profit in money, property or services or (2) a judgment or other final adjudication adverse to the person is entered in a proceeding based on a finding that the person s action, or failure to act, was the result of active and deliberate dishonesty and was material to the cause of action adjudicated at the proceeding.

The MTGE charter authorizes MTGE and the MTGE bylaws obligate MTGE to indemnify its directors and officers, including the advancement of expenses, to the full extent permitted by Maryland law. The MGCL permits a corporation to indemnify its present and former directors and officers against judgments, penalties, fines, settlements and reasonable expenses actually incurred by them in connection with any proceeding to which they may be made a party by reason of their service in those or other capacities, unless it is established that (1) the act or omission of the director or officer was material to the matter giving rise to the proceeding and (a) was committed in bad faith, or (b) was the result of active and deliberate dishonesty; (2) the director or officer actually received an improper personal benefit in money, property or services; or (3) in the case of any criminal proceeding, the director or officer had reasonable cause to believe that the act or omission was unlawful. In addition, the MGCL permits a corporation to advance reasonable expenses to a director or officer upon the corporation s receipt of (a) a written affirmation by the director or officer of such person s good faith belief that he or she has met the standard of conduct necessary for indemnification by the

Annaly

money damages except to the extent that (1) it is proved that the person actually received an improper benefit or profit in money, property or services or (2) a judgment or other final adjudication adverse to the person is entered in a proceeding based on a finding that the person s action, or failure to act, was the result of active and deliberate dishonesty and was material to the cause of action adjudicated at the proceeding.

The Annaly charter and bylaws obligate Annaly to indemnify its directors and officers, including the advancement of expenses, to the full extent permitted by Maryland law. The MGCL permits a corporation to indemnify its present and former directors and officers against judgments, penalties, fines, settlements and reasonable expenses actually incurred by them in connection with any proceeding to which they may be made a party by reason of their service in those or other capacities, unless it is established that (1) the act or omission of the director or officer was material to the matter giving rise to the proceeding and (a) was committed in bad faith, or (b) was the result of active and deliberate dishonesty; (2) the director or officer actually received an improper personal benefit in money, property or services; or (3) in the case of any criminal proceeding, the director or officer had reasonable cause to believe that the act or omission was unlawful. In addition, the MGCL permits a corporation to advance reasonable expenses to a director or officer upon the corporation s receipt of (a) a written affirmation by the director or officer of such person s good faith belief that he or she has met the standard of conduct necessary for indemnification by the corporation and (b) a written

corporation and (b) a written undertaking undertaking by such person or on his by such person or on his

Annaly

Table of Contents

MTGE

	or her behalf to repay the amount paid or reimbursed by the corporation if it shall ultimately be determined that the standard of conduct was not met.	or her behalf to repay the amount paid or reimbursed by the corporation if it shall ultimately be determined that the standard of conduct was not met.
Exclusive Forum Provision	MTGE s bylaws provide that unless MTGE consents in writing to the selection of an alternative forum, the Circuit Court for Baltimore City, Maryland, or if that court does not have jurisdiction, the United States District Court for the District of Maryland, Baltimore Division, will be the sole and exclusive forum for (1) any derivative action or proceeding brought on behalf of MTGE, (2) any action asserting a claim of breach of any duty owed by a director or officer or other employee of MTGE to MTGE or its stockholders, (3) any action asserting a claim against MTGE or any director or officer or other employee of MTGE arising pursuant to any provision of the MGCL or the MTGE charter or MTGE bylaws, or (4) any other action asserting a claim against MTGE or any director or officer or other employee of MTGE that is governed by the internal affairs doctrine.	Annaly s bylaws provide that unless Annaly consents in writing to the selection of an alternative forum, the Circuit Court for Baltimore City, Maryland, or if that court does not have jurisdiction, the United States District Court for the District of Maryland, Baltimore Division, will be the sole and exclusive forum for (1) any derivative action or proceeding brought on behalf of Annaly, (2) any action asserting a claim of breach of any duty owed by an director or officer or other employee of Annaly to Annaly or its stockholders, (3) any action asserting a claim against Annaly or any director or officer or other employee of Annaly arising pursuant to any provision of the MGCL or the Annaly charter or Annaly bylaws, or (4) any other action asserting a claim against Annaly or any director or officer or other employee of Annaly that is governed by the internal affairs doctrine.

LEGAL MATTERS

Certain matters relating to the issuance of shares of Annaly common stock and Annaly preferred stock will be passed upon for Annaly by DLA Piper LLP (US).

EXPERTS

Ernst & Young LLP, an independent registered public accounting firm, has audited Annaly s consolidated financial statements included in Annaly s Annual Report on Form 10-K for the year ended December 31, 2017, and the effectiveness of Annaly s internal control over financial reporting as of December 31, 2017 as set forth in their reports, which are incorporated by reference in this document. Annaly s financial statements are incorporated by reference in reliance on Ernst & Young LLP s reports, given on their authority as experts in accounting and auditing.

Ernst & Young LLP, an independent registered public accounting firm, has audited MTGE s consolidated financial statements included in MTGE s Annual Report on Form 10-K for the year ended December 31, 2017, and the effectiveness of MTGE s internal control over financial reporting as of December 31, 2017 as set forth in their reports, which are incorporated by reference in this document. MTGE s financial statements are incorporated by reference in reliance on Ernst & Young LLP s reports, given on their authority as experts in accounting and auditing.

WHERE TO OBTAIN MORE INFORMATION

Annaly and MTGE file annual, quarterly and current reports, proxy statements and other information with the SEC. MTGE stockholders may read and copy any reports, statements or other information that Annaly or MTGE file with the SEC at the SEC s public reference room at 100 F Street, N.E., Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information regarding the public reference room. Annaly s and MTGE s public filings also are available to the public from commercial document retrieval services and may be obtained without charge at the SEC s website at www.sec.gov.

Annaly has filed a registration statement on Form S-4 with the SEC to register the offer and sale of shares of Annaly common stock to be issued in the offer and the merger and shares of Annaly preferred stock to be issued in the merger. This document is a part of that registration statement. Annaly may also file amendments to such registration statement. In addition, on the date of the initial filing of the registration statement on Form S-4 of which document is a part, Annaly and the Offeror filed with the SEC a Tender Offer Statement on Schedule TO under the Exchange Act, together with exhibits, to furnish certain information about the offer. Annaly and the Offeror may file amendments to the Schedule TO. As allowed by SEC rules, this document does not contain all of the information in the registration statement or the Schedule TO, or the exhibits to the registration statement or the Schedule TO. Copies of the Form S-4 and the Schedule TO (and any amendments to those documents) may be obtained by contacting the information agent as directed elsewhere in this document.

The SEC allows Annaly to incorporate information into this document by reference, which means that Annaly and the Offeror can disclose important information to MTGE stockholders by referring to another document or information filed separately with the SEC. The information incorporated by reference is deemed to be part of this document, except for any information amended or superseded by information contained in, or incorporated by reference into, this document. This document incorporates by reference the documents and information set forth below that Annaly and MTGE have previously filed with the SEC. These documents contain important information about Annaly and MTGE and their financial conditions.

Annaly Filings:

Annaly Information Incorporated by Reference Annual Report on Form 10-K	Period Covered or Date of Filing Fiscal year ended December 31, 2017, as filed with the SEC on February 16, 2018
Quarterly Reports on Form 10-Q	Fiscal quarter ended March 31, 2018, as filed with the SEC on May 3, 2018
Definitive Proxy Statement on Schedule 14A	Filed with the SEC on April 10, 2018
The description of Annaly common stock set forth or incorporated by reference in Annaly s Registration Statement on Form 8-A, including all amendments and reports filed for the purpose of updating such description.	Filed with the SEC on October 6, 1997
The description of Annaly Series C Preferred Stock set forth or incorporated by reference in Annaly s Registration	Filed with the SEC on May 16, 2012

Table of Contents

Statement on Form 8-A, including all amendments and reports filed for the purpose of updating such description.

Annaly Information Incorporated by Reference	Period Covered or Date of Filing
The description of Annaly Series D Preferred Stock set forth or incorporated by reference in Annaly s Registration Statement on Form 8-A, including all amendments and reports filed for the purpose of updating such description.	Filed with the SEC on September 13, 2012
The description of Annaly Series F Preferred Stock set forth or incorporated by reference in Annaly s Registration Statement on Form 8-A, including all amendments and reports filed for the purpose of updating such description	Filed with the SEC on July 27, 2017
The description of Annaly Series G Preferred Stock set forth or incorporated by reference in Annaly s Registration Statement on Form 8-A, including all amendments and reports filed for the purpose of updating such description	Filed with the SEC on January 10, 2018
Current Reports on Form 8-K	Filed on:
	January 3, 2018
	January 9, 2018
	January 12, 2018
	May 2, 2018
MTGE Filings:	May 3, 2018
MTGE Information Incorporated by Reference Annual Report on Form 10-K	Period Covered or Date of Filing Fiscal year ended December 31, 2017, as filed with the SEC on February 27, 2018
Quarterly Report on Form 10-Q	Fiscal quarter ended March 31, 2018, as filed with the SEC on May 10, 2018

Edgar Filing: ANNALY CAPITAL MANAGEMENT INC - Form 424B3

Definitive Proxy Statement on Schedule 14A	Filed with the SEC on March 22, 2018
The description of MTGE s common stock set forth or incorporated by reference in MTGE s Registration Statement on Form 8-A, including all amendments and reports filed for the purpose of updating such description.	Filed with the SEC on August 1, 2011
The description of MTGE Series A preferred stock set forth or incorporated by reference in MTGE s Registration Statement on Form 8-A, including all amendments and reports filed for the purpose of updating such description.	Filed with the SEC on May 16, 2014

MTGE Information Incorporated by Reference

Current Reports on Form 8-K

Period Covered or Date of Filing

February 13, 2018

May 2, 2018

May 3, 2018

Annaly also hereby incorporates by reference any additional documents that either it or MTGE may file with the SEC pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act from the date of this document until the termination of the offering. Such additional documents, however, are not automatically incorporated by reference into the Schedule TO. Annaly will file amendments to the Schedule TO, to the extent required, specifically to include information that is filed from the date of this document and incorporated by reference herein. Nothing in this document shall be deemed to incorporate information furnished but not filed with the SEC or the contents of Annaly s and MTGE s websites.

MTGE stockholders may obtain any of these documents without charge upon request to the information agent, D.F. King & Co., Inc., the information agent, at (800) 669-5550 (toll free for all stockholders) or (212) 269-5550 (banks and brokers), or from the SEC at the SEC s website at www.sec.gov.

ANNEX A

AGREEMENT AND PLAN OF MERGER

BY AND AMONG

ANNALY CAPITAL MANAGEMENT, INC.

MOUNTAIN MERGER SUB CORPORATION

AND

MTGE INVESTMENT CORP.

DATED AS OF

MAY 2, 2018

Table of Contents

<u>ARTICLE</u>	I THE OFFER	Page A-2
1.1	The Offer	A-2
1.2	Company Actions	A-7
ARTICLE	II THE MERGER	A-9
2.1	The Merger: Effects of the Merger	A-9
2.2	The Closing	A-9
2.3	Effective Time	A-10
2.4	Governing Documents	A-10
2.5	Directors and Officers of the Surviving Corporation	A-10
2.6	Tax Consequences	A-10
ARTICLE	III TREATMENT OF SECURITIES	A-11
3.1	Treatment of Capital Stock	A-11
3.2	Payment for Securities; Surrender of Certificates	A-13
3.3	Dissenters Rights	A-16
3.4	Treatment of Company RSUs	A-16
3.5	Withholding	A-16
3.6	Fractional Shares	A-16
ARTICLE	IV REPRESENTATIONS AND WARRANTIES OF THE COMPANY	A-16
4.1	Qualification, Organization, Subsidiaries, etc.	A-17
4.2	Capitalization	A-17
4.3	Corporate Authority	A-19
4.4	Governmental Consents; No Violation	A-19
4.5	SEC Reports and Financial Statements	A-20
4.6	Internal Controls and Procedures	A-20
4.7	No Undisclosed Liabilities	A-21
4.8	Absence of Certain Changes or Events	A-21
4.9	Compliance with Laws; Permits	A-21
4.10	Employee Benefit Plans	A-23
4.11	Tax Matters	A-24

4.12	Labor Matters	A-26
4.13	Investigation: Litigation	A-27
4.14	Intellectual Property	A-27
4.15	Real Property	A-27

A-i

Table of Contents

(continued)

4.16	Material Contracts	Page A-28
4.17	Mortgage Backed Securities	A-30
4.18	Insurance	A-30
4.19	Information Supplied	A-31
4.20	Opinion of Financial Advisor	A-31
4.21	State Takeover Statutes	A-31
4.22	Investment Company Act	A-31
4.23	Finders and Brokers	A-31
4.24	No Other Representations	A-31
ARTICLE	V REPRESENTATIONS AND WARRANTIES OF PARENT AND PURCHASER	A-32
5.1	Qualification, Organization, Subsidiaries, etc.	A-32
5.2	Capitalization	A-33
5.3	Corporate Authority	A-33
5.4	Governmental Consents: No Violation	A-34
5.5	SEC Reports and Financial Statements	A-34
5.6	Internal Controls and Procedures	A-35
5.7	No Undisclosed Liabilities	A-35
5.8	Absence of Changes or Events	A-35
5.9	Compliance with Laws: Permits	A-36
5.10	Investigations; Litigation	A-36
5.11	Information Supplied	A-36
5.12	Sufficient Funds	A-37
5.13	No Purchaser Activity	A-37
5.14	Tax Matters	A-37
5.15	Stock Ownership	A-39
5.16	Investment Company Act	A-39
5.17	No Other Representations	A-39
ARTICLE	VI COVENANTS RELATING TO CONDUCT OF BUSINESS PENDING THE MERGER	A-39

6.1	Conduct of Business by the Company	A-39
6.2	Conduct of Business by Parent	A-44
6.3	Non-Solicitation by the Company	A-45
6.4	Additional Dividends	A-48

A-ii

Table of Contents

(continued)

ARTICLE VII ADDITIONAL AGREEMENTS		Page A-49
7.1	Access; Confidentiality; Notice of Certain Events	A-49
7.2	Reasonable Best Efforts	A-50
7.3	Publicity	A-51
7.4	D&O Insurance and Indemnification	A-51
7.5	Takeover Statutes	A-53
7.6	Obligations of Purchaser	A-53
7.7	<u>Rule 16b-3</u>	A-53
7.8	Security Holder Litigation	A-53
7.9	Delisting	A-53
7.10	Director Resignations	A-53
7.11	Certain Tax Matters	A-53
7.12	Stock Exchange Listing	A-55
7.13	14d-10 Matters	A-55
7.14	Treatment of Management Agreement	A-55
7.15	Existing Indebtedness	A-55
7.16	Employees and Employee Benefit Plans	A-55
ARTICLE	VIII CONDITIONS TO CONSUMMATION OF THE MERGER	A-56
8.1	Conditions to Each Party s Obligations to Effect the Merger	A-56
ARTICLE	IX TERMINATION	A-57
9.1	Termination	A-57
9.2	Effect of Termination	A-58
ARTICLE	X MISCELLANEOUS	A-59
10.1	Amendment and Modification: Waiver	A-59
10.2	Non-Survival of Representations and Warranties	A-59
10.3	Expenses	A-59
10.4	Notices	A-59
10.5	Interpretation	A-60

10.6	Counterparts	A-60
10.7	Entire Agreement Third-Party Beneficiaries	A-61
10.8	Severability	A-61
10.9	Governing Law; Jurisdiction	A-61

A-iii

Table of Contents

		Page
10.10	Waiver of Jury Trial	A-62
10.11	Assignment	A-62
10.12	Enforcement: Remedies	A-62
Annex A	Certain Defined Terms	A-A-1
Annex B	Conditions to the Offer	A-B-1
Annex C	Form of Articles Supplementary Classifying Parent Series H Preferred Stock	A-C-1

A-iv

AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER (this *Agreement*), dated as of May 2, 2018, is by and among ANNALY CAPITAL MANAGEMENT, INC., a Maryland corporation (*Parent*), MOUNTAIN MERGER SUB CORPORATION, a Maryland corporation and a wholly owned subsidiary of Parent (*Purchaser*), and MTGE INVESTMENT CORP., a Maryland corporation (the *Company*). All capitalized terms used in this Agreement shall have the meanings ascribed to such terms in Annex A or as otherwise defined elsewhere in this Agreement. Parent, Purchaser and the Company are each sometimes referred to herein as a *Party* and collectively as the *Parties*.

RECITALS

WHEREAS, the Company and Parent are Maryland corporations operating as real estate investment trusts for U.S. federal income tax purposes;

WHEREAS, on the terms and subject to the conditions set forth herein, Purchaser has agreed to commence an exchange offer (the *Offer*) to acquire any (subject to the Minimum Condition) and all of the outstanding shares of common stock, \$0.01 par value per share, of the Company (the *Company Common Stock* or, such shares, *Company Shares*) for the consideration and upon the terms and subject to the conditions set forth herein;

WHEREAS, following the consummation of the Offer, the Parties intend that the Company will be merged with and into Purchaser, with Purchaser being the surviving corporation of such merger (the *Merger*), without the approval of the Company s stockholders in accordance with Section 3-106.1 of the Maryland General Corporation Law (the *MGCL*), on the terms and subject to the conditions set forth in this Agreement, and the Merger shall be governed by and effected pursuant to Section 3-106.1 of the MGCL;

WHEREAS, the Parties intend that the Offer and the Merger, taken together, will qualify as a reorganization under Section 368(a) of the Code, and this Agreement is intended to be and is adopted as a plan of reorganization for the Merger for purposes of Sections 354 and 361 of the Code and within the meaning of Section 1.368-2(g) of the Treasury Regulations;

WHEREAS, in connection with the Merger, (a) each Company Share issued and outstanding immediately prior to the Effective Time (other than any Cancelled Shares) will be automatically converted into the right to receive the Common Merger Consideration upon the terms and conditions set forth in this Agreement and in accordance with the MGCL, and (b) each Company Preferred Share issued and outstanding immediately prior to the Effective Time will be automatically converted into the right to receive the Preferred Merger Consideration upon the terms and conditions set forth in this Agreement and in accordance with the MGCL;

WHEREAS, the board of directors of the Company (the *Company Board of Directors*), acting upon the unanimous recommendation of the special committee of independent, disinterested directors of the Company (the *Company Special Committee*) formed for the purpose of, among other things, evaluating and making a recommendation to the Company Board of Directors with respect to this Agreement and the transactions contemplated hereby (the

Transactions), has (i) determined that the terms of this Agreement and the Transactions, including the Offer, the Merger and the issuance of shares of Parent Common Stock in connection therewith, are in the best interests of, the Company and its stockholders (the **Company Stockholders**), (ii) declared the Offer, the Merger and the other Transactions advisable, (iii) approved the execution and delivery by the Company of this Agreement, the performance by the Company of its covenants and agreements contained herein and the consummation of the Offer, the Merger and the other Transactions upon the terms and subject to the conditions contained herein, and (iv) resolved to recommend that the Company Stockholders accept the Offer and tender their shares of Company Common Stock to Purchaser

pursuant to the Offer (including such Company Special Committee recommendation, the *Company Board Recommendation*);

WHEREAS, the respective boards of directors of Parent and Purchaser have unanimously approved and declared advisable this Agreement, the Offer, the Merger, the issuance of shares of Parent Common Stock and Parent Series H Preferred Stock in connection therewith and the other Transactions upon the terms and subject to the conditions contained herein; and

WHEREAS, the Parties desire to make certain representations, warranties, covenants and agreements in connection with the Offer and the Merger and also prescribe various terms of and conditions to the Offer and the Merger.

Now, THEREFORE, in consideration of the mutual covenants and agreements contained in this Agreement and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the Parties hereto, intending to be legally bound, agree as follows:

ARTICLE I

THE OFFER

1.1 The Offer.

(a) Terms and Conditions of the Offer. Subject to the terms and conditions of this Agreement and provided that this Agreement shall not have been terminated pursuant to Article IX and that the Company shall have complied with its obligations under this Section 1.1 and Section 1.2, as promptly as practicable after the date hereof (but in no event more than ten (10) business days thereafter), Purchaser shall (and Parent shall cause Purchaser to) commence (within the meaning of Rule 14d-2 promulgated under the Exchange Act) the Offer. In the Offer, each Company Share accepted by Purchaser in accordance with the terms and subject to the conditions of the Offer shall be exchanged for the right to receive, at the election of the holder: (i) \$19.65 in cash (the *Cash Consideration*), (ii) 1.9037 shares of Parent Common Stock (the *Stock Consideration*), or (iii) \$9.82 in cash and 0.9519 shares of Parent Common Stock (the *Mixed Consideration*), in each case subject to proration as set forth in Section 1.1(c) and the other provisions of this Article I. The Offer shall be made by means of an offer to purchase (the *Offer to Purchase*) that is disseminated to all of the holders of Company Shares as and to the extent required by the Exchange Act. The Offer to Purchase will contain the terms and conditions set forth in this Agreement and in Annex B. Each of Parent and Purchaser shall use its reasonable best efforts to consummate the Offer, subject to the terms and conditions hereof and thereof. The Offer shall be subject only to:

(i) the condition that, prior to the expiration of the Offer, there being validly tendered (excluding Company Shares tendered pursuant to guaranteed delivery procedures that have not yet been received by the depositary for the Offer) and not validly withdrawn in accordance with the terms of the Offer a number of Company Shares that, together with the Company Shares then owned by Parent and Purchaser (if any), represents at least a majority of all then-outstanding Company Shares (the *Minimum Condition*); and

(ii) the other conditions set forth in Annex B.

(b) Purchaser expressly reserves the right to waive or modify any of the conditions to the Offer and to make any change in the terms of, or conditions to, the Offer; *provided, however*, that notwithstanding the foregoing or anything to the contrary set forth herein, without the prior written consent of the Company, in its sole discretion, Purchaser may not (and Parent shall not permit Purchaser to) (i) waive any of the conditions set forth in clauses (A), (B), (C), (D), (E)(5)(ii), (E)(6)(i), (G)(1), (G)(2) or (G)(3) of Annex B (provided that Parent shall (and shall cause Purchaser to) waive any of the conditions set forth in clauses (E)(5)(ii), (E)(6)(i), (G)(2) or (G)(3) of Annex B (provided that Parent shall (and shall cause Purchaser to) waive any of the conditions set forth in clauses (E)(5)(ii), (E)(6)(i), (G)(2) or (G)(3) of Annex B (provided that Parent shall (and shall cause Purchaser to) waive any of the conditions set forth in clauses (E)(5)(ii), (E)(6)(i), (G)(2) or (G)(3) of Annex B (provided that Parent shall (and shall cause Purchaser to) waive any of the conditions set forth in clauses (E)(5)(ii), (E)(6)(i), (G)(2) or (G)(3) of Annex B upon the

written request of the Company), or (ii) make any change in the terms of or

conditions to the Offer that (A) changes the form of consideration to be paid in the Offer, (B) decreases the amount of consideration in the Offer or the number of Company Shares sought in the Offer (unless an appropriate adjustment is made pursuant to Section 1.1(d)), (C) extends the Offer, other than in a manner required or permitted by the provisions of Section 1.1(e), (D) imposes conditions to the Offer other than those set forth in Annex B, or (E) amends or modifies any other term of, or condition to, the Offer in any manner that is adverse or would reasonably be expected to be adverse to the holders of Company Shares in their capacities as such.

(c) Elections; Proration; Fractional Shares.

(i) Subject to the other provisions of this Section 1.1(c), each holder of Company Shares validly tendered in the Offer shall be entitled to elect (1) the number of Company Shares which such holder desires to exchange for the right to receive the Mixed Consideration (a *Mixed Election*, and such shares, the *Mixed Election Shares*), (2) the number of Company Shares which such holder desires to exchange for the right to receive the Cash Consideration (a *Cash Election*, and such shares), and (3) the number of Company Shares which such holder desires to exchange for the right to receive the Stock Consideration (a *Stock Election*, and such shares, the *Stock Election Shares*). Any Cash Election, Stock Election or Mixed Election shall be referred to herein as an *Election*, and shall be made on a form for that purpose prepared by Parent that is reasonably acceptable to the Company (a *Form of Election in Offer*), included as part of the letter(s) of election and transmittal accompanying the Offer. Holders of record of Company Shares who hold such Company Shares as nominees, trustees or in other representative capacities may submit multiple Forms of Election in Offer on behalf of their respective beneficial holders.

(ii) Any Election pursuant to the Offer shall have been properly made only if the depositary for the Offer shall have actually received a properly completed Form of Election in Offer by the expiration date of the Offer. Any Form of Election in Offer may be revoked or changed by the authorized Person properly submitting such Form of Election in Offer, by written notice received by the depositary for the Offer prior to the expiration date of the Offer. In the event an Election is revoked prior to the expiration date of the Offer, the Company Shares represented by such Form of Election in Offer and are not validly withdrawn from the Offer), except to the extent such Company Shares remain tendered in the Offer and are not validly withdrawn from the Offer, the depositary for the Offer. Subject to the terms of this Agreement and of the Form of Election in Offer, the depositary for the Offer shall have reasonable discretion to determine whether any election, revocation or change has been properly or timely made and to disregard immaterial defects in any Form of Election in Offer, and any good-faith decisions of the depositary for the Offer regarding such matters shall be binding and conclusive. None of Parent, Purchaser, the Company or the depositary for the Offer shall be under any obligation to notify any Person of any defect in a Form of Election in Offer.

(iii) Notwithstanding any other provision contained in this Agreement, the Cash Elections and the Stock Elections shall be subject to proration as described in this Section 1.1(c).

(1) The maximum number of Company Shares validly tendered and not validly withdrawn in the Offer which shall be eligible to receive the Cash Consideration shall equal 50% of the aggregate number of Company Shares tendered in the Offer (and not validly withdrawn) (excluding Mixed Election Shares and No Election Shares) (the *Maximum Cash Shares in Offer*).

(2) If the total number of Cash Election Shares validly tendered and not validly withdrawn in the Offer exceeds the Maximum Cash Shares in Offer, such Cash Elections shall be subject to proration as follows: for each such Cash Election, the number of Company Shares that shall be exchanged for the right to receive the Cash Consideration shall be (A) the total number of Cash Election Shares validly tendered and not validly withdrawn in the Offer multiplied by (B) the Offer Cash Proration Factor, rounded down to the nearest Company Share. The *Offer Cash Proration Factor*

means a fraction (x) the numerator of which shall be the

Maximum Cash Shares in Offer and (y) the denominator of which shall be the total number of Cash Election Shares validly tendered and not withdrawn in the Offer. The Cash Election Shares validly tendered and not validly withdrawn in the Offer that are not exchanged for the right to receive the Cash Consideration in accordance with this Section 1.1(c)(iii)(2) shall instead be exchanged for the right to receive the Stock Consideration.

(3) The maximum number of Company Shares validly tendered and not validly withdrawn in the Offer which shall be eligible to receive the Stock Consideration shall equal 50% of the aggregate number of Company Shares tendered in the Offer (and not validly withdrawn) (excluding Mixed Election Shares and No Election Shares) (the *Maximum Stock Shares in Offer*).

(4) If the total number of Stock Election Shares validly tendered and not validly withdrawn in the Offer exceeds the Maximum Stock Shares in Offer, such Stock Elections shall be subject to proration as follows: for each such Stock Election, the number of Company Shares that shall be exchanged for the right to receive the Stock Consideration shall be (A) the total number of Stock Election Shares validly tendered and not validly withdrawn in the Offer multiplied by (B) the Offer Stock Proration Factor, rounded down to the nearest Company Share. The *Offer Stock Proration Factor* means a fraction (x) the numerator of which shall be the Maximum Stock Shares in Offer and (y) the denominator of which shall be the total number of Stock Election Shares validly tendered and not withdrawn in the Offer. The Stock Election Shares validly withdrawn in the Offer that are not exchanged for the right to receive the Stock Consideration in accordance with this Section 1.1(c)(iii)(4) shall instead be exchanged for the right to receive the Cash Consideration.

(iv) All prorations resulting from either Section 1.1(c)(iii)(2) or Section 1.1(c)(iii)(4) shall be applied on a pro rata basis, such that each holder of Company Shares who validly tenders Cash Election Shares or Stock Election Shares, as applicable, bears its proportionate share of the proration, based on a percentage of the total Cash Election Shares or Stock Election Shares, as applicable, validly tendered in the Offer (and not validly withdrawn) by such holder of Company Shares bears to all Cash Election Shares or Stock Election Shares, as applicable, validly tendered or Stock Election Shares, as applicable, validly tendered in the Offer (and not validly withdrawn) by such holder of Company Shares bears to all Cash Election Shares or Stock Election Shares, as applicable, validly tendered in the Offer (and not validly withdrawn) by all holders of Company Shares.

(v) Each Company Share validly tendered into the Offer (and not validly withdrawn) but which is not the subject of a valid Election (a *No Election Share*) received prior to the expiration of the Offer shall be deemed to be a Mixed Election Share. In no event shall any Mixed Election Shares or No Election Shares be subject to proration pursuant to Section 1.1(c)(iii)(2) or Section 1.1(c)(iii)(4).

(vi) In lieu of any fractional share of Parent Common Stock that otherwise would be issuable pursuant to the Offer, each holder of Company Shares who otherwise would be entitled to receive a fraction of a share of Parent Common Stock pursuant to the Offer (after aggregating all Company Shares validly tendered in the Offer (and not validly withdrawn) by such holder) will be paid an amount in cash (without interest) equal to such fractional part of a share of Parent Common Stock multiplied by the Parent Trading Price, rounded to the nearest cent.

(d) Adjustments to the Offer. The Mixed Consideration, the Cash Consideration, the Stock Consideration, the Maximum Cash Shares in Offer, the Offer Cash Proration Factor, the Maximum Stock Shares in Offer and the Offer Stock Proration Factor shall be adjusted appropriately, without duplication, to reflect the effect of any stock split, reverse stock split, stock dividend (including any dividend or other distribution of securities convertible into Company Common Stock or Parent Common Stock, as applicable), reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to the number of Company Shares or shares of Parent Common Stock outstanding after the date hereof and prior to Purchaser s acceptance for payment of, and payment for, Company Shares that are validly tendered pursuant to the Offer. Nothing in this Section 1.1(d) shall be construed to permit the Company or Parent to take any action with respect to its securities that is prohibited by the

terms of this Agreement.

(e) Expiration and Extension of the Offer.

(i) Unless the Offer is extended pursuant to and in accordance with this Agreement, the Offer shall expire at 5:00 p.m., Eastern Time, on the twenty-third (23rd) business day (for this purpose calculated in accordance with Section 14d-1(g)(3) promulgated under the Exchange Act) after the date the Offer is first commenced (within the meaning of Rule 14d-2 promulgated under the Exchange Act). In the event that the Offer is extended pursuant to and in accordance with this Agreement, then the Offer shall expire on the date and at the time to which the Offer has been so extended.

(ii) Notwithstanding the provisions of Section 1.1(e)(i) or anything to the contrary set forth in this Agreement:

(A) Purchaser shall (and Parent shall cause Purchaser to) extend the Offer for any period required by any Law, or any rule, regulation, interpretation or position of the SEC or its staff or of the NYSE or NASDAQ, in any such case, which is applicable to the Offer or the Merger, or to the extent necessary to resolve any comments of the SEC or its staff applicable to the Offer or the Offer Documents or the Form S-4;

(**B**) in the event that any of the conditions to the Offer (other than the Minimum Condition) have not been satisfied or waived as of any then-scheduled expiration of the Offer, Purchaser shall (and Parent shall cause Purchaser to) extend the Offer for one or more consecutive increments of not more than ten (10) business days each (or for such longer period as may be agreed by the Company) in order to permit the satisfaction of such conditions to the Offer (other than the Minimum Condition);

(C) if as of any then-scheduled expiration of the Offer each condition to the Offer (other than the Minimum Condition and those conditions that by their nature are to be satisfied by action to be taken at the relevant expiration date of the Offer) shall have been satisfied or waived and the Minimum Condition shall not have been satisfied, Purchaser may, and, at the written request of the Company, Purchaser shall (and Parent shall cause Purchaser to), extend the Offer for one or more consecutive increments of ten (10) business days each (or for such shorter or longer period as may be agreed by the Company); *provided*, that in no event shall Purchaser be required to extend the expiration date of the Offer more than two times pursuant to this clause (C); and

(**D**) Purchaser may extend the Offer for any period necessary to satisfy the notice requirements contained in Section 3-106.1(e)(1) of the MGCL (the *Maryland Short Form Merger Notice*);

provided, however, that (1) any such extension shall not be deemed to impair, limit, or otherwise restrict in any manner the right of the Parties to terminate this Agreement pursuant to the terms of Article IX (subject, in the case of Section 9.1(h), to Purchaser s rights and obligations to extend the Offer pursuant to this Section 1.1(e)(ii)), and (2) Purchaser shall not be required to extend the Offer beyond the Outside Date.

(iii) Neither Parent nor Purchaser shall extend the Offer or provide a subsequent offering period (within the meaning of Rule 14d-11 promulgated under the Exchange Act) in any manner other than pursuant to and in accordance with the provisions of Section 1.1(e)(ii) without the prior written consent of the Company.

(iv) Neither Parent nor Purchaser shall terminate or withdraw the Offer prior to the then-scheduled expiration of the Offer unless this Agreement is validly terminated in accordance with Article IX, in which case Purchaser shall (and Parent shall cause Purchaser to) irrevocably and unconditionally terminate the Offer promptly (but in no event more than one (1) business day) after such termination of this Agreement.

(f) Payment for Company Shares. On the terms of and subject to the conditions set forth in this Agreement and the Offer, Purchaser shall (and Parent shall cause Purchaser to) accept for payment, and pay for, all Company Shares that are validly tendered and not validly withdrawn pursuant to the Offer promptly (within the meaning of Section 14e-1(c) promulgated under the Exchange Act) after the expiration of the Offer (as it may

A-5

be extended in accordance with Section 1.1(e)(ii)) (or, at Parent s election, concurrently with the expiration of the Offer if all conditions to the Offer have been satisfied or waived) (such time of acceptance, the *Acceptance Time*). Without limiting the generality of the foregoing, Parent shall provide or cause to be provided to Purchaser on a timely basis the funds and shares of Parent Common Stock necessary to pay for any Company Shares that Purchaser becomes obligated to purchase pursuant to the Offer; *provided, however*, that without the prior written consent of the Company, Purchaser shall not accept for payment, or pay for, any Company Shares if, as a result, Purchaser would acquire less than the number of Company Shares necessary to satisfy the Minimum Condition. The consideration in the Offer payable in respect of each Company Share validly tendered and not validly withdrawn pursuant to the Offer shall be paid net to the holder thereof in cash or shares of Parent Common Stock, as applicable, without interest and subject to reduction for any applicable withholding Taxes payable in respect thereof. The Company shall register (and shall use its reasonable best efforts to cause its transfer agent to register) the transfer of Company Shares accepted for payment immediately after the Acceptance Time (such registration to be effective as of the Acceptance Time).

(g) Schedule TO; Offer Documents; Form S-4.

(i) As soon as practicable on the date the Offer is first commenced (within the meaning of Rule 14d-2 promulgated under the Exchange Act), Parent and Purchaser shall:

(1) file with the SEC a Tender Offer Statement on Schedule TO (together with all amendments and supplements thereto, and including all exhibits thereto, the *Schedule TO*) with respect to the Offer, which Schedule TO shall contain as an exhibit the Offer to Purchase and forms of the letter(s) of election and transmittal and summary advertisement, if any, and other customary ancillary documents, in each case, in respect of the Offer (together with all amendments and supplements thereto, the *Offer Documents*);

(2) deliver a copy of the Schedule TO, including all exhibits thereto, to the Company at its principal executive offices in accordance with Rule 14d-3(a) promulgated under the Exchange Act;

(3) give telephonic notice of the information required by Rule 14d-3 promulgated under the Exchange Act, and mail by means of first class mail a copy of the Schedule TO, to NASDAQ in accordance with Rule 14d-3(a) promulgated under the Exchange Act;

(4) subject to the Company s compliance with Section 1.2, cause the Offer Documents to be disseminated to all holders of Company Shares as and to the extent required by the Exchange Act; and

(5) cause the Maryland Short Form Merger Notice to be given to all holders of Company Shares with the Offer Documents, unless the Maryland Short Form Merger Notice has been disseminated to all holders of Company Shares prior to the date the Offer is first commenced.

(ii) Concurrently with the filing of the Offer Documents with the SEC, Parent shall file with the SEC a registration statement on Form S-4 to register under the Securities Act the offer and sale of Parent Common Stock pursuant to the Offer and the Merger and to register the shares of Parent Series H Preferred Stock that will be issued in the Merger (the *Form S-4*). The Form S-4 will include a preliminary prospectus containing the information required under Rule 14d-4(b) promulgated under the Exchange Act.

(iii) The Offer Documents and the Form S-4 shall include the Maryland Short Form Merger Notice (whether as an exhibit or otherwise) if the Maryland Short Form Merger Notice has not been disseminated to all holders of Company Shares prior to the date the Offer is first commenced and may include a description of the determinations, approvals and recommendations of the Company Board of Directors, the Company Special Committee and the Company

described in this Agreement (and a change to such determinations, approvals and recommendations to the extent applicable at the relevant time). Each of the Company and Parent shall use its reasonable best efforts to (A) have the Form S-4 declared effective under the Securities Act as promptly as

practicable after such filing, (B) ensure that the Form S-4 complies as to form in all material respects with the applicable provisions of the Exchange Act or Securities Act, and (C) keep the Form S-4, if the Form S-4 is declared effective by the SEC, effective for so long as necessary to complete the Merger. The Company shall furnish in writing to Parent and Purchaser all information concerning the Company and the Company Subsidiaries (including financial information) that is required by applicable Law or the rules of the SEC and its staff to be included in the Offer Documents and the Form S-4 so as to enable Parent and Purchaser to comply with their obligations under this Section 1.1(g), including providing such reasonable cooperation and information as may be necessary or appropriate to include any pro forma financial statements (to the extent required) in the Offer Documents and the Form S-4 in accordance with the rules of the SEC and its staff. Parent, Purchaser and the Company shall reasonably cooperate in good faith to determine the information regarding the Company that is necessary to include in the Offer Documents and the Form S-4 in order to satisfy applicable Laws. Each of Parent, Purchaser and the Company shall promptly correct any information provided by it or any of its respective Representatives for use in the Offer Documents and the Form S-4 if and to the extent that such information shall have become false or misleading in any material respect. Parent and Purchaser shall take all steps necessary to cause the Offer Documents and the Form S-4, as so corrected, to be filed with the SEC and to be disseminated to the holders of Company Shares, in each case as and to the extent required by applicable Laws, or by the SEC or its staff or by NASDAQ or the NYSE. Unless the Company Board of Directors or the Company Special Committee has effected a Change of Recommendation to the extent permitted by Sections 6.3(d) and (e), Parent and Purchaser shall provide the Company and its counsel a reasonable opportunity to review and comment on the Offer Documents and the Form S-4 prior to the filing thereof with the SEC, and Parent and Purchaser shall give reasonable and good faith consideration to any comments made by the Company and its counsel (it being understood that the Company and its counsel shall provide any comments thereon as soon as reasonably practicable). Unless the Company Board of Directors or the Company Special Committee has effected a Change of Recommendation to the extent permitted by Sections 6.3(d) and (e), Parent and Purchaser shall provide in writing to the Company and its counsel any and all written comments or other material communications that Parent, Purchaser or their counsel receive from the SEC or its staff with respect to the Offer Documents and the Form S-4 promptly after such receipt, and Parent and Purchaser shall provide the Company and its counsel a reasonable opportunity to participate in the formulation of any response to any such comments of the SEC or its staff (including a reasonable opportunity to review and comment on any such response, to which Parent and Purchaser shall give reasonable and good-faith consideration to any comments made by the Company and its counsel (it being understood that the Company and its counsel shall provide any comments thereon as soon as reasonably practicable)) and to participate in any discussions with the SEC or its staff regarding any such comments. Purchaser and Parent shall as promptly as practicable respond to any comments of the SEC or its staff regarding the Offer Documents or the Form S-4. Parent shall notify the Company promptly of the time when the Form S-4 or any amendment or supplement thereto has become effective and of the issuance of any stop order or suspension of the qualification of the shares of Parent Common Stock issuable in connection with the Offer and the Merger for offering or sale in any jurisdiction. Parent shall also take any other action required to be taken under the Securities Act, the Exchange Act, any applicable foreign or state securities or blue sky Laws and the rules and regulations thereunder in connection with the issuance of the Parent Common Stock in the Offer or the Merger, and the Company shall furnish all information concerning the Company and the holders of the Company Common Stock as may be reasonably requested in connection with any such actions.

1.2 Company Actions.

(a) Company Determinations, Approvals and Recommendations. The Company hereby approves and consents to the Offer upon the terms and subject to the conditions contained in this Agreement and represents and warrants to Parent and Purchaser that, at a meeting duly called and held prior to the date hereof, the Company Board of Directors, upon the unanimous recommendation of the Company Special Committee, has, upon the terms and subject to the conditions set forth herein:

(i) determined that the terms of the Offer, the Merger and the other Transactions are in the best interests of, the Company and its stockholders;

(ii) determined that the Offer, the Merger and the other Transactions are advisable;

(iii) approved the Offer, the Merger and the other Transactions and the execution and delivery by the Company of this Agreement, the performance by the Company of its covenants and agreements contained herein and the consummation of the Offer, the Merger and the other Transactions, upon the terms, and subject to the conditions, contained herein; and

(iv) resolved to make the Company Board Recommendation.

The Company hereby approves and consents to the inclusion of the foregoing determinations and approvals and the Company Board Recommendation in the Offer Documents and the Form S-4; provided that if the Company Board of Directors has effected a Change of Recommendation to the extent permitted by Section 6.3(d) and Section 6.3(e), Parent shall cause the Offer Documents and the Form S-4 to disclose any such event. In addition, the Offer Documents and Form S-4 may include the fairness opinion of the Company s financial advisor referenced in Section 4.20 and any information that may be required to be provided to holders of Company Shares by the MGCL.

(b) Schedule 14D-9. The Company shall (i) file with the SEC, concurrently with the filing by Parent and Purchaser of the Schedule TO, a Solicitation/Recommendation Statement on Schedule 14D-9 pertaining to the Offer (together with all amendments and supplements thereto, and including all exhibits thereto, the Schedule 14D-9) and (ii) cause the Schedule 14D-9 to be mailed to the holders of Company Shares promptly after commencement of the Offer. The Company shall cause the Schedule 14D-9 to comply as to form in all material respects with requirements of applicable Law. To the extent requested by the Company or Parent, Parent shall cause the initial Schedule 14D-9 to be mailed or otherwise disseminated to the holders of Company Shares (to the extent required by applicable Laws) together with the Offer Documents. Each of Parent and Purchaser shall furnish in writing to the Company all information concerning Parent and Purchaser that is required by applicable Laws to be included in the Schedule 14D-9 so as to enable the Company to comply with its obligations under this Section 1.2(b). Parent, Purchaser and the Company shall cooperate in good faith to determine the information regarding the Company that is necessary to include in the Schedule 14D-9 in order to satisfy applicable Laws. Each of the Company, Parent and Purchaser shall promptly correct any information provided by it or any of its Representatives for use in the Schedule 14D-9 if and to the extent that such information shall have become false or misleading in any material respect. The Company shall take all steps necessary to cause the Schedule 14D-9, as so corrected, to be filed with the SEC and disseminated to the holders of Company Shares, in each case as and to the extent required by applicable Laws. Unless the Company Board of Directors or the Company Special Committee has effected a Change of Recommendation to the extent permitted by Section 6.3(d) and Section 6.3(e), the Company shall provide Parent, Purchaser and their counsel a reasonable opportunity to review and comment on the Schedule 14D-9 prior to the filing thereof with the SEC, and the Company shall give reasonable and good faith consideration to any comments made by Parent, Purchaser and their counsel (it being understood that Parent, Purchaser and their counsel shall provide any comments thereon as soon as reasonably practicable). Unless the Company Board of Directors has effected a Change of Recommendation to the extent permitted by Section 6.3(d) and Section 6.3(e), the Company shall provide in writing to Parent, Purchaser and their counsel any written comments or other material communications the Company or its counsel receive from the SEC or its staff with respect to the Schedule 14D-9 promptly after such receipt, and the Company shall provide Parent, Purchaser and their counsel a reasonable opportunity to participate in the formulation of any response to any such comments of the SEC or its staff (including a reasonable opportunity to review and comment on any such response, to which the Company shall give reasonable and good-faith consideration to any comments made by Parent, Purchaser and their counsel (it being understood that Parent, Purchaser and their counsel shall provide any comments thereon as soon as reasonably practicable)) and to participate in any discussions with the SEC or its staff regarding any such comments. The Company shall as promptly as practicable respond to any comments of the SEC or its staff with respect to the Schedule 14D-9. Unless the Company Board of Directors has effected a Change of Recommendation to

the extent permitted by Section 6.3(d) and Section 6.3(e), the Company shall include the Company Board

Recommendation in the Schedule 14D-9. The Schedule 14D-9 shall include the fairness opinion of the Company s financial advisor referenced in Section 4.20 and any information that may be required to be provided to holders of Company Shares by the MGCL.

(c) Company Information. In connection with the Offer and the Merger (including to enable Parent to disseminate the Maryland Short Form Merger Notice promptly after the date hereof), the Company shall, or shall use reasonable best efforts to cause its transfer agent to, promptly furnish Parent and Purchaser with such assistance and such information as Parent or its agents may reasonably request in order to disseminate and otherwise communicate the Offer, the Merger and the Maryland Short Form Merger Notice to the record and beneficial holders of Company Shares, including a list, as of the most recent practicable date, of the stockholders of the Company, mailing labels and any available listing or computer files containing the names and addresses of all record and beneficial holders of Company Shares, and lists of security positions of Company Shares held in stock depositories (including lists of stockholders, mailing labels, listings or files of securities positions), and shall promptly furnish Parent and Purchaser with such additional information and assistance (including updated lists of the record and beneficial holders of Shares of Company Common Stock, mailing labels and lists of security positions) as Parent and Purchaser or their agents may reasonably request in order to communicate the Offer, the Merger and the Maryland Short Form Merger Notice to the holders of Company Shares. Subject to applicable Laws, and except for such steps as are necessary to consummate the Offer and the Merger, Parent and Purchaser (and their respective agents) shall:

(i) hold in confidence the information contained in any such lists of stockholders, mailing labels and listings or files of securities positions;

(ii) use such information only in connection with the Offer and the Merger; and

(iii) if this Agreement shall be terminated pursuant to Article IX, promptly return to the Company or destroy (and shall use their respective reasonable efforts to cause their agents to deliver or destroy) any and all copies and any extracts or summaries from such information then in their possession or control.

ARTICLE II

THE MERGER

2.1 The Merger; Effects of the Merger. Promptly following the Acceptance Time, the Parties shall take all necessary and appropriate actions to cause the Merger to become effective as promptly as reasonably practicable without a meeting of the stockholders of the Company as permitted by Section 3-106.1 of the MGCL or otherwise as provided in this Article II. Without limiting the foregoing, the Merger shall be governed by Section 3-106.1 of the MGCL and the Merger shall be effected at the Closing in accordance herewith. Upon the terms and subject to the satisfaction or waiver of the conditions set forth in this Agreement, and in accordance with the MGCL, the Company shall be merged with and into Purchaser, whereupon the separate existence of the Company will cease, with Purchaser surviving the Merger (Purchaser, as the surviving corporation of the Merger, sometimes being referred to herein as the

Surviving Corporation). As a result of the Merger, the Surviving Corporation shall be a wholly owned Parent Subsidiary. The Merger shall have the effects set forth in this Agreement and specified in the applicable provisions of the MGCL. Without limiting the generality of the foregoing and subject thereto, at the Effective Time, all the assets, property, rights, privileges, immunities, powers and franchises of the Company and Purchaser shall transfer to, vest in and devolve on the Surviving Corporation and all debts, obligations, liabilities and duties of the Company and Purchaser shall become the debts, obligations, liabilities and duties of the Surviving Corporation.

2.2 The Closing. The closing of the Merger (the *Closing*) will take place at 8:00 a.m., Eastern Time, at the offices of Wachtell, Lipton, Rosen & Katz, 51 West 52nd Street, New York, New York 10019, as promptly

as practicable following the Acceptance Time, and in any case no later than the second (2nd) business day after the satisfaction or, to the extent permitted by applicable Law, waiver of the last of the conditions set forth in Article VIII to be satisfied or waived (other than any such conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or, to the extent permitted by applicable Law and this Agreement, waiver of such conditions at the Closing), unless another date or place is agreed to in writing by the Company and Parent; provided, that, for the avoidance of doubt, the Closing shall in any event not occur earlier than thirty (30) days following the date of dissemination of the Maryland Short Form Merger Notice as provided in this Agreement. The date on which the Closing actually takes place is referred to as the *Closing Date*.

2.3 Effective Time. At the Closing, the Parties shall cause the Merger to be consummated by filing with the State Department of Assessments and Taxation of Maryland (the *Maryland Department*) articles of merger (the *Articles of Merger*) in connection with the Merger, in such form as is required by, and executed in accordance with, the MGCL, and the Parties shall make all other filings or recordings required under the MGCL in connection with the Merger. The Merger shall become effective at the time the Articles of Merger are accepted for record by the Maryland Department or such later date and time (not to exceed thirty (30) days after the Articles of Merger are accepted for record by the Maryland Department) as shall be agreed to by the Company and Parent and specified in the Articles of Merger (the date and time the Merger becomes effective being the *Effective Time*), it being understood and agreed that the Parties shall cause the Effective Time to occur on the Closing Date.

2.4 Governing Documents. At the Effective Time, subject to Section 7.4, the charter and bylaws of Purchaser, as in effect immediately prior to the Effective Time, shall be the charter and bylaws of the Surviving Corporation, until thereafter amended in accordance with applicable Law and the applicable provisions of such charter and bylaws.

2.5 Directors and Officers of the Surviving Corporation.

(a) The directors of Purchaser immediately prior to the Effective Time, from and after the Effective Time, shall be the directors of the Surviving Corporation.

(b) The officers of Purchaser immediately prior to the Effective Time, from and after the Effective Time, shall be the officers of the Surviving Corporation.

2.6 Tax Consequences.

(a) It is intended that, for U.S. federal income tax purposes, (a) the Offer and the Merger, taken together, shall qualify as a reorganization within the meaning of Section 368(a) of the Code and (b) this Agreement be, and is hereby adopted as, a plan of reorganization for purposes of Sections 354 and 361 of the Code and within the meaning of Section 1.368-2(g) of the Treasury Regulations.

(b) The parties acknowledge and agree that for purposes of determining the value of Parent Common Stock to be received by Company Stockholders pursuant to the transactions contemplated by this Agreement under Revenue Procedure 2018-12, 2018-6 IRB 349 (Rev. Proc. 2018-12), (i) the Safe Harbor Valuation Method will be the Average of the Daily Volume Weighted Average Prices as described in Section 4.01(1) of Revenue Procedure 2018-12; (ii) the

Measuring Period (within the meaning of Section 4.01 of Rev. Proc. 2018-12) will be the ten (10) consecutive trading days ending on April 30, 2018; (iii) the specified exchange (within the meaning of Section 3.01(4) of Rev. Proc. 2018-12) will be the New York Stock Exchange; and (iv) the authoritative reporting source within the meaning of Section 3.01(4) of Rev. Proc. 2018-12) will be Bloomberg Finance L.P.

A-10

ARTICLE III

TREATMENT OF SECURITIES

3.1 Treatment of Capital Stock.

(a) **Treatment of Company Common Stock**. At the Effective Time, by virtue of the Merger and without any action on the part of the Parties or holders of any securities of the Company or of Purchaser, subject to any applicable withholding Tax, each Company Share issued and outstanding immediately prior to the Effective Time (other than any Cancelled Shares) shall be automatically converted into the right to receive, without interest, at the election of the holder: (i) the Cash Consideration, (ii) the Stock Consideration, or (iii) the Mixed Consideration (in each case, the

Common Merger Consideration), in each case subject to proration as set forth in Section 3.1(e) and the other provisions of this Article III. From and after the Effective Time, all such Company Shares shall no longer be outstanding and shall automatically be cancelled and shall cease to exist, and each applicable holder of such Company Shares shall cease to have any rights with respect thereto, except the right to receive the applicable portion of Common Merger Consideration therefor upon the surrender of such Company Shares in accordance with Section 3.2, including the right to receive, pursuant to Section 3.6, cash in lieu of fractional shares of Parent Common Stock, if any, into which such Company Shares have been converted pursuant to this Section 3.1(a) (the *Fractional Share Consideration*), together with the amounts, if any, payable pursuant to Section 3.2(f).

(b) **Treatment of Certain Company Common Stock**. At the Effective Time, each Company Share issued and outstanding immediately prior to the Effective Time that is owned by Parent or Purchaser (including Company Shares accepted for payment by Purchaser pursuant to the Offer) shall be cancelled and shall cease to exist, and no consideration shall be delivered in exchange therefor (collectively, the *Cancelled Shares*).

(c) Treatment of Purchaser Shares. At the Effective Time, each issued and outstanding share of common stock, par value \$0.001 per share, of Purchaser issued and outstanding immediately prior to the Effective Time shall remain outstanding as shares of common stock of the Surviving Corporation.

(d) **Treatment of Company Preferred Stock**. At the Effective Time, by virtue of the Merger and without any action on the part of the Parties or holders of any securities of the Company or of Purchaser, subject to any applicable withholding Tax, each share of the Company s 8.125% Series A Cumulative Redeemable Preferred Stock, \$0.01 par value per share (the *Company Preferred Stock* or such shares, the *Company Preferred Shares*) issued and outstanding immediately prior to the Effective Time shall be automatically converted into the right to receive one newly issued share of Parent Series H Preferred Stock (the *Preferred Merger Consideration*), subject to the provisions of this Article III. From and after the Effective Time, all such Company Preferred Shares shall no longer be outstanding and shall automatically be cancelled and shall cease to exist, and each applicable holder of such Company Preferred Merger Consideration therefor upon the surrender of such Company Preferred Shares in accordance with Section 3.2.

(e) Elections; Proration.

(i) Subject to the other provisions of this Section 3.1(e), each holder of Company Shares as of immediately prior to the consummation of the Merger (other than Cancelled Shares) shall be entitled to elect (1) a number of Cash Election Shares, (2) a number of Stock Election Shares, and (3) a number of Mixed Election Shares. Any Cash Election, Stock Election or Mixed Election shall be made on a form for that purpose prepared by Parent that is reasonably acceptable to the Company (a *Form of Election in Merger*), which shall be mailed to such holders of Company Shares promptly after the Closing Date (such date, the *Mailing Date*). The deadline for submitting an effective, properly completed

Form of Election in Merger to the Exchange Agent shall be 5:00 p.m., Eastern Time, on the 20th day following the Mailing Date (or such other time and date as the

Parties may mutually agree) (the *Election Deadline*). Holders of record of Company Shares who hold such Company Shares as nominees, trustees or in other representative capacities may submit multiple Forms of Election in Merger on behalf of their respective beneficial holders.

(ii) Any election shall have been properly made only if the Exchange Agent shall have actually received a properly completed Form of Election in Merger by the Election Deadline. Any Form of Election in Merger may be revoked or changed by the authorized Person properly submitting such Form of Election in Merger, by written notice received by the Exchange Agent prior to the Election Deadline. In the event a Form of Election in Merger is revoked prior to the Election Deadline, the Company Shares represented by such Form of Election in Merger shall become Mixed Election Shares, except to the extent a subsequent election is properly made with respect to any or all of such Company Shares prior to the Election Deadline. Subject to the terms of this Agreement and of the Form of Election in Merger, the Exchange Agent shall have reasonable discretion to determine whether any election in Merger, and any good-faith decisions of the Exchange Agent regarding such matters shall be binding and conclusive. None of Parent, Purchaser, the Company or the Exchange Agent shall be under any obligation to notify any Person of any defect in a Form of Election in Merger.

(iii) Notwithstanding any other provision contained in this Agreement, the Cash Election Shares and Stock Election Shares eligible to receive Common Merger Consideration pursuant to Section 3.1(a) shall be subject to proration as described in this Section 3.1(e).

(1) The maximum number of Company Shares which shall be eligible to receive the Cash Consideration pursuant to Section 3.1(a) shall equal 50% of the aggregate number of Company Shares entitled to receive Common Merger Consideration pursuant to Section 3.1(a) (excluding Mixed Election Shares and No Election Shares) (the *Maximum Cash Shares in Merger*).

(2) If the total number of Cash Election Shares eligible to receive Common Merger Consideration pursuant to Section 3.1(a) exceeds the Maximum Cash Shares in Merger, such Cash Elections shall be subject to proration as follows: for each such Cash Election, the number of Company Shares that shall be converted into the right to receive the Cash Consideration shall be (A) the total number of Cash Election Shares eligible to receive Common Merger Consideration pursuant to Section 3.1(a) multiplied by (B) the Merger Cash Proration Factor, rounded down to the nearest Company Share. The *Merger Cash Proration Factor* means a fraction (x) the numerator of which shall be the Maximum Cash Shares in Merger and (y) the denominator of which shall be the aggregate number of Cash Election Shares eligible to receive Common Merger Consideration pursuant to Section 3.1(a). The Cash Election Shares that were not converted into the right to receive the Cash Consideration in accordance with this Section 3.1(e)(iii)(2) shall be converted into the right to receive the Stock Consideration.

(3) The maximum number of Company Shares which shall be eligible to receive the Stock Consideration pursuant to Section 3.1(a) shall equal 50% of the aggregate number of Company Shares entitled to receive Common Merger Consideration pursuant to Section 3.1(a) (excluding Mixed Election Shares and No Election Shares) (the *Maximum Stock Shares in Merger*).

(4) If the total number of Stock Election Shares eligible to receive Common Merger Consideration pursuant to Section 3.1(a) exceeds the Maximum Stock Shares in Merger, such Stock Elections shall be subject to proration as follows: For each such Stock Election, the number of Company Shares that shall be converted into the right to receive the Stock Consideration shall be (A) the total number of Stock Election Shares eligible to receive Common Merger Consideration pursuant to Section 3.1(a) multiplied by (B) the Merger Stock Proration Factor, rounded down to the nearest Company Share. The *Merger Stock Proration Factor* means a fraction (x) the numerator of which shall be the

Maximum Stock Shares in Merger and (y) the denominator of which shall be the aggregate number of Stock Election Shares eligible to receive Common

Merger Consideration pursuant to Section 3.1(a). The Stock Election Shares that were not converted into the right to receive the Stock Consideration in accordance with this Section 3.1(e)(iii)(4) shall be converted into the right to receive the Cash Consideration.

(iv) All prorations resulting from either Section 3.1(c)(iii)(2) or Section 3.1(c)(iii)(4) shall be applied on a pro rata basis, such that each holder of Company Shares eligible to receive Common Merger Consideration pursuant to Section 3.1(a) bears its proportionate share of the proration, based on a percentage of the total Cash Election Shares or Stock Election Shares, as applicable, elected by such holder of Company Shares bears to all Cash Election Shares or Stock Election Shares, as applicable, elected by holders of Company Shares in the Merger.

(v) Each Company Share eligible to receive the Common Merger Consideration pursuant to Section 3.1(a) for which an election shall not have been properly made by the Election Deadline shall be deemed to be a Mixed Election Share and shall only be entitled to the right to receive the Mixed Consideration in accordance with Section 3.1(a). In no event shall any Mixed Election Shares be subject to proration pursuant to Section 3.1(e)(iii)(2) or Section 3.1(e)(iii)(4).

(f) Adjustment to Merger Consideration. The Merger Consideration, Maximum Cash Shares in Merger, Merger Cash Proration Factor and Merger Stock Proration Factor shall be adjusted appropriately, without duplication, to reflect the effect of any stock split, reverse stock split, stock dividend (including any dividend or other distribution of securities convertible into Company Common Stock, Company Preferred Stock or Parent Common Stock, as applicable), reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to the number of Company Shares, Company Preferred Shares or shares of Parent Common Stock outstanding after the date hereof and prior to the Effective Time. Nothing in this Section 3.1(f) shall be construed to permit the Company or Parent to take any action with respect to its securities that is prohibited by the terms of this Agreement. For the avoidance of doubt, for Tax purposes, no adjustment shall be made to the Merger Consideration, Maximum Cash Shares in Merger, Merger Cash Proration Factor or Merger Stock Proration Factor to reflect the effect of any cash distribution permitted to be made by the Company under this Agreement prior to the Effective Time, including any cash dividend paid pursuant to Section 6.4(a).

3.2 Payment for Securities; Surrender of Certificates.

(a) Exchange Fund. Prior to the Effective Time, Parent or Purchaser shall designate a bank or trust company reasonably acceptable to the Company to act as the exchange agent in connection with the Merger (the Exchange Agent). The Exchange Agent shall also act as the agent for the holders of Company Shares and Company Preferred Shares for the purpose of receiving and holding their Certificates and Book-Entry Shares and shall obtain no rights or interests in the shares represented thereby. At or immediately after the Effective Time, Parent or Purchaser shall deposit, or cause to be deposited, with the Exchange Agent (i) evidence of Parent Common Stock and Parent Series H Preferred Stock issuable pursuant to Section 3.1(a) and Section 3.1(d), respectively, in book-entry form equal to the aggregate Parent Common Stock portion of the Common Merger Consideration (excluding any Fractional Share Consideration) and the aggregate Preferred Merger Consideration, respectively, and (ii) cash in immediately available funds in an amount sufficient to pay the aggregate cash portion of the Merger Consideration, Fractional Share Consideration and any dividends under Section 3.2(f) (such evidence of book-entry shares of Parent Common Stock and Parent Series H Preferred Stock and cash amounts, together with any dividends or other distributions with respect thereto, the *Exchange Fund*), in each case, for the sole benefit of the holders of Company Shares and Company Preferred Shares. In the event the Exchange Fund shall be insufficient (other than as a result of payment of the Merger Consideration in accordance with this Agreement) to pay the aggregate cash portion of the Merger Consideration, Fractional Share Consideration and any dividends under Section 3.2(f), Parent shall, or shall cause the Surviving Corporation to, promptly deposit, or cause to be deposited, additional funds with the Exchange Agent in an amount

which is equal to the shortfall that is required to make such payment. Parent shall cause the Exchange Agent to make, and the Exchange Agent shall make, delivery of the Merger Consideration, including payment of the Fractional Share

Consideration, and any amounts payable in respect of dividends or other distributions on shares of Parent Common Stock in accordance with Section 3.2(f) out of the Exchange Fund in accordance with this Agreement. The Exchange Fund shall not be used for any purpose that is not expressly provided for in this Agreement. The cash portion of the Exchange Fund shall be invested by the Exchange Agent as reasonably directed by Parent; *provided, however*, that any investment of such cash shall in all events be limited to direct short-term obligations of, or short-term obligations fully guaranteed as to principal and interest by, the U.S. government, in commercial paper rated P-1 or A-1 or better by Moody s Investors Service, Inc. or Standard & Poor s Corporation, respectively, or in certificates of deposit, bank repurchase agreements or banker s acceptances of commercial banks with capital exceeding \$10 billion (based on the most recent financial statements of such bank that are then publicly available), and that no such investment or loss thereon shall affect the amounts payable to holders of Certificates or Book-Entry Shares pursuant to this Article III. Any interest and other income resulting from such investments shall be paid to the Surviving Corporation on the earlier of (A) one (1) year after the Effective Time or (B) the full payment of the Exchange Fund.

(b) Procedures for Surrender. Promptly after the Effective Time, Parent shall cause the Exchange Agent to mail (and make available for collection by hand) to each holder of record of a certificate or certificates which immediately prior to the Effective Time represented outstanding Company Shares or Company Preferred Shares (as applicable the Certificates), or non-certificated Company Shares or Company Preferred Shares represented by book-entry (as applicable, **Book-Entry Shares**) and whose Certificates and Book-Entry Shares, as applicable, were converted pursuant to Section 3.1 into the right to receive the applicable Merger Consideration (i) a letter of transmittal, which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon delivery of the Certificates (or affidavits of loss in lieu thereof and, if required by Parent, an indemnity bond) to the Exchange Agent and shall be in such form and have such other provisions as Parent may reasonably specify (it being understood that the forms of Letter of Transmittal to be mailed to the holders of Company Common Stock and Company Preferred Stock may vary in certain respects due to differences in the respective securities) and (ii) instructions for effecting the surrender of the Certificates (or affidavits of loss in lieu thereof and, if required by Parent, an indemnity bond) or Book-Entry Shares in exchange for payment of the applicable Merger Consideration into which such Company Shares or Company Preferred Shares, as applicable, have been converted pursuant to Section 3.1, including any amount payable in respect of Fractional Share Consideration in accordance with Section 3.6, and any dividends or other distributions on shares of Parent Common Stock in accordance with Section 3.2(f). Upon surrender of a Certificate (or an affidavit of loss in lieu thereof and, if required by Parent, an indemnity bond) or Book-Entry Share for cancellation to the Exchange Agent or to such other agent or agents as may be appointed by Parent or the Surviving Corporation, together with such letter of transmittal duly completed and validly executed in accordance with the instructions thereto, and such other documents as may be required pursuant to such instructions, the holder of such Certificate or Book-Entry Share shall be entitled to receive in exchange therefor the applicable Merger Consideration pursuant to the provisions of this Article III, any Fractional Share Consideration that such holder has the right to receive pursuant to the provisions of Section 3.6, and any amounts that such holder has the right to receive in respect of dividends or other distributions on shares of Parent Common Stock in accordance with Section 3.2(f) for each Company Share formerly represented by such Certificate or Book-Entry Share, to be mailed (or made available for collection by hand if so elected by the surrendering holder) within five (5) business days following the later to occur of (x) the Election Deadline and the determination of proration pursuant to Section 3.1(e) or (y) the Exchange Agent s receipt of such Certificate (or affidavit of loss in lieu thereof and, if required by Parent, an indemnity bond) or Book-Entry Share, and the Certificate (or affidavit of loss in lieu thereof and, if required by Parent, an indemnity bond) or Book-Entry Share so surrendered shall be forthwith cancelled. The Exchange Agent shall accept such Certificates (or affidavits of loss in lieu thereof and, if required by Parent, an indemnity bond) or Book-Entry Shares upon compliance with such reasonable terms and conditions as the Exchange Agent may impose to effect an orderly exchange thereof in accordance with normal exchange practices. Notwithstanding anything herein to the contrary, at Parent s election, Parent may instruct the Exchange Agent to automatically convert Book-Entry Shares into the applicable Merger Consideration without any required action on the part of the holders of such Book-Entry Shares. If

payment of the applicable Merger Consideration is to be made to a Person other than the Person in whose name

the surrendered Certificate is registered, it shall be a condition precedent of payment that (A) the Certificate so surrendered shall be properly endorsed or shall be otherwise in proper form for transfer and (B) the Person requesting such payment shall have paid any transfer and other similar Taxes required by reason of the payment of the applicable Merger Consideration to a Person other than the registered holder of the Certificate surrendered or shall have established to the satisfaction of Parent that such Tax either has been paid or is not required to be paid. Payment of the applicable Merger Consideration with respect to Book-Entry Shares shall only be made to the Person in whose name such Book-Entry Shares are registered. Until surrendered as contemplated by this Section 3.2, each Certificate and Book-Entry Share shall be deemed at any time after the Effective Time to represent only the right to receive the applicable Merger Consideration in accordance with Section 3.6, and any dividends or other distributions on shares of Parent Common Stock in accordance with Section 3.2(f), in each case without interest thereon.

(c) Transfer Books; No Further Ownership Rights in Company Shares or Company Preferred Shares. At the Effective Time, the stock transfer books of the Company shall be closed and thereafter there shall be no further registration of transfers of Company Shares or Company Preferred Shares on the records of the Company. From and after the Effective Time, the holders of Certificates or Book-Entry Shares outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such Company Shares or Company Preferred Shares, as applicable, except as otherwise provided for herein. If, after the Effective Time, Certificates or Book-Entry Shares are presented to the Surviving Corporation for any reason, they shall be cancelled and exchanged as provided in this Agreement.

(d) Termination of Exchange Fund; No Liability. At any time following the first (1st) anniversary of the Effective Time, Parent shall be entitled to require the Exchange Agent to deliver to it any funds (including any interest received with respect thereto) remaining in the Exchange Fund that have not been disbursed, or for which disbursement to holders of Certificates or Book-Entry Shares is pending subject only to the Exchange Agent s routine administrative procedures, and thereafter such holders shall be entitled to look only to the Surviving Corporation and Parent (subject to abandoned property, escheat or other similar Laws) as general creditors thereof with respect to the applicable Merger Consideration, including any amount payable in respect of Fractional Share Consideration in accordance with Section 3.6, and any dividends or other distributions on shares of Parent Common Stock in accordance with Section 3.2(f), payable upon due surrender of their Certificates or Book-Entry Share for any Corporation, Parent or the Exchange Agent shall be liable to any holder of a Certificate or Book-Entry Share for any Merger Consideration or other amounts delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law.

(e) Lost, Stolen or Destroyed Certificates. In the event that any Certificates shall have been lost, stolen or destroyed, the Exchange Agent shall issue in exchange for such lost, stolen or destroyed Certificates, upon the making of an affidavit of that fact by the holder thereof and, if required by Parent, an indemnity bond, the applicable Merger Consideration payable in respect thereof pursuant to Section 3.1 hereof, including any amount payable in respect of Fractional Share Consideration in accordance with Section 3.6, and any dividends or other distributions on shares of Parent Common Stock in accordance with Section 3.2(f).

(f) Dividends or Other Distributions with Respect to Parent Common Stock. No dividends or other distributions with respect to Parent Common Stock with a record date after the Effective Time shall be paid to the holder of any unsurrendered Certificate or Book-Entry Share with respect to the shares of Parent Common Stock issuable hereunder (for the avoidance of doubt, only with respect to the Common Merger Consideration), and all such dividends and other distributions shall be paid by Parent to the Exchange Agent and shall be included in the Exchange Fund, in each case until the surrender of such Certificate or Book-Entry Share (or affidavit of loss in lieu thereof) in accordance with

this Agreement. Subject to applicable Laws, following surrender of any such Certificate or Book-Entry Share (or affidavit of loss in lieu thereof) there shall be paid to the holder thereof, without interest, (i) the amount of dividends or other distributions with a record date after the

Effective Time theretofore paid with respect to such shares of Parent Common Stock to which such holder is entitled pursuant to this Agreement and (ii) at the appropriate payment date, the amount of dividends or other distributions with a record date after the Effective Time but prior to such surrender and with a payment date subsequent to such surrender payable with respect to such shares of Parent Common Stock.

3.3 Dissenters Rights. No dissenters or appraisal rights shall be available with respect to the Merger or the other Transactions.

3.4 Treatment of Company RSUs.

(a) Each Company RSU that is outstanding as of immediately prior to the Effective Time, whether vested or unvested, shall be cancelled and converted into the right to receive the Mixed Consideration in respect of each share of Company Common Stock underlying such Company RSU immediately prior to the Effective Time (the *RSU Consideration*), which amount shall be paid in accordance with Section 3.4(c). Any holder of Company RSUs who becomes entitled to receive a fraction of a share of Parent Common Stock (after aggregating all shares represented by Company RSUs held by such individual) shall receive, in lieu thereof, cash, without interest, in an amount equal to such fractional part of a share of Parent Common Stock, multiplied by the Parent Trading Price, less any applicable Tax withholding, payable in accordance with Section 3.4(c).

(b) Prior to the Effective Time, the Company Board of Directors (or the appropriate committee(s) thereof) shall pass resolutions as are necessary for the treatment of the Company RSUs as contemplated by this Section 3.4.

(c) The Parties will coordinate to cause any cash payments required pursuant to this Section 3.4 to be made no later than twenty-one (21) days after the Effective Time.

3.5 Withholding. Parent, Purchaser, the Surviving Corporation and the Exchange Agent shall be entitled to deduct and withhold from amounts otherwise payable pursuant to this Agreement, any amounts as are required to be withheld or deducted with respect to such payment under the Code, or any other applicable Tax Law. To the extent that amounts are so deducted or withheld and remitted to the appropriate Governmental Entity, such amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction or withholding was made.

3.6 Fractional Shares. No certificate or scrip representing fractional shares of Parent Common Stock shall be issued upon the surrender for exchange of Certificates or Book-Entry Shares, and such fractional share interests shall not entitle the owner thereof to vote or to any other rights of a stockholder of Parent. Notwithstanding any other provision of this Agreement, each holder of Company Shares converted pursuant to the Merger who would otherwise have been entitled to receive a fraction of a share of Parent Common Stock (after aggregating all shares represented by the Certificates and Book-Entry Shares delivered by such holder) shall receive, in lieu thereof, cash, without interest, in an amount equal to such fractional part of a share of Parent Common Stock multiplied by the Parent Trading Price.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as disclosed (a) in the Company SEC Documents filed with or furnished to the SEC since December 31, 2016 (including exhibits and other information incorporated by reference therein) and publicly available prior to the date hereof (but excluding any forward-looking disclosures set forth in any risk factors section, any disclosures in any forward-looking statements section and any other disclosures included therein to the extent they are predictive or

forward-looking in nature) or (b) in the applicable section of the disclosure

A-16

letter delivered by the Company to Parent immediately prior to the execution of this Agreement (the *Company Disclosure Letter*) (it being understood that any information set forth in one section or subsection of the Company Disclosure Letter shall be deemed to apply to and qualify the representation and warranty set forth in this Agreement to which it corresponds in number and, whether or not an explicit reference or cross-reference is made, each other representation and warranty set forth in this Article IV for which it is reasonably apparent on its face that such information is relevant to such other section), the Company represents and warrants to Parent as set forth below:

4.1 Qualification, Organization, Subsidiaries, etc.

(a) The Company is a corporation duly incorporated, validly existing and in good standing under the Laws of the State of Maryland and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted. The Company is gualified to do business and is in good standing as a foreign corporation or other entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so qualified or, where relevant, in good standing, has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Each of the Company Subsidiaries is a legal entity duly organized, validly existing and, where relevant, in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing as a foreign corporation or other entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, validly existing, qualified or, where relevant, in good standing, or to have such power or authority, has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. The Company has filed with the SEC, prior to the date of this Agreement, a complete and accurate copy of the Company Governing Documents as amended to the date hereof. The Company Governing Documents are in full force and effect and the Company is not in violation of either of the Company Governing Documents. The Company has made available to Parent complete and accurate copies of the charter and bylaws, or equivalent organizational or governing documents, of each of the Company s significant subsidiaries (as defined in Regulation S-X promulgated under the Securities Act), each as currently in effect.

(b) Section 4.1(b) of the Company Disclosure Letter sets forth an accurate and complete list of each Company Subsidiary and each Person in which the Company or any Company Subsidiary owns an equity or other economic interest, including a list of each Company Subsidiary that is a qualified REIT subsidiary within the meaning of Section 856(i)(2) of the Code (*Qualified REIT Subsidiary*), or a taxable REIT subsidiary within the meaning of Section 856(l) of the Code (*Taxable REIT Subsidiary*), together with (i) the jurisdiction of incorporation or organization, as the case may be, of each Company Subsidiary, (ii) the type and percentage of interest held, directly or indirectly, by the Company in each Company Subsidiary or in each such Person, (iii) the names and the type of and percentage of interest held by any Person other than the Company or a Company Subsidiary in each Company Subsidiary, and (iv) the classification for United States federal income tax purposes of each Company Subsidiary. All the issued and outstanding shares of capital stock of, or other equity interests in, each Company Subsidiary have been validly issued and are fully paid and nonassessable and are wholly owned, directly or indirectly, by the Company free and clear of all Liens, other than Permitted Liens.

4.2 Capitalization.

(a) The authorized capital stock of the Company consists of 300,000,000 Company Shares and 50,000,000 shares of preferred stock, par value \$0.01 per share, of the Company (the *Company Base Preferred Stock*). As of April 30, 2018 (the *Company Capitalization Date*), (i)(A) 45,797,687 Company Shares were issued and outstanding,

(B) 2,200,000 shares of the Company Base Preferred Stock were issued and outstanding, consisting of 2,200,000 shares of Company Preferred Stock, and (C) no Company Shares were held, directly or

indirectly, by the Company Subsidiaries, and (ii) (A) 26,011 Company Shares were underlying Company RSUs (excluding those held in the nominee account), and (B) 829,383 Company Shares were reserved for issuance pursuant to the Company Equity Plan. All the outstanding Company Shares and Company Preferred Shares are, and all Company Shares reserved for issuance as noted above shall be, when issued in accordance with the respective terms thereof, duly authorized, validly issued, fully paid and non-assessable and free of pre-emptive rights.

(b) Section 4.2(b) of the Company Disclosure Letter sets forth a true and complete list, as of the Company Capitalization Date, of (i) the name of each holder of Company RSUs, (ii) the number of unvested shares of Company Common Stock subject to Company RSUs held by such holder, (iii) the number of deferred shares of Company Common Stock subject to Company RSUs held by such holder (excluding any such shares included in the number of issued and outstanding Company Shares set forth in Section 4.1(a) above), (iv) the number of shares of Company Common Stock accrued as dividends with respect to the Company RSUs held by such holder, (v) the Company Equity Plan under which the Company RSUs held by such holder were granted, and (vi) the vesting schedule with respect to the Company RSUs held by such holder, including any right of acceleration of such vesting schedule.

(c) Except as set forth in Section 4.2(a) and Section 4.2(b) above, as of the date hereof: (i) the Company does not have any shares of capital stock or other equity interests issued or outstanding other than the Company Shares that have become outstanding after the Company Capitalization Date, but were reserved for issuance as set forth in Section 4.2(a) above, and (ii) there are no outstanding subscriptions, options, warrants, puts, calls, exchangeable or convertible securities or other rights, agreements or commitments for the issuance of, or that correspond to, capital stock to which the Company or any of the Company Subsidiaries is a party obligating the Company or any of the Company Subsidiaries to (A) issue, transfer or sell, or make any payment with respect to, any shares in the capital or other equity interests of the Company or any Company Subsidiary or securities convertible into or exchangeable for, or that correspond to, such shares or equity interests (in each case other than to the Company or a wholly owned Company Subsidiary); (B) grant, extend or enter into any such subscription, option, warrant, put, call, exchangeable or convertible securities or other similar right, agreement or commitment; (C) redeem or otherwise acquire any such shares in its capital or other equity interests; or (D) provide any amount of funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any Company Subsidiary that is not wholly owned or in any other Person. Except as set forth in Section 4.2(a) and Section 4.2(b) above, there are no outstanding obligations of the Company or any Company Subsidiary (1) restricting the transfer of, (2) affecting the voting rights of, (3) requiring the repurchase, redemption or disposition of, or containing any right of first refusal or right of first offer with respect to, (4) requiring the registration for sale of, or (5) granting any preemptive or anti-dilutive rights with respect to, any shares of capital stock or other equity interests of the Company or any Company Subsidiary.

(d) Neither the Company nor any Company Subsidiary has outstanding bonds, debentures, notes or other similar obligations, the holders of which have the right to vote (or which are convertible into or exercisable for securities having the right to vote) with the Company Stockholders on any matter.

(e) There are no voting trusts or other agreements or commitments to which the Company or any Company Subsidiary is a party with respect to the voting of the capital stock or other equity interests of the Company or any Company Subsidiary.

(f) Assuming that Parent and Purchaser comply with the provisions of Section 1.1(g)(i)(5) and the terms of the Parent Series H Preferred Stock to be issued to the holders of the Company Preferred Stock pursuant to Section 3.1(d) are as set forth in the articles supplementary in the form set forth in Annex C, no holder of Company Common Stock or Company Preferred Stock has the right to demand as a result of the Merger or otherwise the fair value of their shares in accordance with the MGCL.

A-18

4.3 Corporate Authority.

(a) The Company has the corporate power and authority to execute and deliver this Agreement and to consummate the Transactions, including the Offer and the Merger. The execution and delivery of this Agreement and the consummation of the Transactions have been duly and validly authorized by the Company Board of Directors and no other corporate proceedings on the part of the Company are necessary to authorize the consummation of, and to consummate, the Transactions, except, with respect to the Merger, for the filing of the Articles of Merger with and acceptance for record of the Articles of Merger by the Maryland Department. On or prior to the date hereof, at a meeting duly called and held, the Company Board of Directors, acting upon the unanimous recommendation of the Company Special Committee, has (i) determined that the terms of the Offer, the Merger and the other Transactions are in the best interests of, the Company and its stockholders, (ii) declared the Offer, the Merger and the other Transactions advisable, (iii) approved the execution and delivery by the Company of the Offer, the Merger and the other Transactions upon the terms and subject to the conditions contained herein, and (iv) resolved to recommend that the holders of shares of Company Common Stock accept the Offer and tender their shares of Company Common Stock to Purchaser pursuant to the Offer. As of the date hereof, none of the foregoing actions by the Company Board of Directors have been rescinded or modified in any way.

(b) Assuming the Minimum Condition is satisfied and the terms of the Parent Series H Preferred Stock to be issued to the holders of the Company Preferred Stock pursuant to Section 3.1(d) are as set forth in the articles supplementary in the form set forth in Annex C, no vote of the holders of Company Shares, Company Preferred Shares or other capital stock of the Company is necessary to adopt this Agreement and consummate the Merger or other Transactions under applicable Law and the Company Governing Documents.

(c) This Agreement has been duly and validly executed and delivered by the Company and, assuming this Agreement constitutes the valid and binding agreement of Parent and Purchaser, constitutes the valid and binding agreement of the Company, enforceable against the Company in accordance with its terms, except that (i) such enforcement may be subject to applicable bankruptcy, insolvency, examinership, reorganization, moratorium or other similar Laws, now or hereafter in effect, relating to creditors rights generally and (ii) equitable remedies of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of the court before which any proceeding therefor may be brought (collectively, the *Enforceability Limitations*).

4.4 Governmental Consents; No Violation.

(a) Other than in connection with or in compliance with (i) the provisions of the MGCL, (ii) the Securities Act, (iii) the Exchange Act, (iv) the Investment Advisers Act, (v) any applicable requirements of NASDAQ or FINRA, (vi) state and local transfer Taxes or state securities or blue sky Laws, and (vii) Required Consents and any other filings, notifications, consents or approvals required by any applicable Governmental Entity or Program Lenders as set forth in Schedule 4.4 of the Company Disclosure Letter, no authorization, permit, notification to, consent or approval of, or filing with, any Governmental Entity or Program Lender is necessary or required, under applicable Law, for the consummation by the Company of the Transactions, except for such authorizations, permits, notifications, consents, approvals or filings that, if not obtained or made, would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

(b) The execution and delivery by the Company of this Agreement do not, and, except as described in Section 4.4(a) of the Company Disclosure Letter, the consummation of the Transactions and compliance with the provisions hereof will not (i) conflict with or result in any violation or breach of, or default or change of control (with or without notice or lapse of time, or both) under, or give rise to a right of, or result in, termination, modification, cancellation or

acceleration of any obligation or to the loss of a benefit under, or right of amendment to a third party or vesting of any Contract binding upon the Company or any of the Company

Subsidiaries or result in the creation of any Lien upon any of the properties, rights or assets of the Company or any Company Subsidiaries, other than Permitted Liens, (ii) conflict with or result in any violation of any provision of the Company Governing Documents or the organizational documents of any Company Subsidiary, or (iii) conflict with or violate any Laws applicable to the Company or any of the Company Subsidiaries or any of their respective properties, rights or assets, other than in the case of clauses (i) and (iii), any such violation, conflict, default, termination, cancellation, acceleration, right, loss or Lien that has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

4.5 SEC Reports and Financial Statements.

(a) From December 31, 2015, the Company has timely (including following any extensions of time for filing provided by Rule 12b-25 promulgated under the Exchange Act) filed or furnished all forms, documents and reports required to be filed or furnished by it with the SEC under the Securities Act or the Exchange Act (such forms, documents and reports, the *Company SEC Documents*). As of their respective filing dates, or, if amended prior to the date hereof, as of the date of (and giving effect to) the last such amendment made prior to the date hereof, the Company SEC Documents complied as to form in all material respects with the requirements of the Sarbanes-Oxley Act, the Securities Act and the Exchange Act, as the case may be, and the applicable rules and regulations promulgated thereunder and the listing and corporate governance rules and regulations of NASDAQ, and none of the Company SEC Documents of a material fact or omitted to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

(b) Each of the consolidated financial statements contained or incorporated by reference in the Company SEC Documents (as amended, supplemented or restated, if applicable), including the related notes and schedules, was prepared (except as indicated in the notes thereto including, in the case of interim financial statements, for normal and recurring year-end adjustments and as may be permitted by the SEC on Form 10-Q or Form 8-K under the Exchange Act and absence of all notes thereto) in accordance with United States Generally Accepted Accounting Principles (*GAAP*) applied on a consistent basis throughout the periods indicated, and each such consolidated financial statement presented fairly, in all material respects, the consolidated financial position, results of operations, stockholders equity and cash flows of the Company and its consolidated Subsidiaries as of the respective dates thereof and for the respective periods indicated therein (subject, in the case of unaudited quarterly financial statements, to absence of notes and normal year-end adjustments).

(c) Other than the off-balance sheet financings disclosed in the Company SEC Documents filed or furnished prior to the date hereof or disclosed in Section 6.1 of the Company Disclosure Letter, neither the Company nor any Company Subsidiary is a party to, or has any Contract to become a party to, any joint venture, off-balance sheet partnership or any similar Contract, including any Contract relating to any transaction or relationship between or among the Company or any Company Subsidiary, on the one hand, and any unconsolidated affiliate, including any structured finance, special purpose or limited purpose entity or Person, on the other hand, or any off-balance sheet arrangements (as defined in Item 303(a) of Regulation S-K of the SEC) where the purpose of such Contract is to avoid disclosure of any material transaction involving, or material liabilities of, the Company in the Company s published financial statements or any Company SEC Documents.

(d) To the knowledge of the Company, none of the Company SEC Documents is as of the date hereof the subject of ongoing SEC review and the Company has not received any comments from the SEC with respect to any of the Company SEC Documents which remain unresolved, nor has it received any inquiry or information request from the SEC as of the date hereof as to any matters affecting the Company which has not been adequately addressed. None of the Company SEC Documents as of the date hereof is the subject of any confidential treatment request by the

Company.

4.6 Internal Controls and Procedures. The Company has established and maintains disclosure controls and procedures and internal control over financial reporting (as such terms are defined in paragraphs

A-20

(e) and (f), respectively, of Rule 13a-15 under the Exchange Act) as required by Rule 13a-15 under the Exchange Act. The Company s disclosure controls and procedures are reasonably designed to ensure that all material information required to be disclosed by the Company in the reports that it files or furnishes under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC, and that all such material information is accumulated and communicated to the Company s management as appropriate to allow timely decisions regarding required disclosure and to make the certifications required pursuant to Sections 302 and 906 of the Sarbanes Oxley Act of 2002 (the Sarbanes-Oxley Act). Since December 31, 2015, the Company s principal executive officer and its principal financial officer have disclosed to the Company s auditors and the audit committee of the Company Board of Directors (which disclosure (if any) has been made available to Parent prior to the date hereof) (a) all known significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting that are reasonably likely to adversely affect in any material respect the Company s ability to record, process, summarize and report financial information and (b) any known fraud, whether or not material, that involves management or other employees who have a significant role in the Company s internal controls over financial reporting. Since December 31, 2015, neither the Company nor any Company Subsidiary has received any material, unresolved, written complaint, allegation, assertion or claim regarding the accounting or auditing practices, procedures, methodologies or methods of the Company or the Company Subsidiaries or their respective internal accounting controls. Since the enactment of the Sarbanes-Oxley Act, none of the Company or any Company Subsidiary has made any prohibited loans to any director or executive officer of the Company (as defined in Rule 3b-7 promulgated under the Exchange Act).

4.7 No Undisclosed Liabilities. Neither the Company nor any Company Subsidiary has any liabilities of any nature, whether or not accrued, contingent or otherwise, except (a) as and to the extent disclosed or reserved against in the Company s consolidated balance sheet (or the notes thereto) as of December 31, 2017 included in the Company SEC Documents filed or furnished prior to the date hereof, (b) for off-balance sheet financing as and to the extent specifically disclosed in the Company SEC Documents filed or furnished prior to the date hereof, (b) for off-balance sheet financing as and to the extent specifically disclosed in the Company SEC Documents filed or furnished prior to the date hereof, or liabilities incurred in the ordinary course of business consistent with past practice since December 31, 2017 (other than any liability for any breaches of Contracts), (c) as incurred in connection with the preparation, negotiation and consummation of the Transactions contemplated under this Agreement, and (d) for liabilities which, individually or in the aggregate, have not had and would not reasonably be expected to have a Company Material Adverse Effect.

4.8 Absence of Certain Changes or Events.

(a) From December 31, 2017 through the date of this Agreement, there has not occurred any event, development, occurrence, or change that has had, or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

(b) From December 31, 2017 through the date of this Agreement, except as for events giving rise to and the discussion and negotiation of this Agreement and consideration and negotiation of Acquisition Proposals prior to April 15, 2018, the business of the Company and the Company Subsidiaries has been conducted in all material respects in the ordinary course of business.

(c) From December 31, 2017 through the date of this Agreement, neither the Company nor any Company Subsidiary has taken any action that, if taken after the date of this Agreement, would constitute a breach of, or require consent of Parent under, Sections 6.1(a), (b), (g), (l), (r) or (x) or, as it relates to any of the foregoing, (bb).

4.9 Compliance with Laws; Permits.

(a) The Company and each Company Subsidiary are and have been since December 31, 2015 in compliance with and are not in default under or in violation of any Laws (including Environmental Laws)

applicable to the Company, such Subsidiaries or any of their respective properties or assets, except where such non-compliance, default or violation has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Notwithstanding anything to the contrary in this Section 4.9(a), the provisions of this Section 4.9(a) shall not apply to matters addressed in Section 4.9(c), Section 4.10, Section 4.11 and Section 4.12.

(b) The Company and the Company Subsidiaries are and since December 31, 2015 have been in possession of all grants, authorizations, licenses, permits, exceptions, consents, certificates, approvals, registrations, clearances and orders of any Governmental Entity and/or Program Lender or pursuant to any applicable Law necessary for the Company and the Company Subsidiaries to own, lease and operate their properties and assets or to carry on their businesses as they are now being conducted (the *Company Permits*), except where the failure to have any of the Company Permits has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, all Company Permits are in full force and effect, no default (with or without notice, lapse of time or both) has occurred under any such Company Permit and none of the Company or any Company Subsidiary has received any written notice from any Governmental Entity threatening to suspend, revoke, withdraw or modify any such Company Permit.

(c) Except as has not been and would not reasonably be expected to be, individually or in the aggregate, material to the Company and the Company Subsidiaries, taken as a whole, since December 31, 2015, neither the Company nor the Company Subsidiaries, in connection with the business of the Company or any Company Subsidiary, or, to the knowledge of the Company, any other third party, in each case, acting on behalf of the Company or any Company Subsidiary, have taken any action in violation of the Foreign Corrupt Practices Act of 1977, as amended (the *FCPA*), or any other applicable domestic or foreign anti-bribery or anti-corruption laws (collectively, *Bribery Legislation*).

(d) Except as has not been and would not reasonably be expected to be, individually or in the aggregate, material to the Company and the Company Subsidiaries, taken as a whole, since December 31, 2015, neither the Company nor the Company Subsidiaries have been subject to any actual, pending, or, to the Company sknowledge, threatened civil, criminal, or administrative actions, suits, demands, claims, hearings, notices of violation, investigations, proceedings, demand letters, settlements, or enforcement actions, or made any voluntary disclosures to any Governmental Entity, involving the Company or any Company Subsidiary in any way relating to applicable Bribery Legislation, including the FCPA.

(e) The Company and the Company Subsidiaries have taken all necessary action, including filing all necessary claims, and have met all the necessary requirements to qualify for the relief set forth in the no-action letter (No. 12-44) issued by the Commodity Futures Trading Commission Division of Swap Dealer and Intermediary Oversight on December 7, 2012 (the *No-Action Letter*). Prior to the date hereof, the Company has provided Parent with accurate and complete copies of all claims submitted by the Company or any of the Company Subsidiaries pursuant to the No-Action Letter and any other written communications with respect thereto.

(f) The Operators currently have all Permits which are required for the operation of their respective businesses as presently conducted and as presently intended to be conducted (the *Operator Permits*), other than those the failure of which to possess has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. To the Knowledge of the Company, no Operator is in default or violation, and no event has occurred which, with notice or the lapse of time or both, would constitute a default or violation of any term, condition or provision of any Operator Permit, and there are no facts or circumstances which could form the basis for any such default or violation, in each case except as had not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. To the Knowledge of the Company, there are no

legal proceedings pending or threatened, relating to the suspension,

revocation or modification of any material Operator Permit. To the Knowledge of the Company, none of the Operator Permits will be impaired or in any material way affected by the consummation of the transactions contemplated by this Agreement.

4.10 Employee Benefit Plans.

(a) Section 4.10(a) of the Company Disclosure Letter sets forth each material Company Benefit Plan. For purposes of this Agreement, *Company Benefit Plans* means each employee benefit plan (as defined in Section 3(3) of ERISA), whether or not subject to ERISA, and each bonus, stock, stock option or other equity-based compensation arrangement or plan, incentive, deferred compensation, retirement or supplemental retirement, severance, employment, change-in-control, collective bargaining, profit sharing, pension, vacation, cafeteria, dependent care, medical care, employee assistance program, education or tuition assistance programs, and each insurance and other similar fringe or employee benefit plan, policy, program, agreement or arrangement, in each case, for the benefit of current employees, directors or consultants (or any dependent or beneficiary thereof) of the Company or any Company Subsidiary or any of their ERISA Affiliates or with respect to which the Company or any Company Subsidiary has any obligation or liability including any plan, program, policy, or practice that is sponsored by a professional employer organization (each, a *PEO*) under which an employee of the Company or any of its subsidiaries may be eligible to receive benefits in connection with the Company sengagement of a PEO (each, a

PEO Company Benefit Plan); provided, however, that the Company Benefit Plans shall not include any plans that both (x) do not cover current or former employees of the Company or any of its Subsidiaries and (y) are sponsored or maintained by MTGE Management, LLC or any of its ERISA Affiliates; provided further, however, that the Parties agree that neither the Company nor any of its Subsidiaries shall be deemed an ERISA Affiliate of MTGE Management, LLC for purposes of this Section 4.10(a). Each Company Employee Plan that is not a PEO Company Employee Plan is referred to in this Agreement as a **Sponsored Company Benefit Plan**. With respect to each material Company Benefit Plan (and, in the case of any PEO Company Benefit Plan, to the extent provided to the Company by the PEO and in the possession of the Company), the Company has made available to Parent correct and complete copies of (or, to the extent no such copy exists, a description of), in each case, to the extent applicable, (i) all plan documents, summary plan descriptions, summaries of material modifications, and amendments related to such plans and any related trust agreement; (ii) the most recent Form 5500 Annual Report; (iii) the most recent audited financial statement and actuarial valuation; (iv) all material filings and correspondence with any Governmental Entity; and (v) all material related agreements, insurance contracts and other agreements which implement each such Company Benefit Plan.

(b) Except as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect, each of the Sponsored Company Benefit Plans, and to the knowledge of the Company each PEO Company Benefit Plan, has been operated and administered in compliance with applicable Laws, including, but not limited to, ERISA, the Code and in each case the regulations thereunder. No liability under Title IV of ERISA that would reasonably be expected to be material to the Company and the Company Subsidiaries taken as a whole has been incurred by the Company, the Company Subsidiaries or any of their respective ERISA Affiliates that has not been satisfied in full, and to the knowledge of the Company no condition exists that is likely to cause the Company, the Company Subsidiaries taken as a whole thereunder. Except as would not result in a material liability to the Company all contributions or other amounts payable by the Company or the Company Subsidiaries pursuant to each Company Benefit Plan in respect of current or prior plan years have been timely paid or accrued in accordance with GAAP or applicable international accounting standards. Except as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect, neither the Company nor any of the Company Subsidiaries has engaged in a transaction in connection with which the Company or any of the Company subsidiaries has engaged in a transaction in connection with which the Company or any of the Company subsidiaries has engaged in a transaction in connection with which the Company or 502(i) of ERISA

or a Tax imposed pursuant to Section 4975 or 4976 of the Code. There are no pending, or to the knowledge of the Company, threatened or anticipated claims, actions, investigations or audits (other than routine claims for benefits) by, on behalf of or

against any of the Company Benefit Plans or any trusts related thereto that would reasonably be expected to result in a material liability.

(c) No Company Benefit Plan (i) is subject to Title IV or Section 302 of ERISA or Section 412 or 4971 of the Code, (ii) is a multiemployer plan (as such term is defined in Section 3(37) of ERISA), or (iii) a plan that has two or more contributing sponsors at least two of whom are not under common control, within the meaning of Section 4063 of ERISA.

(d) No Company Benefit Plan provides retiree medical benefits (whether or not insured), with respect to current or former employees or directors of the Company or the Company Subsidiaries following their retirement or other termination of service, other than coverage mandated by the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, or comparable U.S. state Law.

(e) (i) Each of the Sponsored Company Benefit Plans that is intended to be qualified within the meaning of Section 401(a) of the Code has received a favorable determination letter or opinion letter as to its qualification, and (ii) to the knowledge of the Company there are no existing circumstances or any events that have occurred that would reasonably be expected to adversely affect the qualified status of any such plan. Each such favorable determination letter has been provided or made available to Parent.

(f) Except as provided under Section 3.4, neither the execution and delivery of this Agreement nor the consummation of the Transactions (either alone or in conjunction with any other event) will (i) entitle to compensation or any benefit or result in any payment (including severance, unemployment compensation, excess parachute payment (within the meaning of Section 280G of the Code), forgiveness of Indebtedness or otherwise) becoming due to any current or former director or any employee of the Company or any Company Subsidiary under any Company Benefit Plan, (ii) increase any compensation, equity award or any other benefits otherwise payable under any Company Benefit Plan, (iii) result in any acceleration of the time of payment, funding or vesting of any such compensation, equity award or other benefits or trigger any other material obligation under any Company Benefit Plan, or (iv) result in any breach or violation of, or default under or limit the Company s right to amend, modify, terminate or transfer the assets of, any Company Benefit Plan.

(g) Each Sponsored Company Benefit Plan has been maintained and operated in documentary and operational compliance in all materials respects with Section 409A of the Code or an available exemption therefrom.

(h) Neither the Company nor any Company Subsidiary is a party to nor does it have any obligation under any Company Benefit Plan to compensate any person for excise Taxes payable pursuant to Section 4999 of the Code or for additional Taxes payable pursuant to Section 409A of the Code.

(i) This Section 4.10 constitutes the exclusive representations and warranties of the Company with respect to the subject matter set forth in this Section 4.10.

4.11 Tax Matters.

(a) The Company and each Company Subsidiary have (i) duly and timely filed (or there have been filed on their behalf) with the appropriate Governmental Entity all U.S. federal income and all other material Tax Returns required to be filed by them, taking into account any extensions of time within which to file such Tax Returns, and all such Tax Returns are true, correct and complete in all material respects, and (ii) duly and timely paid in full (or there has been duly and timely paid in full on their behalf), or made adequate provision for, all material amounts of Taxes required to be paid by them. True and materially complete copies of all U.S. federal income Tax Returns that have been filed with

the IRS by the Company and each Company Subsidiary with respect to the taxable years ending on or after December 31, 2011 have been provided or made available to representatives of Parent.

(b) The Company (i) for its taxable years commencing with the Company s taxable year that ended on December 31, 2011 and through and including its taxable year ended December 31, 2017 has been subject to taxation as a real estate investment trust within the meaning of and under the provisions of Sections 856 through 860 of the Code (a *REIT*) and has satisfied all requirements to qualify as a REIT in such years; (ii) has operated since January 1, 2018 until the date hereof in a manner consistent with the requirements for qualification and taxation as a REIT; (iii) intends to continue to operate in such a manner as to qualify as a REIT for its taxable year that will end with the Merger; and (iv) has not to its knowledge taken or omitted to take any action that could reasonably be expected to result in a successful challenge by the IRS or any other Governmental Entity to its qualification as a REIT, and to the knowledge of the Company, no such challenge is pending or threatened.

(c) Each Company Subsidiary has been since the later of its acquisition or formation treated for U.S. federal income (and applicable state) Tax purposes as (i) a partnership (or a disregarded entity) and not as a corporation or an association, a publicly traded partnership taxable as a corporation, or a taxable mortgage pool, (ii) a Qualified REIT Subsidiary of the Company, or (iii) a Taxable REIT Subsidiary of the Company.

(d) Neither the Company nor any Company Subsidiary holds, directly or indirectly, any asset the disposition of which would be subject to (or to rules similar to) Section 1374 of the Code.

(e) (i) There are no disputes, audits, examination, investigations or proceedings pending (or threatened in writing), or written claims asserted, for and/or in respect of any material Taxes or material Tax Returns of the Company or any Company Subsidiary and neither the Company nor any Company Subsidiary is a party to any litigation or administrative proceeding relating to Taxes; (ii) no deficiency for Taxes of the Company or any Company Subsidiary has been claimed, proposed or assessed in writing or, to the knowledge of the Company, threatened, by any Governmental Entity, which deficiency has not yet been settled, except for such deficiencies which are being contested in good faith or with respect to which the failure to pay, individually or in the aggregate, has not had and would not reasonably be expected to have a Company Material Adverse Effect; (iii) neither the Company nor any Company Subsidiary has extended or waived (nor granted any extension or waiver of) the limitation period for the assessment or collection of any Tax that has not since expired; (iv) neither the Company nor any Company Subsidiary currently is the beneficiary of any extension of time within which to file any material Tax Return that remains unfiled other than automatic extensions of time obtained in the ordinary course of business; (v) neither the Company nor any Company Subsidiary has received a written claim by any Governmental Entity in any jurisdiction where any of them does not file Tax Returns or pay any Taxes that it is or may be subject to taxation by that jurisdiction, and (vi) neither the Company nor any Company Subsidiary has entered into any closing agreement as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign income Tax Law).

(f) Neither the Company nor any Company Subsidiary has incurred any liability for excise Taxes under Sections 857(b), 860(c) or 4981 of the Code which have not been previously paid. Neither the Company nor any Company Subsidiary (other than a Taxable REIT Subsidiary or a Subsidiary of a Taxable REIT Subsidiary) has engaged at any time in any prohibited transactions within the meaning of Section 857(b)(6) of the Code. Neither the Company nor any Company Subsidiary has engaged in any transaction that would give rise to redetermined rents, redetermined deductions, excess interest or redetermined TRS service income described in Section 857(b)(7) of the Code. To the knowledge of the Company, no event has occurred, and no condition or circumstance exists, which presents a material risk that any material Tax described in the previous sentences will be imposed upon the Company or any Company Subsidiary.

(g) The Company and each Company Subsidiary have complied in all material respects with all applicable Laws relating to the payment and withholding of Taxes (including withholding of Taxes pursuant to Sections 1441, 1442, 1445, 1446, 1471-1474, 3102 and 3402 of the Code or similar provisions under any state and foreign Laws) and have

duly and timely withheld and, in each case, have paid over to the appropriate Governmental Entities any and all material amounts required to be so withheld and paid over on or prior to the due date thereof under all applicable Laws.

(h) There are no material Tax Liens upon any property or assets of the Company or any Company Subsidiary except Liens for Taxes not yet due and payable or that are being contested in good faith by appropriate proceedings and for which adequate reserves have been established in accordance with GAAP.

(i) Neither the Company nor any Company Subsidiary has requested, has received or is subject to any written ruling of a Governmental Entity or has entered into any written agreement with a Governmental Entity with respect to any Taxes that is still in effect.

(j) There are no Tax allocation or sharing agreements or similar arrangements with respect to or involving the Company or any Company Subsidiary, and after the Closing Date neither the Company nor any Company Subsidiary shall be bound by any such Tax allocation agreements or similar arrangements or have any liability thereunder for amounts due in respect of periods prior to the Closing Date, in each case, other than any agreements entered into in the ordinary course of business, the principal purpose of which is unrelated to Taxes.

(k) Neither the Company nor any Company Subsidiary (i) has been a member of an affiliated group filing a consolidated U.S. federal income Tax Return (other than a Company Subsidiary taxable as a Taxable REIT Subsidiary) or (ii) has any liability for the Taxes of any Person (other than the Company or any Company Subsidiary) under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local, or foreign law), as a transferee or successor, or otherwise.

(I) Neither the Company nor any Company Subsidiary has participated in any listed transaction within the meaning of Treasury Regulations Section 1.6011-4(b)(2).

(m) Neither the Company nor any Company Subsidiary (other than Taxable REIT Subsidiaries) has or has had any earnings and profits attributable to such entity or any other corporation in any non-REIT year within the meaning of Section 857 of the Code.

(n) The Company is not aware of any fact or circumstance that would reasonably be expected to prevent the Offer and Merger, taken together, from qualifying as a reorganization within the meaning of Section 368(a) of the Code.

(o) Neither the Company nor any of the Company Subsidiaries has constituted either a distributing corporation or a controlled corporation (within the meaning of Section 355(a) of the Code) in a distribution of stock qualifying for tax-free treatment under Section 355 of the Code in the two (2) years prior to the date of this Agreement.

(**p**) No Company Subsidiary that is a Taxable REIT Subsidiary directly or indirectly (i) operates or manages a lodging facility or a healthcare facility or (ii) provides to any other Person (under a franchise, license or otherwise) rights to any brand name under which any lodging or healthcare facility is operated, in each case, within the meaning of Section 856(1)(3) of the Code.

(q) This Section 4.11 constitutes the exclusive representations and warranties of the Company with respect to Tax matters.

4.12 Labor Matters.

(a) Neither the Company nor any Company Subsidiary is a party to, or bound by, any collective bargaining agreement or other Contract with a labor union or labor organization. Neither the Company nor any Company Subsidiary is (or has during the past two years been) subject to a material labor dispute, strike or work stoppage. There are no organizational efforts with respect to the formation of a collective bargaining unit presently being made or, to the

knowledge of the Company, threatened involving employees of the Company or any Company Subsidiary.

(b) Except as has not had and not would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, the Company and each Company Subsidiary are and have been since December 31, 2015 in compliance with all applicable Laws respecting labor, employment, immigration, fair employment practices, terms and conditions of employment, workers compensation, occupational safety, plant closings, mass layoffs, worker classification, exempt and non-exempt status, compensation and benefits, wages and hours and the Worker Adjustment and Retraining Notification Act of 1988, as amended, within the six (6) months prior to the date of this Agreement that remains unsatisfied.

(c) This Section 4.12 constitutes the exclusive representations and warranties of the Company with respect to the matters set forth in this Section 4.12.

4.13 Investigation; Litigation. (a) To the knowledge of the Company, there is no investigation or review pending or threatened by any Governmental Entity with respect to the Company or any Company Subsidiary or any of their respective properties, rights or assets, and (b) there are no claims, actions, suits or proceedings pending (or, to the knowledge of the Company, threatened) against the Company or any Company Subsidiary or any of their respective properties, rights or assets before any Governmental Entity, which, in the case of clause (a) or (b), would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Neither the Company nor any Company Subsidiary is subject to any outstanding order, judgment or decree that would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

4.14 Intellectual Property. Except as, individually or in the aggregate, has not had and would not reasonably be expected to have a Company Material Adverse Effect, (i) the Company or the Company Subsidiaries own or are licensed or otherwise possess valid rights to use all Intellectual Property used in the conduct the business of the Company and the Company Subsidiaries as it is currently conducted, (ii) to the knowledge of the Company, the conduct of the business of the Company and the Company and the Company Subsidiaries as it is currently rights of any Person, (iii) there are no pending or, to the knowledge of the Company, threatened claims with respect to any of the Intellectual Property rights owned by the Company or any Company Subsidiary, and (iv) to the knowledge of the Company, no Person is currently infringing or misappropriating Intellectual Property owned by the Company or any Company Subsidiaries are taking all actions that are reasonably necessary to maintain and protect each material item of Intellectual Property that they own.

4.15 Real Property.

(a) Section 4.15(a) of the Company Disclosure Letter sets forth a complete list as of the date hereof of: (i) all real property and interests in real property owned in fee by the Company and the Company Subsidiaries (individually, a *Healthcare Owned Property* and collectively, the *Healthcare Owned Properties*); (ii) all real property and interests in real property leased by the Company and the Company Subsidiaries as lessee (individually, a *Real Property Lease* and collectively, the *Real Property Leases* and, together with the Healthcare Owned Properties, being referred to herein individually as a *Company Property* and collectively as the *Company Properties*); and (iii) the Operator Leases, including the name of the third party lessor under such Real Property Lease or lessee under such Operator Lease and the date of such Real Property Lease or Operator Lease and all amendments thereto. The Company and the Company Subsidiaries have good and marketable fee title to all Healthcare Owned Property, free and clear of all Liens, except for Permitted Liens. The Company Properties constitute all real property currently owned or leased by the Company and the Company Subsidiaries. To the knowledge of the Company, the Company Properties and buildings, fixtures and improvements thereon (i) are in good operating condition (ordinary wear and tear excepted), in all respects, and all mechanical and other systems located thereon are in good operating condition in all respects; and (ii) are suitable, sufficient and appropriate in all respects for their current uses, except, in each case as have not had

and would not reasonably be expected to have a Company Material Adverse Effect. The Company has delivered or made available to Parent (or its Representatives) true, correct and complete copies of: (i) the Existing Title Policies and Existing Surveys

for the Healthcare Owned Properties, in each case in the possession of the Company, the Company Subsidiaries or the Company s Representatives; (ii) the Real Property Leases and Operator Leases, together with all amendments, modifications, supplements and guarantees, if any, relating thereto and (iii) all material environmental reports relating to conditions at the Healthcare Owned Properties, including material reports relating to compliance with Environmental Laws, which are in the possession or control of the Company, the Company Subsidiaries or the Company s Representatives. Except for Permitted Liens, the Company Properties are not subject to any leases, rights of first refusal, options to purchase or rights of occupancy entered into or granted by the Company and the Company Subsidiaries, and neither the Company nor any Company Subsidiary has granted any party other than the applicable Operator (and any sublease or license entered into by any Operator) any right to use or occupy the Company Properties or any portion thereof except for permitted subtenants or assignees.

(b) The Company or a Company Subsidiary, as applicable, has a valid, binding and enforceable leasehold interest under each of the Real Property Leases under which it is a lessee, free and clear of all Liens other than Permitted Liens. Each of the Real Property Leases is in full force and effect except as would not be reasonably expected to have a Company Material Adverse Effect. Neither the Company nor any Company Subsidiary is in default in any material respect under any Real Property Lease or Operator Lease, and no event has occurred and no circumstance exists which, if not remedied, and whether with or without notice or the passage of time or both, would result in such a default. Neither the Company nor any Company Subsidiary has received any written or, to the knowledge of the Company, oral notice of any material default which remains in existence or any event that with notice or lapse of time, or both, would constitute a material default by the Company or any Company Subsidiary under any of the Real Property Leases or any of the Operator Leases and, to the knowledge of the Company, no other party is in material default thereof. No party to any Real Property Lease or Operator Lease has exercised any termination rights with respect thereto.

(c) There does not exist any actual, ongoing, or, to the knowledge of the Company, threatened or contemplated condemnation or eminent domain proceedings that affect any material Company Property, and neither the Company nor any Company Subsidiary has received any written notice of the intention of any Governmental Entity or other Person to take or use all or any material part thereof.

(d) Neither the Company nor any Company Subsidiary has received any written notice from any insurance company since January 1, 2017 that has issued a policy with respect to any Company Property requiring performance of any structural or other repairs or alterations to such Company Property which the failure to perform such repairs or alterations would void the applicable insurance policy and which repairs or alterations have not been made.

(e) Neither the Company nor any Company Subsidiary owns, holds, is obligated under or is a party to, any option, right of first refusal or other contractual right to purchase, acquire, sell, assign or dispose of any real estate or any portion thereof or interest therein.

4.16 Material Contracts.

(a) Except for this Agreement or exhibits to the Company SEC Documents filed prior to the date hereof, Section 4.16 of the Company Disclosure Letter contains a complete and correct list, as of the date of this Agreement, of each Contract described below in this Section 4.16(a) under which the Company or any Company Subsidiary has any current or future rights, responsibilities, obligations or liabilities (in each case, whether contingent or otherwise) or to which the Company or any Company Subsidiary or any of their respective properties or assets is subject, in each case as of the date of this Agreement (all Contracts described in this Section 4.16(a) being referred to herein as a *Material Contract*):

(i) each Contract that contains any non-compete or exclusivity provisions or that otherwise limits in any respect the freedom of the Company, any of the Company Subsidiaries or any of their respective affiliates

(including Parent and its affiliates after the Effective Time) to compete or engage in any line of business or geographic region or with any Person or to negotiate or, except for provisions requiring notice or consent to assignment by the counterparty thereto, consummate any of the Transactions;

(ii) any partnership, joint venture, limited liability company agreement (other than any such agreement solely between or among the Company and its wholly owned Subsidiaries), strategic alliance Contract;

(iii) other than (A) Contracts providing for the acquisition, purchase, sale, funding, pledging or divestiture of mortgage backed securities and credit risk transfer securities entered into by the Company or the Company Subsidiaries in the ordinary course of business consistent with past practice, and (B) repurchase Contracts entered pursuant to the Company s existing master repurchase agreements (as in effect as of the date hereof) to finance the purchase price of assets or refinance the Company s repurchase obligations pursuant to such master repurchase agreements, in each case in the ordinary course of the Company s business consistent with past practice, each merger, business combination, acquisition, purchase, sale or divestiture Contract that contains representations, covenants, indemnities or other obligations (including earnout or other contingent payment obligations) that would reasonably be expected to result in the receipt or making of future payments in excess of \$750,000;

(iv) any settlement agreement or similar Contract with a Governmental Entity or Program Lender imposing operational restrictions or conduct requirements on the Company or any Company Subsidiary or any of their respective affiliates (including Parent and its affiliates after the Effective Time);

(v) each Contract not otherwise described in any other subsection of this Section 4.16(a) pursuant to which the Company or any Company Subsidiary is obligated to pay, or entitled to receive, payments in excess of \$5,000,000 in the twelve (12) month period following the date hereof, excluding (A) Contracts providing for the acquisition, purchase, sale or divestiture of mortgage backed securities entered into by the Company or the Company Subsidiaries in the ordinary course of business consistent with past practice, (B) repurchase Contracts entered pursuant to the Company s existing master repurchase agreements (as in effect as of the date hereof) to finance the purchase price of assets in the ordinary course of the Company s business consistent with past practice or refinance the Company s repurchase obligations pursuant to such master repurchase agreements in the ordinary course of the Company s business consistent with past practice;

(vi) any Contract that obligates the Company or any Company Subsidiary to make any capital investment or capital expenditure outside the ordinary course of business and in excess of \$500,000;

(vii) each Contract that grants any right of first refusal or right of first offer or that limits the ability of the Company, any Company Subsidiary or any of their respective affiliates to own, operate, sell, transfer, pledge or otherwise dispose of any businesses, securities or assets (other than provisions requiring notice of or consent to assignment by any counterparty thereto);

(viii) each Contract relating to outstanding Indebtedness (or commitments or guarantees in respect thereof) of the Company or the Company Subsidiaries (whether incurred, assumed, guaranteed or secured by any asset) in an amount in excess of \$5,000,000, other than obligations in respect of repurchase agreements, dollar roll transactions and similar financing arrangements, in each case entered into in the ordinary course of the Company s business and consistent with past practice;

(ix) each Contract that involves or constitutes an interest rate cap, interest rate collar, interest rate swap or other contract or agreement relating to a forward, swap or other hedging transaction of any type, other than Contracts

Table of Contents

entered into for bona fide hedging purposes in the ordinary course of the Company s business consistent with past practice;

(x) each repurchase agreement that the Company or the Company Subsidiaries have used in their business (or had any amounts outstanding under) or entered into since January 1, 2018;

(xi) other than advances for reimbursable employee expenses or mortgage backed securities made, purchased or sold, or subservicing advances (including without limitation in respect of legal fees), in each case, in the ordinary course of the Company s business consistent with past practice, each Contract that constitutes a loan to any Person (other than a wholly owned Company Subsidiary) by the Company or any Company Subsidiary in an amount in excess of \$50,000;

(xii) each management or similar Contract between or among the Company or any Company Subsidiary, on the one hand, and a third party manager or operator of the Company Healthcare Owned Property, on the other hand (other than any Operator Leases);

(xiii) each Contract between or among the Company or any Company Subsidiary, on the one hand, and the Company Manager, or any officer, director or affiliate (other than a wholly owned Company Subsidiary) of the Company or any Company Subsidiary or any of their respective associates or immediate family members (as such terms are defined in Rule 12b-2 and Rule 16a-1 of the Exchange Act) or of the Company Manager, on the other hand;

(xiv) any Contract that provides for the acquisition or disposition, directly or indirectly (including by merger, purchase of equity, business combination or otherwise) of any real or personal property for aggregate consideration under such Contract in excess of \$750,000 that is pending or pursuant to which the Company or its Subsidiaries have continuing earn-out or similar contingent obligations relating to purchase price adjustments;

(xv) any Real Property Lease or Operator Lease that provided for payments to or by the Company or any of the Company Subsidiaries in excess of \$750,000 in 2017 or provides for future payment to or by the Company or any of the Company Subsidiaries in the next 12 months in excess of \$750,000; and

(xvi) any Contract not otherwise described in any other subsection of this Section 4.16(a) that would constitute a material contract (as such term is defined in Item 601(b)(10) of Regulation S-K of the SEC) with respect to the Company.

(b) True and complete copies of each Material Contract in effect as of the date hereof has been made available to Parent or publicly filed with the SEC prior to the date hereof. Neither the Company nor any Company Subsidiary is in breach of or default under the terms of any Material Contract where such breach or default has not had and would reasonably be expected to have, individually or in the aggregate, a Company Material Contract is in breach of or default under the terms of the date hereof, no other party to any Material Contract is in breach of or default under the terms of any Material Contract where such breach or default has had or would reasonably be expected to have, individually or in the aggregate, a Company Material Contract is in breach of or default under the terms of any Material Contract where such breach or default has had or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, each Material Contract is a valid, binding and enforceable obligation of the Company or the Company Subsidiary which is party thereto and, to the knowledge of the Company, of each other party thereto, and is in full force and effect, subject to the Enforceability Limitations.

4.17 Mortgage Backed Securities. As of April 30, 2018, the Company or a Company Subsidiary is the sole owner of each of the mortgage backed securities set forth in Section 4.17 of the Company Disclosure Letter (collectively, the

Company MBS) and the related certificates and other instruments evidencing ownership of the Company MBS, free and clear of any Liens, except for Permitted Liens that have not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.